REMEDIAL INVESTIGATION WORK PLAN RG&E West Station Former Manufactured Gas Plant (MGP) Site Rochester, New York

> Site No. V00593-8 Index No. B-0535-97-07





Rochester Gas and Electric 89 East Avenue Rochester, NY 14649

Submitted by:



Geomatrix Consultants. Inc. 90B John Muir Drive, Suite 104 Amherst, NY 14228

February 2008

TABLE OF CONTENTS

Page

1.0	INTRODUCTION		
	1.1	SITE LOCATION AND DESCRIPTION	2
	1.2	SITE HISTORY	2
	1.3	SUMMARY OF PREVIOUS INVESTIGATIONS	5
	1.4	SITE GEOLOGY AND HYDROGEOLOGY	.10
	1.4.1	Site Geology	10
	1.4.2	Site Hydrogeology	11
	1.5	REMEDIAL INVESTIGATION APPROACH	.12
	1.5.1	Known Site Conditions	12
	1.5.2	Remedial Investigation Objectives	13
	1.5.3	Investigation Data Needs	13
2.0	REME	DIAL INVESTIGATION WORK SCOPE	.17
	2.1	MOBILIZATION	.17
	2.2	HEALTH AND SAFETY	.18
	2.3	COMMUNITY AIR MONITORING PROGRAM	.18
	2.4	GEOPHYSICAL SURVEY	.18
	2.5	SOIL INVESTIGATION	.19
	2.5.1	Test Pit Excavation	19
	2.5.2	Soil Borings	19
	2.5.3	Surface Soil Sample Collection	22
	2.6 GF	ROUNDWATER INVESTIGATION	.23
	2.6.1	Existing Well/Piezometer Assessment	23
	2.6.2	Groundwater Monitoring Well Installation	23
	2.6.3	Groundwater Monitoring Well Development, Sampling and Hydraulic Tests	27
	2.7	NEAR SHORE SEDIMENT PROBING AND SAMPLE COLLECTION	.27
	2.8	INVESTIGATION DERIVED WASTE	.28
	2.9	SAMPLE LOCATIONS AND SURVEY	.28
	2.10	ENVIRONMENTAL SAMPLE ANALYSES	.28
3.0	DATA	EVALUATION AND RI REPORT	.30
4.0	SCHE	DULE	.32
5.0	REFE	RENCES	.33

TABLE OF CONTENTS (Continued)

FIGURES

Figure 1	Site Location Map
Figure 2	Site Boundaries
Figure 3	Current Site Features
Figure 4A	Historic Site Features with Recent Aerial Photograph
Figure 4B	Historic Site Features without Aerial Photograph
Figure 5	Previous Investigation Sampling Locations
Figure 6	Visual Sheens and Tars
Figure 7	Geologic Section - Genesee River Valley – Rochester, New York
Figure 8 A	Proposed Soil Boring and Test Pit Locations in the North Plant Area
Figure 8 B	Proposed Soil Boring and Test Pit Locations in the South Plant Area
Figure 9 A	Proposed Surface Soil Sample Locations in the North Plant Area
Figure 9 B	Proposed Surface Soil Sample Locations in the South Plant Area
Figure 10A	Proposed Groundwater Monitoring Well Locations in the North Plant Area
Figure 10B	Proposed Groundwater Monitoring Well Locations in the South Plant Area
Figure 11	Proposed Near Shore Sediment Sampling Locations

APPENDIXES

Appendix A	Historical Information – Plat/Sanborn Maps
Appendix B	Previous Investigation Boring Logs/Well-Piezometer Completion Details
	B-1 – Ish, Inc. Oil Tar Separator Investigation Borings/Wells
	B-2 – AES/META Plant Area Angle Cores in Upper Bedrock
	B-3 – AES/META Plant Area Borings and Piezometers
Appendix C	Historical Sampling Locations and Tabulated Analytical Results
Appendix D	Remedial Investigation Field Sampling Plan
Appendix E	Quality Assurance Project Plan

Appendix F Compilation of Known Underground Utilities at the Plant Area

REMEDIAL INVESTIGATION WORK PLAN RG&E West Station Former Manufactured Gas Plant (MGP) Site Rochester, New York Site No. V00593-8 Index No. B-0535-98-07

1.0 INTRODUCTION

Rochester Electric & Gas Corporation (RG&E) retained Geomatrix Consultants, Inc. to prepare this Remedial Investigation Work Plan (RIWP) for the West Station Former Manufactured Gas Plant (MGP) Site (Site No. V00593-8). The RIWP has been prepared consistent with the requirements of a Voluntary Cleanup Agreement (VCA) Index Number *B-0535-98-07* between RG&E and the New York State Department of Environmental Conservation (NYSDEC) effective April 10, 2003.

This RIWP identifies the means for further characterization of the nature and extent of environmental impact from potential MGP material at the Site and describes the rational for, and methods to assess the impacts. It also strives to identify the needed information and data requirements to assist in the preparation of a remedial action work plan.

The following information is presented in this Work Plan:

- The remainder of Section 1 provides a description of the Site and its location, a summary of the site history, a review of previous site investigations; and identification of remedial investigation (RI) objectives and data needs;
- Section 2 presents the RI scope of work and references the appended Field Sampling Plan (FSP) and Quality Assurance Project Plan (QAPP);
- Section 3 describes the content of the RI Report; and
- Section 4 presents a target duration schedule to complete the work.

A "stand alone" project Community Air Monitoring Plan (CAMP) and Odor Mitigation Plan (OMP) are also attached and will be used throughout the project during tasks that involve subsurface sampling. A Health and Safety Plan (HASP) will be prepared as another separate, stand alone project document and will be submitted under separate cover prior to the start of field activities.

1.1 SITE LOCATION AND DESCRIPTION

The West Station Site is a former MGP located at 254 Mill Street in Rochester, Monroe County, New York (Figure 1). The Site is presently owned by RG&E and, as shown on Figure 2, is bounded by RG&E Power Station No. 3 (Beebe Station electric generation plant) situated on Mill Street and Falls Street to the west, the closed City of Rochester trash incinerator plant to the north, the Genesee River to the east, and the Platt Street Bridge on the south. The area of the Site south of the Platt Street Bridge was called the "Park Area" of the West Station Site (which was remediated by RG&E in accordance with VCA Index No. D8-0001-95-10), while the remaining portion of the West Station Site was referred to as the "Plant Area". The former MGP processes were located on the Plant Area portion of the Site and encompass approximately 7 acres. The Site is relatively flat, with an average ground elevation of approximately 412 feet above mean sea level (MSL). Several buildings and structures owned by RG&E remain on the property. These include: an electric substation; two oil tanks with secondary containment structures; a natural gas-fired turbine; and several historic buildings used for RG&E equipment storage. The RG&E property is zoned "CCD-R – Center City District – Riverfront". The zoning also applies to surrounding properties and allows industrial, commercial, and residential property use. The RG&E property and other properties in its immediate vicinity, have long histories of industrial and commercial use. Existing site features on RG&E property are shown on Figure 3.

1.2 SITE HISTORY

During the development of this Work Plan, the history of the site was compiled based on maps, photographs, and documents identified and reviewed in RG&E files, as well as, historic information documented in previous site investigation reports. Additional information has been inferred from 1875, 1892, 1911, 1950, and 1971 Plat and Sanborn Fire Insurance maps (see Appendix A).

A number of businesses, such as mills and foundries, formerly occupied the West Station Site and adjacent properties. Brown's Race, a man-made channel which conveyed water from above the 90 foot high Upper Falls (just south of the site), provided the hydropower to these businesses. Water was returned to the river via numerous tail races which flowed from the base of the cliffs to the River. Several of these races crossed the Site from a west to east direction. From 1875 to 1915, the many small parcels of land in this area supported a large number of businesses. During the 1890s, utilities, which eventually merged and became RG&E, acquired many of the small parcels of land and established power, steam, and coal gas manufacturing facilities.

From 1875 to the present, the landscape at and around the West Station Site changed significantly. The "flats" (i.e., that area between the river and top of the Genesee River Gorge) were gradually filled and expanded into what was formerly the river channel. Individual industrial buildings were removed and replaced with RG&E facilities, and existing RG&E facilities were modified or replaced. Several of the tail races used by the historical industries were either filled as their use as tail races discontinued or piped and filled to make use of the land near and above the races. The approximate historical locations of the former head and tail races are shown on Figures 4A and 4B. Historical site features are transposed over a recent aerial photograph in Figure 4A.

In summary, the area of what is currently the RG&E West Station site has undergone almost continual change over the past century. A timeline based on RG&E files and historical site maps, depicting those changes is described briefly below.

In 1875, many businesses bordered Brown's Race, occupying the area between the race and the top edge of the cliff. Grist mills were the most common industry, but foundries, machine shops, and the Rochester Water Works were also present in the area. Approximately one dozen tail races exited the buildings into notches in the cliff and flowed across the flats and into the Genesee River.

From the 1870s to the early 1890s, changes in this area were minor. The configuration of some of the tail races changed, some new buildings were constructed, and the edge of the river channel was partially filled. In some cases, the property ownership changed hands. However, the type of industry on the site, primarily mills, remained the same.

By 1900, a predecessor company of today's RG&E had obtained land north and south of the Platt Street Bridge. Station No. 2, a hydroelectric plant, was constructed south of the bridge prior to 1900. The RG&E predecessor company also owned a parcel north of the Platt Street Bridge and adjacent to the Citizens Light and Power Company. Mills, foundries, and machine shops also continued to operate in the area. The configuration of

the tail races associated with these industries was altered during this time and the edge of the river channel was further filled.

From 1910 to 1926, major changes took place on the West Station Site. During this time, the companies that would become the present day RG&E acquired the majority of the property bounded by Brown's Race, Mill Street, Falls Street, and the Genesee River. Power and gas generating facilities (coal fired steam, coal gas and water gas) were all constructed during this time period. In 1910 much of the area was vacant, but by 1926 the area north of the Platt Street Bridge was completely occupied. Station No. 3 (also known as "Beebe Station") was constructed between 1900 and 1910 and provided steam power from coal combustion. Surrounding land use remained heavily industrial.

In 1911, a garbage incinerator was constructed by the City of Rochester on property immediately north of the generally undeveloped West Station property. The garbage incinerator operated until some time between 1935 and 1950 based on Sanborn mapping information. The 1911 Sanborn map also shows an iron gasometer (gas holder) close to the riverbank on the West Station property. Since gas production at West Station MGP site did not begin until 1916, it is assumed this holder was used for gas generated at the East Station MGP located northeast of the West Station site across the Genesee River, which began gas production around 1872.

In the summer of 1913, a concrete retaining wall (approximately 500 feet long) was built along the riverbank from the north end of the West Station property to the gasometer, and the area behind the wall and existing ground surface westward toward Falls Street was backfilled with 20 or more feet of fill material. In 1916 and 1917, the West Station coal carbonization gas plant was constructed on the northern portion of the filled area. This original plant consisted of coal and coke storage and transport facilities, a retort house, a producer gas house, an ammonia still building, an office building, a scrubber house, a decanter, cooling coils, and tar and ammonia liquor storage tanks. A 675,000 gallon tank labeled "tar storage tank" was on the location previously identified as the iron gasometer. Gas generated at the West Station MGP was transported to the East Station MGP site via a pipe bridge across the Genesee River. Historic information suggests that limited purification processes, such as tar and ammonia removal, took place at the West Station property and further purification of manufactured gas was completed at the East Station site. The water gas plant at the West Station site was built in 1923 and occupied the area of riverbank just south of the coal gas plant near the present location of the Central Water Treatment Facility. An oil-tar separation structure was located on the south side of the water gas plant. Coal storage was located in an open area west of the retort and transported to different MGP process areas via overhead conveyors.

An RG&E publication, *Gas Manufacture and Distribution*, was originally published in 1928 and subsequently revised in 1934, 1936, and 1938. These historical pamphlets contain information about the type of the gas manufacturing operations at the former West Station MGP site. In addition, dated photographs from RG&E's files provide expansion details for facility development.

A coke oven containing the 60 (later 97) Becker-type ovens, was constructed directly south of the existing facilities and just north of the Platt Street bridge in 1925 and 1926. The Koppers producer house was also built at this time and a creosote tank was constructed east of the coke oven battery. Thus, by 1928 the old retort house and the new coke ovens were functioning. There is no evidence of further expansion of gas manufacturing facilities. The 1951 map of the site, the 1935 Plat maps and the 1950 Sanborn map confirm that no further expansion occurred during this time period.

Gas manufacturing continued at the West Station MGP until 1952. From 1952 to 1961 the gas manufacturing facilities were dismantled and a few buildings, such as Numbers 16, 17, and 18, were converted for other purposes. The water gas house was still present on the 1971 Sanborn map. It was subsequently demolished and replaced by the Central Water Treatment Facility in the late 1970s to support the Beebee Station electric generation power plant.

Following decommissioning of MGP processes, RG&E utilized the property to support operations at Beebe Station. During the 1980s, a 1.3 million gallon fuel oil tank was constructed on a concrete containment structure to fuel power generation at Beebe Station. The tank originally stored fuel oil No. 6 and later stored fuel oil No. 2. By May 1987, the tank was no longer used for fuel oil storage. RG&E records indicated the tank was decommissioned and removed from the property in 1994.

1.3 SUMMARY OF PREVIOUS INVESTIGATIONS

A number of investigations have been performed at the West Station Site, including a 1977 investigation by Rochester Drilling Company, Inc., an investigation in 1985 and 1986 by Empire Soils, a 1991 investigation by META Environmental, Inc. (META) and Atlantic Environmental Services (AES), a 2003 survey of the riverbank as summarized in

the 2004 River Bank Observation Report by Ish Inc. and META (Ish/META), and an investigation at the former oil tar separator by Ish/META in 2006. A brief summary of the findings of the previous work is provided in the following sections.

Rochester Drilling Company, Inc. Investigation (1977)

A geotechnical investigation was conducted prior to the construction of the Central Water Treatment Facility (CWTF) building on the site. The CWTF was constructed partially in the footprint of the former carbureted water gas MGP plant at the site. Soil borings advanced with hollow stem augers in the vicinity of the present CWTF encountered fill materials consisting of brown to black silt, sand, gravel, brick, concrete, cinders, ash, and wood. Petroleum odors were noted at all boring locations.

Empire Soils Investigations, Inc. Borings (1985 and 1986)

In November 1985 and February 1986, limited investigation activities were carried out by Empire Soils Investigations, Inc. A total of five borings were drilled to depths ranging from 7.8 to 19.4 feet bgs as part of a coal pile leachate study. The soils in the borings consisted of fill (gray, brown and black silt, sand, gravel, cinders, brick, slag, and wood) overlying a native soil layer (alluvium). The alluvium consisted of gray, tan, and brown fine to coarse sand with some gravel and clay.

Atlantic Environmental Services, Inc. and META (AES/META), Characterization of the Former Manufactured Gas Plant (MGP) at the RG&E West Station Site (March 1993)

Between September and December 1991, a site investigation was performed at the West Station Site which included examination for the presence of surface tar and subsurface investigation. Visible areas of surface tar were mapped. There was evidence of hardened tar drippings from pipes along the retaining wall along the river bank, as well as, hardened tar at the sewer pipe openings in the retaining wall. The mapped areas of tar at the Plant Area are shown on Figure 5.

The subsurface investigations included: soil gas sampling at 76 locations; soil borings at 42 locations; installation of 2 monitoring wells and 10 piezometers; and drilling of three angled bedrock core holes. Much of the subsurface investigation was completed at the Park Area beneath and south of the Platt Street Bridge. The subsurface investigation at the Plant Area focused on three areas:

- 1. former tar storage tank area
- 2. former oil-tar separator/water gas plant area
- 3. former creosote storage tank area

Work completed in each area is summarized below:

Former Tar Storage Tank Area:

Borings, piezometers, and angled rock core holes completed in the former tar storage tank area include:

- borings: TST-01; TST-02, R33, R34
- piezometers: P32, P35, P36
- angled bedrock core hole: AB3

The depth to bedrock in the former tar tank area was in range of 28 to 35 feet. Tar staining and/or petroleum odors were noted at intermittent depths on boring log descriptions of fill and alluvial soils. Tar saturated soils were not described in the boring logs. No odors or staining was noted in the bedrock (AB3). During boring advancement, soil samples were collected at each boring location and analyzed for total PAHs (polynuclear aromatic hydrocarbon compounds). Typically 7 to 14 samples were analyzed per location. Samples where total PAH concentrations were detected above 500 ppm included borings TST-02, R34, and P36. Samples analyzed from TST-01, R32, R33, and P35 did not detect total PAHs above 500 ppm.

Former Oil-Tar Separator/Water Gas Plant Area:

Borings, piezometers, and angled rock core holes completed in the former oil-tar separator/water gas plant area include:

- borings: TWS-01, TWS-02
- angled bedrock core hole: AB2

The depth to bedrock in the former oil-tar separator/water gas plant area was approximately 27 feet at each boring. Occasional tar staining, sheens, and/or tar odors were noted at intermittent depths on boring log descriptions of fill and alluvial soils. Tar saturated soils were not described in the boring logs. Occasional tar odors were noted in the bedrock (AB2). During boring advancement, soil samples were collected at each boring location and analyzed for total PAHs. The highest concentrations of total PAHs were detected in the upper 10 feet of fill at TWS-02. Four out of seven samples analyzed from the upper fill detected Total PAH concentrations above 500 ppm. Samples analyzed from TWS-01 did not detect total PAHs above 500 ppm.

Former Creosote Storage Tank Area:

Borings, piezometers, and angled rock core holes completed in the former creosote storage tank area include:

- piezometer: PCST-1
- angled bedrock core hole: AB1

The depth to bedrock in the area of the former creosote tank was approximately 12 feet. Occasional tar staining and/or tar odors were noted at intermittent depths on boring logs descriptions of fill and alluvial soils. Tar saturated soils were not described in the boring logs. A tar odor and sheen was noted at a depth of approximately 25 feet below the bedrock surface (AB1). During boring advancement, soil samples were collected from PCST-1 and analyzed for total PAHs. Total PAH concentrations were detected above 500 ppm in five of nine samples analyzed. No recognizable pattern of PAH distribution was noted.

The locations of soil borings and piezometers installed on the Plant Area of the Site are shown on Figure 6. Boring logs and well completion details for this investigation are presented in Appendix B.

Ish Inc., Observation Report, Riverbank Survey, Former West Station MGP Site (January 2004)

In November 2003, Ish Inc. and META personnel performed a riverbank survey from the northern property boundary south to the Platt Street Bridge. The river sediments immediately off-shore were examined by moving rocks and debris and probing any soft sediment with a hand auger. The sediments were probed approximately every 25 feet or less. NAPL was considered present if sheen developed on the water. The NAPL was identified as former MGP material if a distinctive coal tar-like odor was noted. At several locations, dried tar-like material (TLM) was present. A heavy sheen and NAPL globules were produced in near shore sediment by moving rocks and debris or by augering into the riverbank deposits near the oil-tar separator area. Observations are summarized on Figure 5.

Ish Inc., Oil-Tar Separator Area Investigation (November, 2006)

An investigation for a proposed Interim Remedial Measure (IRM) was undertaken by Ish, Inc. at the oil-tar separator area near the former water gas plant. The investigation included test pit excavation (TP01) to locate and characterize construction materials and tar presence in the subgrade oil-tar separator, sampling of subsurface soil (borings SB01 and SB15/15A) to characterize soils near the oil-tar separator and the former water gas plant, and installation of monitoring wells MW201 (at SB05) and MW202 (at SB09) and sampling to characterize groundwater quality. A recovery well (RW-1) was installed to evaluate potential recovery of tar encountered at SB03. The three wells were sampled on two occasions.

The test pit excavation was successful in locating the former oil/tar separator (TP-1A). During the work, the excavation was advanced outside of the separator walls to further investigate and characterize the surrounding soils and to determine if mobile or residual NAPL was present. In addition, two other test pits were advanced westward (TP-1B and TP-1C), following the outside wall of the former oil/tar separator. The former separator was found to be visually free of MGP residuals. Borehole logs for borings completed in close proximity to the oil-tar separator (SB02, SB03, and SB13) describe several samples collected from the lower portion of the overburden having viscous, tar-like material and NAPL sheens or globules. TLM and NAPL was also described in samples collected from borings SB-15/15A located north of the former central water treatment facility, at the bottom of boring SB-14 which was terminated at 15.5 feet on top of the concrete pipe for Outfall #3 (Brown's Race discharge), and boring SB08 located south of the oil-tar separator between the two former oil tanks. Tabulated laboratory detection summaries for samples analyzed during the investigation are provided in Appendix C. The detections were compared to Technical and Administrative Guidance Memorandum (TAGM) #4046 and Part 375 SCOs for Industrial Property Use. Eight of 22 samples analyzed detected individual volatile organic compounds above one of the comparison criteria. Individual polynuclear aromatic hydrocarbon (PAHs) compounds were detected in one of the comparison criteria in 20 of 21 samples. Total PAHs were detected above 500 ppm in 9 of the samples.

Groundwater samples were collected from overburden monitoring wells MW201 and MW202 and recovery well RW01 on two occasions after well development. A slight sheen and moderate MGP/fuel odor was noted during well development at MW201. A slight MGP odor was noted and one small NAPL globule was observed during

development of MW202. A strong odor and sheens were reported during development of RW01. A measurable thickness of NAPL was not identified in any of the wells during sampling although the sample tubing was stained with a light colored NAPL, which extended for 5 to 7 feet up from the pump during both sampling rounds.

Benzene, toluene, ethyl benzene, and xylenes (BTEX) and naphthalene were not detected at well MW201. BTEX and naphthalene were detected at elevated concentrations (compared to groundwater quality standards in TOGS 1.1.1) at RW01 and MW202. Naphthalene was detected at concentrations ranging from 1,700 to 5,400 ug/L with MW202 having the higher concentrations.

1.4 SITE GEOLOGY AND HYDROGEOLOGY

1.4.1 Site Geology

The former MGP is located at the base of the Genesee River Gorge. Adjacent to much of the river and at the Site, a layer of fill material overlies alluvial deposits. Bedrock beneath the overburden, which forms sidewalls and floor of the gorge at the base of the High Falls, is the Rochester Shale formation. Descriptions of geologic materials encountered during previous Site investigations follow.

Fill Material

Fill material descriptions in logs for borings drilled on the Site during previous investigations indicate a heterogeneous composition. In some areas of the Site a 6 to 8 inch layer of top soil (brown, sandy loam) was described. Common fill materials included, but were not limited to: brick, ash, slag, coke, coke breeze, cinder, coal, building rubble, gravel, and occasional solidified TLM. The TLM encountered in borings was distributed randomly in the fill and generally above the water table (upper 15 feet). The thickness of the fill is indicated to range from 15 to 30 feet (AES/META, 1993). The existing data suggest the thickest areas of fill are present close to the retaining wall at the northern portion of the Site and in areas where the tail races were historically filled.

Alluvium

The alluvium deposits were described as fine to coarse sands with varying mixtures of silt and clay with rounded gravel. Silt and clay lenses also were occasionally encountered. The thickness of alluvium varied from 0 to approximately 15 feet across the Site (AES/META, 1993). At the bottom of the alluvium, a layer of angular cobble size rock fragments was often encountered. AES/META (1993) speculated that these angular rock fragments may correspond to the bottoms of the tail races that once crossed the Site in numerous locations.

Bedrock

The uppermost layer of bedrock, Rochester Shale, was encountered at depths ranging from approximately 20 to 30 feet below ground (AES/META, 1993). The Rochester Shale is a dark gray, calcareous mudstone, which contains frequent, thin, interbedded layers of dolomite and limestone. In many borings, 5 to 10 feet of weathered and severely weathered bedrock was encountered above the competent bedrock. Shale-type bedrock weathers to form clay, which has the appearance of bedrock but the texture and hardness of clay. The top of the weathered bedrock was brown clay, generally dry and compact, which graded to gray as the weathering lessened with depth. Although not described in Site rock core descriptions, the Irondequoit Limestone occurs beneath the Rochester Shale. The bedrock stratigraphy in the area of Genesee River Gorge is shown on Figure 7.

1.4.2 Site Hydrogeology

Groundwater flows in the lower portion of the fill and the alluvial deposits. The depth to the water table was approximately 18 to 22 feet during the investigation activities performed in November and December 1991. AES/META (1993) report a southerly groundwater flow direction in the overburden and speculated that the groundwater is redirected or influenced by the retaining wall which is keyed-in to the upper surface of the bedrock and the coarser-grained materials used to in-fill the historic tail races.

Although the surface of the bedrock is reported to be weathered, some groundwater flow from the overburden may enter the upper bedrock as indicated by the presence of tar odors and sheens in bedrock which continued from the overburden above. The predominant groundwater flow direction in the shallow bedrock is expected to be toward the Genesee River which is an area of regional groundwater discharge. However, the proximity of the Middle Falls to the Site (approximately ¹/₂ mile downstream) suggests that deeper groundwater flow may occur in a northward direction toward the bedrock groundwater discharge area associated with the lower head at the base of the Middle Falls.

1.5 REMEDIAL INVESTIGATION APPROACH

1.5.1 Known Site Conditions

Prior investigations focused on the characterization of soils, and to a lesser extent, groundwater at three former MGP process areas within the Plant Area:

- 1. former tar tank area
- 2. former oil-tar separator area
- 3. former creosote tank area

These historic structures are shown on Figures 4A and 4B. In these areas, approximately 15 to 20 feet of historic fill material was placed over native alluvial soil which directly overlies shale bedrock.

Previous investigations did not identify wide-spread tar presence in these suspected source areas. The 1991 and 2006 investigations of the oil-tar separator area did not identify a sufficient quantity of MGP byproducts in the soil to warrant an Interim Remedial Measure (IRM). Extensive NAPL saturated areas were not identified and tar presence was characterized as NAPL sheens or globules in fewer than 25% of the borings completed in this area. Laboratory analytical results for soil samples exhibiting MGP-impact detected VOCs at concentrations above Part 375 SCOs for Industrial Property use in 5 of 22 samples analyzed. One or more PAHs were detected at concentrations above Part 375 SCOs in all but one of the samples analyzed. However, some PAH detections were likely attributable to the placement of historic fill across the site in the early 1900s based on the absence or comparatively low detected concentrations of naphthalene and pyrogenic PAHs which are typically associated with MGP processes.

These investigations did not characterize site-wide groundwater conditions in detail. However, a preliminary site conceptual model has been developed for the Plant Area based on data collected from previous investigations. The investigation data indicate that MGP byproducts near former MGP process areas impacted overburden soil and groundwater. MGP impacts occur in the form of stained soil and soil having sheens with occasional tar-like material. VOCs, PAHs, and total cyanide were detected in unsaturated and saturated soil. The presence of MGP process-related constituents in the groundwater suggests that MGP materials in the soil may have migrated downward through the vadose zone to overburden groundwater. The groundwater in the overburden flows in the direction of the River (regional groundwater discharge area) but may be diverted in a southerly direction by the concrete retaining wall to the in-filled area of the former Brown's Race discharge. The overburden groundwater will ultimately discharge to the Genesee River. Some groundwater may migrate vertically downward into the bedrock as indicated by petroleum-type odors and sheen observed in the shale bedrock beneath the Site. Some solidified tar is observed in and around sewers that discharge to the ground surface between the retaining wall and the river. The presence of tar in and near those structures indicates sewers and other subsurface piping may be involved as preferential migration pathways for Site-related constituents.

1.5.2 Remedial Investigation Objectives

The remedial investigation activities described in Section 2.0 will supplement prior investigation results and will generate a combined data set sufficient to assess:

- Extent of on-site MGP-related material (i.e, NAPL, tar-like material);
- Nature and extent of chemical constituents in Site media (i.e., soil, groundwater); and
- Potential for human health and ecological risks posed by MGP and/or non-MGP derived chemical constituents at the Site.

1.5.3 Investigation Data Needs

The previous investigations provide an indication of the nature and possible extent of soil impacts associated with the tar tank, oil-tar separator, and creosote tank areas. However, additional data would be necessary to refine and/or verify the limits of impacted soil and to evaluate associated groundwater impacts. In addition, there is little information available from previous investigations concerning other areas of the former Plant and Site-wide soil and groundwater conditions. In order to complete the RI, additional field investigations will be required to characterize the following:

- The limits of impacted soils and groundwater in and around the three areas previously investigated (tar tank, oil-tar separator, and creosote tank areas)
- Soil and groundwater impacts at former MGP process areas not previously investigated
- Soil and groundwater impacts between the Site concrete retaining wall and the River

- Near shore sediments impacts adjacent to the Site
- Deeper bedrock geology, hydrogeology, and extent of MGP impacts in bedrock groundwater beneath the Site
- Transport mechanisms including preferential pathways (e.g., sewers)

The table below summarizes MGP process locations and areas of the Site where additional characterization of the extent of MGP impacts is recommended and includes a brief rationale for investigation.

Investigation Area (Historical Locations)	Rationale	
Septic Pool Area	Hardened tar observed at TAR-E1 on river side of concrete retaining wall near opposite side of former septic pool area discharge	
Ammonia Still Bldg. Area (Bldg No. 16) to Scrubber Bldg. Area (Bldg No. 18)	Hardened tar observed on the river side of the retaining wall opposite buildings No. 16, No. 17, and No. 18	
Liquor Cooling Area	Hardened tar observed on river side of concrete retaining wall opposite side of liquor cooling coils	
Oil Pit at Primary/Secondary Coolers	To assess possible impacts from primary/secondary coolers	
Tar Storage Tank Area	To assess impacts from tar storage in 675,000 gallon storage tank and other MGP impacts. Two tanks located north of tar tank are labeled concentrated ammonia tanks, spare tanks, or creosote tanks on different drawings and from smaller tanks identified as cyanogens tanks. Prior investigations identified MGP impacts in this area and the extent of impact requires further characterization.	
Retort House Area	To assess possible impacts of MGP process materials	
Hot Drain & Flush Tank Area	To assess possible impacts of MGP process liquids	
Non-MGP Fuel Oil Tank/Pump House	To assess possible oil impacts to subsurface	
In-filled Northern Race (former Brown's Race)	To assess possible preferential pathway for migration of MGP-derived materials	
Gas Producer House	To assess possible impacts of MGP process materials	

Investigation Area (Historical Locations)	Rationale	
Koppers Producer House	To assess possible impacts of MGP process materials	
Area near substation east of producer gas holder above in-filled race	To assess possible impacts of MGP process materials and potential transport in preferential pathway	
Inside eastern end of Central Water Treatment Facility	To assess possible impacts of MGP process materials from former water gas producer house	
North and South side of Central Water Treatment Facility	Prior investigations identified MGP impacts from Water Gas Plant in this area and the extent of impact requires further characterization.	
Former Tar Separator Area	Prior investigations identified MGP impacts in this area and the extent of impact requires further characterization.	
Coke oven battery area	To assess possible impacts of coking process materials	
100,000 gal. Creosote Storage Tank Area	Prior investigations identified MGP impacts in this area and the extent of impact requires further characterization.	
Pitch Hopper Area	To assess possible impacts of MGP process materials	
Sewer outfall piping exposed along the retaining wall	To assess impacts in pipe bedding materials and former tail races as a result of preferential migration pathways	
Land between river and retaining wall	To assess possible impacts to river sediment	
Near shore river sediments	To assess possible impact to near shore sediment from Site operations	

MGP process areas and subsurface structures from: Drawing of Pipe Lines at West Station (date unknown); Drawing of West Station Gas Works Pant Layout (August 1925)

In addition to the investigation areas above, bedrock conditions beneath the site require further characterization and understanding. The bedrock is fractured and existing rock core logs describe petroleum odors and sheens in the upper bedrock. The nature and extent of MGP related process materials and bedrock groundwater quality impacts have not been fully assessed. Characterization of the bedrock during the RI will include an assessment of bedrock stratigraphy, head distribution, and water quality in the bedrock to the top of the Reynales Limestone anticipated to be approximately 90 to 100 feet below ground surface at the property. The bedrock investigation will be conducted using a phased approach where deeper exploratory borings will be completed in the first phase.

The second phase of the bedrock investigation would be targeted as appropriate to assess the nature, extent, and potential migration of MGP constituents in bedrock water-bearing zones beneath the Site.

Although VOCs were detected during previous investigations in on-Site soil and groundwater, vapor intrusion sampling is not proposed during this RI since no occupied structures exist on or immediately adjacent to the Site. Therefore, vapor migration from impacted soil/groundwater into the vadose zone and subsequent intrusion into an occupied structure does not represent a potentially complete exposure pathway. Should building occupancy at the site occur, the need for vapor intrusion will be reevaluated at that time.

The remedial investigation activities are described in Section 2 and detailed in the Field Sampling Plan (FSP) presented in Appendix D. USEPA and NYSDEC-approved sample collection and handling techniques will be used throughout the investigation fieldwork and laboratory analyses will be conducted in accordance with the Quality Assurance Project Plan (QAPP) presented in Appendix E.

2.0 **REMEDIAL INVESTIGATION WORK SCOPE**

This section of the RI Work Plan describes the field investigation activities and rationale for implementation. The planned field activities include:

- Completion of an electromagnetic (EM) geophysical survey to locate buried structures (i.e., pipes, wires, foundations, etc.) containing metal in an effort to locate unknown below grade features. Note, since the Site is comprised of fill the results of the EM survey may or may not provide data supportive of the RI
- Completion of subsurface soil sampling to investigate the nature and extent of onsite impacts from potential MGP related process materials, and to evaluate potential migration pathways
- Collection and analysis of surface soil samples to assess the nature and extent of MGP impacts and support qualitative risk assessment
- Identification of historic MGP process locations and completion of soil borings including collection and analyses of subsurface soil samples to supplement the previous Site characterization data
- Assessment of existing piezometer/monitoring well integrity for use during the RI
- Installation of overburden and bedrock monitoring wells and collection of groundwater samples from new and existing wells to: 1. supplement previous groundwater characterization data and assessment; 2. identify areas where mobile separate phase liquid is present, if any; and 3. characterize groundwater flow directions in overburden and bedrock where Site-related constituents are identified
- Characterization and sampling of near shore-stream bed sediment in the Genesee River to provide initial identification of nature and extent of sediment impacts and to compare with the sediment criteria guidance by NYSDEC Division of Fish and Wildlife

To facilitate viewing the mapped locations of test pits, borings, and monitoring well installations proposed to be completed during RI implementation, the Plant Area is divided into a North Plant Area and a South Plant Area. The in-filled Brown's Race separates the two areas.

2.1 MOBILIZATION

Prior to conducting remedial investigation work involving intrusive field activities, Dig Safely New York will be contacted and RG&E staff will be consulted to mark known on-Site underground utilities. Geophysical survey results (see Section 2.4) and historical drawings will also be used to assess the presence of underground structures and utilities. A compilation of identified underground utilities and current structures based on RG&E file searches is presented in Appendix F.

2.2 HEALTH AND SAFETY

A Site-Specific Health and Safety Plan (HASP) will be prepared pursuant to 40 CFR 300.150 of the NCP and 29 CFR 1910.120 and submitted under separate cover prior to the commencement of the field activities. The HASP will address worker safety and will include monitoring of air in the worker breathing zone using a photoionization detector (PID) and Real-Time Mini-Ram Aerosol Monitor. Additional perimeter air monitoring during the investigation is described in the Community Air Monitoring Plan (CAMP).

2.3 COMMUNITY AIR MONITORING PROGRAM

A project community air monitoring plan (CAMP) is attached and will be implemented during intrusive project activities which have the potential to generate VOC and/or dust fugitive emissions. The CAMP is consistent with the New York State Department of Health's (NYSDOH) generic program requirements.

2.4 GEOPHYSICAL SURVEY

An electromagnetic (EM) survey will be conducted to facilitate the identification of buried structures (pipes, wires, foundations, etc.) which contain metal and that such areas may help target investigation locations. Both MGP and non-related MGP infrastructure needs to be known before intrusive work activities begin. The identification of subsurface structures remaining from the decommissioned MGP and those associated with the former Beebee generating station and/or active piping wires associated with the active on-site facilities will facilitate the establishment of investigation targets at the former MGP. Electromagnetic (EM) surveys have demonstrated success in the identification of buried metallic objects. EM anomalies suggesting the location of former foundations or buried piping would therefore be the target of soil borings and/or test pit excavations. Note, since the Site is comprised of fill the results of the EM survey may or may not provide useful or clear data.

An electromagnetic survey of accessible areas of the Site will be conducted using a GEONICS EM-61 time domain metals detection system. A reference grid will be established across the entire Site prior to data acquisition which will aid in subsequently field locating EM anomalies. A description of the EM survey methods and data

presentation techniques are included in the FSP presented in Appendix D. The EM survey would be completed prior to subsurface intrusive investigations.

2.5 SOIL INVESTIGATION

The investigation of Site soils will include collection of surface and subsurface soil for laboratory analysis, identification of MGP related materials (i.e., coal tar, NAPL, purifier waste, etc.), and characterization of soil types, soil saturation, relative permeability, and potential chemical migration pathways in the overburden. Soil samples collected for laboratory analysis will be analyzed for Target Compound List (TCL) volatile organic compounds (VOCs), TCL semi-volatile organic compounds (SVOCs), polychlorinated biphenyls (PCBs), and Target Analyte List (TAL) inorganics including total cyanide. Methods of investigation are described below.

2.5.1 Test Pit Excavation

At appropriate locations, test pits will be excavated to explore for potential MGP related materials, determine subsurface material composition, and investigate subsurface features (i.e., pipes, foundations, impacts, etc.). Seven definitive proposed test pit locations are shown on Figures 8A and 8B. Additional test pits may be proposed following the interpretation of the electromagnetic geophysical survey and those needed to meet the objectives of this RI. The rationale for relocating and/or proposing additional test pit locations and the nature of any geophysical anomalies targeted will be documented and discussed with the Department's project manager.

Soils and materials encountered during test pit excavations will be examined and logged in accordance with the FSP included in Appendix D. One or more soil samples will be collected from each test pit location and submitted for laboratory analyses for TCL VOCs, TCL SVOCs, PCBs, and TAL inorganics including total cyanide. The selection of soil samples for laboratory analysis will be based on the presence and magnitude of suspected chemical impact from historic MGP operations or the suspected absence of chemical impact to document "unimpacted" soil conditions (depending on the purpose of the sample).

2.5.2 Soil Borings

Soil borings were completed during previous investigations to characterize certain MGP process areas (tar storage tank area, oil-tar separator area, and creosote storage tank area). Based on the findings of MGP impacts in soil in these areas and the need to characterize

former MGP process areas previously not investigated (see Section 1.5.3), additional soil borings and soil sample collection will be necessary to characterize the Site. The primary objectives of the soil borings will be to:

- identify potential MGP related material in the subsurface
- assess the extent of on-site MGP or non-MGP residuals (both at the Plant Area and the area between the concrete retaining wall and the river shoreline)
- collect samples of subsurface soil for laboratory analysis

Soil borings will be completed at the process areas of the former MGP as well as between, to the extent it's feasible and safe, the MGP process areas and the river shoreline (i.e., riverside of the retaining wall). A total of 43 proposed boring locations are shown on Figures 8A and 8B, however the exact number (more or less) and locations of borings needed to satisfy the objectives of the RI will be based on the field observations and the need to satisfy investigation objectives. The investigation area of the former MGP processes area is generally topographically flat which would allow the use of standard truck-mounted drilling equipment. The area between the MGP process areas and the river is sloped and tree-covered and will require site preparation before drilling can commence. ATV-mounted or skid-mounted drilling equipment will be needed to access those locations. Boring locations are coincident with investigation areas summarized in the table presented in Section 1.5.3. Investigation rationale is also provided in the table.

Drilling at Plant Area MGP Process Areas:

Soil borings completed at the Plant Area former process areas will use a conventional truck-mounted drill rig equipped with a pneumatic auto-hammer capable of driving split spoon samplers and/or macrocore samplers advanced ahead of hollow stem augers. Soil samples collected from split spoon samplers and macrocore samplers will be examined and logged in accordance with the FSP presented in Appendix D. Soil boring depths will extend through the fill material and the alluvium to the bedrock until refusal occurs. Existing drilling logs indicate the shale bedrock is sufficiently soft and weathered to allow the advancement of augers several feet into bedrock. Split spoon samplers will be used to collect weathered bedrock samples where the bedrock is soft. At boring locations where auger advancement occurs in weathered bedrock, split-spoon samples will be collected until sample recovery is less than 6-inches per 100 blows of the auto-hammer.

Drilling between the MGP Process Areas and River:

To the extent that soil borings can be completed safely on the river side of the retaining wall and those on the slope south of the retaining wall terminus (near the former Central Water Treatment Building), borings will be completed with an ATV-mounted or skid-mounted drill rig capable of either macro-core or split spoon sample collection and hollow stem auger drilling. Site access may require use of a crane to lower the drilling equipment to the boring locations or constructing a temporary access road on the slope between the Site and the west bank of the river. Any necessary permits will be obtained prior to site mobilization. The borings will be drilled and continuously sampled until equipment refusal. The soils will be examined and logged in accordance with the FSP presented in Appendix D. Should it be deemed unsafe or otherwise not feasible to obtain samples in this location with a drill rig, data needs will be discussed with the Department's project manager.

Supplemental Borings:

Supplemental borings may be needed to establish the extent of on-site MGP related materials. The supplemental boring locations will be established in the field with concurrence with the Department's project manager.

Subsurface Soil Sampling Strategy:

One or more soil samples are contemplated to be collected and analyzed from each soil boring and analyzed for TCL VOCs, TCL SVOCs, PCBs, TAL Metals, and total cyanide. Where MGP materials are visible and impacts from MGP or non-MGP materials are not suspected in the fill, a sample will be collected of "unimpacted" fill from soil within the upper 8 feet of the ground surface. At some boring locations, samples will be collected from soil immediately above the water table (interface sample) which is anticipated to be approximately 15 feet or more below ground surface. Rationale for these sampling depths is two fold: to support human receptor exposure assessments and assess the leaching potential of constituents in the vadose zone soil to groundwater. Human receptor exposure assessments at industrial properties commonly involve construction worker scenarios that engaged in subsurface excavation for subsurface utility installation/repair and building foundation construction. These workers are frequently exposed to soils in the upper 8 feet of the ground surface.

If chemical impacts are suspected in the fill, a sample will be collected from the impacted zone and analyzed for the previously described analytical suite. A second sample will be collected from the boring exhibiting impact at a depth below the potentially impacted zone. The objective of this sampling strategy is to characterize the vertical extent of potential chemical impacts. This sampling strategy is presented graphically below:

Soil Sampling Rationale



A summary of subsurface soil sampling is provided in Section 2.10 and detailed in the QAPP presented in Appendix E.

2.5.3 Surface Soil Sample Collection

Although Site access is restricted by RG&E, it is possible for Site workers and trespassers to come in contact with surface soil. Therefore, Site characterization will include an assessment of chemical constituents in surface soil. The majority of the Site is paved or covered with dense, crushed stone. Surface soil samples will be collected from areas of the Site where asphalt pavement or crushed stone does not cover the ground surface. The surface soil samples will be collected from the upper 2-inches of soil beneath the sod layer (if present) at 5 locations across the Plant Area. An additional 4 samples will be collected from the sloped land between the MGP process areas and the river's edge (east of the retaining wall) at locations shown on Figures 9A and 9B. Samples will be collected in accordance with the FSP presented in Appendix D and analyzed for TCL VOC, TCL SVOCs, PCBs, TAL inorganics including total cyanide. The results will be compared to constituents detected as background in the City of Rochester (e.g. EPRI 2003 background data) or other appropriate background values in the literature.

2.6 GROUNDWATER INVESTIGATION

The investigation of groundwater will include: monitoring well installation; collection of overburden and bedrock groundwater samples for laboratory analysis; identification of separate phase liquids (including dissolved phase and NAPL); and characterization of hydrogeologic conditions that affect groundwater flow and chemical migration. Methods of investigation are described below.

2.6.1 Existing Well/Piezometer Assessment

Each of the existing monitoring wells and piezometers installed at the Plant Area during previous investigations will be located and evaluated for usability during the RI. It is anticipated that wells installed during the 2006 oil-tar separator investigation (Ish, 2006) will be usable for the assessment of overburden groundwater quality during the remedial investigation. These wells include: MW201, MW202, and RW01. Piezometers installed by AES/META (1993) are anticipated to be used to monitor heads in the overburden to support the assessment of groundwater flow direction in the overburden. These piezometers include: P32, P35, P36, and PCST-1.

The well integrity will be evaluated by inspecting the surface seal and sounding the bottom depth of the well to compare the measured depth with well installation records. If the well is considered potentially usable based on this inspection, the well will be purged and its recovery gauged. If the rate of recovery is consistent with the geologic description contained in the well log, and the purged water does not contain suspended solids that would be considered too large to pass through the well screen (i.e., coarse-grained materials suggest the well screen may not be intact), the well will be considered in Appendix D. Monitoring wells and piezometers determined to be unusable will be decommissioned by overdrilling and tremie-grouting the borehole or backfilled with bentonite pellets.

2.6.2 Groundwater Monitoring Well Installation

Monitoring wells will be installed in the overburden and bedrock. As conceptualized, the location and rationale for installing overburden and bedrock monitoring wells is provided below. The actual well locations will be based on observations made in the field and as discussed with the Department's PM and/or on-site project representative.

Overburden Monitoring Wells:

Monitoring wells will be installed at select soil boring locations across the Plant Area. The Genesee River is an area of regional groundwater discharge and Site groundwater will ultimately discharge to the River. Although the proximity of the Site to the Genesee River strongly influences the groundwater flow direction, man made structures such as the retaining wall and foundations and piping constructed below the water table are expected to locally affect the direction of on-Site groundwater flow. AES/META (1993) speculated that the retaining wall constructed in the north Plant Area diverts some overburden groundwater in a southerly direction. The selected monitoring well locations for this RI will assess the direction of groundwater flow in the overburden and bedrock, as well as, groundwater quality.

Monitoring wells installed to characterize groundwater quality and groundwater flow direction will be located between the river and the former MGP process areas and within/near the former MGP process areas. Monitoring wells may also be located in areas where NAPL is encountered in soil borings to assess separate phase liquid mobility. An estimated 19 overburden monitoring wells are proposed to be installed at locations shown on Figures 10 A and 10B. The depth to groundwater measurements presented in previous investigation reports indicate that the lower 10 feet of overburden material is saturated. Therefore, overburden monitoring wells will consist of 10-foot screen lengths with screens set at the bottoms of the borings. A sump will be added to the base of well screens where NAPL is encountered in soil borings. The monitoring wells will be installed and developed in accordance with the FSP presented in Appendix D. Each monitoring well (new and existing) will be developed and sampled as described in Section 2.6.3.

Bedrock Monitoring Wells:

Descriptions of bedrock core from previous investigations identified sheens and petroleum-type odors in shallow bedrock. However, unlike the overburden, a site conceptual model describing groundwater flow and contaminant distribution has not been established. Therefore, a dynamic decision making process is needed during investigation activities to properly characterize Site bedrock as a site conceptual model is developed. This strategy will provide flexibility in identifying appropriate data collection methods. The approach described in this work plan involves the collection of a sufficient amount of bedrock characterization data that should allow development of a preliminary site conceptual model. The preliminary site conceptual model will be strengthened with data from additional bedrock investigation to satisfy RI objectives.

The investigation of bedrock will be completed in two consecutive phases at the Site. The first phase will involve drilling three exploratory bedrock core holes and installation of multi-level monitoring systems to: assess bedrock stratigraphy, assess NAPL presence/absence in bedrock beneath the Site; locate transmissive zones in the bedrock to identify discrete zones to be monitored by well screens; and install an appropriate depth discrete multilevel monitoring system to measure heads (calculate vertical gradients as wells as triangulate horizontal flow directions) and collect groundwater samples vertically in the bedrock. The second phase of bedrock investigation will involve bedrock well installation at various on-Site locations to monitor depth-specific zones in the bedrock for flow and water quality. The locations and depth-specific zones to be monitored will be identified after the evaluation of hydraulic data and potential NAPL presence during the first phase of bedrock investigation.

Phase I Bedrock Investigation: The three exploratory bedrock core holes will be drilled at locations shown on Figure 10A and 10B. The core holes will be advanced through the Rochester Shale and deeper bedrock formations to the top of the Reynales Formation estimated to be approximately 100 feet below Site grade (see Figure 7 for bedrock stratigraphy) using an HQ core barrel. During coring, the rock core and drilling return water will be examine closely for NAPL. Assuming NAPL is not visually identified in the rock core and/or the drilling return water, the core hole will be advanced through the deeper bedrock formations into the upper Reynales Formation. However, if NAPL is observed in the bedrock, advancement of the core hole will be terminated at the top of the next bedrock formation having different rock type and/or hydraulic characteristics. The core hole would then be reamed with a larger diameter roller bit and a steel casing grouted in-place to minimize the potential downward migration of NAPL during further core hole advancement. After allowing the grout to set for a minimum of 12 hours (overnight), the core hole will continue to be advanced. Description of the core, including the Rock Quality Designation (RQD), will be completed as described in the FSP (Appendix D). Core samples will be retained in wooden core boxes for future reference. It is anticipated that the bottom of the core hole will extend approximately 90 to 100 feet below grade.

Packer testing using a double packer assemblage will be conducted to facilitate the identification of discrete zones to be monitored with multi-level well screens. Depending upon the degree of fracturing and drilling water loss records during coring, the packer assemblage will isolate 5 to 10 foot zones in the core hole. Packer tests will be conducted by potable water injection into the isolated portion of the bedrock at various pressures until the water injection rate remains constant. Results of the packer injection tests will be used to quantify the hydraulic character of the isolated zones in the bedrock. Packer test protocols are included in the FSP presented in Appendix D.

It is conceptualized that four of the most transmissive zones identified by packer testing of the core hole will be screened in a single borehole. A multi-level monitoring system will be constructed in the borehole using one-inch diameter PVC risers and screens. The installation of the multi-level monitoring system can occur only if NAPL is not encountered in both shallow and deep bedrock which would require the use of casings to seal-off zones containing NAPL. Prior to installation, the borehole will be pumped to remove drilling water lost to the formation during drilling and packer testing. The volume of water to be removed will be field determined but is expected to be not less than the volume of water lost to the formation during drilling and packer testing and will include the volume of water standing in the borehole. Monitoring system screen lengths will be determined after evaluating rock coring data, water loss records, and packer testing results. Screen lengths may range from 2 to 10 feet in length. Multi-level monitoring system construction methods are included in the FSP. If casings are used to seal-off NAPL containing fracture zones, individual bedrock monitoring wells will be installed with uncased monitoring intervals open to NAPL containing fractures in the bedrock.

The multi-level monitoring systems (or individual bedrock wells) will be sampled approximately two weeks after installation and analyzed for TCL VOCs, TCL SVOCs, PCBs, and TAL inorganics including total cyanide. The results will be used to progress the next phase of bedrock investigation.

Phase II Bedrock Investigation: Additional bedrock monitoring wells will be installed to assess the nature and extent of MGP impacts in bedrock beneath the Site and provide additional data for hydrogeologic characterization. The additional bedrock wells will be installed to monitor depth-specific intervals identified as target zones for monitoring from the Phase I bedrock investigation. An addendum to the RI Work Plan will be prepared in

a project memorandum and submitted to the NYSDEC for approval that describes the rationale and locations for additional bedrock monitoring wells. The wells will be installed after concurrence from the Department is obtained.

2.6.3 Groundwater Monitoring Well Development, Sampling and Hydraulic Tests

Each monitoring well will be developed in accordance with well development methods described in the FSP presented in Appendix D. Two rounds of groundwater samples will be collected from overburden and bedrock monitoring wells. Presence and accumulation of NAPL, if any, will be measured and water levels will be recorded during groundwater sampling events. Sample collection procedures are described in the FSP presented in Appendix D. The groundwater samples will be analyzed for TCL VOCs, TCL SVOCs, PCBs, TAL inorganics including total cyanide, and field measured parameters. The analytical parameter list is summarized in Section 2.10 and detailed in the QAPP presented in Appendix E. Slug tests will be conducted in each well to provide estimates of hydraulic conductivity. Slug test methods and data analysis are described in the FSP presented in Appendix D.

2.7 NEAR SHORE SEDIMENT PROBING AND SAMPLE COLLECTION

Near shore sediment samples will be collected from the Genesee River approximately 5 feet from the shoreline where water levels are less than 3 feet deep and flow velocities are sufficiently low to pose minimal safety risk to field personnel. The samples will be collected in the general locations shown on Figure 11. Sampling locations extend from the northern property line to the Platt Street Bridge and are spaced at an approximate 100 foot sample interval. Sample locations may be adjusted as appropriate to correspond with Site-related piping discharge points (former and/or active) from the Site to the river. At each location, the upper six inches of sediment will be sampled. The sediment samples will be analyzed for TCL VOCs, TCL SVOCs, MGP chemical fingerprint, PCBs, TAL inorganics including total cyanide, total organic carbon (TOC) and grain size. At approximately half of the locations, pore water samples will be collected by advancing a screen length of 1-foot length) into the shallow sediment (upper 18-inches), purging of 3 well volumes using a peristaltic or inertial pump at a very low flow rate, and collection of a representative sample of water flowing through the sediment pore space.

In addition to the sediment sampling, the near shore sediment will be probed with a steel rod at 50 foot intervals to identify locations that produce a sheen on the surface water.

Probing will extend as deep as feasible (i.e., top of bedrock or minimum three feet below the top of sediment). Where sheens are observed, a sediment sample will be collected from the upper six inches of sediment and a second sample collected from 18 to 24 inches, if possible. The samples will be analyzed for the same parameter list above.

Sediment probing and sample collection procedures are described in the FSP presented in Appendix D. The analytical parameter list is summarized in Section 2.10 and detailed in the QAPP presented in Appendix E.

2.8 INVESTIGATION DERIVED WASTE

Excess soil derived from intrusive field activities (investigation derived waste) will be containerized in DOT-approved 55-gallon steel drums. Groundwater derived from field activities will be containerized in an appropriately sized tank. Containers will be appropriately labeled with the contents, location, and date and staged on-Site at a location selected by RG&E for off-site transportation and disposal. Container contents will be sampled and analyzed for disposal characterization.

2.9 SAMPLE LOCATIONS AND SURVEY

Subsequent to the completion of field activities, a surveyor will field survey sample locations and any other pertinent locations. Ground surface elevations and appropriate reference elevations will be obtained and tied to an appropriate datum.

2.10 Environmental Sample Analyses

Environmental samples collected during the supplemental RI will be submitted to a New York State Department of Health (NYSDOH) accredited laboratory. At this time, TestAmerica (formerly known as Severn-Trent Laboratories) located in Amherst, New York is conceptualized to perform laboratory analyses, however the exact laboratory will be determined at the time of the field activities based on availability and analysis turn around time. Analytical methods, sample handling, chain of custody procedures, and laboratory protocols are described in QAPP presented in Appendix E. Sample analyses will be conducted in accordance with NYSDEC Analytical Services Protocol (ASP). Field and laboratory QC samples are described in the QAPP. Analytical results generated by the laboratory will be reported using NYSDEC ASP Category B data deliverables. The laboratory-generated data will be validated by Geomatrix or a qualified third party and evaluated to assess achievement of data quality objectives described in the QAPP. A Data Usability Summary Report (DUSR) will be prepared to summarize these findings. The table below presents a summary of environmental sample anticipated to be collected during the RI.

Sampling Activity	Anticipated Laboratory Analyses	Estimated Number of Samples to be collected*
Subsurface Soil Samples from Test Pits	TCL VOCs, TCL SVOCs, PCBs, TAL Metals, Total Cyanide	7
Subsurface Soil Samples from Borings	TCL VOCs, TCL SVOCs, PCBs, TAL Metals, Total Cyanide	86
Surface Soil Samples	TCL VOCs, TCL SVOCs, PCBs, TAL Metals, Total Cyanide	9
Groundwater Sampling: Overburden Wells Bedrock Wells	TCL VOCs, TCL SVOCs, PCB, TAL Metals, Total Cyanide, Field Measured Parameters	44 overburden groundwater samples (2 rounds of 22 overburden wells – 3 existing & 19 new) # not defined for bedrock wells (2 rounds – w/number of samples based on number of wells installed)
Sediment Sampling: Sediment and Pore Water	<u>Sediment:</u> TCL VOCs, TCL SVOCs, MGP chemical fingerprint, PCBs, TAL Metals, Total Cyanide, TOC, Grain Size <u>Pore water</u> : TCL VOCs, TCL SVOCs, PCBs, TAL Metals, Total Cyanide	6
Investigation-Derived Waste Soil/Water	Waste Characterization Sampling as required by the disposal facility	1/1

Note: Field Measured Parameters - Dissolved oxygen, Redox, pH, temperature, conductivity

*The sample numbers shown do not include QA/QC samples.

3.0 DATA EVALUATION AND RI REPORT

Site characterization data from the RI and previous investigations will be compiled and interpreted to: 1. establish the extent of MGP and non-MGP material that may be encountered at the Site; 2. assess the nature and extent of chemical constituents detected in Site media with delineation of impacts that exceed regulatory criteria as needed support analysis of potential remedial alternatives; and 3. complete a qualitative exposure pathway assessment for human and ecological receptors. An enhanced conceptual site model will be developed integrating Site hydrogeologic conditions, chemical constituent presence in Site media, and potential migration pathways.

The exposure pathways analysis will address potential risk to human and ecological receptors utilizing the RG&E property and surrounding properties. The analysis will include characterizing the exposure setting (including the physical environment and potentially exposed populations) and identifying the means by which an individual may be exposed to constituents originating from the Site.

A RI Report will be prepared after evaluation of the site characterization data. The report will include the following information and documentation:

- A description of the Site in its current state;
- A summary of field activities used to characterize the Site during the RI with a discussion of variances from the scope of work described in this Work Plan and reference to work plan addendums;
- A Data Usability Summary Report;
- Laboratory reports will be included on a CD in an Appendix
- Maps showing RI sample locations and other information such as maps showing areas of source material and constituent distribution in soil and groundwater, hydrogeologic cross-sections, overburden and bedrock potentiometric surface maps, vertical profile of constituent concentrations in bedrock groundwater, and other maps beneficial toward data summary and comprehension;
- Tabulated data summaries with comparison to regulatory guidance criteria and values;

- Soil boring logs, test pit excavation logs, monitoring well construction diagrams, laboratory analytical reports, sampling information and other supportive Site characterization data;
- Discussion of and conclusions regarding the extent and nature of environmental impact in the various media being investigated;
- A conceptual site model describing constituent distribution of chemical constituents of potential concern at the Site in various media;
- Completion of an exposure pathways assessment to qualitatively assess Human Health Risks and complete a Fish & Wildlife Impact Analysis through Part 1 Resource Characterization of Draft DER-10; and
- Recommendations regarding the performance of Quantitative Risk Assessment, if considered necessary, and remedial alternatives analysis.

4.0 SCHEDULE

Following agency approval of the Work Plan, the activities described in the preceding sections will be implemented. The Department will be provided with two weeks notice before initiating the first on-Site work activity and will be provided with periodic updates regarding field activity scheduling during the course of the investigation. A target project duration schedule is provided below.

<u>RI Activity</u>	Target Duration
Work Plan Approval	Written Notice from Agency
Implementation of Field Activities	<u>16 weeks ⁽¹⁾</u>
Laboratory Sample Analysis	<u>12 weeks</u>
Data Validation	<u>6 weeks</u>
Field/Lab Data Compilation-Interpretation	12 weeks
Preparation of Draft RI Report	20 weeks

Notes:

(1) Includes Phase I and Phase II Bedrock Investigations.

5.0 **REFERENCES**

Boring logs from Rochester Drilling Company, Inc. Investigation (1977)

Boring logs from Empire Soils Investigations, Inc. Borings (1985 and 1986)

EPRI, 2003, *Polycyclic Aromatic Hydrocarbons (PAHs) in Surface Soils in Western New York: Follow-up Report for New York State Soils*. EPRI, Palo Alto, CA, New York State Electric & Gas Corporation, Binghamton, NY, Rochester Gas and Electric Corporation, Rochester, NY, and Public Service Enterprise Group Services, LLP, Newark, NJ: 2003. 1005296.

Preliminary Site Review, NYS Superfund Site No. 828044, prepared by Morrison-Knudsen Engineers, Inc. in June 1986

Characterization of the Former Manufactured Gas Plant (MGP) at the RG&E West Station Site, prepared by META Environmental, Inc. and Atlantic Environmental Services, Inc. dated March 1993

Interstate Technology & Regulatory Council (ITRC) Dense Nonaqueous Phase Liquids Team September 2003 An Introduction to Characterizing Sites Contaminated with DNAPLs

Observation Report, Riverbank Survey, Former West Station MGP Site, prepared by Ish Inc. dated January 2004

NYSDEC, 2002, Draft DER-10, Technical Guidance for Site Investigation and Remediation.

NYSDEC, 1999, Technical Guidance for Screening Contaminated Sediments

NYSDEC, 1991, Analytical Services Protocol.

Parker, B.L., Gillham, R.W., and Cherry, J.A., 1994. "Diffusive Disappearance of Immiscible-Phase Organic Liquids in Fractured Geologic Media" *Groundwater*, 32(5):805-20.

Puls, R.W. and Barcelona, M.J., 1996, USEPA Groundwater Issue Report "Low-Flow (Minimal Drawdown) Groundwater Sampling Procedures.

Riha, B.D., J. Rossabi, C.A. Eddy-Dilek, D. Jackson, and C. Keller, 2000. DNAPL characterization using the ribbon NAPL sampler: Methods and results. In *Proceedings of the Second International Conference on Remediation of Chlorinated and Recalcitrant Compounds*, 33.40. Columbus, Ohio: Battelle Press

USEPA, 1986, Test Methods for Evaluating Solid Waste Physical/Chemical Methods. SW-846 Third Edition. Office of Solid Waste, USEPA, Washington DC.

USEPA, July 2001, *The State-of-the Practice of Characterization and Remediation of Contaminated Ground water at Fractured Rock Sites* EPA542-R-01-010 July 2001 www.epa.gov/tio www.cluin.org
USEPA. 2001, Fractured Bedrock Focus Area. http://cluin.org/databases/#Fractured_Bedrock_Focus_Area

APPENDIX A Sanborn Maps























(







ĺ











APPENDIX B Boring Logs

APPENDIX B-1 Ish Inc. Oil Tar Separator Investigation

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/5/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Harry Lyon DRILLING METHOD: Direct Push SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: 414.496 ft. above MSL WELL ELEVATION: NA OUTER CASING ELEV.:NA DEPTH TO WATER:NA BOREHOLE DEPTH: 20.8 ft. below grade WEATHER: Cool, overcast, breezy GEOLOGIST: Amanda Bissell

> sh nc.

Page 1 of 1

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	WELL CONSTRUCTION	REMARKS	
		1	1	[1 14 14	-0		1
E1			0.0		sand and silt with some red clay	1		
2	0-4	75	0.0		and root material	2		
<u> </u> [3			0.0		FILL: Dry to moist, black to gray			
E -4			0.0		like material (CLM), brick pieces,			
-5			0.0		and other debris			
E 6	4-8	80	0.0			6		
E			0.0			7		
E -8			0.0			8		
	8-12 25		0.0			9		
		25	0.0			- 10		
E 11		2.5	0.0			_ 11		
			0.0		FILL: Dry to moist, black to gray			
			0.0		fill with silt, sand, gravel, coal-	12		
	10.10		0.0		and other debris	13		
E -14	12-16	40	0.0	40.000		14		
E -15			0.0		Ell L: Majot brown to rod longe	15		
⊢ -16 E			0.0		medium sand and gravel with	16		
⊢ -17 F			0.0		CLM and trace clay	17		
-18 -	16-20	50	0.0			18		
E -19			0.0	19-20		19		
-20	20-20.8	20	-			20		
1	J				WEATHERED BEDROCK: Moist, light gray, compact fine to medium sand and silt with broken fragments of weathered shale. Refusal on bedrock at 20.8'.			

PID (ppm) = Photoionization Detector, readings in ppm	
(ppin) i notoionzation beteetel, redainge in ppin	

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/5/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Harry Lyon DRILLING METHOD: Direct Push SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: Not Surveyed WELL ELEVATION: NA OUTER CASING ELEV.:NA DEPTH TO WATER:NA BOREHOLE DEPTH: 27.5 ft. below grade WEATHER: Cold, snow flurries, overcast GEOLOGIST: Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	% RECOVERY	(mqq) OIq	ANALYTICAL SAMPLE	SOIL DESCRIPTION a LITHOLOGIC SYMBO	nd L	WELL CONSTRUCTION	REMARKS	
0 1			0.0		TOPSOIL: Moist, compact, brown, fine sand with silt, clay and gravel				
- 	0-4	75	3		FILL: Dry to moist, loose, brown, medium sand with clay, silt and gravel		****		
			5		FILL: Moist, semi-compact, black, fine sand with clay and silt, some gravel and sheens				
- - 			10 		FILL: Moist, sticky, compact, black, fine sand with some				
	4-8 S	90	51	51	gravel and silt, NAPL and a tar- like odor present throughout interval FILL: Moist, semi-compact, brown, fine to medium sand and gravel with shale fragments, and NAPL				
			54						
		70	9	FILL: Moist, compact, brown, fine sand with silt, clay, shale fragments and some weathered					
	8-12	70	4		brick, slag-like material throughout				
			5	u conte e e e e e e e e e e e e e e e e e e					
	12-16	35	10						
	12-10		0.0		FILL: Same as the above unit with a very slight, tar-like odor and NAPL between 14 and 15.5'				
L -16			4			10 <u>4</u> /			

sh nc. META Page 1 of 2

WS-SB02 SAMPLING METHOD:4' Geoprobe Macro-cores PROJECT: West Station - RG&E GROUND ELEVATION: Not Surveyed PROJECT NO: 103039 WELL ELEVATION: NA LOCATION: Rochester, NY OUTER CASING ELEV .: NA DATE: 4/5/2006 DEPTH TO WATER:NA DRILLING CONTRACTOR: Lyon Drilling BOREHOLE DEPTH: 27.5 ft. below grade DRILLER: Harry Lyon WEATHER: Cold, snow flurries, overcast **DRILLING METHOD: Direct Push GEOLOGIST: Amanda Bissell** CONSTRUCTION NTERVAL (ft.) ANALYTICAL SAMPLE RECOVERY DEPTH (ft.) REMARKS (mqq) OI9 SOIL DESCRIPTION and SAMPLE LITHOLOGIC SYMBOL WELL % -16 FILL: Moist, compact, brown, 3 2 medium to fine sand with silt, -17 trace clay, gravel, coal-like material (CLM), shale fragments 5 and weathered brick 16-20 70 - -18 -18 9 -19 -19 14 -20 -20 15 -21 -21 5 -22 20-24 90 .22 11 FILL: Moist, semi-compact, -23 -23 black, fine to coarse sand with $^{\circ}$ 9 black, viscous, sticky, tar-like material throughout interval

	sh nc.	NET/
PID (ppm) = Photoionization Detector, readings in ppm	P	,age

FILL: Wet, loose, black, fine to coarse sand with clay, gravel

FILL: Wet, loose, black, fine to

medium sand with some gravel and NAPL present throughout. Refusal at 27.2' - likely on

and shale fragments, tar-like

odor and NAPL present

throughout interval

bedrock.

-24

-25

-26

-27

24-27.2

75

12

16

17

25-26

\$Q 2 of 2

-24

-25

-26

 $^{\circ}$

WS-SB03/RW01

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/6/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Craig Brown SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: Not Surveyed WELL ELEVATION: Not Surveyed OUTER CASING ELEV.:415.89 ft. above MSL DEPTH TO WATER:average=22.82 ft below grade BOREHOLE DEPTH: 27.5 ft. below grade WEATHER: Cloudy, cool, light breeze GEOLOGIST: Amanda Bissell

DRILLING METHOD: Direct Push/Recovery Well Drilled w/10" HSAs GEOLOGIST: Amanda Bissell

DEPTH (î .)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL		WELL CONSTRUCTION	REMARKS
-2						2		Steel outer casing
-0			0.0		FILL: Moist, loose, dark brown,	-0		
1 			0.0		FILL: Dry, loose, black very fine	1		Bentonite grout
2	0-4	50	0.0		clasts	2	000 000	
3			0.0				000	
			0.4				000	
6	4-8	60	0.0					6-inch PVC riser
- 7			7.0		silt and fine sand with fractured shale clasts	-7		
			0.0			8		
9		na na mang wana na mang	0.0			9		
-10	8-12	20	0.0			10	0000	
-11		and a local processing to the second s	0.0					
-12		and an and the second	0.0			12		Bentonite seal
-13			0.0			13		
-14	12-16	20	0.0	*A ver ver ver ver ver		14		Sand pack
E -15		1111111111111	0.0	(Investment of the second of the	LNOž4	15		
PID (ppm) =	Photoio	nization (Detector, rea	idings in ppm		nc. META	I of 2

WS-SB03/RW01

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/6/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Craig Brown SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: Not Surveyed WELL ELEVATION: Not Surveyed OUTER CASING ELEV.:415.89 ft. above MSL DEPTH TO WATER:average=22.82 ft below grade BOREHOLE DEPTH: 27.5 ft. below grade WEATHER: Cloudy, cool, light breeze

DRILLING METHOD: Direct Push/Recovery Well Drilled w/10" HSAs GEOLOGIST: Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (mad)	ANALYTICAL SAMPLE	SOIL DESCRIPTION a	nd L	WELL CONSTRUCTION	REMARKS			
-15 			2.6		FILL: Moist, compact, brown, fine sand and clay with fractured shale clasts, coal-like material (CLM) and slag-like material						
	16-20	20	0.0								
19		Y MY TA THE	0.0		FILL: Moist, compact, brown, fine sand and clay with fractured shale clasts			0.020-inch slotted screen			
			12					Sand pack			
22	20-24	55	55	55	55	53	22.5-23.5	FILL: Moist, loose, gray shale fragments and gray clay with NAPL and sheens			Water level (7/25/06) Water level (5/24/06)
24			113		FILL: Shale clasts with fine, black sand and black, sticky, tar- like material with a tar-like odor	0 4 - 24		Water level (4/28/06)			
25			87		FILL: Moist, loose, tan, gravel with fine sand and clay	-25 					
-26	24-27.5	90	16		FILL: Wet, loose, black, medium to coarse sand, gravel, trace						
			13	an distant and the second s	clay and blick with NAPL, CLW, sheens and a mix of tar-like and fuel-like odors	-27					
-29		v po po po na contra da contra			FILL: Moist, compact, very fine sand with trace clay and a slight tar-like odor. Direct push refusal at 27.5'			2-feet x 6-inch PVC sump			
t 20	1	1	1	1		L					

PID (ppm) = Photoionization Detector, readings in ppm

Page 2 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/24/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:414.663 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 22 ft. below grade WEATHER: Cool, light breeze, drizzling rain, overcast GEOLOGIST:Amanda Bissell

WELL CONSTRUCTION RECOVERY £ ANALYTICAL SAMPLE Blow Count (ff) SAMPLE INTERVAL (REMARKS (mqq) DEPTH SOIL DESCRIPTION and LITHOLOGIC SYMBOL * 0 -0 ASPHALT: Auger through asphalt 0.0 58 H-Weight of \Diamond 0-2 11,17,15 75 -1 Hammer 0.0 FILL: Dry to damp, light brown and gray, fine sand with clay, some shale -2 0.0 clasts, weathered brick, gravel and 8,15,8,8 75 -3 -3 2-4 coal-like material (CLM) 0.0 -4 0.0 5,6,5,7 45 5 -5 4-6 0.0 -6 -6 0.0 6.7-7 F 50 -7 -7 6-8 6.13.6.7 0.0 -8 -8 0.0 $^{\circ}$ -9 -9 8-10 5,3,5,8 45 0.0 -10 -10 FILL: Dry, loose, light gray, medium to \diamond 0.0 coarse sand with angular clasts of 10-12 5,4,3,9 40 10 -11 -11 shale 0.0 10 -12 -12 FILL: Dry to damp, loose, black, \diamond 0.0 medium to fine sand with slag-like Δ 4,2,2,3 50 -13 -13 12-14 material, trace fine tan sand with 0.0 11 natural iron staining -14 -14 0.0 FILL: Moist, compact, light brown, 1 50 -15 -15 14-16 2,2,2,5 medium sand, clay and shale clasts 0.0 with some (CLM) -16 -16 0.0 FILL: Moist, compact, light gray, 16-18 H,H,H,3 60 -17 medium to fine sand with clay and 0.0 sub-rounded gravel -18 -18 0.0 H,3,5,12 40 -19 -19 18-20 0.0 -20 -20 . Not 0.0 WEATHERED BEDROCK: Damp to 14,32,20,50 60 dry, very hard, compact, light brown -21 -21 20-22 0.0 clay with shale clasts throughout, 21.5-22 2 F -22 -22 weathered bedrock in shoe. Refusal on bedrock at 22.0

PID (ppm) = Photoionization Detector, readings in parts per million

WS-SB05/MW201

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/5/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Harry Lyon DRILLING METHOD: Direct Push/ Well Drilled with 4 1/4" HSAs SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: Not Surveyed WELL ELEVATION: Not Surveyed OUTER CASING ELEV.:416.15 ft. above MSL DEPTH TO WATER:average=23.25 ft below grade BOREHOLE DEPTH: 28 ft. below grade WEATHER: Cool, overcast, breezy GEOLOGIST: Amanda Bissell

بور معنی است اور می اورد اور اورد اورد اورد اورد اورد اورد	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOI	nd L	WELL CONSTRUCTION	REMARKS
						3		Steel outer
			0.0		TOPSOIL: Moist, loose brown, medium to fine sand with gravel		0000	casing
na ja kaj kaj kaj kaj kaj kaj kaj kaj kaj	0-4	50	0.0				<u>ଚ୦୭୦୭୦</u> ୭୦୭୦୭୦	
			0.0		FILL: Dry to moist, loose, gray silt, clay and medium to fine sand with broken shale fragments, coal-like material (CLM), and weathered brick		<u> </u>	Bentonite grout
	4-8	30	0.0				000000 000000	
			0.0		FILL: CLM and slag-like material FILL: Dry, loose, gray shale		0000 0000	2-inch PVC riser
0	8-12	80	0.0		and medium to fine sand			
1	~ 14a		0.0	No man () () () ()	FILL: CLM and slag-like material	0-1-11	5050 1050	
2			0.0	and the second	FILL: Weathered brick FILL: CLM and slag-like		0000	
3		almanul, cu adolera esta foi la referencia	0.0		material, very slight tar-like odor present	-13 		
4	12-16	30	0.0	n () - a granning () - a () -				Bentonite seal

PID (ppm) = Photoionization Detector, readings in ppm

Page 1 of 2

WS-SB05/MW201

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/5/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Harry Lyon DRILLING METHOD: Direct Push/ Well Drilled with 4 1/4" HSAs SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: Not Surveyed WELL ELEVATION: Not Surveyed OUTER CASING ELEV .: 416.15 ft. above MSL DEPTH TO WATER:average=23.25 ft below grade BOREHOLE DEPTH: 28 ft. below grade WEATHER: Cool, overcast, breezy GEOLOGIST: Amanda Bissell

Page 2 of 2

DEPTH (û.)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL		REMARKS
-15			0.0		FILL: Dry, loose, gray, medium		
10			0.0		with broken shale fragments		
1/			0.0				
18	16-20	00	0.0				Sand nack
19	: -		0.0		FILL: Dry to moist, semi-		ound paok
20			1.0		with silt, fine sand, gravel and		0.010-inch
21	20.24	60	4.0				slotted screen
-22	20-24		9.0		FILL: Dry to moist, loose, gray		(7/25/06)
23			4.0		shale fragments, CLM and an aromatic odor, NAPL present in		Water level (5/25/06) Water level
-24			4.0		shoe material		(4/28/06)
25	24.00	6	4.0	24.7-26.5	FILL: Wet, loose, gray silt with trace clay, some shale fragments and brick pieces		
26	24-28	50	3.0		throughout. Unit is very unconsolidated and has NAPL.		
27			3.0	-	Refusal at 26.5' likely caused by encountering bedrock.		Well Cap

nc. PID (ppm) = Photoionization Detector, readings in ppm

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/6/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Craig Brown DRILLING METHOD: Direct Push SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: 413.895 ft. above MSL WELL ELEVATION: NA OUTER CASING ELEV.:NA DEPTH TO WATER:NA BOREHOLE DEPTH: 28 ft. below grade WEATHER: Sunny, cool, light breeze GEOLOGIST: Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	% RECOVERY	(mɑd) Qid	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL		WELL	REMARKS	
0 1			0.0		TOPSOIL: Moist, loose, fine to coarse, brown sand with trace clay, sub-rounded gravel and root material				Non-Advancementaries and the state of the st
2	0-4	60	0.0		FILL: Moist, loose, fine to coarse, brown sand with trace clay, sub-rounded gravel and				
			0.2		coal-like material (CLM)				a management of the second space of the
		4-8 50 0.0 0.0 0.0 0.0	0.0						nan a magana na manana ing manana na ma
6	4-8						ale cale and a second		
7			0.0	FILL: Moist, loose brown, medium to coarse sand with large angular rock clasts, fine				an a	
			0.0		gravel, and hard, cement-like aggregate between 7.35 and 8 ft bgs				and we want to be a state of the
	8-12	40	0.0						
			0.0						
12			0.0	resource in the second seco	FILL: Moist, loose, brown and black clay with medium to coarse sand, CLM, weathered brick shale fragments and				
13		0.0		gravel		A CALIFORNIA A LA CALIFORNIA A			
	12-16	12-16 30	0.0				none version and the second		
			0.0			10 <u>°</u> 4	And and an and a second se		
		***					sh nc.	18#Ringensile_	

PID (ppm) = Photoionization Detector, readings in ppm

Page 1 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/6/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Craig Brown DRILLING METHOD: Direct Push SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: 413.895 ft. above MSL WELL ELEVATION: NA OUTER CASING ELEV.:NA DEPTH TO WATER:NA BOREHOLE DEPTH: 28 ft. below grade WEATHER: Sunny, cool, light breeze GEOLOGIST: Amanda Bissell

DEPTH (A.)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (mqq) Olq	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL		WELL	REMARKS	
			0.0		FILL: Moist, loose, brown and black clay with medium to coarse sand, CLM, weathered brick, shale fragments and	-15 			
			0.0		gravel				
17	16-20 2	25	0.1						
18		25	0.2						
L -20			0.2		FILL: Moist, loose, light brown and white clay with some silt,				
		an off the second s	19		medium to coarse sand and large shale clasts				
	20.04	24 60 28							
	20-24		28		FILL: Wet, loose, brown and gray clay with medium to coarse sand CI M, weathered brick				
	ar management of the second	and the same of the sam	20		shale fragments and gravel, fuel- like odor present				
	24-25.6	90	163						
25 			52	24.6-25.6					
	25.6-28	50	0.0	WEATHERED BEDROCK: Wet, loose, gray, weathered shale with trace clay and trace medium to coarse sand, boring					
			0.0		terminated in bedrock at 28.0	0300 -27 0300 -28			

Page 2 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/6/2006 DRILLING CONTRACTOR: Lyon Drilling DRILLER: Craig Brown DRILLING METHOD: Direct Push SAMPLING METHOD:4' Geoprobe Macro-cores GROUND ELEVATION: 413.777 ft. above MSL WELL ELEVATION: NA OUTER CASING ELEV.:NA DEPTH TO WATER:NA BOREHOLE DEPTH: 20.7 ft. below grade WEATHER: Cool, light breeze, cloudy GEOLOGIST: Amanda Bissell

DEPTH (tt.)	SAMPLE INTERVAL (ft.)	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL		WELL CONSTRUCTION	REMARKS	
E ⁻⁰			0.0		FILL: Dry to moist, loose, brown,				
1			0.0		medium to coarse sand with clay, sub-rounded gravel, shale				
2	0-4	60	0.0		fragments and weathered brick				1
3			0.0			04F-3			
E-4			0.4				Character in the second s		
5			04						
6 E	4-8	70	02						
E-7			02						
			0.0		FILL: Dry to damp, hard, tight,		versam filter every store engine		
-9		95	0.0	Ye have been a second and a second a	brown, medium to coarse sand with sub-rounded gravel, shale				
-10	8-12		0.0		clasts and weathered brick				
E 11 -			0.0			10 % - 11			
E -12			0.0	91 - 11 - 12 - 12 - 12 - 12 - 12 - 12 -		-12 	real and a second s		
F -13			0.0			-13			
E -14	12-16	90	0.0	And the second		10×4-14	ver and the second s		
15			0.0		FILL: Dry to damp, brittle, loose, black, coal-like material (CLM).	04-15	an ang ang ang ang ang ang ang ang ang a		
-16			1.5		with slag-like material, and a slight fuel-like odor	-16			
E -17		A for a first state part of the	85	17 deservatives were too			and the state state of the stat		Non-
E -18	16-20	40	25	- And an and a second		104 - 18	And a second		
E -19	Non-real and the second se		2.5 FILL: Moist, brittle, black CLM						
E -20	20-20.7	2 2 20 7 95 1 20-20 7	clay and fine to medium sand and gravel with a fuel-like odor	R. 00/4 -20	an owner and the second se				
- -	<u></u>	1	`	1	Direct push refusal at 20.7'		ł		1

 PID (ppm) = Photoionization Detector, readings in ppm
 Pid Mathematical Action of 1

WS-SB07A

PID

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:413.967 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION:NA DEPTH TO WATER: NA BOREHOLE DEPTH: 28 ft. below grade WEATHER: Warm, breezy, clear sky GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	1	WELL CONSTRUCTION	REMARKS
-20	20-22	5,4,4,6	40	300		SB07A is a continuation of SB07 which hit refusal before bedorck was reached. The location of SB07A is about 15 feet west of SB07. Augers were advanced to 21' below ground surface where sampling began.	20		
- - 22				300		FILL: Moist to wet, loose, gray clay with fine to medium sand, shale clasts and a very slight fuel-like odor			
23	22-24	4,10,9,10	35	571	23-24	FILL: Wet, loose, black stained clay			
- 24 - -				0.0		sheen a few NAPL globules between 23.3 and 23.6, fuel-like odor present FILL: Wet, cohesive, gray clay with medium sand, weathered brick and shale clasts			
25 - - - 26	24-26	26 Not recorded	led 20	0.0					
27	26-28	Not recorded	>5						
L -28				l		WEATHERED BEDROCK: Wet, gray clay and shale clasts. Boring terminated in bedrock at 28.0'	02201 -28		

	Sh META KRammeria In-
(ppm) = Photoionization Detector, readings in parts per million	Page 1 of 1

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/25/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:415.086 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 28.2 ft. below grade WEATHER: Cool, light breeze, overcast

GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	(mqq) Olq	ANALYTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOL	d	WELL	REMARKS
	0-2	1,1,2,3	75	0.0		TOPSOIL: Damp, loose, brown, fine sand with rounded gravel and root material		Very service and a se	H-Weight of Hammer
2				0.0		FILL: Dry, loose, black, coal-like material (CLM), clinker-like material, and ash-like material (ALM)			
3	2-4	3,3,3,4	70	0.0	en og generalet er en	FILL: Damp, cohesive, brown clay with fine to coarse sand and gravel		named and the state of the stat	
	4-6	2,2,2,4	40	0.0		FILL: Dry to damp, loose, black, fine sand with trace brown clay and CLM, ALM, weathered brick and slag-like material		de forção formada do alta do a	
7	6-8	2,3,2,1	25	0.0	n gyr			n management of management of the state of the	
	8-10	1,1,H,H	50	0.0				la mangana mang	
	10-12	1,1,H,1	70	0.0					
	12-14	1,2,2,2	25	0.0					
	14-16	1,4,14,7	50	0.0	dumunum dumunum dumunum du	FILL: Dry to moist, loose, gray and brown, medium to coarse sand with clay, gravel, shale clasts, weathered brick and CLM		The manufacture and the state of the state o	
, -10 [1			· · · · · · · · · · · · · · · · · · ·			sh	HETA \$	to summer in

	nc. permana comm
PID (ppm) = Photoionization Detector, readings in parts per million	Page 1 of 2
PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/25/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:415.086 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 28.2 ft. below grade WEATHER: Cool, light breeze, overcast GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	ł	WELL	REMARKS	
-16	40.40	50744	70	874					H-Weight of Hammer	the first and the first state of
E -1/	16-18	5,3,7,14	70	275		FILL: Moist, cohesive, clay with fine, yellow and brown sand, small angular				
18 - -				383		gravel, CLM and shale clasts	O2118			
19 -	18-20	4,3,4,8	60	425		HLL: Moist, loose, black, fine sand with a fuel-like odor				
20				7654	20-21.6	FILL: Wet to moist, loose, black, coarse sand and shale clasts with NAPL present, sheens and a tar-fuel-	-20 			or way for the second se
'21	20-22	5,9,10,7	65	913	9) 	cil mix odor				
22 	*****			615		like material (TLM) that is very viscous, sticky, and black with sheens				
23	22-24	12,15,18,8	80	455	serving locate communications	FILL: Wet, black, silty material with coarse sand and shale clasts with				
				913		FILL: Dry, brown, fine sand with shale	ion i			
25 -	24-26	5,5,2,9	70	2001		clasts, slight tar-like odor	02/0025			
-26				325	26-27.2	HILL: Dry to moist, brown silt with black, sticky, viscous, TLM, some shale clasts and a tar-like odor	-26 04 04			
-27 -	26-28	9,12,10,10	30	175		WEATHERED BEDROCK: Wet, fine, loose, black, sand with shale clasts.	-27 -27			
-28			<u> </u>	175	1979 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -	NAPL and sheens. Refusal on bedrock at 28.2'	0 <u>7</u> -28			
	28-30	10, 50 for 2*	10	170	an and an					

ish Inc. WELL PR Page 2 of 2

WS-SB09/MW202

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/26/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:Not Surveyed WELL ELEVATION:Not Surveyed OUTER CASING ELEVATION:417.821 ft. above MSL DEPTH TO WATER: average=24 ft. below top of casing BOREHOLE DEPTH: 25 ft. below grade WEATHER: Warm, sunny, breezy GEOLOGIST:Amanda Bissell

Page 1 of 2

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANAL YTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOL	d	WELL	REMARKS	
-2							-2		H-Weight of Hammer Steel outer casing	
	0-2	4,6,8,6	70	0.0		TOPSOIL: Dry to damp, loose, gray, fine to medium sand with gravel, root material and trace clay		<u>Jacobol</u> Jacobol		
2	2-4	7,6,7,5	5	0.0		FILL: Darip, black, the sand whet the gravel and natural iron staining FILL: Dry to damp, loose, brown, medium to coarse sand with sub- rounded, fine gravel, broken shale		0000000	Bentonite grout	
	4-6	3,2,1,H	30	0.0		clasts, with trace clay, weathered brick material and coal-like material (CLM) at 7.5		2050505 050505	2-inch PVC	ferrer many a grant a second and a second a seco
-7	6-8	1,2,6,9	20	0.0	and a second			000000 000000	nser	
	8-10	4,5,4,6	40	0.0	en meneral de la meneral est en contracte de la meneral	FILL: Moist to damp, cohesive, loose,		<u>006050</u> 606050		e managemente de la construcción d
	10-12	2,3,7,4	30	0.0	- And a second	clasts and weathered brick			Bentonite seal	
	12-14	2,4,4,11	35	0.0						na la fai de la fai na fai an la fai na f
							Ish Inc.	META PR	<u>(n.c. 1996) (n. 1</u> 00)	

WS-SB09/MW202

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/26/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:Not Surveyed WELL ELEVATION:Not Surveyed OUTER CASING ELEVATION:417.821 ft. above MSL DEPTH TO WATER: average=24 ft. below top of casing BOREHOLE DEPTH: 25 ft. below grade WEATHER: Warm, sunny, breezy GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOL	d	WELL CONSTRUCTION	REMARKS
-14 	14-16	4,11,9,5	30	0.0		FILL: Moist to damp, cohesive, loose, brown, clay with fine sand, shale clasts and weathered brick	-14		
16	16-18	8,5,4,11	30	-					0.010-inch slotted screen
18 18 19	18-20	3,2,3,10	40			FILL: Wet, loose, black, coarse sand with broken shale clasts and a moderate fuel-like odor	04 -18 		
-20					and the Contemporary and the second	FILL: Wet, black, porous, medium to coarse sand with slag-like material, and a moderate fuel-like odor			Sand pack
21	20-22	10,13,7,5	55	150	21-22	FILL: Wet to moist, gray clay with some yellow staining, with medium to	-21 		
	22-24	8,21,23,22	90	250	Amount and the second se	moderate fuel-like odor	04		Water level
-24				2556	and a second and a s	coarse sand with slag-like material, and a moderate fuel-like odor	0001- 0001- 000		(4/28 and 5/24/06)
25	24-26	14, 50 for 3"	50	0.0	and the second second	WEATHERED BEDROCK: Wet to moist, gray clay with angular shale	-25		(7/26/06)
L -26				0.0		WEATHERED BEDROCK: Weathered shale with a fuel-like odor. Refusal on bedrock at 25.0'	-26		PVC sump

sh nc,	META PRESS ACCOUNTS IN
p	ane 2 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/24/2006 **DRILLING CONTRACTOR: Nothnagle Drilling** DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers

SAMPLING METHOD:2' Split Spoons GROUND ELEVATION: 414, 451 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION. NA DEPTH TO WATER: NA BOREHOLE DEPTH: 22.5 ft. below grade WEATHER: Cool, light breeze, drizzling rain, overcast GEOLOGIST:Amanda Bissell

no.

Page 1 of 2

WELL CONSTRUCTION RECOVERY £ ANALYTICAL SAMPLE low Count (mqq) OI REMARKS SAMPLE INTERVAL (DEPTH (ft.) SOIL DESCRIPTION and LITHOLOGIC SYMBOL % ā ā 0 -0 ASPHALT: Auger through asphalt 0.0 FILL: Dry to damp, loose, black, fine 0-2 1,4,9, 30 - 1 sand and sub-rounded gravel with slag-like material and coal-like 0.0 material (CLM) -2 0.0 60 -3 2-4 12,13,14,11 0.0 FILL: Weathered brick, some CLM, -4 shale clasts and ash-like material 0.0 FILL: Dry, compact, light brown, clay, silt, fine sand with shale fragments -5 4-6 4,4,3,2 50 0.0 .8 -6 0.0 6-8 2,2,3,2 21 - -7 0.0 -8 0.0 15 9 - -9 8-10 2,2,3,2 0.0 -10 10 0.0 -11 10-12 2,7,6,6 20 0.0 -12 META & R sh

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/24/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:414.451 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 22.5 ft. below grade WEATHER: Cool, light breeze, drizzling rain, overcast GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	(mqq) Old	ANALYTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOL	d	WELL CONSTRUCTION	REMARKS
-12	12-14	2.7.4.2	45	0.0		FILL: Dry, compact, light brown, clay, silt, fine sand with shale fragments			
-				0.0				ne for far a fan fan fan fan fan fan fan fan fan f	
	14.16	3585	45	0.0				A summary restriction of the second se	
10	14-10	3,3,0,3	*2	0.0				and and a first second and a second	
			76	0.0					
	16-18	2,14,31,15	/5	0.0		FILL: Dry, loose, tan and yellow, medium to coarse, clumpy sand		the and	
18				0.0		FILL: Dry loose, black, brittle CLM with a tar-like odor, some slag-like material and weathered brick		an fer an	
	18-20	6,9,8,10	75	0.0		FILL: Moist, loose, black, medium sand with CLM, ash-like material, and rock clasts with sheens from 20.5- 20.9 tar-like odor present		and a second	
					20.6.20.9		-20		
	20-22	4,12,30,15	70		21-22	WEATHERED BEDROCK: Damp to moist, cohesive, light brown and gray clay with medium to fine sand, with	-21 000000-21	na si sina na mafa ngi kana na	
22 		6, 50 for 2"				snale fragments with a fuel-like odor. Refusal on bedrock at 22.5'	10000F-22		

PID (ppm) = Photoionization Detector	, readings in parts per million
--------------------------------------	---------------------------------

Sh nc. Page 2 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/26/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:414.779 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 24 ft. below grade WEATHER: Warm, sunny, breezy GEOLOGIST:Amanda Bissell

WELL CONSTRUCTION RECOVERY Ę. ANAL YTICAL SAMPLE low Count DEPTH (A.) (mqq) REMARKS SAMPLE INTERVAL SOIL DESCRIPTION and Ω LITHOLOGIC SYMBOL % n $\overline{\mathfrak{m}}$ -0 0 FILL: Dry, loose, brown, medium to H-Weight of 0.0 coarse sand with gravel, rock clasts Hammer and trace clay 50 0-2 9,5,7,5 -1 0.0 FILL: Dry, loose, black, fine sand with -2 -2 slag-like material, fine gravel, rock 0.0 fragments, coal-like material (CLM) and trace clay 3,2,2,3 75 -3 2-4 -3 0.0 4 -4 FILL: Moist, loose, light brown, medium to coarse sand with trace clay 0.0 \triangle and rock fragments 60 4-6 5,6,5,6 -5 - -5 0.0 FILL: Moist, loose, cohesive, brown, medium to fine sand with clay, shale -6 -R clasts, natural iron staining and CLM 0.0 FILL: Dry, hard, loose, fine, brown 6-8 7,6,4,4 30 -7 sand and wilt with shale clasts 0.0 FILL: Damp, cohesive, brown, clay -8 -8 and silt, fine sand and shale clasts 0.0 with CLM and slag-like material 25 8-10 2,4,3,3 -9 . .9 0.0 -10 -10 0.0 10-12 3,2,1,H 25 -11 0.0 .12 -12 H-Weight of 0.0 Hammer 12-14 1,1,H,3 10 -13 17 0.0 10 sh META & Ramanana in

PID (ppm) = Photoionization Detector, readings in parts per million

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/26/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoons GROUND ELEVATION:414.779 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION:NA DEPTH TO WATER: NA BOREHOLE DEPTH: 24 ft. below grade WEATHER: Warm, sunny, breezy GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	(mqq) Olq	ANALYTICAL SAMPLE	SOIL DESCRIPTION an LITHOLOGIC SYMBOL	d	WELL CONSTRUCTION	REMARKS
-14	14-16	4,2,2,2	10	0.0		FILL: Damp, cohesive, brown, clay and silt, fine sand and shale clasts with CLM and slag-like material	-14	and in the foreign and the second	
-16				0.0		FILL: Moist, loose, brown clay,	-16	A state of the sta	
	16-18	1,1,H,1	10	0.0		medium to coarse sand, fine gravel, shale clasts and weathered brick		a for a manufacture of the second	
18 	18-20	2,2,1,H	10	0.0	a granda a service a s			a na manana manana na	
-20				0.0	and an and a second	FILL: Wet to moist, loose, brown, broken shale clasts with clay, silt and fine sand	0 <u>1</u> 0 <u>1</u> -20	an a	
- 	20-22	1,1,H,1	<5	0.0	and and a second se	FILL: Dry, loose, tan, clumpy, coarse sand with ash-like material, and fine gravel		en valanse de la fugação do "Provincio" provincio magemento	
	22-24	1.2.4.50 for 3*	20	0.0	and a second	FILL: Wet, loose, brown, fine sand and silt with ash-like material and cinder-like material		n manana ng mga ng m	
-23	+2-27	1,2,4, 50 101 0	£0	50	23.5-24	WEATHERED BEDROCK: Moist, gray, compact silt, clay, fine sand and shale with a faint fuel-like odor. Refusal on bedrock at 24.0'			

	nc. <u>META JA Beneveration</u>
PID (ppm) = Photoionization Detector, readings in parts per million	Page 2 of 2

WS-SB12 A

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:414.841 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 10 ft. below grade WEATHER: Warm, cloudy, light breeze GEOLOGIST:Amanda Bissell

WELL CONSTRUCTION RECOVERY £ ANALYTICAL SAMPLE Blow Count REMARKS DEPTH (ft.) SAMPLE PID (ppm) SOIL DESCRIPTION and LITHOLOGIC SYMBOL 8 0 -0 FILL: Loose, dry to damp, gray silt, 0.0 fine sand, clay and angular shale clasts and sub-angular gravel 0-2 7,11,12,9 90 - - 1 0.0 -2 0.0 FILL: Dry to damp, loose, porous, -3 2-4 9,6,4,4 80 black, coal-like material (CLM) and -3 cinder-like material with fine to 300 medium brown sand and trace clay with a slight tar-like odor throughout -4 interval 0.0 -5 4-6 2,1,5,10 50 -5 0.0 6 -6 1,637 -7 6-8 13,13,14,8 80 567 FILL: Tight, cohesive, damp, gray clay with silt and large shale clasts .8 -8 8,657 FILL: Damp, black, loose, fine sand with fine gravel with trace clay 8-10 72 .0 -9 8-10 Not Recorded FILL: Broken, loose, dry shale clasts 2.809 \diamond FILL: Moist, cohesive, tan day with 1 -10 black staining, CLM, with a fuel-oil and tar-like odor, broken shale clasts and concrete material in shoe. Refusal at 10' on large drainage pipe (02 discharge pipe)

1	sh .nc.	METAL Recovering in
PID (ppm) = Photoionization Detector, readings in parts per million	Ŧ	Page 1 of 1

WS-SB12 B

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:415.014 WELL ELEVATION:NA OUTER CASING ELEVATION.:NA DEPTH TO WATER: NA BOREHOLE DEPTH: 8 ft. below grade WEATHER: Warm, cloudy, light breeze GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	ł	WELL CONSTRUCTION	REMARKS	
				0.0		FILL: Loose, dry to damp, gray silt, fine sand, clay and angular shale clasts and sub-angular gravel				
	0-2	7,11,12,9	90	0.0						
2	2-4	9.6.4.4	80	0.0		FILL: Dry to damp, loose, porous, black, coal-like material (CLM) and				
				300	na mananan kananan kana kana kana kana k	with a slight tar-like odor				
	4-6	2 1 5 10	50	0.0	an and a second seco					
		2, , , , , , , , , , , , , , , , , , ,		0.0	And and a second se					
	6-8	13.13.14.8	80	1,637	na na mana na mana mana na fato na mangana na mangana					
				567		FILL: Tight, cohesive, damp, gray clay with silt and large shale clasts.				

	sh nc.	METAPRissemente in
PID (ppm) = Photoionization Detector, readings in parts per million	Ŧ	Page 1 of 1

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Auger SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:414.512 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 26.3 ft. below grade WEATHER: Sunny, breezy, partly cloudy GEOLOGIST:Amanda Bissell

WELL CONSTRUCTION RECOVERY ŧ ANALYTICAL SAMPLE Blow Count DEPTH (ft.) SAMPLE INTERVAL (REMARKS (mqq) SOIL DESCRIPTION and ۵d LITHOLOGIC SYMBOL 8 -0 0 FILL: Loose, dry to damp, gray silt, $^{\circ}$ 0.0 fine sand, clay and angular shale 0-2 5.7.7.7 80 - -1 clasts and sub-angular gravel 0.0 FILL: Dry to damp, loose, fine sand -2 -2 and porous, black, coal-like material 0.0 (CLM) ∇a -3 2-4 8,6,4,6 80 -3 Δ FILL: Dry, cohesive, brown clay 0.0 XV -4 FILL: Dry, loose, brown clay, medium -4 $^{\prime}$ sand, fine angular gravel, shale clasts, 0.0 weathered brick, CLM and white ALM -5 -5 4-6 10,9,7,7 45 with some slag-like material -6 0.0 $\langle \cdot \rangle$ C -6 -6 FILL: Damp, black, medium sand and 10 0.0 shale clasts D30 5,5,3,2 -7 6-8 FILL: Dry, hard, brown, clay, silt and 0.0 fine to medium sand with fine, angular -8 gravel and broken shale -8 \sim 0.0 FILL: Moist, loose, brown, clay and 30 -9 8-10 5,2,2,3 -9 silt, fine to coarse sand with broken 0.0 shale clasts, weathered brick and CLM -10 -10 0.0 10-12 3,2,3,2 30 -11 -11 $^{\circ}$ 0.0 -12 -12 FILL: Moist, loose, brown, clay and 0.0 silt, fine to coarse sand with broken -13 25 -13 12-14 2,3,10,4 shale clasts, weathered brick and CLM 0.0 -14 0.0 30 15 4,3,4,2 -15 14-16 0.0 -16 -18

PID (ppm) = Photoionization Detector, readings in parts per million	Page 1 of 2

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Auger SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:414.512 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION:NA DEPTH TO WATER: NA BOREHOLE DEPTH: 26.3 ft. below grade WEATHER: Sunny, breezy, partly cloudy GEOLOGIST:Amanda Bissell

MER & R

Page 2 of 2

sh nc.

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	(mqq) Olq	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	d	WELL CONSTRUCTION	REMARKS
r16 r			1		7		-16	1	1
17	16-18	2,6,1,4	25	0.0		FILL: Moist, loose, brown, clay and silt, fine to coarse sand with broken shale clasts, weathered brick and	-04 		
				0.0		CLM	iO社		
	18-20	17 16 14 8	95	0.0		FILL: As above with wet, sticky, white clay covering the shale clasts			
20	10.20	v, r,		994		FILL: Damp, loose, black, fine sand with CLM, some slag-like material and a fuel-like odor	······································		
E -21	20-22	4667	75	767				of the second	
E -22	€a 52° (sa fa	.,0,7,*		859	21.5	FILL: Moist, tan clay with shale clasts and interior black staining and a fuel- like odor	O_{4}^{*}		
	22.24	9.12.10.5	60	1637		FILL: Wet, loose, porous, black CLM		A TO A TO A MARINA AND A TO A MARINA AND A MAR	
	22-2 4	5,12,10,5		859	22.5-23.8	tar-like odor, NAPL globules on water between 22.5-23.8			
-24	24-26	10653	50	0.0		FILL: Moist, gray clay with interior staining and black, stringy material in			
20	24-20	10,0,0,0		0.0		clay, with a strong fuel-like odor and black, sticky, viscous, tar-like material, brick in shoe			
-20	26-28	50 for 3"	<5	1	1		Marial 20	[]	J
						FILL: vvet, loose, gray, tine to medium sand with gray clay, silt and rock clasts, slight fuel-like odor and organic, wood			
						WEATHERED BEDROCK: Moist, cohesive, gray clay with shale clasts. Refusal on bedrock at 26.3'			

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/27/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:413.848 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 15.5 ft. below grade WEATHER: Warm, light breeze, clear sky

GEOLOGIST:Amanda Bissell

SAMPLE	Blow Count	% RECOVERY	PID (ppm)	ANAL YTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	rj	WELL CONSTRUCTIO	REMARKS
0-2	2,7,12,13	60	0.0		TOPSOIL: Damp, loose, gray and brown, coarse to medium sand with root material and trace, brown clay			
2-4	16,29,25,39	75	0.0		FILL: Damp to moist, loose, brown, medium to coarse sand with trace clay, fine, angular gravel, shale clasts			
4-6	19.17.23.20	80	0.0		and weathered brick		of the general state and the state of the st	
3			0.0			-6	n stran en de la constant en la cons	
7 6-8 3	11,22,20,17	75	0.0				() () () () () () () () () () () () () (
8-1 10	0 6,15,14,14	60	0.0			0 4 -9 		
11 10-1	12 12,13,14,12	60	0.0				a de la constante de la constan	
13 12-1	4 12,13,8,9	80	300				n biyl si shekara ta biya da an ti ci da an	
14 15 ¹⁴⁻¹	12,27, 50 for 1"	50	973	14-15.5	FILL: Damp, loose, black, very fine sand with coal-like material (CLM) and a slight fuel-like odor		an under son and a source of the	
	1	1	1		FILL: Dry, very hard, yellow and white medium to coarse, clump sand FILL: Dry, hard, black, sticky tar-like material (TLM) with CLM and a tar-like odor with brown sand, trace, brown clay and some concrete. Refusal at 15.5' below grade on 5' concrete pipe (03 discharge pipe)		<u>L</u>	

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/28/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:413.730 ft. above MSL WELL ELEVATION:NA OUTER CASING ELEVATION.NA DEPTH TO WATER: NA BOREHOLE DEPTH: 28 ft. below grade

WEATHER: Sunny, clear sky, light breeze, mild temperatures GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	(mdd) DId	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	3	WELL	REMARKS	
14						FILL: Continuation of SB14, auger down to 12' bgs and begin sampling.	-14 104			policity (minute interview)
-				767		Dry, hard, black, sticky tar-like material (TLM) with CLM and some				
15	14-16	14,11,3,1	50			odor with fine, black sand and	-15			
				973	15.3	weathered brick.		A PARA LA CALLANCE A RANK A		مر می از این از این از این
-16				0.000				4.00 × 4.00 × 100		
÷				2,609				i den i de la della d		
-17	16-18	2,1,2,2	40				10417	Sharen Amerika (andara 1		
				300	17-18	FILL: As above with ash-like material				
						FILL: Moist, loose, black sand with	1: ACT -18			
-			an fan fan fan fan fan fan fan fan fan f	300		some slag-like material, brown clay and a very slight tar-like odor				G and contraction and contraction
			a a Froma the solution of the				1024			17 Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y
⊢ -19 ⊢	18-20	Not recorded	>5				19 10 19	15 TE 1000		
1			a managana na mangana ka	2,159						and the second
- 20					and the second second second second		10×4 -20			- And a second
	20-21	11,12	>5	973		FILL: Compact, hard, gray, coarse to medium sand with sandstone clasts and apparent concrete. Refusal at 21.2		na prima na mana na kata kata na mana n		No for the second s
							-21			

	sh nc,	META KREssenantes in
PID (ppm) = Photoionization Detector, readings in parts per million	149. 149.	Page 1 of 1

WS-SB15 A

PROJECT: West Station - RG&E PROJECT NO: 103039 LOCATION: Rochester, NY DATE: 4/28/2006 DRILLING CONTRACTOR: Nothnagle Drilling DRILLER: Kevin Busch DRILLING METHOD: 4 1/4" Hollow-Stem Augers SAMPLING METHOD:2' Split Spoon GROUND ELEVATION:Not surveyed WELL ELEVATION:NA OUTER CASING ELEVATION:NA DEPTH TO WATER: NA BOREHOLE DEPTH: 28 ft. below grade WEATHER: Sunny, clear sky, light breeze, mild temperatures

> sh nc,

META

Page 1 of 1

GEOLOGIST:Amanda Bissell

DEPTH (ft.)	SAMPLE INTERVAL (ft.)	Blow Count	% RECOVERY	PID (ppm)	ANALYTICAL SAMPLE	SOIL DESCRIPTION and LITHOLOGIC SYMBOL	WELL	REMARKS
21 21	21-22	13,10	25	300	21-22	FILL: Continuation of SB15, auger down to 21' bgs and begin sampling.Wet, loose, brown and gray clay, coarse-fine sand, silt and shale	21	
22				070		clasts with black staining, green-tinted sheens and a fuel-like odor	22	
	22-24	11,13,12,10	95	9/3		FILL: Wet, loose shale clasts with black-stained clay, sheens present, NAPL globules present, some CLM	23	
				300			4	
				300				
25 - -	24-26	23,17,9,11	50	300			:5	
- 				300		WEATHERED BEDROCK: Wet, loose shale clasts with black-stained clay, sheens and NAPL globules present. Refusal on bedrock at 28.0' bgs.	6	
-27	26-28	23, 50 for 4"	90	300			7	

APPENDIX B-2 AES/META Plant Area Angle Cores in Upper Bedrock

BORING AB-1

Page 1 of 1

PROJECT: EPRI RG&E PROJECT NO: 1734-01-02 LOCATION: 15' SOUTH OF CST-1 DATE STARTED: 12/18/91 DATE COMPLETED: 12/17/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND JEFF GRANT ORILLING METHOD: HOLLOW STEM AUGER WITH ROLLER BIT & CASING SAMPLING METHOD: 10' CORE BARREL

GROUND ELEVATION (FT): 397.2 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): NA DATUM: 397.2 NORTHING: 1153768.0532 EASTING: 757777.3205 WEATHER: UNRECORDED INSPECTOR: VASILE RUSU, BOB RUSSO CHECKED BY: DOUGLAS BONOFF

			VERY		SOIL DESCRIPTION	VI CC	(S))N		ALYSIS	0GY	(FT.)	WELL CONSTRUCTION
	SPLIT SPOON SAMPLE DEPTH (ft)	BLOWS PER 8	RECO	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	SAMPLE AN	ГІТНОЦ	DEPTH	
					Overburden Boring was angled at 30D to Intersect vertical fractures. Linear depths are reduced to their vertical component for representation in log format.						0	
12	2.1-20.8				Top of Bedrock. Gray SHALE with horizontal bedding planes. Voids present with diameter imm to 1.5cm. Some voids filled with calcite crystals. Vertical Fracture. No stain or odor.						15-	
					Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor.						20-	
20	.8-29.4				Vertical Fracture. No stain or odor.						25-	
					Vertical Fracture. No stain or odor.						30-	
29.	.4-38.				Bedding plane with strong tar odor. Vertical Fracture. Bottom of Core. Strong sheen and petroleum odor noted in return water.						35-	
L <u>-</u> ,	L	I	<u>L</u>				1		L		-0-	ATLANTIC

BORING AB-2

ſ

Page 1 of 1

PROJECT: EPRI RG&E GROUND ELEVATION (FT): 413.83 PROJECT NO: 1734-01-02 PROTECTIVE CASING ELEVATION (FT): NA LOCATION: 15' SOUTH OF CTF WELL ELEVATION (FT): NA DATE STARTED: 12/18/91 DEPTH TO WATER (FT): NA DATE COMPLETED: 12/18/91 DATUM: 413.83 DRILLING CONTRACTOR: NORTH STAR DRILLING NORTHING: 1153945.9887 DRILLER: HARRY LYON AND JEFF GRANT EASTING: 757803.1837 DRILLING METHOD: HOLLOW STEM AUGER WITH ROLLER BIT & CASING WEATHER: 4150 F SNOW SAMPLING METHOD: 10' CORE BARREL INSPECTOR: VASILE RUSU, BOB RUSSO CHECKED BY: DOUGLAS BONOFF

			ЕВУ		SOIL DESCRIPTION	,	VIS CON	IU/ IT A	AL M.	AL YSIS	0GY	(FT.)	WELL CONSTRUCTION
	SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 8	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN		STAIN	SHEEN	HEAVY	SAMPLE AN	ГІТНОГ	DEPTH (
					Overburden Boring was angled at 30P to Intersect vertical fractures. Linear depths are reduced to their vertical component for representation in log format.							0- 5- 10- 15- 20-	
29	.0-37.7 7-48.2				Top of Bedrock. Gray SHALE with horizontal bedding planes. Fewer volds and calcite crystals than AB-1. Vertical Fracture. Tar Odor. Vertical Fracture. Tar Odor. Vertical Fracture. Vertical Fracture. Vertical Fracture. Vertical Fracture. Tar Odor. Vertical Fracture. Tar Odor. Vertical Fracture. Vertical Fracture. Ver							25- 30- 35- 40- 45- 	
								_					ATLANTIC

BORING AB-3

Page 1 of 1

PROJECT: EPRI RGSE PROJECT NO: 1734-01-02 LOCATION: 70' NORTH OF R32 DATE STARTED: 12/19/91 DATE COMPLETED: 12/19/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND JEFF GRANT DRILLING METHOD: HOLLOW STEM AUGER WITH ROLLER BIT & CASING SAMPLING METHOD: 10' CORE BARREL

(

GROUND ELEVATION (FT): 414.58 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): NA DATUM: 414.58 NORTHING: 1154158.8582 EASTING: 757414.9730 WEATHER: UNRECORDED INSPECTOR: VASILE RUSU, BOB RUSSO CHECKED BY: DOUGLAS BONOFF

			ERY		SOIL DESCRIPTION	N C	/IS	IU/	۹L M.	L YSIS	GΥ	(1.	WELL CONSTRUCTION
	SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 0	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	ΗΕΑΥΥ	SAMPLE ANA	ГІТНОГО	DEPTH (F	
					Overburden Boring was angled at 30Þ to intersect vertical fractures. Linear depths are reduced to their vertical component for representation in log format.							0 5 10 15 20 25	
30.	7-38. \$				Top of Highly Weathered Gray SHALE with CLAY lenses.					V 		30- - 35-	
38.	5-45.0				Gray Consolidated SHALE with some Weathered SHALE. Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor.							+ + + + + +	
5.0	0-49.5				Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor. Vertical Fracture. No stain or odor. Bottom of Core. No stain or odor anywhere in boring.						5	15-	
											5	51	ATLANTIC

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: NEAR SOUTHEAST CORNER OF CTF DATE STARTED: 10/30/91 DATE COMPLETED: 10/30/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

BORING TWS-1

Page 1 of 1

GROUND ELEVATION (FT): 413.710 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 8.2 DATUM: 0 NORTHING: 1153962.342438 EASTING: 757622.723500 WEATHER: CLEAR, LOW 30'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

			ЕВҮ		SOIL DESCRIPTION	v C	IS ON	IUA TA	NL M.	AL YSIS	0G Y	(FT.)	WELL CONSTRUCTION
1	SPLIT SPOON GAMPLE DEPTH (1t)	BLOWS PER 8*	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONF	STAIN	SHEEN	HEAVY	SAMPLE AN	ГІТНОЦ	DEPTH	
											~~~~	0-	
a day a sa a sa ang ang ang ang ang ang ang ang ang an	0-2	5, 4 5, 8	20	0	Cobbles and brick, FILL. Asphalt fragments, brown clay lense with some sand at 2 ft, FILL.							-	
	2-4	3, 2 2, 2	25	1.7	Brown silt and clay, some coarse to pebbly sand, trace cobbles and clinker, FILL.						$\bigotimes$		
and the second se	4-8	3, 3 3, 3	80	1.2	Sandstone and quartzite cobbles, some silt and coarse sand, wet at 8.2 ft, FILL.							5	
N (MAR) MARY, STATE WAS INTO A 14 (1997)	8-8	3, 2 3, 2	25	1.3	Pebbles and cobbles, trace brick, wet, FILL.								
8	-8.4	1		1.9	Spoon Refusal, In Tar/Water Separator. End of Boring at 8.4 Feet.					×		10	
		I	l			1			1		<b>I</b>	10-	ATLANTIC

PROJECT: EPRI RG&E PROJECT NO: 1309-02 LOCATION: IN GRASSY AREA NEXT TO UNIT #12 DATE STARTED: 11/8/91 DATE COMPLETED: 11/8/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

BORING R39

Page | of |

GROUND ELEVATION (FT): 413.030 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): NA DATUM: 0 NORTHING: 1153902.303088 EASTING: 757375.880724 WEATHER: CLOUDY, 20'S INSPECTOR: BOB RUSSO, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

			K VERY		SOIL DESCRIPTION			SIUA	I F F		0G Y	(FT.)	WELL CONSTRUCTION
	SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 8	RECO	PID (ppm)	color, density, SOIL.admixture, moisture, ather notes, ORIGIN	NONE	STAIN	SHEEN	HEAVY SAMPLE AN		ГІТНОL	DEPTH	
												0-	
	0-2	2, 4 8, 8	55		FILL. Black ash, some pebbles, trace sand, FILL.					X	$\bigotimes$	-	
	2-4	5, 8 5, 5	40		some rock flour and red brick, FILL.					$\mathbf{i}$			
	4-8	1, 2 8, 2	40		Grey SILT, some sand, pebbles, and cobbles, fill.					$\langle$		5-	
	8-8	7, 7 5, <b>5</b>	85							X		4	
1	3-10	3, 1 2, 10	15	-	Brown CLAY, trace cobbles, silt, and wood fragments,FILL, BLACK STAIN AND SLIGHT PETROLEUM ODOR.					$\left\langle \right\rangle$			
10	)-12	11, 4 9, 18	75		Brown weathered chain trace				K	$\left\{ \begin{array}{c} \\ \\ \\ \\ \\ \\ \end{array} \right\}$		10-	
12	2-14	8, 8 12, 14	70		pebbles, BEDROCK. SLIGHT PETROLEUM ODOR.							-	
14	-18	20, 48 50/.2	25									5	
					No sample taken.							-	
7.8-	-19.8	45, 50/.1	30		Unweathered SHALE. End of Boring at 18.4 Feet.					7			
											20	)-]	ATLANTIC

# APPENDIX B-3 AES/META Plant Area Borings and Piezometers

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: NEAR RIVER AND DISCHARGE AREA FUEL TANKS DATE STARTED: 10/30/91 DATE COMPLETED: 10/30/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

Т

(

#### BORING CST-1

GROUND ELEVATION (FT): NA PROTECTIVE CASING ELEVATION (FT): 397.21 WELL ELEVATION (FT): 397.210 DEPTH TO WATER (FT): 4.6 DATUM: 0 NORTHING: 1153772.208103 EASTING: 757773.961988 WEATHER: CLEAR, LOW 30'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

			ЛЕВҮ		SOIL DESCRIPTION	v c	IS ON	IUA TAP		0GY	(FT.)	WELL CONSTRUCTION
SI SF SA DE	PLIT POON MPLE EPTH (ft)	BLOWS PER 8	RECO	PID (ppm)	color, density, SOIL,admixture, moisture, ather nates, ORIGIN	NONE	STAIN	SHEEN	SAMPLE AN	ГІТНОГ	DEPTH	
					Brown SAND, some reads and						0-1	<b>T</b>
с	)-2	3, 19 28, 4	75	5	pebbles, trace red brick, moist, FILL. Pebbles, cobbles, black ash, and tar, trace coal, FILL. MODERATE TAR ODOR.							VC
2	-4	8, 11 11, 9	50	2	Cobbles, no odor, FILL. Brown-black SAND, some silt, cobbles, pebbles, and organics, moist, FILL. TAR ODOR.					$\bigotimes$		
4	-8	3, 4 8, 2	50	18 0	Grey-black, very fine SAND, some organics, wet, FILL, PETROLEUM ODOR.						5-	
8-	-8	I, 2 8, 25	75	1.8	Crushed yellow brick and cobbles,				K K			IED PVC
8-	10	4, 8 17, 8	25	0	some fine sand and weathered shale, FILL. SLIGHT TAR ODOR AND SHEEN. Black SILT, some sand and roots. SLIGHT TAR ODOR.							- 0.02 In. SL01
10-	12	25, 50/.2	40	41	Cobbles and pebbles, trace sand. STRONG TAR ODOR AND SHEEN. Auger Refusal. End of Boring at 11.2 Feet.						10-	
											<b>.</b>	
											15-	ATLANTIC

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: NEAR GAS TURBINE DATE STARTED: 10/30/91 DATE COMPLETED: 10/30/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

#### BORING TST-1

GROUND ELEVATION (FT): 414.510 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 18 DATUM: 0 NORTHING: 1154120.718840 EASTING: 757448.829525 WEATHER: CLEAR, LOW 30'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

			/FBY		SOIL DESCRIPTION	v Ci	ISI 0N1	UAL	AL YSIS	0GY	(FT.)	WELL CONSTRUCTION
	SPLIT SPOON SAMPLE DEPTH (ft)	BLOWS PER 8	RECO%	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	SAMPLE AN	ГІТНОГ	DEPTH	
	0-2 2-4 4-8 8-8 8-10 10-12 12-14 14-18 18-18 8-20	BLOWS PER 8' 10, 15 10, 6 15, 22 22, 20 8, 8 5, 11 7, 8 8, 6 7, 3 8, 7 3, 8 8, 4 4, 10 10, 14 7, 14 10, 10 10, 8 5, 10 3, 5 5, 9	75 40 65 10 40 60 50 30 80 75	PID (ppm) 2.7 0 4.2 4 0 3.8 0 0 5.7 3.8	Color, density, SOIL, admixture, moisture, other notes, ORIGIN Brown-grey, fine SAND, some pebbles, FILL. Brown-grey, SILT, some clay,rock fragments,pebbles, FILL. SLIGHT ODOR. Grey pebbles, sand, and rock flour, FILL. Brown ash and slag, some cobbles and shell fragments, FILL. Black, fine sand and slit, some pebbles and cobbles, FILL. Brown sand, ash, and pebbles, some brick, coal, and shell, FILL. Brown-grey, ash and sand, some cobbles, moist, FILL. SLIGHT PETROLEUM ODOR. Grey, fine SAND, wet at I8 feet. PETROLEUM ODOR. Brown-black, fine SAND, some	NO						
2	0-22	0, 0 11, 5	50	4	Brown-black, tine SANU, some cobbles, no odor. Black, rock and brick fragments,						20-	
2	2-24	22, 48 32, 40	95	3.8	trace shell, wood, and sand. LIGHT ODOR.							
2	5-28	18, 20	30	4.3	PETROLEUM ODOR. Tan SILT and CLAY, weathered						25-	
2	8-28	17, 15	40	3.9	shale, some pebbles. LIGHT PETROLEUM ODOR AND STAINING FROM 25.4 FT TO 20.8 FT.							
28	3-30	4,5 8,8	25	4	Tan-brown CLAY, some silt and pebbles.						30-	
30	0-32	48, 30	25	0								
32	2-34	8,9 9,9	0		Na recovery.							
3.	1-38	14, 34 50/.1	100	3.2	Brown-tan CLAY, some pebbles. Brown coarse SAND, some cobbles and pebbles. Possible BEDROCK. Auger Refusal and End of Boring at 38.5 Feet.					4	0-	
												ATLANTIC

BORING TWS-2

Page 1 of 1

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: SOIL GAS POINT F14 DATE STARTED: 10/29/91 DATE COMPLETED: 10/29/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"1D HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

{

GROUND ELEVATION (FT): 413.830 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 23 DATUM: 0 NORTHING: 1153947.211189 EASTING: 757597.888434 WEATHER: CLEAR, LOW 30'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

Shift Show Chill Chill Step Chill Chill Step Chill Chill Step Chill Chill Step Chill Chill Step Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill Chill C			ERY		SOIL DESCRIPTION		/IS CON	IU/	AL M.	AL YSIS	0G Y	FT.)	WELL CONSTRUCTION
0-2     2.14     00     0     Brown, SANDY, SILTY LOAH, trace       2-4     8, 6     55     96     Block ash, pebbles, and coarse     0       4-8     3, 4     50     10     Block ash, pebbles, and coarse     10       7-8     5.2     60     30     0     10     Block ash, pebbles, and coarse       9-10     7, 18     60     24     Algered through cables, fill, STROK TAA 000R, and coarse     10-12     5-1       9-10     7, 18     60     24     Algered through cables, fill, STROK TAA 000R, and other cables, fill, STROK STAA 000R, and other cables, fill, STROK TAA 000R, and other cables, fill, STROK TAA 000R, and other cables, fill, STROK STAA 000R, and other cables, filll	SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 8	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	AIONE	STAIN	SHEEN	ΗΕΑΥΥ	SAMPLE AN	ГІТНОГО	ОЕРТН (	
0-2     2,14     00     0     Brown, SANCY, SILTY LOAM, trace       2-4     8,8     55     86     Black ash, pebbles, and coarse       4-9     16,8     55     86     Black ash, pebbles, and coarse       3-4     50     0     Prover, SANCY, SILTY LOAM, trace     Fill       4-9     16,8     50     0     10     Fill       7-8     5,2     00     30     Direct fill are and cobles, fill, SINC fill and coold, coarse sand, fill, SINC fill and coold, coarse sand, fill, SINC fill and coold, fill an												~	
2-4   8, 6   55   09   Fill.     4-8   3, 4   50   10   File.   State ship pebbles, and coarse sing and brick, Fill.     4-8   16, 8   50   10   File.   State ship pebbles, and coarse sing and brick, Fill.     7-6   5.2   90   30   File.   State ship pebbles, sing ash trans     8-10   7.18   90   24   State ship pebbles, sing ash trans   File.     10-12   9, 12   90   33   Field and block-forwn sing, fracture ship erables, sing and block-forwn sing, fracture ship erables, sing and block-forwn sing, fracture ship erables, sing and pebbles, fill.   State ship erables, fill.     10-12   9, 12   80   33   Fracture ship erables, sing and solg, some pebbles, sing and solg, some pebbles, sing and solg, some pebbles, fill.   State ship erables, fill.     12-14   8, 7   60   1.2   Brown-black sig and solg, some pebbles, fill.   State ship erables, fill.   State ship erables, fill.     10-18   8, 9   10   1.2   Brown-black sig and solg, some pebbles, fill.   State ship erables, fill.   State ship erables, fill.     12-14   8, 7   10   6   1.2   Brown-orange sand coarse	0-2	2, 14 14, 8	80	0	Brown, SANDY, SILTY LOAM, trace pebbles, FILL.	7				$\left  \right\rangle$	$\mathbf{X}$	 [	
4-8     3.4     50     27     Bleck far and cobles. Ine ash from 27' to 3.17 ILLSTRONG TAR 000R.       7-8     5.2     80     30     Code. Are observed. The observed and observed. The observed and observed and observed. The observed and o	2-4	8, 8 8, 8	55	68	FILL.	┦					$\bigotimes$	4	
Image: Constraint of the state in	4-8	3, 4 18, 8	50	27 10	Black tar and cobbles, fine ash from 2.7' to 3.1', FILL.STRONG TAR ODOR.	4				K	$\bigotimes$	5-	
B-I0     7, IB     BO     24     Grey cobbles and sit, FILL. SLIGHT       I0-I2     8, I2     80     38     Intervention of the sector of t	7-8	5, 2	80	30	Black-brown, ash, pebbles, slag, coarse sand,FILL. TAR STAIN AND ODOR.						$\bigotimes$		
IO-12     8, 12     80     38       12-14     4, 7     55     1.8     Term of Diack-Prown sand, trace Pebbles, Fill.     Stand Pebbles, Fill.	8-10	7, 18 10, 9	60	24	Grey cobbles and silt, FILL. SLIGHT	-				Ŕ	$\mathbf{X}$		
12-14     4. 7     55     1.8     Grey, fractured shale, trace sand and pebbles, FILL. IGHT 000R.       14-18     8. 9     80     1.2     Brown, silt, some sand, orange mottles, coal and cobbles, FILL.     1600R.       18-18     8. 10     70     0     Brown, silt, some sand, orange mottles, coal and cobbles, FILL.     1600R.       18-20     7. 10     0     Brown silt, some clay, cobbles, pebbles, trace brock, FILL.     20-       20-22     7. 7     65     8.9     0     Orange-brown silt, some clay, cobbles, pebbles, trace brock, FILL.     20-       22-24     12, 10     70     21     Brown-grey, silt, some clay, cobbles, pebbles, trace brock, FILL.     20-       24.5-     10, 8     25     .1     Brown-orange sandstone.     25-       27.8-     20, 15     21     Brown-orange sandstone.     25-     36-       27.8-     50/.15     1.9     Shale chips in basket.     30-     35-	10-12	8, 12 9, 12	80	38	Tar and black-brown sand, trace pebbles, FILL, STRONG TAR ODOR.					K	$\mathbf{X}$	10-1	
14-18   7, 7 8, 9   80   1.2   Smew silt, FILL. TAR ODOR. Brown, silt, some sand, orange mottles, coal and cobiles, FILL.   15-     18-18   8, 9 70   0   Brown, silt, some sand, orange mottles, coal and cobiles, FILL.   15-     18-20   4, 8 4, 8   45   0   Orange-Torwn silt, some clay, cobles, pebbles, trace brick, FILL.   16-     20-22   7, 7 7, 4   85   8.9   12.   Brown-grey, silt, some clay, FIL.     22-24   12, 10   70   21   Brown-grey, silt, some clay, FIL.   Brown-grey, silt, some clay, FIL.     24.5-   10, 9   25   .1   Brown-orange sandstone.   25-     24.5-   14, 28   25   .1   Brown-orange sandstone.   25-     27.8-   50/.15   20   0   Shale chips in basket.   25-     27.8-   50/.15   20   0   Shale chips in basket.   30-     30-   35-   30-   35-	12-14	4,7 8,5	55	1.8	Grey, fractured shale, trace sand and pebbles, FILL, LIGHT ODOR. Grey-brown, cobbles and sand,					K	$\bigotimes$		
18-18   6, 9 8, 10   70   0   White-tan, line ash and sigg, some pebbles, FilL. SLIGHT 000R.     18-20   4, 8 7, 10   45   0   Brown-black sigg and ash, trace coarse sand, FilL. SLIGHT 000R.   0     20-22   7, 7 7, 4   65   6.9   Brown-grey, silt, some clay, cobbles, pebbles, trace brick, FilL.   20- Brown-grey, silt, some clay, FilL.   20- Brown-grey, silt, some clay, FilL.     22-24   12, 10   70   21   Tar and cobbles, some brown silt and wood, FilL. STRONG 0DOR.   26- Grey, file sand, some cobbles, moist. SLIGHT 0DOR.     24.5-   10, 9   25   .1   Brown-orange sandstone.   25- 27.4   3, 15/.4   20   0     Shale chips in basket.   End of Boring at 27.95 Feet.   30-   30-	14-18	7, 7 8, 9	60	1.2	some silt, FILL, TAR ODOR. Brown, silt, some sand, orange mottles, coal and cobbles, FILL,					R		15-	
18-20   4, 8   7, 10   45   0   Brown-Diack Siag and ash, trace clask, course sand, FILL. SLIGHT ODOR.   20-22   7, 7   85   8.9   Coarse sand, FILL. SLIGHT ODOR.   20-22   7, 7   85   8.9   Brown-orrange sand, FILL. SLIGHT ODOR.   20-22   20-22   7, 7   85   8.9   BLACK TAR STAINING AND ODOR.   20-22   7, 7   85   8.9   BLACK TAR STAINING AND ODOR.   20-22   7   7   85   8.9   BLACK TAR STAINING AND ODOR.   20-22   7   7   85   8.9   BLACK TAR STAINING AND ODOR.   20-22   7   7   85   8.9   BLACK TAR STAINING AND ODOR.   7   7   7   7   85   8.9   Tar and cobbles, some clavy.   20-22   20-22   20-22   7   7   8.9   7   7   7   7   8.9   7   8.9   7   7   8.9   7   8.9   7   7   8.9   7   8   7   9   7   9   7   9   7   9   7   9   7   9   7   9   7   9   7   9   7   9 <t< td=""><td>18-18</td><td>8,9 8,10</td><td>70</td><td>0</td><td>White-tan, fine ash and slag, some pebbles, FILL. SLIGHT ODOR.</td><td></td><td></td><td></td><td></td><td>Ŕ</td><td>$\langle \rangle$</td><td></td><td></td></t<>	18-18	8,9 8,10	70	0	White-tan, fine ash and slag, some pebbles, FILL. SLIGHT ODOR.					Ŕ	$\langle \rangle$		
20-22   7, 7   85   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   8.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9   9.9 <td< td=""><td>18-20</td><td>4, 8 7, 10</td><td>45</td><td>0</td><td>Brown-black slag and ash, trace coarse sand, FILL. SLIGHT ODOR. Orange-brown silt, some clay,</td><td></td><td></td><td></td><td></td><td>K</td><td>$\mathbf{X}$</td><td></td><td></td></td<>	18-20	4, 8 7, 10	45	0	Brown-black slag and ash, trace coarse sand, FILL. SLIGHT ODOR. Orange-brown silt, some clay,					K	$\mathbf{X}$		
22-24   12, 10 14, 28   70   21   Tar and cobbles, some brown silt and wood, FILL. STRONG ODOR. Grey, fine sand, some cobbles, molst. SLIGHT ODOR.   25   .1     24.5- 28.5   8, 11 20   25   .1   Brown-orange sandstone.   25- 27.4   25- 27.4   20   0     27.8- 27.95   50/.15   50/.15   20   0   Shale chips in basket. End of Boring at 27.95 Feet.   30- 35-	20-22	7, 7 7, 4	85	8.9	Cobbles, pebbles, trace brick, FILL. Brown-grey, silt, some clay, FILL. BLACK TAR STAINING AND ODOR.					K		20-	
24.5- 10, 9 25 .1 Brown-orange sandstone. 25-   28.5 8, 11 20 0 Shale chips in basket. 25-   27.4 3, 15/.4 20 0 Shale chips in basket. 25-   27.8- 50/.15 50/.15 End of Boring at 27.95 Feet. 30-   30- 35- 35-	22-24	12, 10 14, 28	70	21	Tar and cobbles, some brown silt and wood, FILL. STRONG ODOR.					Ŕ			
20.5     0, 11       20.5-     3, 15/.4     20       27.4     3, 15/.4     20       27.8-     50/.15       27.95     50/.15	24.5-	10, 9	25	.1	moist. SLIGHT ODOR. Brown-orange sandstone.							25-	
27.4     Oriority       27.8- 27.95     50/.15       End of Boring at 27.95 Feet.       30- 35-	20.5	3 15/ 4	20	0					-			1	
	27.4	0,10/11		F	Shale chips in basket.							1	
	27.95	50/.15			End of boring at 21.95 reet.							30-	
												4	
				1								35-1	ATLANITIC

BORING TST-2

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: NEAR GAS TURBINE DATE STARTED: 10/30/91 DATE COMPLETED: 10/30/91 DRILLING CONTRACTOR: NORTH STAR DRILLING ORILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): 413.800 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 20.4 DATUM: 0 NORTHING: 1154184.013087 EASTING: 757437.853888 WEATHER: CLEAR, LOW 30'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

		ΈRΥ		SOIL DESCRIPTION	V C	ISI ON 1	UAI FAN	AI YSTS	36Y	(.T.)	WELL CONSTRUCTION
SPLIT SPOON SAMPLI DEPTH (1t)	BLOWS	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	SAMPLE AN	ГІТНОГ	DEPTH	
										0	
0-2	2, 14	100	0	and pebbles, black ash from .4it to .8it and 2it to 3.5it, FILL.					$\times$	-	
2-3.5	5, 5 Ø	60	0	Augered to 5 feet.					$\bigotimes$	4	
5-8	7, 5	100	0	Brown SAND, some pebbles, trace					$\times$	5-	
8-8	5, 5 4, 4	45	0	FILL. Brown SILT, some sand and pebbles,					$\times$	-	
01-B	8, 11 14, 14	85	0	trace coal flecks, FILL. Brown, fine SAND, some cobbles,					$\times$	1	
10-12	8, II 8, II	80	0	Orange-brown-black SILT, some sand,pebbles, ash,FILL, SLIGHT TAR ONOR FROM 9 TO UR FT						10	
12-14	3, I 1, 10	30	٥	Tan-brown, very fine SAND, some cobbles, perched water table, FILL.						1	
14.5-18	5, 11 10	75	0	Grey cobbles and shale fragments, some silt and sand, FILL.				K		15-	
18-18	6, 8 8, 9	50	3.2	Sandy tar and CLAY lenses, some pebbles, FILL, TAR ODOR AND STAIN				┥		4	
18-20	4,5 8,8	45		Grey cobbles, tar, and red brick, trace sand,FILL. TAR ODOR AND				K		20	
20-22	8, 8 12, 20	80	0	Brown SAND and CLAY, some pebbles, trace tar, wet at 20.4,				K		20	
22-24	8, 10 11, 12	55	.1	Grey weathered shale and tar,				R	$\mathbf{X}$		
25-28	8, 30 18, 5	55	0	Pebbles and liquid tar, trace sand, FILL. SHEEN AND TAR ODOR.				Ķ		25-	
28-28	1, 1 2, 7	55	0	Brown-grey weathered shale				ŀ		4	
28-30	10, 23 30, 25	55	0	BEDROCK. TAR FILLED FRACTURE AT 30.5 FT.						-	
30-32	18, 27 31, 35	80	з.9							30-	
32-34	12, 37 50/.2	50	0	End of Boring at 33 Feet.							
										35	ATLANTIC

PROJECT: EPRI RGGE PROJECT NO: I309-02 LOCATION: ALONG TRANSECT D DATE STARTED: 10/4/91 DATE COMPLETED: 10/4/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND DAVE LYONS DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

٠

GROUND ELEVATION (FT): 410.550 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 18 DATUM: 0 NORTHING: 1153650.873417 EASTING: 757934.845482 WEATHER: CLOUDY, 50'S INSPECTOR: FRAN DUMONT, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

				u, Lon						00001		
		уевү		SOIL DESCRIPTION	VI CC	SI	UAL	IAL VETC	1010	.06Y	(FT.)	WELL CONSTRUCTION
SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 8	RECO	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	SIAIN	SHEEN	SANDIE AN		СЛНОГ	DEPTH	
											0-	•
0-2	2, 8 12, 12	80		Organic SAND and SILT, FILL. Black-brown SAND and ash, some slit brick and gravel FILL					$\bigotimes$	$\mathbf{X}$		
2-4	14, 8 8, 8	80		Black-grey, tine brick, ash, slag, and cinder, FILL.					$\bigotimes$		4	
4-8	4, 3	65							X	$\mathbf{X}$	5-	
6-8	4, 3	55		Black-red, ash, brick, and cinder, FILL.					$\bigotimes$	$\mathbf{X}$	4	
8-10	2, 2 3, 4	50							$\bigotimes$	$\mathcal{A}$	10-	
10-12	3, 1	40	-	Brown-black, fine SAND, some ash, trace brick and concrete, FILL.					$\bigotimes$		-	
12-14	8,7	70		fragments, trace brick, FILL. Black-brown, fine ash and silty clay,					$\bigotimes$	$\langle \rangle$		
14-18	9, 18	80		some rock, brick, and concreté					$\bigotimes$	X	15-	
18-18	8,7 4,3	40		CONTAMINATION AND ODOR.					$\bigotimes$	$\langle \rangle$		
18-20	4, 3	55		Trock, and brick fragments, FILL.				K	$\times$	X	20-	
21-23	2, 3 8 4	0		brick fragments, wet, FILL. SLIGHT				k	$\bigotimes$		-	
23-25	8,9 10,17	20		No recovery. Black ash, wood, and rock fragments, some coal, FILL. SLIGHT				k	$\bigotimes$	X	-	
25-27	37, 33 18, 15	55		ODOR. Black, fine SAND and cobbly rock fragments, some fine, silty sand.					· · · · · ·		25	
				SLIGHT TAR/FUEL ODOR.								
				No recovery, at BEDROCK. End of Boring at 30.5 feet.				- M			-08	
										3	15-	

#### Page 1 of 1

ATLANTIC

1

PROJECT: EPRI RG&E PROJECT NO: 1309-02 LOCATION: ALONG TRANSECT D DATE STARTED: 10/4/91 DATE COMPLETED: 10/4/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND DAVE LYONS DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON GROUND ELEVATION (FT): 408.870 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 16 DATUM: 0 NORTHING: 1153879.319750 EASTING: 757895.417738 WEATHER: CLOUDY, 50'S INSPECTOR: FRAN DUMONT, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

				SOIL DESCRIPTION		/ISIU	AL	AL YSIS	<b>3</b> 6Υ	FT.)	WELL CONSTRUCTION
SPLI SPOC SAMPI DEPT (ft)	T DN E H BLOWS PER B	RECOV	PID (ppm	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONF	STAIN	HEAVY	SAMPLE AN	LITHOLO	DEPTH (	
				Augered through FILL to 4 feet.					$\bigotimes$		
4-8	2, 1 8, 8	55		Brown, silty CLAY, trace organics, moist, FILL. Black ash , cinder, and brick, FILL.	<u>_</u>				$\bigotimes$	5-	
8-8	5, IO 18, 23	35		Black-grey, ash and cinder, trace brick and coke, FILL, SLIGHT ODOR FROM 10.3 TO 10.8 FT.				K			
8-10	7, 8 7, 8	35						ß		10	
10-12	3, 5 4, 4	50		Black-brown, coarse ash, slag, and coal, trace silty clay, dry, no odor, F ILL.				R			
12-14	3, 7 15, 20	70		Brown-rust, fine SAND and coal, trace cinder and wood fragments, FILL.						1	
				Augered to 18 feet.				K		15-	
18-18	3, 4 18, 30	25		Coarse slag, wood, rock, and concrete fragments, wet, FILL. STRONG ODOR AND SHEEN. Green-grey, silty CLAY some sand						4	
18.5-20	3, 3 18	70		brick, and organics, FILL.						4	
20-22	4, 18 18, 25	60		Black slag and cinder, some cobbles, FILL. Dark grey, rock fragments, slag, cobbles, glass, and metal, FILL. MODERATE ODOR, SLIGHT SHEEN FROM 20.5 TO 20.8 FT.						20-	
				Auger refusal. End of Boring at 21.8 feet.						25-	

ATLANTIC

BORING R12A

PROJECT: EPRI RGGE PROJECT NO: 1309-02 LOCATION: ALONG TRANSECT D DATE STARTED: 10/8/91 DATE COMPLETED: 10/8/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND DAVE LYONS DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): NA PROTECTIVE CASING ELEVATION (FT): 408.41 WELL ELEVATION (FT): 408.210 DEPTH TO WATER (FT): 12.3 DATUM: 0 NORTHING: I153672.102372 EASTING: 757891.389920 WEATHER: CLEAR, 40'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

			ГЕНУ	SOIL DESCRIPTION	VI CO	SIU	AL M.	AL YSIS	0G Y	(FT.)	WELL CONSTRUCTION
SPLIT SPOON SAMPLI DEPTH (1t)	BLOWS PER 8	94 S	PI PI (pp	color, density, SOIL,admixture, m) moisture, other notes, ORIGIN	NONE	SHEEN	НЕАVY	SAMPLE AN	LITHOL	ОЕРТН	
0-2	3, 7 7, 8 3, 5 3, 5	40	0	Brown, fine SANDY LOAM, trace organics and rock fragments, FILL. Black, fine ash, some coal, cinder, concrete, and brick fragments, FILL. Light to dark grey, coarse ash, trace cinder and brick fragments, FILL. Augered to 12 feet.						0 - 5 -	- 1 in. Bl.ank PVC
12-14 14-18	3, 3 50/.2 0, 0 0, 0	45 20	0	Black, coarse to pebbly ash, some clay, trace coal, slag, and wood, wet at 12.3 feet, FILL. Green-black, coarse to cobbly slag and cinder, some fine sand, FILL. SLIGHT TAR ODOR AND SHEEN.						10-	22 in. SLOTTED PVC
18-18	Ι, Ο	25	DEF	Auger refusal and end of boring at 17 feet.						20-	

ATLANTIC

PROJECT. EPRI RGGE PROJECT NO: 1309-02 LOCATION: ALONG TRANSECT D DATE STARTED: 10/8/91 DATE COMPLETED: 10/8/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLER: HARRY LYON AND DAVE LYONS DRILLING METHOD: 3/1/4"ID HOLLOW STEM AUGER BAMPLING METHOD: 2"X24" SPLIT SPOON GROUND ELEVATION (FT): 409.450 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 16.8 DATUM: 0 NORTHING: 1153071.858507 EASTING: 757915.377557 WEATHER: CLEAR, 40'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

		ΈRΥ		SOIL DESCRIPTION	v C	ISIU CONT	JAL AM	AL YSIS	0G Y	(FT.)	WELL CONSTRUCTION
SPLI SPOO SAMPL DEPT	T N E H BLOWS PER 8	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONF	STAIN	HEAVY	SAMPLE AN	ГІТНОГ	DEPTH	
1											
0-2	7, 7 14, 21	90	0	Brown-grey, fine SAND and CLAY, some rock fragments, FILL. Black-rust, fine SAND and SILT.					$\bigotimes$	07	
2-4	4, 11 5, 5	55	0	some rock and brick fragments, trace coal and pebbles, FILL. Grey, very fins SAND, some brick				ľ	$\mathbf{X}$	1	
4-8	7.7 8.8	45	0	Black, coarse to pebbly ash and cinder, some brick and coal, FILL.					$\bigotimes$	5-	
8-8	I, 2 I, 2	40	0								
8-10	0, 14 18, 11	40	0	Brown, coarse SAND, some rock fragments, trace brick and clay, FILL.				K		-01	
:0-12	8, 17 12, 14	85	0	Black, coarse asn, some rock fragments, trace clay and brick, FILL.						4	
12-14	5, 10 2 <b>2, 11</b>	75	3	Brown, coarse to peobly SAND, trace brick, coal, and rock fragments, line sand from 12 to 12.5				K		-	
14-18	8, 10 12, 7	35	15	Tar and brick, FILL. TAR CONTAMINATION AND ODOR.				Ŕ		15-	
18-18	5, 9 10, 8	55	2	Clay, FILL, HEAVY TAR CONTAMINATION AND ODOR. Black, coarse SAND and tar, some				X			
18-20	5, 10 11, 15	40	0	silt and brick, FILL. TAR CONTAMINATION AND ODOR. Silty, fine SAND, trace pebbles,						20-1	
20-22	10, 9 8, 4	25	5	SHEEN FROM I7.1 TO 18.8 FT.						20	
22-24	8, 7 50	50		Rock fragments, some silt, trace					$\leq$		
				End of Boring at 23.8 feet.					2	25-	
										4	
									3	Ĺ	

#### Page | of |

ATI ANTIC.

PROJECT: EPRI RG&E PROJECT NO: 1309-02 OCATION: TRANSECT I CATE STARTED: 10/31/91 DATE COMPLETED: 10/31/91 DRILLING CONTRACTOR: NORTH STAR DRILLING ORILLER: HARRY LYON AND RICK BECKWITH CRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): NA PROTECTIVE CASING ELEVATION (FT): 411.33 WELL ELEVATION (FT): 411.330 DEPTH TO WATER (FT): 16.8 O MUTAC NORTHING: 1153828.759898 EASTING: 757932.701238 WEATHER: CLOUDY, 50'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

		20		SOIL DESCRIPTION	WELL DNSTRUCTION
SPLI SPOC SAMP DEP1 (ft	IT LE IH BLOWS ) PER 8		PIO (ppr	color, density, SOIL.admixture, moisture, other notes, ORIGIN	
0-2 2-4 4-6 8-8 8-10 10-12 12-14 14-18 18-20 20-22 22-24 24-28 28-28 28-30 30-32 32-34	H BLOWS PER 8* PER 8* 1.1 12.18 3.9 18.12 20.9 3.4 1.2 1.1 0.0 1.1 2.3 2.3 3.3 3.3 3.3 12.3 3.3 3.3 12.3 3.3 12.3 3.3 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0	80 70 40 35 30 40 90 80 40 90 80 45 0 20 50 35 45 35 45 35 40	PID (ppn 0 25.2 50.4 8.9 4.5 3.3 3.8 1.4 1.2 8.2 2 3.4 2.1 5 1.8 2.8	Brown coarse SAND, some pebbles, trace siag, FILL. Black san and red brick, some pebbles, FILL. TAR IN BASKET. Brown-black SAND, asn, trace pebbles, FILL. SIGNERATE TAR OOGR. Tar, some ash, trace sand and white brick, FILL, STRONG TAR ODOR AND CONTAMINATION. Brown-red, brick tragments, FILL. Brown ash, some sig, ash, and pebbles, white specs, FILL. SLIGHT MUSTY ODOR, SLIGHT COAL TAR OOOR. TAR IN BASKET. No recovery. Black siag and ash, trace coarse sand, FILL. MUSTY OOR. Grey, very line SAND, some pebbles, Some glass, FILL, FUEL OID OR FROM 22.8 TO 23 FT. Brown-black, coarse SAND and PEBLES, some glass, FILL, FUEL OID OR FROM 22.8 TO 23 FT. Brown-clack, coarse SAND and PEBLES, rule JU ODOR. Grey-brown, coarse SAND and COBBLES, some silt. Meathered and unweathered SHALE, BEDROCK. Auger Refusal, End of Boring at 34 Feet.	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0

ATI ANITIC

ROJECT ND: 1309-02 DEATION: TRANSECT I, 25 FT NORTH OF R29

ROJECT. EPRI RGSE

(

CATE STARTED: 10/31/91

DATE COMPLETED: 10/31/91

CRILLING CONTRACTOR: NORTH STAR DRILLING

BILLER: HARRY LYON AND RICK BECKWITH

CRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER

DAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

SROUND ELEVATION (FT): 410,730 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA CEPTH TO WATER (FT): 17 DATUM: 0 NORTHING: 1153838.093448 EASTING: 757909.540312 WEATHER: CLOUDY, 50'S INSPECTOR: TERRY TAYLOR, BOB RUSSO CHECKED BY: NICOLE RUDERMAN

		те с ИЕАХ		SOIL DESCRIPTION	v C	ISI ON	UA TAP	AL YSIS	3GY	F1.)	CONSTRUCTION
SPLI SPOO SAMPL DEPT !U	TI DNI LEI 'HI BLOWS I PER 8	BECU,	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONF	STAIN	SHEEN	SAMPLE AN	ГІТНОГ	DEPTH (	
			•								
)-2	8, 8 :4, 8	70	1.3	Grey SAND, some silt and cobbles, moist, FILL. Black-brown ash, some coarse sand,					$\bigotimes$	0-   	
2-4	4,8 25,9	50	1.4	Grey CLAY, some sand, trace wood and organics, FILL.					$\mathbf{X}$		
4.5-8	4.9	100	.8	Black-grey SILT, some sand and brick fragments, FILL.						5-	
8-8	3, 3	45	1.2	Grey SILT, some clay and pebbles. trace sand, FILL.						4	
9-10		45	1.2	coarse sand, brick fragments, FILL. SLIGHT ODOR FROM 5.2 TO 8.9 FT.				K			
·0-12	17,5	 85	.8	Cobbles, trace wood, FILL. TAR FLECKS.				ß		10-	
-2-14	1 1 2	- 30 1	.4	Brown ash and slag, trace brick and coarse sand,FILL. SLIGHT ODOR FROM 12.8 TO 14.9 FT.				$\mathbf{R}$			
'4-18	1, 3	45	.5	Brown-grey slag and ash, some				K		15-	
:8-18	3, 3 3, 3	50	.8	white brick, trace coarse sand, wet at 17 it, FILL.				K			
:9-20	1, 1 1, 5	50	.4	Black-grey SILT, some clay and sand, FILL, SLIGHT ODOR FROM				K		-	
20-22	2, 2 3, 1	70	0	20.1 FT.				Ķ		20-1	
22-24	12, 34 38, 34	80		red brick, FILL. LIGHT ODOR AT 23 FT.				K	$\mathbf{X}$	4	
24-28	507.4	20		oro, Acadiered Shale and CEAT.					2	5	
				Auger Refusal and End of Boring at 25.7 Feet.				<u> </u>		-	
									30		
·	k.		i			-11			<u></u> l		ATLANTIC

²agelofl

ET-ING P31

#ROJECT: EPRI RG&E PROJECT NO: 1309-02 LCCATION: TRANSECT I CATE STARTED: 11/1/91 DATE COMPLETED: 11/1/91 CRILLING CONTRACTOR: NORTH STAR DRILLING CRILLER: HARRY LYON AND RICK BECKWITH CRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER EAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

DROUND ELEVATION (FT): 411.150 PROTECTIVE CASING ELEVATION (FT): 1/A WELL ELEVATION (FT): 1/A DEPTH TO WATER (FT): 14.8 CATUM: 0 'IORTHING: 05832.429970 EASTING: 757920.806140 WEATHER: COUDY, 40'S INSPECTOR: TERRY TAYLOR, 80B RUSSO CHECKED BY: NICOLE RUDERMAN

· · · · · · · · · · · · · · · · · · ·		UF D V		SOIL DESCRIPTION		WELL CONSTRUCTION
SPLI SPOC SAMP DEPT (1t	T DN LE TH BLOWS D PER 8	5 *C	PID (ppm)	eolor, density, š⊖it,admixture, moisture, other notes, öRIGIN	NOHE STAIN STAIN SHEEN HEAVY LITHOL	
0-2 2-4 4-8 8-10 10-12 12-14 14-16 12-14 14-16 18-20 20-22 22-24 24-28 28-30 30-32	4. 8 8. 12 12, 12 17, 9 3. 3 2. 2 2. 1 1. 1 1. 1 2. 2 2. 2 1. 1 2. 2 3. 3 3. 3 1. 3 4. 3 1. 1 2. 1 1. 1 2. 2 3. 3 3. 3 1. 3 4. 3 1. 1 2. 1 1. 1 2. 2 3. 3 3. 3 4. 3 1. 1 2. 1 2. 2 3. 3 3. 3 5. 7 35. 14 7. 14 49. 8 3. 5 7. 7 25. 12 8. 50/.2	75 50 85 45 30 70 30 45 45 50 70 30 50 50 35 45	2.8 8.1 7.3 8.4 8.5 11.2 3.8 8.1 7.5	Brown-grey, tine SALS, some pebbles, trace roots, Fill, Brown ash and stag, trace coal, Fill, Brown, tine SAND, trace wood, coal, and red brick, Fill, Grey, fine silt, some probles and roots, Fill, Black-brown ash and stag, some coke, Fill, Dark grey, fine Silt, some roots, Fill, Black-brown ash and stag, trace coke, Fill, StiGHT (COR FROM 10.4 TO 14.9 FT. Water at 14.8 ft, Trace clay from 18 to 8.3 ft, Trace clay from 18 to 8.3 ft, Dark grey CLAY, some silt, StiGHT PETROLEUM ODOR. Black, coarse SAND, some pebbles, Fill, StiGHT FUEL Sti ODOR. Grey-brown, weathered SHALE, trace sand, STRONG FUEL OIL ODOR. COBBLES and PEBBLES, trace sand, StiGHT ODOR. Grey-brown, weathered SHALE,trace sand.FUEL (COR FROM 28.7 TO 30.8 FT. Top of BEDROCK. Auger Refusal, End of Boring at 32.2 Feet.		ΑΤΙΑΝΙΤΤΟ

PROJECT NO: 1309-02 DCATION: TAR STORAGE TANK AREA CATE STARTED: 11/1/91 DATE COMPLETED: 11/1/91 CRILLING CONTRACTOR: NORTH STAR ORILLING

(

PROJECT: EPRI RGGE

CRILLER: HARRY LYON AND RICK BECKWITH

CRILLING METHOD: 3 1/4" ID HOLLOW STEM AUGER

GROUND ELEVATION (FT): NA PROTECTIVE CASING ELEVATION (FT): 414.14 WELL ELEVATION (FT): 413.690 DEPTH TO WATER (FT): 18 DATUM: 0 'IORTHING: 1154127.098522 EASTING: 757478.332038 WEATHER: COUDY, 40'S INSPECTOR: TERRY TAYLOR, BOB RUSSO



ATIANTIC

Page 1 of 1

PROJECT. EPRI RG&E PROJECT NO: 1309-02 DCATION: TAR STORAGE TANK AREA DAITE STARTED: 11/4/91 CATE COMPLETED: 11/4/91 CRILLING CONTRACTOR: NORTH STAR DRILLING CRILLER: HARRY LYON AND RICK BECKWITH CRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER AMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): 413.940 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 18.7 DATUM: 0 NORTHING: 1154138.580998 EASTING: 757455.944258 WEATHER: SUNNY, HIGH 20'S INSPECTOR: BOB RUSSO, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

			VERY		SOIL DESCRIPTION	v C	ISI ON 1	UAI FAN	AL YSIS	0G Y	F1.)	WELL CONSTRUCTION
SP SPC SAN DEI	LIT DON PLE PTH BLO (1) PER	ws 8⁺	RECO	PID (ppm)	calor, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN HFAVV	SAMPLE AN	ГІТНОГ	DEPTH (	
	- 3. 6				Brown SILT, some time sand, cobbis						تـر	
0-	2 12. I	4	75	0	and pebbles, FILL. Black-brown ash and silt, some slag					$\mathbf{X}$	1	
2-	4 7, 10		15	0	Brown SILT, some cobbles and pebbles, trace white brick and				R	$\mathbf{X}$	-	
4-1	8 12,14	* *	95	30	sand,FILL. SLIGHT TAR ODOR.				K		5-	
8-8	3 3. II 12, 6	3	15	1					╏			
8-10	3.4 5.5				No recovery, rock in basket.				Ŕ	$\langle \rangle$		
10-1;	3, 3 8, 10	5	5	1.1	Brown, coarse SAND, some slag, asn, coal, and clam shells, FILL.SLIGHT				K	$\mathbf{X}$	10-	
2-14	4, 4 8, 10	6	5	.a	Brown SAND, some coal and clamshells, FILL.				Ŕ	$\bigotimes$		
<del>1</del> 8	3, 15 11, 15	26	5	.з Ь	Dark grey, SILT, some peobles, clamshells, and porcelain fragments, trace sand, FILL,				k		15-	
:8-18	8,9			٥ĥ	Brown, coarse SAND, some cobbles, trace shell, FILL.			-	K			
18-20	18, 8	75	1	.7	and cobbles, FILL.				K		7	
20-22	4, 10	35		h	Jense at 18.7 ft. PETROLEUM ODOR.						20-	
22-24	21, 40	80	1.	3	FRÓM 20.5 FT. Black SAND, trace silt, cobbles, and word PETROLEUM ODOR STAIN AT							
24-28	12, 14	40		,	24 FT. Brown-grey weathered SHALE,	-					25-	
28-28	8,5	40	0	K	Trace roots and peoples Street				$\langle \rangle$	$\langle \rangle$		
28-30	3, 8	30	0	-	ODOR. Some cobbles and pebbles.				$\langle$	$\mathbf{i}$		
30-32	7, 11 7, 11	_    30	0						$\sum_{i=1}^{n}$	30	2-1	
32-34	10, 507.1	15	1.1		End of Boring at 32.5 Feet				$\langle \rangle$	N		
										_		
·		<u> </u>								35		ATLANTIC

ATI ANTIC.

PROJECT: EPRI RGGE PROJECT: NO: 1309-02 UCATION: 25 FEET NORTH OF RI2A DATE STARTED: 11/5/91 DATE COMPLETED: 11/5/91 DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLING CONTRACTOR: NORTH STAR DRILLING DRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER JAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): 405.750 PROTECTIVE CASING ELEVATION (FT): NA WELL ELEVATION (FT): NA DEPTH TO WATER (FT): 12.9 DATUM: 0 NORTHING: 1153884.970430 EASTING: 757870.287980 WEATHER: SNOW FLURRIES, 20'S INSPECTOR: BOB RUSSO, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

		ERY		SOIL DESCRIPTION		/IS :0N		AL AM.	AL YSIS	0G Y	(FT.)	WELL CONSTRUCTION
SPLIT SPOON SAMPLE DEPTH (1t)	BLOWS PER 8	RECOV	PID (ppm)	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	HEAVY	SAMPLE AN	ГІТНОГ	DEPTH	
0-2 2-4 4-8 8-8 3-10 (2-14 (4-18 8-20 0-22	BLOWS     PER 8*     5, 4     7, 4     2, 2     3     5, 9     28, 17     22, 20     8, 8     4, 4     4, 5     5, 7     7, 4     4, 5     0, 1     1, 0     1, 0     1, 0     1, 7     50/.1	55 0 45 75 80 70 45 100 55 85	PID (ppm)	Color, density, SOIL, admixture, moisture, other notes, ORIGIN Grey COBBLES and PEBBLES, some brick fragments and sand, trace twigs, FILL. Fire brick, some ash, slag, trace coke, FILL. MUSTY ODOR. Na recovery. Black ash and coke, some siag and coarse sand, FILL. MUSTY ODOR. Brown-black ash and slag, some brick and cobbles, FILL. SLIGHT ODOR. Dark brown SAND, some ash, slag, pebbles and brick, FILL.MUSTY ODOR. Tan, very tine SAND, some ash, and slag, FILL. Grey and red brick, some ash, trace coarse sand, FILL. Black-brown, coarse SAND and ash, FILL. WATER TABLE AT 12.8 FT. Very fine, silty SAND, no ash. Trace pebbles. SLIGHT ODOR. Grey CLAY, trace organic matter, FILL. SLIGHT PETROLEUM ODOR. Coarse SAND, FILL. STAINED BLACK, PETROEUM ODOR. Possible top of BEDROCK at 19.3 ft. Auger Refusal. End of Boring at 20.7 Feet.				HEI				
 	l	<u>_</u>	l	<u>_</u>								ATLANTIC

Page 1 of 1

.

PROJECT. EPRI RG&E PROJECT NO: 1309-02 COCATION: TAR STORAGE TANK AREA DATE STARTED: 11/4/91 DATE COMPLETED: 11/4/91 RILLING CONTRACTOR: NORTH STAR DRILLING CRILLER: HARRY LYON AND RICK BECKWITH CRILLING METHOD: 3 1/4" ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(



			/ERY	10	SOIL DESCRIPTION	VI	SIU NT	JAL AM.	AL YSIS	JGY	(FT.)	WELL CONSTRUCTION
SPL SPC SAM DEF	IT OON PLE TH t)	BLOWS PER 81	RECU	PIO (ppm	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	SHEFN	HEAVY	SAMPLE AN	LITHOL	ОЕРТН	
0-	2	5, 5			Brown-grey SILT, some peobles,					$\propto$	, L	
	 a	5, 14 8, 5	- 85	0	sand lense at 1.1 it., fill. SLIGHT ODOR FROM I TO 2.2 FT.				R	$\mathbf{X}$		
		3,2 8,3			Brown coke, ash, slag, trace coarse sand and pebbles.FILL, LIGHT				$\mathcal{F}$	$\bigotimes$		
4-1		4, 7	- 30	18.2	ODOR. Brown-black SAND, some silt and				K	$\mathbf{X}$	5-	
8-8	3	21, 17			ODOR.				K	$\mathbf{X}$	4	
8-10	2   	4, 10 10, 7	75	44.4	Black SAND, some silt, pebbles, trace cobbles, FILL STAIN AND STRONG				K	$\mathbf{X}$	ر م_	
10-1	2	10, 11 10, 11	95	35	FT. WATER TABLE AT 9.5 FT.				X		107	
12-14	4	e, 8 5, 9	30	4.3	ODOR. Grey SILT and SAND, some clay and Debbles trace coal FILL STRONG				K	$\bigotimes$		
14-16	3	7, 7 9, 17	70	45.5	TAR ODOR.				$\bigotimes$	$\mathbf{X}$	:5-	
18-18		9, 10 3, 11	75	10.4	Brown, coarse SAND, some pebbles, FILL. SLIGHT ODOR.				$\bigotimes$	$\square$		
18-20		4, 7 7, 7	55	0	Brown SILT, some fine sand and cobbles, FILL.							
20-22	2	18, 4 2, 1	90	0	AT 17FT.				•		20-	
22-24		1, 4	70	0	Black, fine SAND, some silt, cobbles. BLACK STAIN AND FUEL OIL ODOR.				••••••		-	
24-28	5	1, 4 5, 14	35		Brown-grey, very fine SAND, some cobbles, LIGHT PETROLEUM ODOR.				•	á	25-	
28-28	10,	50/.4	10	-	Augered through cobbles, hit							
28-30	17 22	, 20 , 35	85		Brown, weathered SHALE.				$\langle$	$\mathbb{N}$	4	
					End of Boring at 30 Feet.				<u> </u>	3	0-    5_	

ATI ANTIC.

Page | of |
BORING R35

Page 1 of 1

PROJECT: EPRI RG&E PROJECT NO: 1309-02 _OCATION: 25 FEET NORTH OF R34 ALONG TRANSECT J DATE STARTED: 11/5/91 DATE COMPLETED: 11/5/91 CRILLING CONTRACTOR: NORTH STAR DRILLING CRILLER: HARRY LYON AND RICK BECKWITH CRILLING METHOD: 3 1/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

(

GROUND ELEVATION (FT): NA PROTECTIVE CASING ELEVATION (FT): 413.98 WELL ELEVATION (FT): 413.700 DEPTH TO WATER (FT): 17.2 DATUM: 0 NORTHING: 1154194.221752 EASTING: 757404.272694 WEATHER: SNOW FLURRIES, 20'S INSPECTOR: BOB RUSSO, BRIAN BITTEKER CHECKED BY: NICOLE RUDERMAN

			/ЕАҮ		SOIL DESCRIPTION	0	VIS CON	IUA ITAI	AL YSIS	0GY	(FT.)	WELL CONSTRUCTION
SPLI SPOC SAMP DEP (1t	IT DN LE TH BLOW ) PER (	S J'	RECO	PID (mq	color, density, SOIL,admixture, moisture, other notes, ORIGIN	NONE	STAIN	SHEEN	HEAVY SAMPLE AN	LITHOL	DEPTH	
	- -		and the second		Brown Still T. room tion cond	-					0-7	
0-2	8,9 8,8	7	5		pebbles, trace glass and ash, FILL.	_			K	$\times$	4	
2-4	12.12	4	5   (	5   	red brick, FILL.					$\langle \langle \rangle \rangle$		
	4, 1				sand and cobbles, FILL.					$\bigotimes$		
4-8	32, 20	85			Black SAND and ash, some cobbles, trace slag, FILL, SLIGHT ODOR.				F	$\bigotimes$	5-	
8-8	10, 14 7, 4	55	C	,   -	Brown-grey color, some silt and coal, no odor.					$\mathbf{X}$		
3-10	8,14	70	0		Dark grey-brown color, trace red brick. SLIGHT PETROLEUM ODOR.				6			ANK PV
	14, 28	_			Grey-orange-brown color,trace coke,no odor.				R	$\otimes$	10-	
:0-12	507.4	- 50	0		Frace porcelain fragments. PETROLEUM ODOR.				_K	$\times$		
12-14	5, 7 8, 50/.3	85	2.3		Brown CLAY, some peobles, trace				X	$\mathbf{X}$	-	
14-18	38, 50	80	4.6	, h	Dark grey SILT, some pebbles and cobbles, FILL.				$\mathbf{\hat{x}}$	$\times$	15-	
18-19	5, 11	- 80			Coke, ash, and slag, some coarse sand, FILL, SLIGHT ODOR.				K	$\sim$	4	
	15, 8		4		Crushed rock,red brick, and clayey silt, FILL. SLIGHT ODOR.				Ŕ	$\bigotimes$		
18-20	5, 7 4, 4	80		$\cap$	PETROLEUN ODOR.WATER TABLE AT					× ×		
20-22	5, 8 12, 8	50	3.2	h	Grey SILT, some sand, cobbles, trace red brick, cobbles from 20.8 ft. PETROLEUM ODOR FROM 19.8 FT.						20-1	
22-24	10, 10 7, 4	20	.4	h	Brown CLAY and COBBLES, trace sand. MODERATE PETROLEUM ODOR AND STAIN.						-	
24-28	8, 10 14, 14	70	1.8		Grey-brown SAND, trace cobbles. BLACK STAIN.						25-	
18-10	10, 22	4.5	<b>.</b>		Brown, coarse SAND, some peobles. SLIGHT ODOR.						1	
20-20	41, 35	45	2.0		Brown, weathered SHALE and CLAY.				$\mathbb{N}$	$\mathbf{N}$	-	
28-30	12, 28 30, <b>45</b>	85							$\backslash$	$\searrow$	-	
				1	End of Boring at 30 Feet.					3	0-]	<u>● • .   ¥</u> .
···	<del></del>			<b></b>		1		<u> </u>		I		ATIANTIC

EL 1.6 R35

FROJECT. EPRI RGGE PROJECT NO: 1309-02 UCATION: 84 FEET EAST OF SE CORNER OF OIL PUMP HOLE CATE STARTED: 11/5/91 DATE COMPLETED: 11/5/91 DRILLING CONTRACTOR: NORTH STAR DRILLING CRILLER: HARRY LYON AND RICK BECKWITH DRILLING METHOD: 31/4"ID HOLLOW STEM AUGER SAMPLING METHOD: 2"X24" & 3"X24" SPLIT SPOON

Į

UROUND ELEVATION (FT): PROTECTIVE CASING ELEVATION (FT): WELL ELEVATION (FT): DEPTH TO WATER (FT): 18.4 DATUM: 0 WORTHING: EASTING: WEATHER: SNOW FLURRIES, 20'S USPECTOR: BOB RUSSO, BRIAN BITTEKER DHECKED BY: NICOLE RUDERMAN

11 11 11 Marcare			VE RY		SOIL DECOMPTION	V IS COI	SIUAL SIUAL	06Y	(F1.)	WELL CONSTRUCTION
SPL SPO SAMF DEP (ft	IT ON   PLE  TH   )	BLOWS PER 8	RECU ²	PID (ppm)	color, density, Solle,admixture, moisture, other notes, ORIGIN	110HE STATN	SHEEN SHEEN HEAVY SAMPLE AN	ГІТНОЦ	ОЕРТН	
									0-1	
0-2	?	9.8 II, II	80	0	Iragments, trace saws and peobles, FILL SLIGHT RETROLEUM					
2-4		7, 8 11, 20	55	0	Black ash, coke, some debbles, trace coarse same, Flour, SciGHT			70	-	
4-8		15, 32 14, 15	40	0	DODR. Dark grey-plack Sill, some sand and pebbles, trace prick, Fill			00	5-	0 0
8-8		8,4 5,4	40	0	COAL TAR ODOR. Brown COBBLES and FEBBLES.			01		0 0 0
8-10		9, 12 14, 9	40	0	FILLSLIGHT COCR. Brown-black SAND, come silt,			201	4	ANK 0 0
10-12		7,8 7,5	:00	0	Brown CLAY, some sand, dobbles, and pebbles, FILL, LIGHT ()DOR,				10-1	
12-14		5, 8 4, 4	30	.3	Grey COBBLES and FEEELES, trace silt, FILL.					
14-18		5, 5	70	4.8	trace clay and coorte cond, FILL, LIGHT PETROLECTION 18.			00	:5-	E F II
18-18		8,8	35	1.7	Red brick fragments, tobbles, and pebbles, trace sational sand, FILL, LIGHT PETROLEUM atoR.				1	NATI 0
18-20		8, 8 5, 30	85	1.9	Grey CLAY, some doct estrace sand, PETROLEUN OCCE, WATER TABLE AT 18.4 FT.				4	
20-22		2, 2	50	3.2	Grey COBBLES and FEEBLES, some sand and silt. SLISHT PETROLEUM ODOR, BLACK STATING FEOM 21 TO				20-	
22-24	2	0, 35	30	1.9	22.0 FT.			00	-	
24-28		4, 14	85	7.4	Some clay.				25-	
28-28	,    	8, 18	80		COBBLES, CLAN, and WEATHERED Shale, Moderate retroleum odor.			0		
28-30	50	D/ .2	25	54	Possible top of SECRICK at 28ft. POSSIBLE TAR STATISTIC. End of Boring at 22.0 Faet.					
								3	-0   	
	·				· · · · · · · · · · · · · · · · · · ·			3	5_]	

## APPENDIX C Historic Data

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet) Laboratory Identification Date Sampled	Part 375 Restricted Use Soil Cleanup Objectives (SCOs) for Industrial Property Use ⁽²⁾	NYSDEC Recommended Soil Cleanup Objective ⁽¹⁾	WS-SB01(19.5-20) A6365201 04/05/2006	WS-SB02(25.5-26) A6365206 04/05/2006	WS-SB02(5.5-6) A6365205 04/05/2006	WS-SB03(22.5-23) A6383903ML 04/06/2006	WS-SB03(25.5-26) A6383905ML 04/06/2006	WS-SB04(21.5-22) A6449003 04/24/2006	WS-SB05(24.7-25.2) A6365203 04/05/2006
Volatile Organic Compounds (µg	g/kg)								
2-Butanone		300	30 U	28 J	40,000 U	780 U	760 U	26 U	7.0 J
Acetone	1,000,000	200	30 U	180 U	40,000 U	780 U	760 U	26 U	38 UJ
Benzene	89,000	60	6.0 U	37 U	22,000	990	150 U	5.0 U	2.0 J
Carbon Disulfide		2700	6.0 U	37 U	8,100 U	160 U	150 U	5.0 U	13 J
Carbon Tetrachloride	44,000	600	6.0 U	37 U	8,100 U	160 U	150 U	5.0 U	8.0 UJ
Chlorobenzene	1,000,000	1,700	6.0 UJ	37 U	8,100 U	160 U	150 U	5.0 U	8.0 UJ
Ethyl Benzene	780,000	5,500	6.0 U	370	230,000	22,000 D	820	5.0 U	8.0 UJ
Methylene Chloride	1,000,000	100	6.0 U	8.0 J	8,100 U	160 U	150 U	5.0 U	8.0 UJ
Styrene			6.0 U	37 U	8,100 U	160 U	200	5.0 U	8.0 UJ
Toluene	1,000,000	1,500	6.0 U	37 U	52,000	570	120 J	5.0 U	8.0 UJ
Total Xylenes	1,000,000	1,200	18 U	190	340,000	14,000	970	16 U	23 UJ
Total VOCs (µg/kg)			ND	600	640,000	38,000	2,100	ND	22

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet) Laboratory Identification Date Sampled	Part 375 Restricted Use Soil Cleanup Objectives (SCOs) for Industrial Property Use ⁽²⁾	NYSDEC Recommended Soil Cleanup Objective ⁽¹⁾	WS-SB06(25-25.5) A6383907 04/06/2006	WS-SB07(20.3-20.5) A6383909 04/06/2006	WS-SB07A(23.2-23.8) 04/27/2006 A6456205	WS-SB08(21.2-21.6) A6449006 04/25/2006	WS-SB08(26.8-27.2) A6449009 04/25/2006	WS-SB09(21-22) A6456302 04/26/2006	WS-SB10(21.5-22) A6449005 04/24/2006
Volatile Organic Compounds (µ	g/kg)								
2-Butanone		300	32 U	29 U	30 U	770 U	35 UJ	18 J	29 U
Acetone	1,000,000	200	38 J	29 U	30 U	770 U	35 U	79	29 U
Benzene	89,000	60	6.0 U	6.0 U	6.0 U	5,300	19,000 D	11 J	6.0 U
Carbon Disulfide		2700	6.0 U	6.0 U	2.0 J	150 U	4.0 J	6.0 U	2.0 J
Carbon Tetrachloride	44,000	600	6.0 U	6.0 U	6.0 U	150 U	7.0 U	6.0 U	6.0 U
Chlorobenzene	1,000,000	1,700	6.0 U	6.0 U	6.0 UJ	150 U	7.0 U	6.0 U	6.0 U
Ethyl Benzene	780,000	5,500	6.0 U	6.0 U	6.0 U	92,000 D	220,000 D	730 DJ	50
Methylene Chloride	1,000,000	100	6.0 U	6.0 U	9.0 U	150 U	7.0 U	7.0 J	6.0 U
Styrene			6.0 U	6.0 U	6.0 U	150 U	7.0 U	6.0 U	6.0 U
Toluene	1,000,000	1,500	6.0 U	6.0 U	6.0 UJ	2,200	18,000 D	60 J	6.0 U
Total Xylenes	1,000,000	1,200	19 U	17 U	18 U	90,000 DJ	260,000 D	2,200 DJ	66
Total VOCs (µg/kg)			38	ND	2.0	190,000	520,000	3,100	120

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet) Laboratory Identification Date Sampled	Part 375 Restricted Use Soil Cleanup Objectives (SCOs) for Industrial	NYSDEC Recommended Soil Cleanup Objective ⁽¹⁾	WS-SB11(23.5-2 A6456301 04/26/2006	24)	WS-SB12(9.5-1 A6456201 04/27/2006	10)	WS-SB13(23-23.5) 04/27/2006 A6456203	WS-SB14(15.1-15 A6466201 04/27/2006	.5)	WS-SB15(17-17.5) A6466203 04/28/2006	WS-SB15A(2 A646620 04/28/200	21-22) 5 06	WS-SB15A(22.7-23.3 A6466206 04/28/2006	) WS- A636 04/04	-TP01 65209 4/2006
	Property Use ⁽²⁾														
Volatile Organic Compounds (µg	g/kg)														
2-Butanone		300	29	U	14	J	740 U	29	U	33 U	28	B U	30 U		42 U
Acetone	1,000,000	200	27	J	79		740 U	29	U	33 U	28	B U	30 U		42 U
Benzene	89,000	60	1.0	J	24		150 U	20		7.0 U	1.0	) J	2.0 J	8	8.0 U
Carbon Disulfide		2700	2.0	J	1.0	J	150 U	6.0	U	7.0 U	1.0	) J	2.0 J	8	8.0 U
Carbon Tetrachloride	44,000	600	6.0	U	6.0	U	150 U	6.0	U	7.0 U	6.0	) U	6.0 U	8	8.0 U
Chlorobenzene	1,000,000	1,700	6.0	UJ	6.0	U	150 U	6.0	U	7.0 U	6.0	) U	6.0 U	8	8.0 U
Ethyl Benzene	780,000	5,500	6.0		4,400	D	1,800	3.0	J	7.0 U	6.0	) U	7.0	4	4.0 J
Methylene Chloride	1,000,000	100	6.0	U	10	J	150 U	14	J	16 J	11	J	10 J	8	8.0 U
Styrene	-		6.0	U	6.0	U	730	18		7.0 U	6.0	) U	6.0 U	8	8.0 U
Toluene	1,000,000	1,500	6.0	UJ	120	J	180 J	29	J	7.0 U	6.0	) U	6.0 U	8	8.0 U
Total Xylenes	1,000,000	1,200	17	U	13,000	D	3,900	78		20 U	17	U U	3.0 J		25 U
Total VOCs (µg/kg)			36		18,000		6,600	160		16	13	3	24	4	4.0

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Notes:

⁽¹⁾ NYSDEC TAGM HWR-94-4046 - Determination of Soil Cleanup Objectives and Cleanup Levels (NYSDEC, Jan. 1994)

⁽²⁾ Part 375 Restricted Use Soil Cleanup Objectives for Restricted Industrial Property Use

-- = Cleanup objective not listed

NA = Not Analyzed

ND = Not Detected

MDL = Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

D = The concentration indicated was obtained from a diluted analytical run.

J = The associated numerical value is an estimated quantity.

R = The result was rejected because one or more QC measures did not meet acceptable criteria.

Bold highlighted value - compound detected above regulatory standard or guidance value in Part 375 SCOs or TAGM 4046.

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet) Laboratory Identification Date Sampled	Part 375 Restricted Use Soil Cleanup Objectives (SCOs) for Industrial	NYSDEC Recommended Soil Cleanup Objective ⁽¹⁾	WS-SB01(19-2 A6365202 04/05/2006	WS-SB02( A63652 04/05/20	5.5-6) 07 006	WS-SB02(25-25.5) A6365208 04/05/2006	WS-SB03(22.5-23.5) A6383904 04/06/2006	WS-SB03(25.5-26.5) A6383906 04/06/2006	WS-SB04(21-22) A6449002 04/24/2006	WS-SB05(24.7-26.5) A6365204 04/05/2006	WS-SB06(24.6-25.6) A6383908 04/06/2006	WS-SB07(20-20.7) A6383910 04/06/2006	WS-SB07A(23-24) A6456206 04/27/2006	WS-SB08(20-21.5) A6449007 04/25/2006
	Filiperty Use			_										
Semi-volatile Organic Compot	inds (µg/kg)	26.400	500	<b>2</b> (00.00		2 000 T	1.40.000	2,400	20 1	2 200	100 11	1.000 11	2.60 11	150.000
2-Methylnaphthalene		36,400	500	2,600,00		2,800 J	140,000	3,400	30 J	2,200 J	420 U	1,800 U	360 U	150,000
Acenaphthene	1,000,000	50,000	2,000	900,00	0	11,000	220,000	7,800	360 U	7,200	84 J	1,800 U	1,200	250,000
Acenaphthylene	1,000,000	41,000	520	86,00	0 J	2,800 J	20,000	2,400	360 U	7,100	23 J	96 J	480 J	29,000 J
Anthracene	1,000,000	50,000	140	450,00	0	8,100 J	92,000	6,400	360 U	17,000	59 J	200 J	1,300	140,000 J
Benz(a)anthracene	11000	224/MDL	840	200,00	0	5,900 J	55,000	4,100	360 U	12,000	240 J	830 J	1,600	100,000 J
Benzo(a)pyrene	1100	61/MDL	920	160,00	0	4,400 J	35,000	2,200	360 U	7,100	190 J	770 J	1,100	76,000
Benzo(b)fluoranthene	11,000	1,100	660	130,00	0	3,800 J	41,000 J	2,200	360 U	5,500	230 J	780 J	1,100	87,000 J
Benzo(g,h,1)perylene	1,000,000	50,000	450	66,00	0 J	1,600 J	15,000 J	770 J	360 U	2,300 J	150 J	520 J	330 J	43,000 J
Benzo(k)fluoranthene	110,000	1,100	290	65,00	0 J	1,700 J	43,000 J	780 J	360 U	2,600 J	78 J	340 J	380	93,000 J
Chrysene	110000	400	750	150,00	0	5,400 J	46,000	3,000	360 U	12,000	210 J	640 J	1,200	90,000 J
Dibenz(a,h)anthracene	1100	14/MDL	140	19,00	0 J	720 J	5,900 J	340 J	360 U	790 J	43 J	140 J	150 J	14,000 J
Fluoranthene	1,000,000	50,000	1,200	480,00	0	12,000	110,000	8,300	360 U	21,000	380 J	1,200 J	3,300	180,000 J
Fluorene	1,000,000	50,000	2,000	510,00	0	7,500 J	120,000	7,200	360 U	10,000	420 U	1,800 U	1,100	180,000
Indeno(1,2,3-cd)pyrene	11,000	3,200	380	58,00	0 J	1,500 J	13,000 J	760 J	360 U	1,900 J	120 J	430 J	380	33,000 J
Naphthalene	1,000,000	13,000	980	3,300,00	0 D	5,700 J	500,000 D	5,900	360 U	4,400	92 J	1,800 U	360 U	330,000
Phenanthrene	1,000,000	50,000	160	1,200,00	0	13,000	320,000	22,000	360 U	14,000	140 J	680 J	1,500	490,000
Pyrene	1,000,000	50,000	1,800	510,00	0	14,000	130,000	9,100	360 U	26,000	340 J	1,100 J	2,900	210,000 J
Total PAHs (µg/kg)		500,000	9,700	11,000,00	0	100,000	1,900,000	87,000	30	150,000	2,400	7,700	18,000	2,500,000
Total CPAHs (µg/kg)		NL	4,000	780,00	0	23,000	240,000	13,000	ND	42,000	1,100	3,900	5,900	490,000
2,4-Dimethylphenol		NL	2,000	J 100,00	0 U	8,200 U	20,000 U	1,900 U	360 U	4,300 U	420 U	1,800 U	360 U	76,000 U
Bis(2-ethylhexyl) phthalate		50,000	2,000	J 100,00	0 U	8,200 U	20,000 U	1,900 U	120 J	4,300 U	420 U	1,800 U	360 U	76,000 U
Dibenzofuran		6,200	2,000	J 150,00	0	930 J	32,000	2,000	360 U	480 J	420 U	1,800 U	820	18,000 J
N-nitrosodiphenylamine		NL	2,000	J 100,00	0 U	8,200 U	20,000 U	1,900 U	360 U	4,300 U	420 U	1,800 U	360 U	76,000 U
Total Cresols	1000000	NL	3,900	J 200,00	0 U	16,000 U	40,000 U	3,700 U	710 U	8,600 U	840 U	3,600 U	710 U	150,000 U
Total SVOCs (µg/kg) ⁽³⁾		500,000	9,700	11,000,00	0	100,000	1,900,000	89,000	150	150,000	2,400	7,700	19,000	2,500,000

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet) Laboratory Identification Date Sampled	Part 375 Restricted Use Soil Cleanup Objectives (SCOs) for Industrial Property Use ⁽²⁾	NYSDEC Recommended Soil Cleanup Objective ⁽¹⁾	WS-SB08(26-27 A6449008 04/25/2006	7.2)	WS-SB09(21- A6456302 04/26/2006	22)	WS-SB10(21-22 A6449004 04/24/2006	2)	WS-SB11(23.5-24) A6456301 04/26/2006	WS-SB1 A645 04/27/	2(8-10) 5202 2006	WS-SB13(22.5-23.8) A6456204 04/27/2006	WS-SB14(14-15.5) A6466202 04/27/2006	WS-SB15(17.5-18) A6466204 04/28/2006	WS-SB15A(22.7-24) A6466207 04/28/2006	WS-TP0 A63652 04/04/20	)1 10 006
Semi-Volatile Organic Compou	unds (µg/kg)																
2-Methylnaphthalene		36,400	31,000		6,400	J	120	J	400 U	7,	800 U	170,000 J	3,200,000 J	77,000 U	360 U	500	J
Acenaphthene	1,000,000	50,000	62,000		83,000	DJ	2,100		3,100	2,	800 J	49,000	290,000 J	16,000 J	8,700 D	2,400	U
Acenaphthylene	1,000,000	41,000	5,500	J	3,800	J	1,000		950	33,	000	62,000	1,700,000	81,000 J	2,500	2,400	U
Anthracene	1,000,000	50,000	33,000		51,000	DJ	800	J	2,700	26,	000	70,000	2,700,000	130,000 J	5,100	2,400	U
Benz(a)anthracene	11000	224/MDL	28,000		39,000	D	620	J	5,300	82,	000	51,000	2,000,000	150,000 J	2,700	2,400	U
Benzo(a)pyrene	1100	61/MDL	15,000		18,000	DJ	420		2,700	70,	000	33,000	1,300,000	110,000	1,900	2,400	U
Benzo(b)fluoranthene	11,000	1,100	17,000		21,000	DJ	470	J	3,400	110,	000 J	48,000 J	2,100,000 J	150,000	1,900 J	2,400	U
Benzo(g,h,i)perylene	1,000,000	50,000	5,600	J	3,500	J	240	J	680	22,	000	11,000 J	430,000 J	38,000 J	510	2,400	U
Benzo(k)fluoranthene	110,000	1,100	5,900	J	9,500	DJ	500	J	1,100	110,	000 J	49,000 J	2,100,000 J	40,000 J	1,900 J	2,400	U
Chrysene	110000	400	21,000		4,800	J	600	J	3,100	80,	000	40,000	1,400,000 J	120,000 J	2,200	2,400	U
Dibenz(a,h)anthracene	1100	14/MDL	2,400	J	2,400		83	J	300 J	11,	000	4,700 J	180,000 J	13,000 J	150 J	2,400	U
Fluoranthene	1,000,000	50,000	68,000		100,000	D	1,200	J	13,000 D	100,	000	110,000	4,600,000	400,000	5,100	2,400	U
Fluorene	1,000,000	50,000	47,000		63,000	DJ	1,200		3,000	9,	700	82,000	2,900,000	98,000	5,700	2,400	U
Indeno(1,2,3-cd)pyrene	11,000	3,200	5,400	J	4,200	J	170	J	830	24,	000	12,000 J	480,000 J	44,000 J	410	2,400	U
Naphthalene	1,000,000	13,000	67,000		32,000	J	240	J	400 U	7,	800 U	320,000 J	12,000,000 J	95,000 U	360 U	1,100	J
Phenanthrene	1,000,000	50,000	140,000		180,000	DJ	3,900		5,100	23,	000 J	210,000	7,800,000	410,000	13,000 D	2,400	U
Pyrene	1,000,000	50,000	54,000		81,000	DJ	1,600		9,800 D	110,	000	100,000	3,000,000	260,000	7,800 D	2,400	U
Total PAHs (µg/kg)		500,000	610,000		700,000		15,000		55,000	810,	000	1,400,000	48,000,000	2,100,000	60,000	1,600	1
Total CPAHs (µg/kg)		NL	95,000		99,000		2,900		17,000	490,	000	240,000	9,600,000	630,000	11,000	ND	,
2,4-Dimethylphenol		NL	9,800	U	370	U	370	U	400 U	7,	800 U	20,000 U	370,000 J	77,000 U	360 U	2,400	U
Bis(2-ethylhexyl) phthalate		50,000	9,800	U	370	U	340	J	400 U	7,	800 U	20,000 U	740,000 U	77,000 U	360 U	2,400	U
Dibenzofuran		6,200	23,000		41,000	DJ	300	J	2,300	7,	800 U	52,000	2,200,000	77,000 U	560 J	2,400	U
N-nitrosodiphenylamine		NL	9,800	U	370	U	370	U	400 U	7,	800 U	3,600 J	740,000 U	77,000 U	360 U	2,400	U
Total Cresols	1000000	NL	20,000	U	750	U	740	U	810 U	16,	000 U	39,000 U	860,000 J	150,000 U	730 U	4,800	U
Total SVOCs (µg/kg) ⁽³⁾		500,000	630,000		740,000		16,000		57,000	810,	000	1,500,000	52,000,000	2,100,000	60,000	1,600	

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Notes:

⁽¹⁾ NYSDEC TAGM HWR-94-4046 - Determination of Soil Cleanup Objectives and Cleanup Levels (NYSDEC, Jan. 1994)

⁽²⁾ Part 375 Restricted Use Soil Cleanup Objectives for Restricted Industrial Property Use

⁽³⁾The total SVOC values include all PAH compounds

-- = Cleanup objective not listed

ND = Not Detected

MDL = Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

D = The concentration indicated was obtained from a diluted analytical run.

J = The associated numerical value is an estimated quantity.

Bold highlighted value - compound detected above regulatory standard or guidance value in Part 375 SCOs or TAGM 4046.

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet)	Part 375 Restricted Use	NYSDEC	Eastern USA	WS-SB01(19-20)	WS-SB02(5.5-6	WS-SB02(25-25	5.5)	WS-SB03(22.5-23.5	) WS-SB03(25.5-26.5)	WS-SB04(21-22)	WS-SB04(6.7-7)	WS-SB05(24.7-26.5)
Laboratory Identification	Soil Cleanup Objectives	Recommended	Background ⁽¹⁾	A6365202	A6365207	A6365208		A6383904	A6383906	A6449002	A6449001	A6365204
Date Sampled	(SCOs) for Industrial	Soil Cleanup	-	04/05/2006	04/05/2006	04/05/2006		04/06/2006	04/06/2006	04/24/2006	04/24/2006	04/05/2006
Dute bumpied	Property Use	Objective (1)		01/00/2000	0110012000	01/00/2000		01/00/2000	01/00/2000	0 112 11 2000	0112112000	01/00/2000
Inorganic Compounds (n	ng/Kg)											
Aluminum		SB	33,000	2,970 J	11,500 J	5,660	J	6,820 J	3,490 J	5,350 J	NA	9,490 J
Antimony		SB	NL	19 J	23 U	J 18	UJ	17 U	J 17 UJ	17 UJ	NA	17 UJ
Arsenic	16	7.5	3 - 12	11	35	4.3		22	8.2	5.1	NA	5.8
Barium	10,000	300	15 - 600	42 U	86 J	47	U	46 U	62 J	44 U	NA	48 J
Calcium		SB	130 - 35,000	36,300	8,350	4,190		9,720	103,000	133,000	NA	98,400
Chromium	6,800	10	1.5 - 40	9.1	36.3	8.1		5.1	4.5	6.0 J	NA	8.0
Cobalt		30	2.5 - 60	14	15 U	12	U	12 U	11 U	11 U	NA	11 U
Copper	10,000	25	1 - 50	117 J	86 J	18	J	38 J	16 J	9.2	NA	39 J
Iron		2,000	2,000 - 550,000	91,300 J	38,800 J	18,000	J	33,700 J	15,200 J	9,520 J	NA	11,000 J
Lead	3,900	SB	200 - 500	849	165	23		106	32	14	NA	23
Magnesium		SB	100 - 5,000	8,640 J	6,280 J	3,440	J	2,010 J	17,500 J	19,600 J	NA	32,000 J
Manganese	10,000	SB	50 - 5,000	280	372	144		105	293	278	NA	275
Mercury	5.7	0.1	0.001 - 0.2	0.12	0.31	0.07		0.14	0.21	0.02 U	NA	0.11
Nickel	10,000	13	0.5 - 25	149 J	33 J	19	J	28 J	9 UJ	10	NA	17 J
Potassium		SB	8,500 - 43,000	1,060 U	1,790 J	1,170	U	1,150 U	1,800 J	3,650	NA	1,500 J
Selenium	6,800	2	0.1 - 3.9	4.2 U	6.2 U	4.7	U	6.6	4.5 U	4.4 U	NA	4.5 U
Vanadium		150	1 - 300	23	19	12	U	12 U	11 U	11 U	NA	11
Zinc	10,000	20	9 - 50	171 J	226 J	40	J	37 J	29 J	10 J	NA	52 J
Total Cyanide	10,000	NL	NL	1.4	3.6	1.1	U	1 U	1 U	0.97 U	1.1 UJ	1.1 U

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

												W10, 00,00 (0.4		THE OF LOCAL A	a)		5
Sample ID(Depth in Feet)	Part 375 Restricted Use	NYSDEC	Eastern USA	WS-SB06(24.6-25	5.6)	WS-SB07(20-20.7)	WS-SB07A(23-24)	•)	WS-SB08(20-21.5)	WS-SB08(26-2	7.2)	WS-SB09(21-	22)	WS-SB10(21-2	2)	WS-SB11(23.5-24	+)
Laboratory Identification	Soil Cleanup Objectives	Recommended	Background ⁽¹⁾	A6383908		A6383910	A6456206		A6449007	A6449008		A6456302		A6449004		A6456301	
Date Sampled	(SCOs) for Industrial	Soil Cleanup		04/06/2006		04/06/2006	04/27/2006		04/25/2006	04/25/2006		04/26/2006		04/24/2006		04/26/2006	
	Property Use	Objective (1)															
Inorganic Compounds (n	ng/Kg)																
Aluminum		SB	33,000	6,190	J	2,890 J	3,670 J		3,660 J	4,410	J	3,040	J	4,160	J	4,950 J	ſ
Antimony		SB	NL	18	UJ	17 UJ	16 UJ	J	17 UJ	18	UJ	17	UJ	15	UJ	19 U	J
Arsenic	16	7.5	3 - 12	4.0		4.5	5.1		25 J	5.3		146		5.5		14	
Barium	10,000	300	15 - 600	48	U	48 J	42 U	J	44 U	48	U	45	U	44	J	51 U	J
Calcium		SB	130 - 35,000	28,800		46,000	99,800 J		47,900 J	37,000	J	56,100	J	120,000	J	88,400 J	ſ
Chromium	6,800	10	1.5 - 40	8.6		7.6	6.9 J		5.8 J	7.1	J	8.0		7.8		7.6 J	ſ
Cobalt		30	2.5 - 60	12	U	12 U	11 U	J	12	12	U	11		10	U	13 U	J
Copper	10,000	25	1 - 50	31	J	103 J	18		48	23		72		26		30	
Iron		2,000	2,000 - 550,000	14,400	J	8,160 J	15,400 J		39,100 J	16,200	J	60,000		13,100	J	23,900	
Lead	3,900	SB	200 - 500	40		37	43		52	45		103		22		41	
Magnesium		SB	100 - 5,000	11,200	J	14,900 J	22,200 J		6,020 J	18,600	J	18,200	J	31,700	J	19,300 J	Γ
Manganese	10,000	SB	50 - 5,000	199		161	407		136	172		244		309		394	
Mercury	5.7	0.1	0.001 - 0.2	0.22		0.21	0.10		0.23	0.11		0.38		0.03		0.14	
Nickel	10,000	13	0.5 - 25	15	J	9.2 UJ	9.2		26	10		29		9.2		11	
Potassium		SB	8,500 - 43,000	1,320	J	1,180 J	1,640		1,110 U	1,340		1,530		2,260		2,180	
Selenium	6,800	2	0.1 - 3.9	4.8	U	4.6 U	4.2 U	J	4.4 U	4.8	U	22.1		3.9	U	5.1 U	J
Vanadium		150	1 - 300	12	U	12 U	11 U	J	12	12	U	11	U	10	U	13 U	J
Zinc	10,000	20	9 - 50	64	J	54 J	28 J		32 J	33	J	61	J	25	J	58 J	j -
																	Τ
Total Cyanide	10,000	NL	NL	1.2	U	1 U	0.97 U	J	1 U	1.1	U	1.2		0.96	U	1 L	J

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample ID(Depth in Feet)	Part 375 Restricted Use	NYSDEC	Eastern USA	WS-SB12(8-1	0)	WS-SB13(22.5-2	3.8)	WS-SB14(14-1	5.5)	WS-SB15(17.5-	-18)	WS-SB15A(22.7	-24)	WS-TP01	i l
Laboratory Identification	Soil Cleanup Objectives	Recommended	Background ⁽¹⁾	A6456202		A6456204		A6466202		A6466204		A6466207		A6365210	)
Date Sampled	(SCOs) for Industrial	Soil Cleanup	_	04/27/2006		04/27/2006		04/27/2006		04/28/2006		04/28/2006		04/04/200	6
	Property Use	Objective (1)													
Inorganic Compounds (n	ng/Kg)														
Aluminum		SB	33,000	3,870	J	4,680	J	4,600	J	4,370	J	3,360	J	12,700	J
Antimony		SB	NL	17	UJ	17	UJ	16	UJ	17	UJ	17	UJ	22	UJ
Arsenic	16	7.5	3 - 12	7.0		8.8		6.5		7.4		4.6		11	
Barium	10,000	300	15 - 600	45	U	47	U	43	U	166	J	45	U	170	J
Calcium		SB	130 - 35,000	61,300	J	11,400	J	76,100		12,500	J	106,000	J	15,000	
Chromium	6,800	10	1.5 - 40	5.7	J	41.7		8.9		4.4	J	5.9	J	8.1	
Cobalt		30	2.5 - 60	11	U	12	U	11	U	11	U	11	U	15	U
Copper	10,000	25	1 - 50	12		33		20		16		15		15	J
Iron		2,000	2,000 - 550,000	13,500	J	131,000	J	10,600		6,050	J	11,300	J	12,900	J
Lead	3,900	SB	200 - 500	52		14		26		30		60		50	
Magnesium		SB	100 - 5,000	27,800	J	2,540	J	37,100	J	3,730	J	34,400	J	3,080	J
Manganese	10,000	SB	50 - 5,000	262		1,470		288		140		332		86	
Mercury	5.7	0.1	0.001 - 0.2	0.04		0.04		0.17		0.08		0.11		0.10	
Nickel	10,000	13	0.5 - 25	9.0	U	14		11		9.2	U	9.0	U	169	J
Potassium		SB	8,500 - 43,000	2,040		1,160	U	1,540		1,140	U	2,070		1,480	U
Selenium	6,800	2	0.1 - 3.9	4.5	U	7.7		4.3	U	4.6	U	4.5	U	5.9	U
Vanadium		150	1 - 300	11	U	78		11	U	11	U	11	U	1,080	
Zinc	10,000	20	9 - 50	144	J	15	J	49	J	40	J	24	J	16	J
Total Cyanide	10,000	NL	NL	1	U	1.2	U	1.6		0.99	U	1	U	1.2	U

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Notes:

⁽¹⁾ NYSDEC TAGM HWR-94-4046 - Determination of Soil Cleanup Objectives and Cleanup Levels (NYSDEC, Jan. 1994)

⁽²⁾ Part 375 Restricted Use Soil Cleanup Objectives for Restricted Industrial Property Use

NA = Not Analyzed

-- = Cleanup objective not listed

MDL = Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

D = The concentration indicated was obtained from a diluted analytical run.

J = The associated numerical value is an estimated quantity.

SB = Site Background.

Bold highlighted value - compound detected above regulatory standard or guidance value in Part 375 SCOs or TAGM 4046.

#### APPENDIX C - TABLE C4 OIL-TAR SEPARATION INVESTIGATION - GROUNDWATER VOC RESULTS

Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample Designation Laboratory Identification	NYSDEC TOGS 1.1.1	WS-MW20 A6594901	1	WS-MW2 A684940	01 1	WS-MW2 A658970	02 2	WS-MW2 A684940	02 3	WS-RW A658970	01 01	WS-RW A68494(	01 04
Date Sampled	Class GA ¹	05/25/2006	5	07/26/200	6	05/24/200	6	07/26/200	6	05/24/20	06	07/25/20	06
Volatile Organic Compounds (µg	/L)												
2-Butanone	NL	25 U		25	U	3.0	J	120	U	25	U	25	U
Acetone	50**	25	U	25	U	25	U	120	U	25	U	4.0	J
Benzene	1	5.0	U	5.0	U	75		65		44		130	D
Ethyl Benzene	5*	5.0	U	5.0	U	220	D	320		86		260	D
Methylene Chloride	5*	5.0	U	5.0	U	5.0	U	3.0	J	5.0	U	5.0	U
Toluene	5*	5.0	U	5.0	U	110	D	130		9.0		22	
Total Xylenes	5*	15	U	15	U	510	D	680		62		160	D
Total VOCs (µg/L)	NL	ND		ND		920		1,200		200		580	

Notes:

NA = Not Analyzed

NL = Not Listed

ND = Not Detected

MDL - Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

D = The concentration indicated was obtained from a diluted analytical run.

J = The associated numerical value is an estimated quantity.

Bold highlighted value - compound detected above regulatory standard or guidance value.

¹Class GA Drinking Water Standard or Guidance Value

*Principal Organic Contaminant Standard

**Class GA Guidance Value

^Applies to the sum of cis- and trans- dichloropropene

#### APPENDIX C - TABLE C5 OIL-TAR SEPARATION INVESTIGATION - GROUNDWATER SVOC RESULTS

Sample Designation	NYSDEC	WS-MW201		WS-MW201		WS-MW202	2	WS-MW202	2	WS-RW01		WS-RW01	
Laboratory Identification	TOGS 1.1.1	A6594901		A6849401		A6589702		A6849403		A6589701		A6849404	
Date Sampled	Class GA ¹	05/25/2006		07/26/2006		05/24/2006		07/26/2006		05/24/2006		07/25/2006	
1													
Semi-volatile Organic Compound	s (µg/L)												
2-Methylnaphthalene	NL	10	U	10	U	300		430	DJ	140		120	
Acenaphthene	20**	12		13		210		300	DJ	160		140	
Acenaphthylene	NL	1.0	J	0.80	J	11	J	11		17	J	16	
Anthracene	50**	10	U	2.0	J	49		18		29	J	22	
Benz(a)anthracene	0.002**	1.0	J	1.0	J	18	J	2.0	J	16	J	9.0	J
Benzo(a)pyrene	ND	1.0	J	0.50	J	11	J	1.0	J	12	J	6.0	J
Benzo(b)fluoranthene	0.002**	1.0	J	10	U	12	J	1.0	J	10	J	6.0	J
Benzo(g,h,i)perylene	NL	10	U	10	U	3.0	J	10	U	4.0	J	3.0	J
Benzo(k)fluoranthene	0.002**	1.0	J	10	U	5.0	J	0.70	J	4.0	J	2.0	J
Chrysene	0.002**	1.0	J	0.70	J	17	J	2.0	J	13	J	7.0	J
Dibenz(a,h)anthracene	NL	10	U	10	U	47	U	10	U	47	U	0.80	J
Fluoranthene	50**	4.0	J	3.0	J	47		13		33	J	22	
Fluorene	50**	1.0	J	3.0	J	94		88		66		64	
Indeno(1,2,3-cd)pyrene	0.002**	10	U	10	U	4.0	J	10	U	4.0	J	2.0	J
Naphthalene	10**	10	U	10	U	4,000	D	5,400	D	47	U	1,700	D
Phenanthrene	50**	10	U	0.60	J	110		73		50		83	
Pyrene	50**	5.0	J	3.0	J	37	J	8.0	J	36	J	22	
Total PAHs (µg/L)	NL	28		28		4,900		6,300		590		2,200	
Total CPAHs (µg/L)	NL	5.0		2.2		67		6.7		59		33	
2,4-Dimethylphenol	50**	10	U	10	U	11	J	27		47	U	3	J
2-Methylphenol	NL	10	U	10	U	18	J	22		47	U	10	U
4-Methylphenol	NL	10	U	10	U	14	J	15		47	U	10	U
bis(2-Ethylhexyl)phthalate	5	10	U	10	U	47	U	10	U	47	U	4.0	J
Dibenzofuran	NL	10	U	1.0	J	80		87		18	J	28	
Phenol	1***	10	U	10	U	9.0	J	5.0	J	47	U	10	U
Total Cresols	NL	19	U	19	U	94	U	37		95	U	19	U
Total SVOCs $(ug/L)^2$	NI	20		20		5 100		( 500		(10		2 200	
10tai 570C8 (µg/L)	INL	28		29		5,100		0,500		610		2,300	

#### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Notes:

NA = Not Analyzed

NL = Not Listed

MDL = Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

D = The concentration indicated was obtained from a diluted analytical run.

J = The associated numerical value is an estimated quantity.

Bold highlighted value - compound detected above regulatory guidance value.

¹Class GA Drinking Water Standard or Guidance Value

²The total SVOC values include all PAH compounds

*Principal Organic Contaminant Standard

**Class GA Guidance Value

***Applies to total chlorinated phenolic compounds

CPAHs - Carcinogenic PAHs which are shown in bold and italics

### APPENDIX C - TABLE C6 OIL-TAR SEPARATION INVESTIGATION - GROUNDWATER INORGANIC RESULTS

### Remedial Investigation Work Plan West Station Former MGP Site, Rochester, NY

Sample Designation Laboratory Identification	NYSDEC TOGS 1.1.1	WS-MW201 A6594901		WS-MW201 A6849401		WS-MW202 A6589702		WS-MW202 A6849403		WS-RW01 A6589701		WS-RW01 A6849404	
Date Sampled	Class GA ¹	05/25/2006		07/26/2006		05/24/2006		07/26/2006		05/24/2006		07/25/2006	
Inorganic Compounds (μg/L)													
Aluminum	NL	706		585		733	J	200	U	2,440		1,400	
Arsenic	25	10	U	10	UJ	10	U	11	J	10	U	10	UJ
Barium	1,000	75.7		55.1		280		323		289		221	
Calcium	NL	118,000		104,000		120,000		136,000		164,000		126,000	
Chromium	50	4.0	U	4.0	UJ	4.0	U	4.0	UJ	24.3		30.5	J
Copper	200	10	U	10	UJ	10	U	10	UJ	20.5		10	UJ
Iron	300	1,390		1,220		20,000		24,500		4,600		2,320	
Lead	25	5.0	U	5.0	UJ	8.9		5.0	UJ	13.1		7.1	J
Magnesium	35,000**	45,000		37,200		26,200		29,300		87,600		70,700	
Manganese	300	107		92.2		148		172		107		81	
Nickel	100	10	U	10	UJ	10	U	10	UJ	21.6		22.2	J
Potassium	NL	9,900		7,280		15,200		14,700		42,200		33,300	
Sodium	20,000	280,000		224,000		35,000		39,400		898,000		735,000	
Zinc	2,000**	10	U	10	UJ	10	U	10	UJ	17.1		10	UJ
Total Cyanide	200	25.4	J	10	UJ	10	UJ	10	U	10	UJ		R

Notes:

NA = Not Analyzed

NL = Not Listed

MDL - Minimum Detection Limit

U = The material was analyzed for but not detected at or above the MDL. The associated numerical value is the sample reporting limit.

J = The associated numerical value is an estimated quantity.

R = The result was rejected because one or more QC measures did not meet acceptable criteria

Bold highlighted values - compound detected above regulatory guidance value.

¹Class GA Drinking Water Standard or Guidance Value

**Class GA Guidance Value

## APPENDIX D Remedial Investigation Field Sampling Plan

## **Remedial Investigation Field Sampling Plan**

West Station Former Manufactured Gas Plant Site Rochester, New York

Prepared for:

Rochester Gas and Electric Corporation 89 East Avenue Rochester, New York 14649-0001

Prepared by:

**Geomatrix Consultants, Inc.** 90B John Muir Drive, Suite 104 Amherst, New York 14228

February 2008

Project No. 12661.002

## TABLE OF CONTENTS

Page
------

1.0	INTRODUCTION						
	1.1	PROJECT OBJECTIVES	1				
	1.2	OVERVIEW OF INVESTIGATION FIELD ACTIVITIES					
	2.1	SAMPLE LABELING, PACKING, AND SHIPPING					
	2.2	EQUIPMENT DECONTAMINATION					
		2.2.1 Back-Hoe and Drill Rig Decontamination	5				
		2.2.2 Sampling Equipment Decontamination	5				
	2.3	GEOPHYSICAL SURVEY METHOD	6				
	2.4	TEST PIT EXCAVATIONS	6				
	2.5	SURFACE SOIL SAMPLE COLLECTION	7				
	2.6	SOIL BORINGS AND SUBSURFACE SOIL SAMPLE COLLECTION	8				
	2.7	EXISTING MONITORING WELL ASSESSMENT AND OVERBURDEN MONITORING					
	WELL	WELL CONSTRUCTION					
	2.9	BEDROCK CORING AND PACKER TESTING	9				
	2.10	MULTI-LEVEL WELL SYSTEM AND/OR NESTED BEDROCK WELL INSTALLATIO	ns11				
	2.11	Well Development and Hydraulic Testing	12				
	2.12	GROUNDWATER ELEVATION MEASUREMENTS AND NAPL DETECTION	12				
	2.13	GROUNDWATER SAMPLING	13				
	2.14	AIR MONITORING	14				
	2.15	SEDIMENT PROBING AND SAMPLE COLLECTION	14				
3.0	FIELD INSTRUMENTS						
	3.1	PORTABLE PHOTO IONIZATION ANALYZER	16				
	3.2	AIR BORNE PARTICULATE (DUST) MONITOR	16				
	3.3	MULTI-PARAMETER FLOW-THROUGH CELL	16				
	3.4	TURBIDITY METER	17				

## ATTACHMENTS

Attachment A	Daily Field Forms
Attachment B	Sample Chain-of-Custody Form
Attachment C	Field Operating Procedures
Attachment D	Drive Point Piezometer Installation

## **REMEDIAL INVESTIGATION FIELD SAMPLING PLAN**

West Station Former Manufactured Gas Plant (MGP) Site Rochester, New York Site No. V00593-8 Index No. B-0535-98-07

## **1.0 INTRODUCTION**

This Field Sampling Plan (FSP) supports the Remedial Investigation Work Plan (RIWP) prepared by Geomatrix Consultants, Inc. (Geomatrix) for the West Station Former Manufactured Gas Plant (MGP) Site, located in Rochester, New York. The FSP addresses the field procedures and sample collection methods to be used during implementation of the investigative field activities. The FSP will be used in conjunction with the Work Plan, the Quality Assurance Project Plan (QAPP), the project specific Health and Safety Plan (HASP), and the Community Air Monitoring Plan (CAMP). The QAPP is provided in Appendix E of the RIWP and the CAMP and HASP are separate projectspecific documents.

## **1.1 PROJECT OBJECTIVES**

The remedial investigation activities described in Section 2.0 if the RIWP will supplement prior investigation results and will generate a combined data set sufficient to assess:

- Extent of on-site MGP-related material (i.e, NAPL, tar-like material);
- Nature and extent of chemical constituents in Site media (i.e., soil, groundwater); and
- Potential for human health and ecological risks posed by MGP and/or non-MGP derived chemical constituents at the Site.

## **1.2 OVERVIEW OF INVESTIGATION FIELD ACTIVITIES**

To obtain information necessary to meet the investigation objectives stated above, the following activities will be conducted:

- Completion of an electromagnetic (EM) geophysical survey to locate buried structures (i.e., pipes, wires, foundations, etc.) containing metal in an effort to locate unknown below grade features. Since the Site is comprised of fill the results of the EM survey may or may not provide data completely supportive of the RI
- Completion of subsurface soil sampling to investigate the nature and extent of on-site impacts from potential MGP materials, and to evaluate potential migration pathways

- Collection and analysis of surface soil samples to assess the nature and extent of MGP impacts and support qualitative risk assessment
- Identification of historic MGP process locations and completion of soil borings including collection and analyses of subsurface soil samples to supplement the previous Site characterization data
- Assessment of existing piezometer/monitoring well integrity for use during the RI
- Installation of overburden and bedrock monitoring wells and collection of groundwater samples from new and existing wells to: 1. supplement previous groundwater characterization data and assessment; 2. identify areas where mobile separate phase liquid is present, if any; and 3. characterize groundwater flow directions in overburden and bedrock where Site-related constituents are identified
- Characterization and sampling of near shore-stream bed sediment in the Genesee River to provide initial identification of nature and extent of sediment impacts and to compare with the sediment criteria guidance by NYSDEC Division of Fish and Wildlife

The types of samples, sampling locations, and sample quantities for each field sampling activity are described in detail in Section 2.0 of the RIWP. Laboratory analytical methods are tabulated in the QAPP presented in Appendix E of the RIWP.

## 2.0 FIELD METHODS

Known underground utilities will be identified prior to any drilling, digging, or subsurface sampling. Publicly owned underground utilities will be located by contacting Dig Safely New York so that underground utilities can be marked at the Site. Other potential hazards such as traffic, over head power lines, and building hazards will be identified during a site reconnaissance visit.

The following is a general list of equipment necessary for sample collection:

- Stainless steel spoons and bowls for compositing samples;
- Appropriate sample containers provided by the laboratory (kept closed and in laboratory supplied coolers until the samples are collected);
- Pre-preserved sample containers for aqueous samples;
- Chain of custody record forms
- Log book, field sampling records, and indelible ink pens and markers;
- Laboratory grade soap (i.e., Alconox), reagent grade solvents, and distilled water to be used for decontaminating equipment between sampling locations;
- Buckets, plastic wash basins, and scrub brushes for decontaminating equipment;
- Camera and film;
- States, flags, and/or spray paint to identify sampling locations;
- Shipping labels and forms;
- Packing/shipping material for sample bottles;
- Clear plastic tape;
- Duct tape;
- Aluminum foil;
- Re-sealable plastic bags; and
- Portable field instruments, including a photoionization detector (PID), airborne particulate monitor, water quality parameter meters (redox potential, dissolve oxygen and specific conductivity meters), and water-level indicator.

Field log books will be maintained by the field team leader and other team members to provide a daily record of significant events, observations, and measurements during the field investigation.

Information pertinent to the field investigation and/or sampling activities will also be recorded in the log books. The books will be bound with consecutively numbered pages. Daily field forms can be found in Attachment A.

Original data recorded in the field log books and Chain of Custody Records will be written with indelible ink. If an error is made on the field forms, that individual will make corrections simply by crossing a single line through the error and entering the correct information. The erroneous information will not be erased. Any subsequent error discovered on a field form will be corrected by the person who made the entry. Subsequent corrections will be initialed and dated.

## 2.1 SAMPLE LABELING, PACKING, AND SHIPPING

Each sample will be given a unique identifier. With this type of identification, no two samples will have the same label.

Samples will be promptly labeled upon collection with the following information:

- Project number and site;
- Unique sample identification;
- Analysis required;
- Data and time sampled;
- Sample type (composite or grab); and
- Preservative, if applicable.

Clear tape will be secured over the sample label and the chain-of-custody will be initiated. A sample chain-of-custody form is included in Attachment B. Appropriate sample containers, preservation methods, and laboratory holding times for each sample will be applied as identified in the QAPP (Appendix E of the RIWP).

Sample bottle/jars will be packed in coolers containing the following:

- A drain plug (if present) that has been sealed with duct tape;
- Water ice packaged in re-sealable plastic bags;
- Appropriate packaging material to help ensure sample integrity while being transported; and
- The completed chain-of-custody in a re-sealable plastic bag, taped to the inside cover of the cooler.

The cooler will then be sealed with tape. Appropriate shipping labels, such as "This End Up" and "Fragile" stickers will be affixed to the cooler. Samples will be received by a courier provided by the laboratory each day samples are collected and transported to the laboratory. A copy of the signed chain-of-custody forms will be retained by the sampler and forwarded to the project files after sample pickup.

## 2.2 EQUIPMENT DECONTAMINATION

## 2.2.1 Back-Hoe and Drill Rig Decontamination

A decontamination pad (decon pad) will be constructed on-site in a manner sufficient to contain material generated during the cleaning of equipment and supplies. The decon pad will be constructed in such away to facilitate collection of water. The equipment that will be decontaminated between each test pit and drilling location will include the back hoe bucket, drilling rig (components of which may have contacted impacted material), augers, bits, rods, tools, split spoon samplers, and tremie pipe. The equipment will be cleaned on the decontamination pad with a high pressure hot water "steam cleaner" unit and scrubbed with a wire brush, as needed, to remove dirt, grease, and oil before beginning work in the project area. If heavy accumulations of tars or oils are present on the equipment, a citrus-based cleaner (e.g., Citra-Solv® or equivalent) may be used to aid in equipment cleaning. Tools, drill rods, and augers will be placed on sawhorses, decontaminated pallets, or polyethylene plastic sheets following steam cleaning. Direct contact with the ground will be avoided. Decontamination water will be contained in a dedicated polyethylene tank or 55-gallon drums staged on site. Open-top drums will remain closed when not in use.

Following decontamination of site equipment, the decontamination pad will be decommissioned. The decommissioning will be completed by:

- Transferring the bulk of the remaining liquids and solids into the drums, tanks, or roll-offs to be provided by RG&E or the drilling subcontractor for these materials; and
- Rolling the sheeting used in the decontamination pad onto itself to prevent discharge of the remaining materials to the ground surface. Once rolled up, the polyethylene sheeting will be placed in the containers used for disposal of personal protective equipment (PPE) and disposable equipment.

PVC monitoring well riser and well screens will be factory sealed in plastic and do not require decontamination.

## 2.2.2 Sampling Equipment Decontamination

Prior to collecting samples to be submitted for chemical analysis, non-dedicated bowls, spoons, hand augers, bailers, and filtering equipment will be washed with potable water and a detergent (e.g., Alconox). Decontamination may take place at the sampling location as long as liquids are contained in pails, buckets, etc. The sampling equipment will then be rinsed with potable water and a distilled water rinse. A 10 percent "pesticide-grade" methanol rinse will be used if oily MGP

source materials are encountered in the samples. When sampling for inorganic constituents in an aqueous phase, an additional rinse step will be added prior to the rinse with methanol. The rinse step will entail a rinse with a 10 percent "ultra pure-grade" nitric acid followed by a distilled water rinse. Between rinses, equipment will be placed on polyethylene sheets or aluminum foil, if necessary. At no time will washed equipment be placed directly on the ground. Equipment will either be used immediately or wrapped in plastic or aluminum foil for storage or transportation from the designated decontamination areas to the sampling locations.

## 2.3 GEOPHYSICAL SURVEY METHOD

This guideline presents a method for completing EM geophysical surveys to identify possible subsurface metallic features that may be associated with MGP source material or may pose a concern to subsurface investigation. The survey will extend to accessible areas of the Site.

The GEONICS EM61 High Sensitivity Metal Detector (EM61) and solid state data logger will be used for surveying the near surface (0 to 11 feet) for the presence of metallic or metal-containing objects. The EM61 is a portable time-domain EM unit. The device detects both ferrous and nonferrous metals. The EM61 is sensitive enough to detect a single 55-gallon drum at a depth of over 10 feet, yet is relatively insensitive to nearby anthropogenic sources of noise such as fences, buildings and power lines. It will be pulled in the trailer mode with an odometer mounted on the axle to trigger the data logger.

The EM61 generates a primary EM field at a rate of 150 pulses per second. After each transmitted pulse, the transmitter turns off and the induced EM field is allowed to decay. The receivers are then turned on; measuring the strength of this decayed secondary EM field between each pulse. Because EM fields decay much more rapidly in normal soils than in metals, the EM61 instrument is relatively insensitive to terrain conductivities and is highly sensitive to metals.

The unit will be configured to digitally collect a data point at 0.62-foot intervals along lines spaced 3 or 5 feet apart. Data will be stored on a digital data logger and archived to a laptop computer. The collected data will be processed using GEOSOFT software and plotted as profile lines, gridded, filtered and color-contoured. Anomalous responses will be annotated and discussed in a report. The report will include a color-contoured map of the EM61 survey results showing any buried USTs, metallic pipes and metallic fill material present.

## 2.4 TEST PIT EXCAVATIONS

Test pits will be excavated with a rubber-tired backhoe with an extend-a-hoe bucket or a trackmounted excavator. The excavation equipment will have a maximum reach greater than 15 feet. Buried utilities will be identified and marked prior to excavation. The depth of the test pit will be determined by the field geologist and will typically be limited by the safe reach of the backhoe, but may also be limited by the stability of the excavated materials (i.e., wall stability). Test pits deeper than 4 feet will not be entered.

The physical dimension of each test pit, scaled sketch of one wall of the pit showing lithologic contacts, zones of groundwater seepage, other special features (jointing, boulders, zones of contamination, etc.), rate of groundwater inflow, and depth to groundwater will be recorded in the field log book (see Attachment A). Soils will be examined by a hydrogeologist and classified in accordance with the Unified Soil Classification System (USCS) as described in the Field Operating Procedure (FOP) included in Attachment C. The hydrogeologist and/or geologist will identify the contrast between the soil properties of the fill and the native soil (if encountered) through color, grain size, texture and moisture content and closely examined for suspected chemical impact and/or MGP source materials. Sample collection will be selected from areas of suspected chemical impact and/or MGP source materials, if present, and suspected "unimpacted" soil as stated in the RIWP.

Soil samples will be collected either by using a stainless steel spade to scrape soil samples from the sidewalls of the excavated test pit or, if the test pit is greater than 4 feet in depth, samples will be collected from the center of the backhoe bucket using the stainless steel hand trowels. Soil samples will be field screened for the presence of vapors containing volatile organic compounds (VOCs) using a field PID. Samples will be analyzed for parameters summarized in Table 2 of the QAPP (Appendix E).

After sample collection is complete, each test pit will be backfilled immediately to grade in the reverse sequence in which soil was removed (i.e. last soil out of the hole will be the first soil back into the hole so that excavated soils will be returned as close as possible to the same elevation from which they originated). The ground surface will be restored to a similar pre-existing condition. A wooden stake or pin flag will be placed to mark each end of the test pit for surveying.

## 2.5 SURFACE SOIL SAMPLE COLLECTION

Surface soil samples will be collected from 0 to 2 inches below the sod layer using dedicated or decontaminated stainless steel sampling equipment (spoons or hand trowels). Samples will be placed in pre-cleaned sample jars and will be analyzed for parameters summarized in Table 2 of the QAPP (Appendix E of the RIWP). Soil samples will be examined by a hydrogeologist and/or geologist and classified in accordance USCS as described in the Field Operating Procedure (FOP) included in Attachment C.

## 2.6 SOIL BORINGS AND SUBSURFACE SOIL SAMPLE COLLECTION

Soil borings will be advanced to assess subsurface soil conditions and facilitate monitoring well installation. Soil borings used to assess subsurface soil conditions will be will be advanced fill/native soil interface soil using 2 1/4-inch ID hollow stem augers (HSA) and sampled continuously using 2-inch O.D. split spoon samplers. Standard penetration tests (in blow counts) will be recorded for estimating the relative in-situ compressive strength of subsurface materials. A macrocore soil sampler may be used in place of the split spoon sampler if more than 80% material recovery can be demonstrated in the field. The macrocore sampler would be advanced using a pneumatic auto-hammer and blow counts would not be recorded.

Soil samples will be examined by a hydrogeologist and classified in accordance with the USCS and Geomatrix protocols on field borehole logs as described in the Field Operating Procedure (FOP) included in Attachment C. The hydrogeologist and/or geologist will identify the contrast between the soil properties of the fill and the native soil through color, grain size, texture and moisture content and closely examined for suspected chemical impact and/or MGP source materials. Soil samples will be field screened for the presence of volatile organic compounds (VOCs) using a field PID. Sample jar head space measurements will be recorded on the filed logs.

Borings that will be converted to an overburden groundwater monitoring well will be advanced using 4 1/4-inch I.D. HSA. The unconsolidated deposits will be sampled continuously using standard 2-inch diameter split spoon samplers or a macro-core sampler advanced with the drilling rig auto-hammer.

For boring/wells completed between the River's edge and upland portion of the Plant Area, ATVmounted or skid-mounted drilling equipment will be used. Brush clearing and grading may be necessary to facilitate access and provide a safe, stable work area. Drilling equipment may be placed on the river side of the concrete retaining wall using a truck-mounted crane. Under no circumstance will drilling occur in an unsafe area (e.g. steep slope, unstable soil). The drilling subcontractor will be required to develop a Health and Safety Plan that considers the safety of the work area and only areas considered safe by the drilling company will be investigated. Truckmounted drilling equipment will be utilized at the Plant Area.

## 2.7 EXISTING MONITORING WELL ASSESSMENT AND OVERBURDEN MONITORING WELL CONSTRUCTION

Monitoring wells installed during previous investigations will be located and evaluated for usability. Monitoring well integrity will be evaluated by inspecting the surface seal and sounding the bottom depth of the well to compare the measured depth with well installation records. Any

accumulated sediment on the bottom of the well will be removed using well development methods described below. If a monitoring well is determined not to be usable, it will be decommissioned by overdrilling the well materials, pulling the well riser and screen, and tremie-grouting the borehole. Decommissioned wells will be replaced with a new well having a similar screen depth interval if they are considered necessary to meet the objectives of the RI.

New overburden monitoring wells will be constructed of 2-inch I.D. flush-joint Schedule 40 PVC riser and screen (0.01-inch slot size). The monitoring well screen will be 10 feet in length. Approximately 6-inches of silica sand will be placed at the bottom of each boring as a base for the well screen and as part of the sand pack. The well casing will be set on the sand layer and the remainder of the sand pack will be placed in the borehole annulus to a level of 2 to 3-feet above the top of the well screen. If DNAPL is anticipated to be present, a two foot deep sump (drilled in to the underlying bedrock) will be threaded onto the bottom of the well screen and the sand layer below the well screen will be replaced with a layer of bentonite so DNAPL entering the sand pack will flow into the well screen and sump rather than sinking into the subscreen sand layer. A bentonite seal of 2 to 3-feet in thickness will be installed immediately above the well screen sand layer. The bentonite seal will be constructed with 3/8-inch bentonite chips or bentonite pellets. The remainder of the borehole annulus will be filled with a cement/bentonite grout using a tremie pipe. The well surface completion will consist of a locking mechanical 'J' plug and a vault-type well cover installed flush with the ground surface or an above grade vented steel protective casing depending upon well location.

## 2.9 BEDROCK CORING AND PACKER TESTING

Three deep exploratory bedrock coreholes and bedrock monitoring wells will be drilled to characterize bedrock stratigraphy, assess degree of fracturing, and identify potential presence of MGP source material (NAPL) in the bedrock during the Phase I bedrock investigation. The exploratory coreholes will be advanced through the Rochester Shale and deeper bedrock formations to the top of the Reynales Formation. The top of the Reynales Formation is estimated to be approximately 100 feet below Site grade. An HQ core barrel (a wire-line coring system will be considered after consultation with the selected drilling firm) will be used to retrieve rock core during advancement of the boring into rock. Description of the core, including the Rock Quality Designation (RQD), will be completed as described in the FOP included in Attachment C. Core samples will be retained in wooden core boxes for future reference. Approximately 7 to 8 core boxes per location are anticipated to contain 70 to 80 feet of bedrock core.

During coring, the rock core and drilling return water will be examined closely for NAPL. If NAPL is not observed in the bedrock, the corehole will be advanced to the top of the Reynales

Formation without installation of casings. However, if NAPL is observed, casings will be grouted in place to minimize potential downward migration of NAPL during deeper coring. The following scenarios describe when casing would be installed.

*NAPL in the Rochester Shale:* If NAPL is encountered in the Rochester Shale, an 8-inch diameter steel casing will be grouted in-place (using bottom-up tremie methods) from the contact between the Irondequoit Limestone and Rochester Shale (estimated to be 20 feet below the top of bedrock) to the ground surface. To facilitate casing installation, 8-1/4-inch ID HSA (12-inch OD) will be used to overdrill the overburden boring, the augers will be removed, and temporary 10-inch diameter casing will be socketed into the top of rock. The corehole will be reamed with a 9-7/8 inch roller bit and an 8-inch diameter steel casing will be grouted in-place to minimize the potential downward migration of NAPL during corehole advancement. After allowing the grout to set for a minimum of 12 hours (overnight), the corehole will continue to be advanced. If NAPL is not visually observed in the rock core from deeper formations (and/or the drilling return water), the corehole will be advanced through the deeper bedrock formations into the upper Reynales Formation.

*NAPL in the Irondequoit Formation:* If NAPL is identified in the Irondequoit Formation or deeper formations, the corehole will be reamed with a 7-7/8 inch roller bit and a 6-inch diameter steel casing will be telescoped through the 8-inch casing and grouted in-place. Coring will resume beyond the cased portion of the borehole.

Following corehole advancement, packer tests will be conducted to estimate fracture permeability of the isolated sections in the corehole. The packer test is a method of monitoring the rate of water injection in a confined, high pressure environment. A portion of the corehole will be sealed-off with a double inflatable packer assembly to isolate specific zones in the corehole. The packer seals allow water pumped into the test zone between the packers to escape only to the available joints and interstices within the rock. The distance between packers will range from 5 to 10-feet during testing and will be determined in the field based on field observations. Packer testing procedures are described in the FOP presented in Attachment C. Packer tests will be conducted in each of the three deep exploratory coreholes. NAPL checks as described in Section 2.12 will be conducted after each packer test.

Upon completion of coring and packer testing, the corehole will be reamed with a roller bit to an appropriate diameter suitable for either multi-level well system installation or individual bedrock well installation. Prior to well installations, water in the reamed bedrock bore hole will be purged and containerized to remove drilling water lost to the formation during drilling and packer testing.

Purge water will be containerized for appropriate disposal. Bedrock well installations are described in Section 2.10.

During the Phase II bedrock investigation, similar drilling procedures utilized for the Phase I bedrock investigation are anticipated.

# 2.10 MULTI-LEVEL WELL SYSTEM AND/OR NESTED BEDROCK WELL INSTALLATIONS

The Phase I bedrock investigation will install multi-level well systems in a single corehole and/or individual open hole bedrock monitoring wells. Assuming steel casings are not required to minimize potential downward NAPL migration during coring, the deep corehole will be reamed with a 5-7/8-inch roller bit to facilitate the installation of a multi-level well system. The multi-level well system will consist of four, 3/4-inch diameter Schedule 40 PVC slotted well screens and risers. Screen lengths will be determined in the field and are expected to range between 5 and 10-feet in length. It is anticipated that screens with be vertically spaced at approximate 20 foot intervals. Precise screen placements will be determined after evaluation of packer testing data and drilling observations.

PVC centralizing templates having diameters similar to the diameter of the borehole will be fixed to the bottom of each well screen to assure proper well screen positioning in the borehole. A diagram of a PVC centralizing template is shown below:



Plan view of a centralizing template placed at the bottom each well screen during installation to ensure proper positioning in the bore hole.

The deepest well screen and riser will be installed with the centralizing template in screen position "1". A silica sand pack will be placed in the borehole annulus to a level of 1 foot above the well screen. A bentonite slurry will be placed above the sand pack to a depth reaching the bottom of the next monitoring interval. A second PVC centralizing template will be fixed to the bottom of the next well screen (screen position "2") and lowered down the borehole to the next monitoring interval. The sand pack will be installed and a bentonite seal will be placed via tremie pipe to the

bottom of the next monitoring interval. The same process will be repeated for screen positions "3" and "4". The tops of the PVC riser pipes will be bundled together and secured to the top of the well vault. A mechanical J-plug will cap the tops of each riser.

At locations where NAPL is encountered and steel casings are telescoped to minimize potential downward migration of NAPL during coring, it will not be possible to install a multi-level well because the casings will isolate the rock formation from the well screens and laterally perforating the steel casing is not considered a feasible alternative. Therefore, to obtain similar data as the multi-level well system, individual bedrock wells will be installed in close proximity to one another (nested wells). An open monitoring interval in the bedrock would be created after grouting in a steel casing and coring a 5 to 10 foot interval below the casing. It is anticipated that at nested well locations, a group of four individually cased wells having open intervals at various depth intervals would provide similar data as the multi-level well system. It is anticipated that Phase II bedrock monitoring well installations would be installed in a similar manner.

## 2.11 WELL DEVELOPMENT AND HYDRAULIC TESTING

Monitoring wells will be developed using a suction-lift pump, air-displacement pump, bottomdischarging bailer, or a Waterra[™] hand pump. Development will involve using a combination of pumps and stainless steel bailers to removed accumulated sediment on the well bottom and withdrawal of groundwater. Development will be considered completed when the pH, specific conductivity and temperature have stabilized; and when the turbidity is at or below 5 NTU, or has stabilized above 5 NTU and approximately 10 well volumes have been removed. Stability is defined as variation between measurements of 10 percent or less and no overall upward or downward trend in the measurements. Water removed during development will be containerized, sampled and characterized, and disposed off-Site. If potable water is utilized during the drilling process, development volumes will be an equivalent volume of water lost to the formation.

Hydraulic testing will be conducted in each monitoring well using slug test methods. Hydraulic testing will use methods described in the FOP presented in Attachment C.

## 2.12 GROUNDWATER ELEVATION MEASUREMENTS AND NAPL DETECTION

Groundwater elevations will be measured in existing and newly installed wells/piezometers from the top of the north side of the PVC riser using an electric water level meter to the nearest 0.01 feet. Groundwater elevation readings will also be made at the time that groundwater sampling is performed and on one other occasion. During each round of groundwater elevations, potential NAPL presence will be detected by a weighted white cotton rope (approximate ¼-inch thickness) lowered to the bottom of the well and retrieved. Dark brown or black staining of the rope with a tar-like odor will be indicative of NAPL presence. The height of staining extending up from the rope bottom will indicate accumulated DNAPL thickness. Care will be taken not to mistake staining on the rope resultant from an LNAPL layer as actually indicating a greater thickness of DNAPL than may actually be present. A clear, PVC, bottom loading bailer may also be used to assess DNAPL thickness and DNAPL well entry recovery rate.

## 2.13 GROUNDWATER SAMPLING

Prior to sampling, the well will be checked for NAPL presence as described in Section 2.12. The NYSDEC will be notified if NAPL is present in a well and a discussion regarding the practicality of obtaining a representative groundwater sample will be discussed. The analytical parameter list for collected groundwater samples is described in the Work Plan and presented in detail in Table 2 of the QAPP presented in Appendix E of the RIWP.

The sampling event will occur approximately, but no sooner than, two weeks following well installation and development. Consistency of groundwater sampling methods is important to the assessment of groundwater quality and, when feasible, low flow sampling with a submersible pump will be the preferred method of sampling in accordance with USEPA Region 2 low flow groundwater sampling procedures. However, the inertial pump will be useful in achieving low-flow purging rates in deep, small diameter wells.

A well will be considered low yielding when purging rates are less than 100 ml/min and more than 2 feet of drawdown occurs in the well. In these instances, groundwater samples may be collected from the well using disposable bailers after adequate purging (e.g., removal of 3 well volumes and stabilization of measured field parameters).

Where feasible, low flow purging will be accomplished using a submersible pump. Groundwater will be discharged to a flow through cell to measure pH, specific conductance, dissolved oxygen, redox, and temperature. In addition, turbidity will be measured using a portable field turbidity meter. Purging will be considered complete when the pH, Eh, specific conductivity and temperature have stabilized; and when the turbidity is below 5 NTU, or has stabilized above 5 NTU. Water removed during purging will be containerized and considered to be investigation derived waste.

Samples will be collected in pre-preserved sample bottles and will be analyzed for parameters summarized in Table 2 of the QAPP (Appendix E). Groundwater samples for metals analysis collected using the USEPA Region 2 Low-Flow sampling method will not be field filtered provided sample turbidity values are at or near 50 NTU. However, groundwater samples that are collected

for total metals analysis from wells that have not achieved a maximum turbidity goal of 50 NTU will include a field filtered sample for soluble metals analysis to compare the difference between total and soluble metals water quality data.

## 2.14 AIR MONITORING

Air monitoring will be conducted with a PID and dust monitor during intrusive activities and only a PID during sampling activities. The PID will be used to monitor organic vapors in the breathing zone and borehole, and to screen samples for analysis and the dust monitor will be used to monitor particulate concentration in the breathing zone for particulates less than 10 microns in diameter. Monitoring will be consistent with the procedures and protocols outlined in the CAMP.

The PID and dust monitoring readings will be recorded in the field book during drilling activities. The instruments will be calibrated at least once each day and more frequently if needed.

## 2.15 SEDIMENT PROBING AND SAMPLE COLLECTION

Near shore sediment samples will be collected at approximately 100 foot-intervals from the Genesee River approximately 5 feet from the shoreline where water levels are less than 3 feet deep and flow velocities are sufficiently low to pose minimal safety risk to field personnel. Sample locations will be field correlated with observations of Site-related piping discharge points (former and/or active) from the Site to the river. Riverbed sediment samples will be collected using stainless steel sampling equipment and a hand auger. A grab sample will be collected by slowly advancing the auger approximately 6-inches into the sediment with minimal disturbance of the surrounding sediment. The collected samples will be removed from the sample barrel of the hand auger using a disposable stainless steel sampling spoon and placed directly into laboratory provided pre-cleaned sample jars and analyzed for parameters summarized in Table 2 of the QAPP (Appendix E). Downstream samples will be collected first to minimize potential impacts caused by disturbance of the sediment.

Pore water samples will be collected from the riverbed sediment at approximately one-half of the sediment sample locations. A pore water sample will be collected from a temporary small diameter well screen and riser driven into the sediment (drive-point piezometer). The piezometer (riser and screen) will be constructed of stainless steel. The drive-point piezometer will be equipped with 3/8" diameter Teflon-lined sample tubing. Drive-point piezometer instructions for a typical drive-point piezometer are included in Attachment D. Pore water samples will be collected after purging of three "well volumes" using a peristaltic pump or inertial pump at a very low flow rate. Sample temperature will be continuously monitored during purging to ensure pore water and not river water

is being collected. If river water is suspected, the drive-point piezometer will be removed and redriven in proximity to the selected pore water sample location.

At 50 foot intervals, a steel "hammer probe" will be driven into the sediment to identify locations that produce a sheen on the surface water. The hammer probe consists of a ¹/₂-inch diameter steel alloy shaft driven into the sediment using a slide hammer. Probing will extend as deep as feasible (i.e., top of bedrock or minimum three feet below the top of sediment). Where sheens are observed, a sediment sample will be collected from the upper six inches of sediment and a second sample collected using the hand auger from a depth of 18 to 24 inches, if possible, unless the precise depth interval of material suspected of causing the sheen is identified. The samples will be analyzed for the parameter list specified in the QAPP.

The probing will be conducted approximately one day after river sediment sample collection.

## 3.0 FIELD INSTRUMENTS

Field-screening equipment will be calibrated prior to each day's use and more frequently if required. The calibration procedures will conform to the manufacturer's standard instructions. Records of instrument calibration and copies of instrument manuals will be maintained by field personnel.

## 3.1 PORTABLE PHOTO IONIZATION ANALYZER

The photoionization analyzer will be a Photovac MicroTip (or equivalent), equipped with a 10.6 eV lamp. The Photovac is capable of ionizing and detecting compounds with an ionization potential of 10.6 eV or less and is suitable for the aromatic hydrocarbons present at the Site. Calibration will be performed according to the manufacturer's requirements.

## **3.2** AIR BORNE PARTICULATE (DUST) MONITOR

The dust monitor will be a MIE DataRAM (or equivalent) and will be calibrated at the start of each day of use. Calibration and maintenance of the dust monitor will be conducted in accordance with the manufacturer's specifications. The calibration data will be recorded in field notebooks.

## 3.3 MULTI-PARAMETER FLOW-THROUGH CELL

A YSI multi-parameter flow-through cell (or equivalent) will be utilized to evaluate water quality characteristics. The flow-through cell will consist of individual sondes which will measure the following parameters:

- pH
- Specific Conductivity
- Dissolved Oxygen
- Oxidation/ Reduction Potential
- Temperature

Sondes comprising the flow through cell (except temperature) are calibrated with a single multicomponent solution. Calibration of the flow-through cell will be performed in accordance with the manufacturer's recommendations. The calibration of the cell will take place at the beginning of each day of sampling, and will be rechecked at the end of each day of sampling to determine instrument drift in the calibration of any individual sonde. Calibration data will be recorded in field notebooks
#### **3.4 TURBIDITY METER**

The turbidity meter will be calibrated daily prior to use. Calibration and maintenance will be conducted in accordance with the manufacturer's specifications. Calibration and maintenance will be recorded in the field notebook.

# ATTACHMENT A Daily Field Forms

DAILY F	IELD RECOR	D		Pace Pace	
Project and Task I	Number:	Date:			
Project Name:		Field Activity:	*****		·
Location:		Weather:			
PERSONNEL:	Name	Company		Time	Time
PERSONAL SAFE	TY CHECKLIST		· · · · · · · · · · · · · · · · · · ·		
Steel-toed	Boots	Hard Hat	Туу	k Coveralls	
Rubber Gl	oves	Safety Goggles	1/2-F	ace Respirat	or
DRUM I.D.	DESCRIPTION	OF CONTENTS AND QUANTITY	Y	LOCATIO	N
TIME					
		DESCRIPTION OF WORK P	CRFURMED		
	· · · · · · · · · · · · · · · · · · ·				
	·····				
			·		
				· · · · · · · · · · · · · · · · · · ·	
				, <u>, , , , , , , , , , , , , , , , </u>	

L

DAILY FIELD REC	ORD (continued)	
Project and Task Number:	Date:	0r
TIME	DESCRIPTION OF WORK PERFORM	1ED

PROJECT:		Log o	of Well I	No.
BORING LOCATION:		ELEVATION AND DA	TUM:	······································
DRILLING CONTRACTOR:		DATE STARTED:		DATE FINISHED:
DRILLING METHOD:		TOTAL DEPTH:		SCREEN INTERVAL:
DRILLING EQUIPMENT:		DEPTH TO FIRST WATER:	COMPL.	CASING:
SAMPLING METHOD:		LOGGED BY:	· · · · ·	
HAMMER WEIGHT:	DROP:	RESPONSIBLE PRO	ESSIONAL:	REG. NO.
DEPTH (feet) Sample Sample Blows/ Foot CVM Reading (ppm)	DESCRIPTION NAME (USCS Symbol): color, moist, % by weight consistency, structure, cementation, react. w/HCL, Surface Elevation:	L, plast., geo. inter.	WELL CO AND/OF	NSTRUCTION DETAILS
				W-1 (1295)

PROJ	ECT:					Lo	og of Well No.
EPTH (feet)	SA Bldro	MPI ejdu	.ES	A Reading (ppm)	DESCRIPTION NAME (USCS Symbol): color, moist, % by weight., plast.,		
	San N	San	Bor	No O	consistency, structure, cementation, react. w/HCI. geo. inter.		AND/ON DRIELING REMARKS
_						-	
_						-	
-							
_							
_							
-							
-						-	
_							
-						-	
4						-	
-							
]						-	
4							
4							
-						-	
ł						-	
4						-	
-							
4							
]							
_							
-							
4							
4							
-							
-						-	
-							
-						-	
···	·······		······································	l		I	W-2 (12/95)
Project N	No.				Geomatrix Consultant	s	Figure



Ì

×.

X = J

New York

)

#### WELL SAMPLING AND/OR DEVELOPMENT RECORD

Well ID:		<u>i</u> _				Initi	al Do	enth to Wa	tor	
Sample ID:	Dupli	cate II	):			Den	th to	Water aft	er Sampli	na
Sample Depth:						Tota	al De	onth to Wel	l.	
Project and Task No.:						Well	Dia	meter:		
Project Name:						1 Ca	sind	a/Borehole	Volume:	
Date:						(Cire	cle o	one)		
Sampled By:						4 Ca	sing	g/Borehole	Volumes	
Method of Purging:								one)		
Method of Sampling: _						Volu	in Ca Imes	sing/Borei s Removec	noie I:	
	Cu	mæ			1	Specif	ic	25		
Time Depth (gp	ite V (m) (a	ol. 👔	lem (°C)	o, – pl (un	H its) C	Electric	cal ance	Oxygen	Potentia	Remarks
	(9)	10				(µS/cn	n) 😤	(mg/l)	(mV; SSC	
						·····				
						No				
	pH CALI	BRATI	ON (c	hoose t	NO)	NA BI			Model	or Unit No.:
Buffer Solution		pН	4.0	pH 7.	0	pH 10.	0		_	
Field Temperature °C										
Instrument Reading										
SPECIFIC EL	ECTRICA	L COI	NDUÇ.	TANCE	- CALI	BRATI	ON		Model	or Unit No.:
KCL Solution (µS/cm=µm	hos/cm)		1413	at 25°C	12880	) at 25°(	c			
Field Temperature °C									-	
Instrument Reading								•		
REDOX CALIBRATION			DIS	SOLVEI	ΟΟΧΥ	GEN C	ALIE	BRATION	Notes:	
Standard Solution	468 mV		Salin	nity %					8	
Field Temperature °C			Altitu	ıde						
Instrument Reading	1		Instru	ument R	eading					
Model or Unit No.:	L		Mode	el or Unit	t No.:					······································
Ag/AgCI Electrode (SSCI	E)									
	-									

G:\FORMATS\WELL SAMPL Rec-DO.doc

BORING LOCATION: DRILLING CONTRACTOR: DRILLING METHOD: DRILLING EQUIPMENT: SAMPLING METHOD: HAMMER WEIGHT: DROP: <u>Hamper of the standard of the standar</u>	ELEVATION AND DATUM: DATE STARTED: TOTAL DEPTH: DEPTH TO WATER: FIRST LOGGED BY: RESPONSIBLE PROFESSIO	DATE FINISHED: MEASURING POINT: COMPL ONAL: REG. NO. REMARKS
DRILLING CONTRACTOR: DRILLING METHOD: DRILLING EQUIPMENT: SAMPLING METHOD: HAMMER WEIGHT: DROP: THUE SAMPLES OF TO BE SA	DATE STARTED: TOTAL DEPTH: DEPTH TO WATER: FIRST LOGGED BY: RESPONSIBLE PROFESSION ructure, cementation, react. W/HCl, geo. inte	DATE FINISHED: MEASURING POINT: COMPL ONAL: REG. NO.
DRILLING METHOD: DRILLING EQUIPMENT: SAMPLING METHOD: HAMMER WEIGHT: DROP: THAMMER WEIGHT: DROP: DESCRIPTION NAME (USCS Symbol): color, moist. % by weight, plast., consistency, str Surface Elevation: 1-	TOTAL DEPTH: DEPTH TO WATER: FIRST LOGGED BY: RESPONSIBLE PROFESSIO	MEASURING POINT: COMPL ONAL: REG. NO. ar. REMARKS
DRILLING EQUIPMENT: SAMPLING METHOD: HAMMER WEIGHT: DROP:	DEPTH TO WATER: FIRST LOGGED BY: RESPONSIBLE PROFESSIO	COMPL ONAL: REG. NO. ar. REMARKS
SAMPLING METHOD: HAMMER WEIGHT: DROP: HAMMER WEIGHT: DROP:	LOGGED BY: RESPONSIBLE PROFESSIon ructure, cementation, react. W/HCl, geo. inte	ONAL: REG. NO. Pr. REMARKS
HAMMER WEIGHT: DROP: <u>H (1994)</u> <u>B (1994</u>	RESPONSIBLE PROFESSI	ONAL: REG. NO. Pr. REMARKS
H     SAMPLES     Discrete       Image: Properties     Image: Properties     Image: Properties       Image: Prope	ructure, cementation, react. W/HCI, geo. inte	Pr. REMARKS
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	:	-
		-
2-     -       3-     -       4-     -       5-     -       6-     -       7-     -       8-     -       9-     -       10-     -       11-     -       12-     -       13-     -       14-     -       15-     -		OVM=Thermo Environmental Instruments calibrated with 100 ppm isobutyle standard.

WATE	R LEVE	L MONIT	ORING I	RECORD	)		
Project Na	ame:			Projec	t and Task N	lumber:	
Date:		Measured	by:		Instrumer	it Used:	
Note: For	you conven	ience, the fo	llowing abbro	eviations ma	y be used.		
P = Pun ST = Stee	nping el Tape	I = Inacce ES = Electric	ssible Sounder	D = Dedica MP = Mease	ated Pump uring Point	WL = Water Level	
Well No.	Time	MP Elevation (feet)	Water Level Below MP (feet)	Water Level Elevation (feet)	Previous Water Level Below MP	Remark	S
							<u> </u>
						- 	
						<u></u>	
							·····
							····
						,,,,,,,,_,,_,,_,	

# FIELD TEST PIT LOG

Job No. Contractor GMX Insp. Weather Location	P C P	Project Operator Elevation	Tes N Equipment Started Completed	t Pit
TEST PIT WIDT	TEST PIT SKETCH			
<b>├</b> ──── <b>├</b> ──	·		NOTES / STRATA D	ESCRIPTIONS
				·
	SAMPLES (2) NOTE	<u>S</u>		
E	XCAVATION NOTES		WATER LEVE TIME WATER	SLS DEPTH (bgs)

GEOMATRIX CONSULTANTS, INC.

SAMF	ILE CONT	TROL LO	G				
Project I	Name:				Laboratory:		
Project ¿	and Task No.					Page	of
Sampling Date	Sampling Time	Sample Number (ID)	C.O.C. Number	Analyses Requested	Turnaround Time, Sample Location, Handling Notes, Chain-of-Custody Remarks, et. (Duplicate, blank info, etc.)	Date Sent to Lab	Date Results Due
	-						
					· · ·		
				-			
orms(PF).008 (Revised	d 12/95)						

# ATTACHMENT B Chain-of-Custody Form

CHAIN-O	F-CUST	<b>FODY RECORD</b>	_						70 7	660
<b>PROJECT I</b>	NAME:						DATE:	PAG	Щ	OF
PROJECT NUMB	ä			LABORATORY NAME:	CLIENT INFORMATION:		REPORTING REQUIREMENT	TS:		
RESULTS TO:				LABORATORY ADDRESS:						
TURNAROUND T	ME:									
SAMPLE SHIPME	NT METHOD:			LABORATORY CONTACT:			GEOTRACKER REQUIRED		YES	ON
				LABORATORY PHONE NUMBER:			SITE SPECIFIC GLOBAL ID N	NO.		
SAMPL	ERS (	(SIGNATURE	:(Ξ	ANALY	SES					
							/ater (W), or Other (O)	ле Туре	າໂສ່ານອາເຮ	
DATE	TIME	SAMPLE NUMBEI	ш к			ССС	NTAINER E AND SIZE Soil (S), W Vapor (V), Vapor (V),	Preservati	MS/MSD No. of Cor	ADDITIONAL COMMENTS
		-								
								2		
	SHEDE	עדערן ->ב				TOTAL NUMBER OF CC	DNTAINERS:			
SIGNATURE:				SIGNATURE:		SAMPLING COMMENTS				
PRINTED NAM	نن			PRINTED NAME:						
COMPANY:				COMPANY:						
SIGNATURE:				SIGNATURE:						
PRINTED NAM	ш			PRINTED NAME:	1					
COMPANY:				COMPANY:	·					
SIGNATURE:				SIGNATURE:		inM nhol. 808	r Drive Suite 104			
PRINTED NAM	نن			PRINTED NAME:		Amherst, New	York 14228-1148		Geo	matrix
COMPANY:				COMPANY:		Tel 716.565.0624	Fax 716.565.0625		)	

# **ATTACHMENT C** Field Operating Procedures (FOPs)

### FIELD OPERATING PROCEDURE

### SOIL BORING LOG DESCRIPTION PROCEDURES USING THE UNIFIED SOIL CLASSIFICATION SYSTEM

#### PURPOSE

This guideline presents a means for insuring proper field identification and description of soils collected from a split-spoon (barrel) sampler. The lithology and moisture content of each soil sample will be visually and physically characterized according to the Unified Soil Classification System. This method of soil characterization describes soil types on the basis of grain size and liquid and plastic limits and includes moisture content.

#### PROCEDURE

Assemble necessary equipment and discuss program requirements with drilling contractor.

- Advance boring in accordance with accepted Geomatrix Split-Spoon Sampling Field Operating Procedure at pre-specified intervals. Samples shall be taken at a minimum from each stratigraphic unit and each screened interval. Record the number of blows necessary to drive the split-spoon sampler per 6-ince interval. If the sampler is not driven the 6-inch interval after 50 blows are delivered, measure the sampler penetration distance and record this distance along with the blow count. Advance augers to the next sample interval and repeat procedure.
- After opening the split-spoon sampler, measure and record the length of the sample. The upper 2 to 3 inches of the sample should be disregarded as the material is likely not representative of the native in-situ materials.
- 3. Shave a thin layer off the entire length of the sample to expose fresh sample. (Note: The outer sample surface of often smeared while the sample barrel is being driven.) The sample should be photographed and screened with HNu at this time, if applicable.

# FIELD OPERATING PROCEDURE

- 4. Describe the sample using terminology presented in Section 3.0 below.
- 5. After the sample has been described, place representative portion of the sample in new, precleaned jars. Label the jar with the borehole number, sample interval, date, number of blow counts and project number and store in a secure location.

#### **DESCRIPTIVE TERMS**

All field soil samples will be classified in accordance with the Unified Soil Classification System (USCS) on the attached pages. It is desirable to supplement the USCS classification with a geologic interpretation of the soil sample that is supported by the soil descriptive terms presented in Table A.

#### **REQUIRED EQUIPMENT**

- knife
- engineer's rule/measuring tape
- permanent marker
- pre-cleaned sample jars (usually provided by the driller)
- 10X hand lens
- hydrochloric acid
- camera
- Munsell soil color chart
- Project Field Book

## FIELD OPERATING PROCEDURE

### UNIFED SOIL CLASSIFICATION SYSTEM IDENTIFICATION TESTS FOR FINE-GRAINED SOILS

Soil Type	Group Symbol	Dry Strength ⁽¹⁾	Dilatancy ⁽¹⁾	Toughness ⁽¹⁾
Silts & Clays	ML	None to slight	Quick to flow	None
Liquid Limit <50	CL	Medium to high	None to very slow	medium
	OL	Slight to medium	Slow	slight
Silts & Clays	MH	Slight to medium	slow to none	slight to medium
Liquid Limit: >50	СН	High to very high	none	high
	ОН	Medium to high	none to very slow	slight medium
Highly Organic	РТ	Identifiable by:	color, order,	Spongy feel
50115		And commonly:	by fibrous texture	

Note

(1) See page of for description test.

# FIELD OPERATING PROCEDURE

#### **ROCK CORE CLASSIFICATION**

#### PURPOSE

This guideline presents rock classification procedures and descriptive terminology for rock core samples collected by standard bedrock coring methods.

#### PROCEDURE

- 1. Place the core sample in good light, remove any extraneous material and wash the sample to clean it of drilling fluid, residue or mudcake.
- 2. Describe the wetted rock according to the following hierarchy:
  - a. Rock Type
  - b. Color
  - c. Bedding Thickness
  - d. Hardness
  - e. Fracturing
  - f. Weathering
  - g. Other Characteristics
- 3. Provide further detail for cored samples based on following steps:
  - recombine the core and measure its length;
  - calculate recovery percentage;
  - count the natural discontinuities and artificial (drilling related) core breaks;
  - calculate rock quality designation (see separate guidance procedure); and calculate fracture frequency.
- 4. Document descriptions in Project Field Book and on the appropriate field forms
  - a. <u>Rock Type</u>

### FIELD OPERATING PROCEDURE

The following basic names are applied to sedimentary rock types common to New York State:

- Sandstone Made up predominantly of granular materials ranging between 1/16 and 2 inches in diameter.
- Siltstone Made up of granular materials less than 1/16 inch in diameter.
   Fractures irregularly. Medium thick to thick bedded.
- (3) Claystone Very fine-grained rock made up of clay and silt-size materials. Fractures irregularly. Very smooth to touch. Generally has irregularly spaced pitting on surface of drilled cores.
- (4) Shale A fissile very fine grained rock. Fractures along bedding planes.
- Limestone Rock made up predominantly of calcite (CaCO₃).
   effervesces upon the application of hydrochloric acid.
- (6) Coal Rock consisting mainly of organic remains.
- Others Numerous other rock types are present in the geologic section.
   Their overall abundance is dependent upon geographical location. These include halite, gypsum, dolomite, anhydrite, lignite, etc.

In most cases a rock will be a combination of several rock types, (e.g., sandy siltstone, or silty sandstone). The modifier indicates that a significant portion of the rock type is composed of the modifier. Other modifiers include carbonaceous, calcareous, siliceous, etc.

Grain size diameters are used for the classification of clastic sedimentary rocks. The attached Table1 specifies the classification that should be applied to sedimentary rocks. For field determination of grain sizes, a scale can be used for the coarse grained rocks. The division between very fine sand and silt is probably not measurable in the field. The boundary should be determined by use of a hand lens. If the grain cannot be seen with

# FIELD OPERATING PROCEDURE

the naked eye but are distinguishable with a hand lens, the rock is a siltstone. If the grains are not distinguishable with a hand lens, the rock is a claystone.

#### b. <u>Color</u>

Rock colors should be described utilizing a Munsell Color Chart. A single color descriptor preceded, when necessary, by a modifier to denote variations in shade or color mixtures should be used when describing. A rock could, therefore be referred to as "gray" or "light gray". Since color can be utilized in correlating units between sampling locations, it is important that color descriptions be kept consistent throughout the field operations.

Rock core samples should be classified while wet.

#### c. <u>Bedding Thickness</u>

Bedding thickness designations for rock classification are given in Table 2.

### FIELD OPERATING PROCEDURE

#### TABLE 1

#### **GRAIN SIZE CLASSIFICATION FOR ROCKS**

Particle Name	Grain Size Diameter
Cobbles	>64 mm
Pebbles	4-64 mm
Granules	2-4 mm
Very Coarse Sand	1 – 2 mm
Coarse Sand	1 – 2 mm
Medium Sand	0.25 – 0.5 mm
Fine Sand	0.125 – 0.25 mm
Very Fine Sand	0.0625 – 0.125 mm
Silt	0.0039 – 0.0625 mm

### FIELD OPERATING PROCEDURE

#### TABLE 2

#### **BEDDING THICKNESS CLASSIFICATION**

Thickness (Metric)	Approx. English Equivalent	Classification
>1.0 meter	>3.3 feet	Massive
30 cm – 1 meter	1.0 ft - 3.3 ft	Thick Bedded
10 cm – 30 cm	4 in0 1.0 ft	Medium Bedded
3 cm - 10 cm	1 in. – 4 in.	Thin Bedded
1 cm – 3 cm	2/5 in. – 1 in.	Very Thin Bedded
3 mm – 1 cm	1/8 in. – 2/5 in.	Laminated
1 mm – 3 mm	1/32 in. – 1/8 in.	Thinly Laminated
<1 mm	<1/32 in.	Micro Laminated

### FIELD OPERATING PROCEDURE

#### d. <u>Hardness</u>

The hardness of a rock is a function of the compaction, cementation, and mineralogical composition of the rock. A relative scale for sedimentary rock hardness is as follows:

Soft – Weathered, considerable erosion of core, easily gouged by screwdriver, scratched by fingernail. Soft rock crushes or deforms under pressure of a pressed hammer. This term is always used for the hardness of the saprolite (decomposed rock which occupies the zone between the lowest soil horizon and firm bedrock).

Medium soft – Slight erosion of core, slightly gouged by screwdriver, or breaks with crumbly edges from single hammer blow.

Medium hard – No core erosion, easily scratched by screwdriver, or breaks with sharp edges from single hammer blow.

Hard – Requires several hammer blows to break and has sharp concoidal breaks. Cannot be scratched with screwdriver.

#### e. Fracturing

Determine the degree of fracturing or brokenness of a rock by measuring the fractures or joint spacing. After eliminating drilling breaks, the average spacing is calculated and the fracture structure is described by the following terms.

Very broken (V.BR.) – Less than 2 in Broken (BR.) –2 in. to 1 ft. Blocky (BL.) – 1 to 3 ft. Massive (M.) – 3 to 10 ft.

# FIELD OPERATING PROCEDURE

The structural integrity of the rock can be approximated by calculating the Rock Quality Designation (RQD) of cores recovered. RQD is described in a separate Field Operating Procedure.

#### f. <u>Weathering</u>

The degree of weathering is a significant parameter that is important in determining weathering profiles and is also useful in engineering designs. The following terms can be applied to distinguish the degree of weathering.

Fresh – Rock shows little or no weathering effect. Fractures or joints have little or no staining and rock has a bright appearance.

Slight – Rock has some staining which may penetrate several centimeters into the rock. Clay filling of joints may occur. Feldspar grains may show some alteration.

Moderate – Most of the rock, with exception of quartz grains, is stained. Rock is weakened due to weathering and can be easily broken with hammer.

Severe – All rock including quartz grains is stained. Some of the rock is weathered to the extent of becoming a soil. Rock is very weak.

g. <u>Other Characteristics</u>

The following items should be including in the rock description:

- (1) Description of contacts between two rock units. These can be sharp or gradational.
- (2) Description of any filled cavities or vugs.
- (3) Description of any joints or open fractures.
- (4) Notation of joints with depth, approximate angel to vertical, any mineral filling or coating, and degree of weathering.

### FIELD OPERATING PROCEDURE

As an attachment to the log, additional information should be provided, including an estimation of the degree of cementation and type of cement for granular sedimentary rocks; a description of the texture of the rock (i.e., the relationship of component particles or crystals); and the structure or megascopic features of the rock mass.

#### **REQUIRED EQUIPMENT**

- Knife
- Ruler
- Permanent Marker
- 10X Handlens
- Hydrochloric acid
- Camera
- Munsell rock color chart

### FIELD OPERATING PROCEDURE

#### **ROCK QUALITY DESIGNATION (RQD)**

#### PURPOSE

This guideline presents a method to describe the Rock Quality Designation (RQD), which is a simple and direct means of indicating rock mass properties. RQD is based on a modified core recovery procedure which, in turn, is based indirectly on the number of fractures and the amount of softening or alternation in the rock mass as observed in the rock cores from the borehole. Core recovery is the ratio of the length of core recovered to the length drilled (i.e., no recovery = 0 and full recovery = 100). This procedure is an indicator of the general quality of rock for engineering purposes and provides a numerical value which is more sensitive and consistent than gross percentage core recovery.

#### PROCEDURE

Sum the total length of core recovered by counting only those pieces of hard and sound core which are 4 inches (10cm) or greater in length and divide that sum by the total length of that run. RQD is presented as a percentage.

#### LIMITATIONS

- RQD should not be applied to core less than 2 inches (5.4 cm) in diameter as a false RQD may be obtained because smaller cores can be frequently broken during the coring operation.
- 2. Care must be taken when removing the core from the core barrel. If a core is broken by handling or during drilling, the fresh broken pieces should be fitted together and counted as one piece.
- 3. Some judgement is necessary in the case of thinly bedded sedimentary rock and foliated metamorphic rocks. The method is most suitable for igneous rock, thick bedded limestone, sandstone, etc. However, this procedure can be applied to shales, although it

# FIELD OPERATING PROCEDURE

is necessary to log the cores immediately upon removal from the core barrel before airslaking and cracking potentially occur.

Although this procedure is less applicable where core recovery is poor, the results are indicative of poor quality rock. It should however be noted that poor drilling techniques and equipment can also cause poor recovery. It is for this reason that proper equipment and procedure along with competent supervision of the drilling procedure are imperative.

#### EXAMPLE

An example RQD calculation is provided on the following page:

### FIELD OPERATING PROCEDURE

Appendix B: Item 6 – ROCK QUALITY DESIGNATION (RQD)

Core Recovery (in.)	Modified Core Recovery (in.)	Rock Quality Designation (RQD)	Description of Rock Quality
25	25	0-25	Very poor
12	12	25 - 50	Poor
5	0	50 - 75	Fair
8	0	75 - 90	Good
8	0	90 - 100	Excellent
8	0		
66	37		

#### **MODIFIED CORE RECOVERY AS AN INDEX OF ROCK QUALITY**

#### NOTES:

84 inches = length of run. Core Recovery =  $(66/84) \times 100 = 79\%$ RQD =  $(37/66) \times 100 = 56\%$ , therefore RQD is Fair.

In this case, the core barrel was advanced 84 inches with a total recovery of 66 inches. However, due to fractures, soft zones, etc., the modified recovery was on 37 inches. This equates to an RQD of 56%, which can be used as a modifier in the geologic description of the rock, which in this case would be "<u>Fair.</u>"

### FIELD OPERATING PROCEDURE

### IN-SITU HYDRAULIC CONDUCTIVITY BY PRESSURE PACKER TESTING PROCEDURE

#### PURPOSE

This procedure is presented for calculating the permeability of the bedrock within a borehole. It is most valid when stratum thickness is at least five times the diameter of the hole tested. The packer test is a method of monitoring the rate of disappearance of water in a confined, high pressure environment. Generally, a portion of the borehole to be tested is sealed off with inflatable packers which expand to the walls of the borehole to prohibit movement of water past the packer. The seal allows water pumped into the test zone between the packers to escape only to the available joints and interstices within the rock.

#### **EQUIPMENT**

#### 1. <u>Pneumatic Packers</u>

Pneumatic packers consist of metal and rubber cylinders whose central rubber portions are a minimum of 12 inches in length and are pneumatically inflatable to seal off a zone of the borehole. The packers are attached together by a perforated pipe to allow water to escape into the test zone.

#### 2. <u>Water Flowmeter</u>

Measures water flow in gallons. This meter shall be easily readable and calibrated to an accuracy within 1.5% (i.e., 0.75 gallons over 50 gallons). The units of graduation shall not exceed 1 gallon.

#### 3. <u>Water Pressure Gauge</u>

Measures water pressure in pounds per square inch (psi). This gauge shall be easily readable and calibrated to an accuracy of  $\pm 2.5$  psi. The units of graduation shall not exceed 5 psi.

### FIELD OPERATING PROCEDURE

#### 4. <u>Surge Chamber (optional)</u>

A heavy, metal, air-filled cylinder installed to prevent the pulsating water delivery yielded by some pumps. The surge chamber should be capable of withstanding the maximum water pressure.

#### 5. <u>Flow Valve (optional)</u>

Controls flow from the pump into the test system and water pressure as shown on the water pressure gauge.

#### 6. <u>Bypass Valve</u>

Permits bypassing any desired portion of the flow back into the pump.

#### 7. <u>Nitrogen Tank or Compressed Air</u>

Nitrogen or compressed air shall be used to inflate the pneumatic packers because of their nonflammable character. The tank shall be equipped with a pressure regulator and gauges.

#### 8. <u>Minimum Nominal ³/₄-inch I.D. Pipe</u>

Used to lower the packer assembly to the appropriate depth and transport water down to the test zone.

#### 9. <u>Air Hose</u>

The air hose shall be made of materials capable of withstanding twice the allowable net pressure of the pneumatic packers.

10. <u>Pumps</u>

A positive displacement or Moyno pump having a minimum capacity of 20 gpm at a pressure of 100 psi should be used.

#### **DATA COLLECTION PROCEDURES**

1. Test the water flow meter prior to use by running a known volume through the meter and checking its calibration.

# FIELD OPERATING PROCEDURE

- 2. Record on the Pressure Test Report form (see Figure 1) all measurements listed below.
  - Length of zone tested (note: this must be <u>at least</u> five times greater than the hole diameter);
  - Length of packer, rubber portion only;
  - Length of portion of hole not tested if double packer assembly is used;
  - Length of entire assembly;
  - Elevation difference between water pressure gauge and midpoint of test segment (the water table should be allowed to stabilize for 24 hours prior to testing when practical); and length of hose or piping used in each test.



#### PRESSURE TEST REPORT

Page ____ of ____

Project:     Si       Location:     G       Contractor:     D		Site: Ground	Elev.:		Total Dej	Date:		Boring I Top of F	No.: Rock: Depth:			
Contractor:     I       Water Level:     Depth       Elevation:		n:	Inspector        Water Pipe Length:				r:	Chek'd By: Water Pipe I.D.:				
Flow Meter No Pressure Gauge No.:		.:	Test Interval: Depth			Elevation		Elevatio	n:			
Gauge Pr	Gauge Pressure: Gauge Pressure:			Gauge Pressure:		Gauge Pressure:			Test Configuration			
Packer Infl't'n Press.: Packer Infl't'n Press.:		: 	Packer Infl't'n Press.:		Packer Infl't'n Press.:							
Elapsed Time (min.)	Flow Reading (gallons)	Δ Flow	Elapsed Time (min.)	Flow Reading (gallons)	Δ Flow	Elapsed Time (min.)	Flow Reading (gallons)	$\Delta$ Flow	Elapsed Time (min.)	Flow Reading (gallons)	Δ Flow	
		-			-							
		-			-							
		-			-						·	-
Remarks	:		Remarks	:		Remarks	:		Remarks	:		
												SINGLE
												Tape/Rule No.

## FIELD OPERATING PROCEDURE

Calculate maximum test pressure. The maximum test pressure will not exceed a value equal to 1 psi per floor depth of hole, calculated at the midpoint of the test segment. Maximum test pressure equals the sum of gauge pressure ( $P_p$ ) and the weight of the water column between the gauge and the midpoint of the test segment ( $P_w$ ). For example, if the depth to the midpoint of the test segment is 50 feet and the gauge is mounted 3 feet above the collar of the hole, then the maximum gauge pressure will be calculated as follows:

50 psi = gauge pressure (psi) +  $(50 + 3) \frac{62.4}{144}$ = gauge pressure (psi) + 53 x 0.433 = gauge pressure + 22.97 (psi) 50-22.97 = 27.03 psi

Maximum gauge pressure = 27 psi

3. Select the packer pressure. The approximate packer pressure can be determined as follows:

 $P_p$  = packer pressure = inflation (resistance of rubber) pressure in air + ½ of hydrostatic head on the packer (approximate static water pressure) + 1.2  $P_w$  max.

The packer pressure shall not exceed the design pressure of the packer. The nitrogen or compressed air tank shall be set at the selected packer pressure and the pressure recorded on the field form.

- 4. Thoroughly clean the borehole and allow the water table 24 hours to stabilize, if practical.
- 5. Insert packer assembly into borehole and inflate. The hole should be tested from the bottom upwards.

### FIELD OPERATING PROCEDURE

- a. Begin water flow at the appropriate pressure and allow the necessary time interval depending upon the grain of test material.
- b. Conduct water pressure testing in four phases as outlined below:

	Test Pressure	Packer Pressure
Phase 1	$(1/2) P_{w}$	$P_p$
Phase 2	$\mathbf{P}_{\mathbf{w}}$	P _p
Phase 3	$\mathbf{P}_{\mathbf{w}}$	P _p plus 20 psi

If the measured flow in Phase 3 is less than that in Phase 2, then:

	Test Pressure	Packer Pressure		
Phase 4	(1/2) P _w	P _p plus 20 psi		

If there is no water taken during Phase 2, the geologist or engineer may end the test.

The third phase is a check on the second phase to determine whether leakage past the packer has occurred. Increase of packer pressure of 20 psi is often sufficient, but higher increases may be used as the situation dictates.

- c. Take gauge measurements every 30 seconds for the first 5 minutes of operation, and every 60 seconds thereafter until stabilization; i.e., until three or more readings of water intake and pressure taken at 5-munute intervals are essentially equal.
- 7. The geologist or engineer may end the test if the following occurs:

# FIELD OPERATING PROCEDURE

- In the event that a zone will not take water in the first 3 minutes of testing;
- In the event that a zone will not hold water test pressure of  $(1/2) P_w$ ; or
- In the event that the measure flow in Phase 3 is greater than that in Phase 2 (i.e., hydraulic jacking of the rock (Reference 1).

#### **DOCUMENTATION OF TEST RESULTS**

- 1. All readings and other pertinent test data shall be recorded on the field form attached.
- A pressure test report shall be completed by the drilling contractor or geologist or engineer, or both, for each test conducted. This includes all measurements taken prior to the test as well as all test results.
- 3. A <u>field log</u> shall also be kept to list all tests performed and their locations.

#### CALCULATION OF HYDRAULIC CONDUCTIVITY

Calculate hydraulic conductivity according to procedures given in <u>Ground Water Manual</u>, 1985, U.S. Department of Interior, Bureau of Reclamation.

# FIELD OPERATING PROCEDURES

#### IN-SITU HYDRAULIC CONDUCTIVITY BY SLUG TEST

#### PURPOSE

This procedure describes the performance of slug tests for evaluating the hydraulic characteristics of an aquifer or water-bearing zone. Slug tests are a non-pumping aquifer test. Slug tests are conducted to evaluate aquifer characteristics when pumping tests are not appropriate or practical.

#### PROCEDURE

- 1. Record all field data in the Project Field Book.
- 2. Measure dimensions of the slug, and calculate its volume.
- 3. Decontaminate slug, water level indicator (e-line) probe and cable, and pressure transducer and cable. Reference the Geomatrix Field Operating Procedure for Sampling Equipment Decontamination.
- 4. Unlock and remove the protective cap or cover and place on clean plastic.
- 5. Measure and record the static water level in the well using the e-line. Slug testing should only be performed at monitoring wells that are believed to be at static conditions. Lower the transducer into the well to a depth of at least 10 feet below the water surface. If water levels in the well prohibit this depth of submergence, place the transducer at the bottom of the well.
- 6. Place the slug into the well and allow the water level to return to its static level.
- 7. Input necessary functions into the datalogger, including monitoring well identification, date, static level, etc.
- 8. Simultaneously withdraw the slug and activate the data logger. Slug withdrawal should be rapid, with the slug emerging completely from the water within 2 to 4 seconds of the test start.
- 9. Continue test until the water level recovers to about 70 percent of the initial level.
- 10. Stop the data acquisition program, remove the pressure transducer, and decontaminate all equipment as discussed above.
# **ATTACHMENT D Drive Point Piezometer Installation**

### **Drive Point Piezometer Instructions**

#### Model 615 & 615S

#### Installation with a Manual Slide Hammer

- 1. Ensure that all components are clean prior to use.
- 2. Cut the piezometer tubing to the depth of the proposed installation plus an additional 5 feet.
- 3. Connect the piezometer tubing to barbed fitting on the Drive Point Piezometer Tip by pushing firmly until the tubing reaches the base of the fitting.
- 4. Slide a length of extension drive pipe over the tubing, and thread it firmly onto the Drive Point Piezometer Tip. Tighten with a pipe wrench.

**WARNING:** When connecting or removing the Tubing Bypass, it is very important to hold the tubing to prevent it from turning. Failure to do so may result in the tube being dislodged from the Drive-Point Tip.

- 5. Hold the tubing to prevent it from turning, then slide the Tubing Bypass over the tubing and tighten it firmly onto the extension pipe, with the tubing extending through the side hole.
- 6. Thread a Drive Extension and the Drive Head onto the Tubing Bypass onto the Tubing Bypass and tighten firmly.
- 7. Thread the Drive Head onto the Drive Extension and tighten firmly.
- 8. Slide the Slide Hammer over the Drive Head and operate the hammer to drive the device until only about 6 inches of the extension pipe below the Tubing Bypass remains above the ground.
- 9. Hold the tubing to prevent it from turning, then remove all sections of the Drive Head Assembly.
- 10.Slide a coupling over the tubing and tighten firmly onto the previous extension pipe. Slide the next extension pipe over the tubing and tighten it securely.
- 11.Repeat steps 4 through 10 until the desired sampling depth is reached.
- 12.Cut the piezometer tubing to fit flush to the top of the extension pipe. Attach a cap to the top of the piezometer to complete the installation.

#### **Component List**

- Drive Point Piezometer Tip
- Extensions (suitable to reach the required depth)
- Couplings (one required for each extension)
- Manual Slide Hammer
- Piezometer tubing (suitable to reach the required depth)
- Drive Head Assembly (3 parts)
  - Includes: Drivehead, Drive Extension & Tubing Bypass
- Cap (101057)



#### Important Note about Shielded Piezometers

- 1. Before driving into the ground, ensure that the shield is on firmly and the o-ring seats properly.
- 2. Once the desired sampling depth is reached, retract the piezometer the length of the shield. At this time, sampling may commence.



## **APPENDIX E Quality Assurance Project Plan**

### **Quality Assurance Project Plan**

West Station Former Manufactured Gas Plant Site Rochester, New York

Prepared for:

Rochester Gas and Electric Corporation 89 East Avenue Rochester, New York 14649-0001

Prepared by:

**Geomatrix Consultants, Inc.** 90B John Muir Drive, Suite 104 Amherst, New York 14228

February 2008

Project No. 12661.002

#### **TABLE OF CONTENTS**

Page
------

1.0	PROJECT DESCRIPTION 1.1 SCOPE AND OBJECTIVES	1 1
2.0	<ul> <li>PROJECT ORGANIZATION AND RESPONSIBILITY.</li> <li>2.1 PROJECT ORGANIZATION</li> <li>2.2 PRINCIPAL INVESTIGATOR</li> <li>2.3 FIELD PROJECT MANAGER.</li> <li>2.4 QUALITY ASSURANCE (QA) RESPONSIBILITIES</li> <li>2.5 FIELD RESPONSIBILITIES</li> <li>2.6 LABORATORY RESPONSIBILITIES</li> </ul>	2 2 2 3 3 3 4
3.0	QUALITY ASSURANCE OBJECTIVES FOR MEASUREMENT DATA3.1PRECISION – FIELD AND LABORATORY3.2ACCURACY3.3COMPLETENESS3.4DATA REPRESENTATIVENESS3.5COMPARABILITY3.6LEVEL OF QC EFFORT FOR SAMPLE PARAMETERS	6 6 7 8 8 8
4.0	SAMPLING PROCEDURES14.1SAMPLING SCHEME14.2SAMPLE COLLECTION14.3EQUIPMENT DECONTAMINATION PROCEDURE14.4SAMPLE CONTAINERS14.5SAMPLE PRESERVATION14.6SAMPLE SHIPMENT1	0 0 0 0 0 1 1
5.0	CUSTODY PROCEDURES1	2
6.0	CALIBRATION PROCEDURES AND FREQUENCY16.1FIELD INSTRUMENT CALIBRATION16.2LABORATORY INSTRUMENT CALIBRATION1	3 3 3
7.0	ANALYTICAL PROCEDURES17.1FIELD ANALYTICAL PROCEDURES17.2LABORATORY ANALYTICAL PROCEDURES1	4 4 4
8.0	DATA REDUCTION, VALIDATION, AND REPORTING18.1DATA REDUCTION18.2DATA VALIDATION18.3DATA REPORTING1	5 5 5 5
9.0	QUALITY ASSURANCE APPROVALS	6
10.0	REFERENCES	7

#### TABLE OF CONTENTS (Continued)

#### TABLES

Table 1Analytical Sampling Parameters for Water, Soils, and Sediment MatricesTable 2Analytical Program Summary

#### ATTACHMENTS

Attachment A TestAmerica -Buffalo -- Certifications and Quality Assurance Manual (electronic format)

#### **QUALITY ASSURANCE PROJECT PLAN**

RG& E West Station Former Manufactured Gas Plant (MGP) Site Rochester, New York Site No. V00593-8 Index No. B-0535-98-07

#### **1.0 PROJECT DESCRIPTION**

This Quality Assurance Project Plan (QAPP) presents the organization, objectives, and specific Quality Assurance/Quality Control (QA/QC) procedures associated with the Remedial Investigation for the West Station Former Manufactured Gas Plant (MGP) Site (herein referred to as the Site) in Rochester, New York. This QAPP provides guidance and specifications to assure that the resulting data are of known quality and meet the needs of the project goals. The types, numbers, and locations of environmental sampling to be performed are described in the site-specific Remedial Investigation Work Plan (RIWP). Field procedures for all environmental sampling activities are detailed in the Field Sampling Plan (FSP) which is provided as Appendix D of the RIWP.

#### **1.1** SCOPE AND OBJECTIVES.

The remedial investigation activities described in Section 2.0 of the RIWP will supplement prior investigation results and will generate a combined data set sufficient to assess:

- Extent of on-site MGP-related material (i.e, NAPL, tar-like material);
- Nature and extent of chemical constituents in Site media (i.e., soil, groundwater); and
- Potential for human health and ecological risks posed by MGP and/or non-MGP derived chemical constituents at the Site.

In order to accomplish these objectives, a plan for the collection and analysis of samples has been prepared.

Field team personnel will collect environmental samples in accordance with the rationale and protocols described in the RIWP. Upon completion of the chemical analysis, the resulting data will undergo a review by a qualified person to ensure that the identification and quantitation of each element or compound were properly done. After the review is completed, the results will

I:\Project\12661.002 RG&E West Station\Work Plan\Appendices\Appendix E - QAPP\Appendix E - QAPP text.doc

be entered into a database and checked for proper data entry. The data will then be used to generate tables and graphs for reporting purposes.

#### 2.0 PROJECT ORGANIZATION AND RESPONSIBILITY

All technical aspects of the performance of the study will be performed by Geomatrix Consultants, Inc. (Geomatrix) who is the principal environmental professional responsible for the performance of all services required to implement each phase, including field operations, laboratory testing, data management, and data analysis and reporting. Geomatrix will perform the field investigation, prepare reports, and perform any subsequent studies. In addition, Geomatrix will also provide project management. The various quality assurance, field, laboratory and management responsibilities of key project personnel are defined below.

#### 2.1 **PROJECT ORGANIZATION**

The lines of authority specific to this investigation are tentatively outlined below. Management responsibilities and experience requirements for key project personnel are also defined.



#### **2.2 PRINCIPAL INVESTIGATOR**

The Principal Investigator (PI) must have technical expertise in the field of study and must have proven capabilities managing environmental investigations. The principal investigator

will be responsible for developing the objectives of the work, for directing the work of others on the project team, and for evaluating and interpreting the results, and for communicating the basis, objectives, and results of the work to interested parties.

#### 2.3 FIELD PROJECT MANAGER

The Field Project Manager (PM) has the overall responsibility for management of the field and laboratory work. The responsibilities of the Field PM are to assist with work plan development, assist with facilitating the understanding of the work plan scope with others on the project team, assist with the formulation of possible work scope revisions if the need arises, manage aspects of project implementation including subcontractors, and compile project data.

#### 2.4 QUALITY ASSURANCE (QA) RESPONSIBILITIES

The QA Officer will remain independent of direct job involvement and day-to-day operations, and have direct access to corporate executive staff as necessary, to resolve any QA dispute. The QA Officer will be responsible for auditing the implementation of the QA program in conformance with the demands of specific investigations and USEPA/NYSDEC requirements. The QA Officer has sufficient authority to stop work on the investigation as deemed necessary in the event of serious QA issues. Specific function and duties include:

- Performing QA audits on various phases of the field operations;
- Reviewing and approving QA plans and procedures;
- Providing QA technical assistance to project staff;
- Reporting on the adequacy, status, and effectiveness of the QA program on a regular basis to the RI Leader for technical operations.
- Responsible for the data validation of all sample results from the analytical laboratory.

#### 2.5 FIELD RESPONSIBILITIES

Field staff for this project is drawn from the Geomatrix pool of qualified resources. The field investigation staff will gather and analyze data, and to prepare various task reports and support materials. All of the designated technical team members are experienced professionals who possess the degree of specialization and technical competence required to effectively and efficiently perform the required work.

#### 2.6 LABORATORY RESPONSIBILITIES

The laboratory tentatively assigned with responsibility for chemical analyses of environmental samples is TestAmerica Laboratories located at 10 Hazelwood Drive, Suite 106, Amherst, New York 14228. TestAmerica follows laboratory procedures in accordance with the USEPA Contract Laboratory Program (CLP) and New York State Department of Environmental Conservation Analytical Services Protocols (NYSDEC ASP) and maintains New York State Department of Health ELAP certification.

#### TestAmerica, Project Manager

The project manager is responsible for the management of the analytical requirements for sample analysis and will report directly to the Principal Investigator. The project manager provides a complete interface with clients from initial project specification to final deliverables.

#### TestAmerica Laboratory Director

The Laboratory Director is a technical advisor and is responsible for summarizing and reporting overall unit performance. Responsibilities of the TestAmerica Laboratory Director include:

- Provide technical, operational, and administrative leadership.
- Allocation and management of personnel and equipment resources.
- Quality performance of the facility.
- Certification and accreditation activities.
- Compliance with audits and corrective actions..

#### TestAmerica Quality Assurance Director (QA Director)

The TestAmerica QA Director has the overall responsibility for data after it leaves the laboratory. The TestAmerica QA Director will be independent of the laboratory but will communicate data issues through the TestAmerica Laboratory Director. In addition, the TestAmerica QA Director will:

- Oversee laboratory QA.
- Oversee QA/QC documentation.
- Conduct detailed data review.

- Determine whether to implement laboratory corrective actions, if required.
- Define appropriate laboratory QA procedures.
- Prepare laboratory SOPs.

Independent QA review will be provided by the TestAmerica Laboratory Director and QA Director prior to release of all data to Geomatrix.

#### TestAmerica Sample Management Office

The TestAmerica Sample Management Office will report to the TestAmerica Laboratory Director. Responsibilities of the TestAmerica Sample Management Office will include:

- Receiving and inspecting the incoming sample containers.
- Recording the condition of the incoming sample containers.
- Signing appropriate documents.
- Verifying chain-of-custody.
- Notifying laboratory manager and laboratory supervisor of sample receipt and inspection.
- Assigning a unique identification number and customer number, and entering each into the sample receiving log.
- With the help of the laboratory manager, initiating transfer of the samples to appropriate lab sections.
- Controlling and monitoring access/storage of samples and extracts.

#### TestAmerica Technical Staff (TS)

The TestAmerica TS will be responsible for sample analyses and identification of corrective actions. The staff will report directly to the TestAmerica Laboratory Director.

#### 3.0 QUALITY ASSURANCE OBJECTIVES FOR MEASUREMENT DATA

The overall objectives and criteria for assuring quality for this effort are discussed below. This QAPP addresses how the acquisition and handling of samples and the review and reporting of data will be documented. The objectives of this QAPP are to address the following:

- The procedures to be used to collect, preserve, package, and transport soil, groundwater, and sediment;
- Field data collection;
- Record keeping;
- Data management;
- Chain-of-custody procedures; and
- Precision, accuracy, completeness, representativeness, decision rules, comparability and level of quality control effort conformance for sample analysis and data management by TestAmerica Buffalo using EPA and NYSDEC ASP analytical methods.

Analytical methods and detection/reporting limits for chemical parameters to be analyzed during this RI for water and soils/sediment are listed in Table 1. A summary of analytical parameters to be analyzed for each matrix and sampling activity is provided in Table 2. Water levels and select water quality parameters (i.e., pH, turbidity, specific conductance, redox potential (Eh), temperature, dissolved oxygen) will be measured in the field as described in the Field Sampling Plan (Appendix D to the RIWP).

The goals for precision, accuracy, and completeness intended for use on this project are discussed in Sections 3.1 through 3.3. Laboratory quality assurance objectives are presented in the analytical laboratory's QA/QC Plan, which is located in Attachment A. TestAmerica - Buffalo is the analytical laboratory tentatively selected to analyze environmental samples for this RI.

Data will be reported completely. No data will be omitted unless an error occurred in the analyses or the run was invalidated because of QC sample recovery or poor precision.

#### 3.1 PRECISION – FIELD AND LABORATORY

Precision is a measurement of the degree to which two or more measurements are in agreement, which is quantitatively assessed based on the standard deviation. Analytical precision in the

laboratory is assessed through the calculation of relative percent difference (RPD) and relative calculation of relative standard deviations (RSD) for three or more replicate samples. Precision can expressed in terms of standard deviation, coefficient of deviation, range, or relative range. Field sample precision will be assessed through collection and analysis of blind duplicate samples collected at a frequency of 1 per 20 samples for each matrix submitted.

Laboratory precision will be assessed through the analysis of matrix spike/matrix spike duplicate (MS/MSD) and field duplicate samples for organic parameters. For inorganic parameters, precision will be assessed through the analysis of field duplicate pairs. The frequency of laboratory precision duplicate analysis will be consistent with NYSDEC ASP requirements.

#### **3.2** ACCURACY

Accuracy is the degree of agreement between an observed value and an accepted reference of true value. Accuracy in the field is assessed through the use of rinsate blanks and trip blanks and through the adherence to sample handling, preservation and holding times. One trip blank will accompany each batch of water matrix sample containers shipped to the laboratory for volatile organic chemical analysis. Laboratory accuracy is assessed through the analysis of a matrix spike/matrix spike duplicate (MS/MSD) (1 per 20 samples), standard reference materials (SRM), laboratory control samples (LCS), and surrogate compounds, and the determination of percent recoveries.

Accuracy for field measured parameters including pH, turbidity, specific conductance, Eh, temperature, and dissolved oxygen will be assessed through instrument calibration standards discussed in instrument calibration and maintenance of the FSP (Appendix D of the RIWP).

#### **3.3 COMPLETENESS**

Data completeness is a measure of the amount of valid data obtained from a prescribed measurement system as compared with that expected and required to meet the project goals. Analytical and field completeness will be addressed by applying data quality checks and assessments and to ensure that the data collected are valid and significant.

In addition, precautions will be taken during packing and shipping of samples to minimize the possibility of breakage. However, some samples may be lost or results deemed questionable due to sample matrix effects or internal QC problems. The QC objective for completeness is generation of valid data for the analyses requested.

#### **3.4 DATA REPRESENTATIVENESS**

Data representativeness expresses the degree to which data accurately and precisely represents a characteristic of a population, parameter variations at a sampling point, a process condition, or an environmental condition within a defined spatial and/or temporal boundary. Proposed field testing and measurement procedures were selected to maximize the degree to which the field data will represent the conditions at the Site, and the matrix being sampled or analyzed.

Sampling will be performed with strict adherence to the sampling methods described in the RIWP and FSP (Appendix D) and procedures defined in Section 4.0 of this QAPP. These sampling procedures were developed to preserve the representativeness of the collected samples. In addition, samples will be properly preserved, and stored prior to, during, and after the shipment to the laboratory. Finally, samples will be prepared and analyzed within holding times so as to preserve the integrity of the samples.

#### **3.5 COMPARABILITY**

Data comparability expresses the confidence with which one data set can be compared to another data set. Procedures for field measurements, contained in the FSP, will ensure that tests performed at various locations across the Site are conducted using accepted procedures, in a consistent manner between locations and over time, and including appropriate QA/QC procedures to ensure the validity of the data. Sampling procedures for environmental matrices are provided in Section 4.0 to ensure that samples are collected using accepted field techniques.

Environmental samples will be analyzed by the laboratory using consistent protocols for sample preservation, holding times, sample preparation, analytical methodology, and QC as described in USEPA SW-846 and NYSDEC ASP.

#### **3.6** LEVEL OF QC EFFORT FOR SAMPLE PARAMETERS

Field blank, method blank, trip blank, field duplicate, laboratory duplicate, laboratory control, standard reference materials (SRM) and matrix spike samples will be analyzed to assess the quality of the data resulting from the field sampling and analytical programs. QC samples are discussed below.

• Rinsate and trip blanks consisting of laboratory provided deionized water will be submitted to the analytical laboratories to provide the means to assess the quality of the data resulting from the field-sampling program.

- Rinsate blank samples are analyzed to check for proper non-disposable sample collection equipment decontamination procedures and assessment of chemical constituents at the Site that may cause sample contamination.
- Trip blanks are used to assess the potential for cross-contamination of samples due to contaminant migration during sample shipment and storage.
- Method blank samples are generated within the laboratory and used to assess contamination resulting from laboratory procedures.
- Duplicate samples are analyzed to check for sampling and analytical reproducibility.
- MS/MSD samples provide information about the effect of the sample matrix on the digestion and measurement methodology. Depending on Site-specific circumstances, one MS/MSD/Dup (for inorganics only) should be collected for every 20 or fewer investigative samples to be analyzed for organic and inorganic chemicals of a given matrix. MS/MSD samples are designated/collected for all samples.

The general level of QC effort will be one field duplicate and one rinsate blank for every 20 or fewer investigative samples of a given matrix. One trip blank consisting of laboratory provided deionized ultra pure water will be included along with each sample delivery group of aqueous samples that requires analysis for volatile organic compounds. Sampling procedures are specified in the FSP (Appendix D of the RIWP).

#### 4.0 SAMPLING PROCEDURES

Quality assurance (QA) practices associated with sampling activities are designed to ensure that samples are collected from documented locations, and that the samples collected are representative of the natural conditions at that location. Also, if samples of a known type, such as from a particular geologic matrix, are desired, the QA program should ensure that those samples are collected.

#### 4.1 SAMPLING SCHEME

The rationale for the selection of sampling locations is described in Section 1.5.3 and Section 2 of the RIWP.

#### 4.2 SAMPLE COLLECTION

Samples will be collected according to the methods described in the FSP (Appendix D). Each sample will be assigned a unique field sample identifier.

#### 4.3 EQUIPMENT DECONTAMINATION PROCEDURE

Reusable sampling equipment, including split spoons, spatulas, bowls, and trowels, will be decontaminated prior to each use, with the addition of a nitric acid rinse for removal of trace inorganic compounds as described in the FSP (Appendix D).

The effectiveness of decontamination activities will be monitored by the analysis of equipment rinsate blanks. Equipment rinsate blanks will be collected at a rate of one per 20 samples by flushing distilled water over the reusable sampling equipment and collecting the liquid in appropriate QC-acceptable containers. Equipment rinsate blanks will be delivered to the laboratory and analyzed for the parameter list consistent with soil/sediment analyses.

Wastewater generated during decontamination activities will be containerized in DOTapproved storage drums or a polyethylene tank, pending characterization and subsequent proper disposal at licensed facility.

#### 4.4 SAMPLE CONTAINERS

Sample containers will be QC-acceptable, pre-cleaned glass containers with Teflon-lined lids or polyethylene bottles. Sample containers appropriate for analysis will be provided to the field team by the laboratory and will include a sample container schematic assuring adequate volume and proper container type for requested analysis.

I:\Project\12661.002 RG&E West Station\Work Plan\Appendices\Appendix E - QAPP\Appendix E - QAPP text.doc

#### 4.5 SAMPLE PRESERVATION

Samples will be preserved by immediately cooling them on ice to approximately 4°C, and maintaining that temperature throughout the chain of custody. Bottles for aqueous samples will be pre-preserved by the laboratory.

#### 4.6 SAMPLE SHIPMENT

Samples shipped to the laboratory will be stored in a cooler at approximately 4°C until they are received by the laboratory. A courier service will be arranged by the laboratory to pick-up samples collected that day. Samples will be packed carefully to ensure the integrity of the sampling containers during shipment. A complete and accurate Chain of Custody (COC) will accompany the samples. The Field Project Manager will be responsible for proper sample shipment and documentation.

#### 5.0 CUSTODY PROCEDURES

Sample custody is controlled and maintained through the chain-of-custody procedures. Chain of custody is the means by which the possession and handling of samples will be tracked from the source (field) to their final disposition, the laboratory. A sample is considered to be in a person's custody if it is in the person's possession or it is in the person's view after being in his or her possession or it was in that person's possession and that person has locked it in a vehicle or room.

Sample containers will be provided by the analytical laboratory and are certified by the manufacturer to be free of contaminants in accordance with Specification and Guidance for Obtaining Contaminant Free Sample Containers, EPA 540/R-93/051 and OSWER Directive 9240.0-05A. The laboratory will preserve sample containers, as necessary, before shipment to the Site.

TestAmerica-Buffalo laboratory custody procedures are discussed in the TestAmerica Quality Assurance Manual located in Attachment A.

#### 6.0 CALIBRATION PROCEDURES AND FREQUENCY

This section describes the calibration procedures and the frequency at which these procedures will be performed for both field and laboratory instruments.

#### 6.1 FIELD INSTRUMENT CALIBRATION

Quantitative field data to be obtained during groundwater and sediment pore water sampling include pH, Eh, turbidity, specific conductance, temperature, and depth to groundwater. Quantitative water level measurements will be obtained with an electronic sounder or steel tape, which require no calibration. Quantitative field data to be obtained during soil sampling includes dust monitoring and organic vapor monitoring and screening soil samples for the presence of volatile organic compounds. This screening will include headspace evaluations using a photoionization (PID) monitor. The instruments will be maintained and calibrated according to the manufacturer's guidelines and recommendations Calibration results will be recorded in the Project Field Book.

#### 6.2 LABORATORY INSTRUMENT CALIBRATION

Equipment and instruments used at the laboratory are operated, maintained and calibrated according to the manufacturer's guidelines and recommendations, as well as to criteria set forth in the applicable analytical methodology. Operation and calibration are performed by personnel who have been properly trained in these procedures. Documentation of calibration information is maintained in appropriate reference files. The frequency of calibration and concentration of calibration standards are determined by the manufacturer's guidelines, the analytical method, or the requirements of special contracts. Generally, purchased standards have a shelf life of 12-36 months and prepared standards have a shelf life of 1-12 months. Recalibration is required at anytime the instrument is not operating correctly or functioning at the proper sensitivity. Brief descriptions of the calibration procedures for major laboratory equipment and instruments are described in the QA Manual.

#### 7.0 ANALYTICAL PROCEDURES

Groundwater, surface water, soil and sediment samples collected during field sampling activities are expected to be analyzed by TestAmerica-Buffalo, 10 Hazelwood, Suite 106, Amherst, New York, 14228, (716) 691-2600.

#### 7.1 FIELD ANALYTICAL PROCEDURES

Field procedures for collecting and preserving samples are described in the FSP located in Appendix D of the Work Plan.

#### 7.2 LABORATORY ANALYTICAL PROCEDURES

This section describes the analytical procedures to be followed in the laboratory. Laboratory analytical procedures will follow the most recent NYSDEC ASP protocols. Analytical methods, method detection limits, and reporting limits selected for use in this RI are listed in Table 1 for groundwater, and soils/sediment. Table 2 (Analytical Program Summary) lists the estimated number of samples and analytical methods anticipated during the RI. The laboratory will provide analytical services necessary to analyze these samples consistent with NYSDEC ASP requirements. General laboratory analytical procedures and sample handling procedures are presented in TestAmerica's QA Manual in Attachment A.

#### 8.0 DATA REDUCTION, VALIDATION, AND REPORTING

#### 8.1 DATA REDUCTION

Analysis results will be reduced to the concentration units specified in the analytical procedures using the equations provided in the analytical references in Section 7. Blank correction will not be performed, but blank analysis results will be documented. Calculations will be independently checked according to the procedures of the laboratory.

#### 8.2 DATA VALIDATION

Data generated by the laboratory will be reviewed by the Quality Assurance Officer and submitted for third party data validation prior to reporting, following the "Guidance for the Development of Data Usability Summary Reports", as documented by the NYSDEC Division of Environmental Remediation. The review will include a check of the accuracy of log-in information, a transcription check, checks of initial, continuing, and QC check standard results, method and field blank results, spiked sample results, replicates, and other QC parameters, as well as checks of compound identifications and calculations. The data validator and QA Officer will attempt to reconcile any QC problems with the laboratory prior to reporting. If certain problems cannot be corrected, the data will be clearly flagged in any reports.

#### 8.3 DATA REPORTING

Laboratory data generated by TestAmerica Buffalo will be reported in NYSDEC ASP Category B deliverables format. The laboratory will also provide a data summary in an electronic format such as MS Excel.

#### 9.0 QUALITY ASSURANCE APPROVALS

<u>Name</u>	<u>Title</u>	<u>Signature</u>	<u>Date</u>
Michael Cummings	Field PM		
Kelly McIntosh	QA Officer		
Richard Frappa	Principal in Charge		

#### **10.0 REFERENCES**

New York State Department of Environmental Conservation. December 1991. Analytical Services Protocol

U.S. Environmental Protection Agency (USEPA), 1986. Test Methods for Evaluating Solid Waste Physical/Chemical Methods. SW-846 Third Edition. Office of Solid Waste, USEPA, Washington DC.

# Table 1 Analytical Sampling Parameters for Water, Soil, and Sediment Matrices West Station Former MGP Site Rochester, New York

	Water			Soil/Se		
	Labonatam, DI	Laboratory		Laboratowy DI	Laboratory	
Analyte	Laboratory KL	MDL	Units	Laboratory KL	MDL	Units
TCL Volatile Organic Comp	ounds 8260 ¹					
1,1,1-Trichloroethane	1	0.26	ug/L	5	0.36	ug/kg
1,1,2,2-Tetrachloroethane	1	0.48	ug/L	5	0.33	ug/kg
1,1,2-Trichloroethane	1	0.42	ug/L	5	0.25	ug/kg
1,1-Dichloroethane	1	0.27	ug/L	5	0.25	ug/kg
1,1-Dichloroethene	1	0.29	ug/L	5	0.61	ug/kg
1,2-Dichloroethane	1	0.46	ug/L	5	0.25	ug/kg
1,2-Dichloroethene (Total)	2	0.70	ug/L	10	2.62	ug/kg
1,2-Dichloropropane	1	0.33	ug/L	5	0.26	ug/kg
2-Butanone	5	2.49	ug/L	25	2.02	ug/kg
2-Hexanone	5	2.39	ug/L	25	1.74	ug/kg
4-Methyl-2-pentanone	5	2.34	ug/L	25	1.64	ug/kg
Acetone	5	2.48	ug/L	25	1.97	ug/kg
Benzene	1	0.35	ug/L	5	0.25	ug/kg
Bromodichloromethane	1	0.39	ug/L	5	0.26	ug/kg
Bromoform	1	0.26	ug/L	5	0.46	ug/kg
Bromomethane	1	0.28	ug/L	5	0.46	ug/kg
Carbon Disulfide	1	0.48	ug/L	5	0.43	ug/kg
Carbon Tetrachloride	1	0.27	ug/L	5	0.18	ug/kg
Chlorobenzene	1	0.32	ug/L	5	0.22	ug/kg
Chloroethane	1	0.32	ug/L	5	0.36	ug/kg
Chloroform	1	0.34	ug/L	5	0.31	ug/kg
Chloromethane	1	0.35	ug/L	5	0.73	ug/kg
cis-1,3-Dichloropropene	1	0.36	ug/L	5	0.29	ug/kg
Dibromochloromethane	1	0.32	ug/L	5	0.28	ug/kg
Ethylbenzene	1	0.34	ug/L	5	0.35	ug/kg
Methylene chloride	1	0.44	ug/L	5	0.35	ug/kg
Styrene	1	0.31	ug/L	5	0.25	ug/kg
Tetrachloroethene	1	0.36	ug/L	5	0.30	ug/kg
Toluene	1	0.35	ug/L	5	1.62	ug/kg
Total Xylenes	3	0.93	ug/L	15	2.94	ug/kg
trans-1,3-Dichloropropene	1	0.37	ug/L	5	0.25	ug/kg
Trichloroethene	1	0.32	ug/L	5	0.35	ug/kg
Vinyl acetate	5	2.23	ug/L	25	1.04	ug/kg
Vinyl chloride	1	0.24	ug/L	10	0.20	ug/kg

# Table 1 Analytical Sampling Parameters for Water, Soil, and Sediment Matrices West Station Former MGP Site Rochester, New York

Laboratory RL         Laboratory NL         Units         Laboratory RL         Laboratory NL         Laboratory NL         Laboratory NL         Laboratory NL         Units           22-Oxybis(1-Chloropropanc)         10         0.42         ug/L         330         17.64         ug/kg           2.4-5-frichlorophenol         10         0.99         ug/L         330         11.14         ug/kg           2.4-Dischlorophenol         10         0.79         ug/L         330         45.60         ug/kg           2.4-Dischlorophenol         10         0.79         ug/L         330         45.60         ug/kg           2.4-Dischlorophenol         10         0.79         ug/L         330         41.30         ug/kg           2.4-Disnityohenol         10         0.51         ug/L         330         41.33         ug/kg           2.4-Disnityohenol         10         0.51         ug/L         330         2.04         ug/kg           2.Methylphenol         10         0.51         ug/L         330         2.04         ug/kg           2.Mitroaniline         50         0.50         ug/L         1600         53.22         ug/kg           2.Nitroaniline         20         0.37		Water			Soil/Sediment ²		
Analyte         Laboratory RL         MDL         Units         Laboratory RL         Units           TCL Semi-Volatile Organic Compounds 8270 ¹ 22:Oxybis1-Chloropropane)         10         0.42         ug/L         330         17.64         ug/kg.           2.4.5-Trichlorophenol         10         0.99         ug/L         800         36.81         ug/kg.           2.4.5-Trichlorophenol         10         0.99         ug/L         330         8.85         ug/kg.           2.4-Dintorophenol         10         0.96         ug/L         330         45.60         ug/kg.           2.4-Dintoroblacne         10         0.45         ug/L         330         26.13         ug/kg.           2.4-Dintoroblacne         10         0.61         ug/L         330         26.13         ug/kg.           2.4-Dintoroblacne         10         0.61         ug/L         330         2.04         ug/kg.           2.Chlorophthalaene         10         0.63         ug/L         330         2.04         ug/kg.           2.Nitrophinol         10         0.60         ug/L         330         7.72         ug/kg.           2.Nitrophinol         10         0.37         ug/L         1600			Laboratory			Laboratory	1
<b>10</b> 0.42         ug/L         330         17.64            2.4°. Srichlorophenol         10         0.99         ug/L         330         17.64         ug/kg           2.4.5. Srichlorophenol         10         0.99         ug/L         330         11.14         ug/kg           2.4.Dimcthylphenol         10         0.99         ug/L         330         45.60         ug/kg           2.4.Dimcthylphenol         10         0.45         ug/L         330         26.13         ug/kg           2.4.Dimitrobluene         10         0.45         ug/L         330         26.13         ug/kg           2.Chlorophenol         10         0.51         ug/L         330         26.13         ug/kg           2.Chlorophenol         10         0.51         ug/L         330         2.04         ug/kg           2.Altrophynaphthalene         10         0.08         ug/L         330         2.04         ug/kg           2.Nitrophenol         10         0.60         ug/L         1600         53.2.6         ug/kg           2.Nitrophenol         10         0.60         ug/L         1600         53.3.71 <t< th=""><th>Analyte</th><th>Laboratory RL</th><th>MDL</th><th>U<u>nits</u></th><th>Laboratory RL</th><th>MDL</th><th>Units</th></t<>	Analyte	Laboratory RL	MDL	U <u>nits</u>	Laboratory RL	MDL	Units
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	TCL Semi-Volatile Organic Co	mpounds 8270 ¹					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2 2'-Oxybis(1-Chloropropage)	10	0.42	119/L	330	17 64	119/kg
24, 6-Trichlorophenol         10         0.99         ug/L         330         11.14         ug/kg $24$ -Dirchlyphenol         10         0.79         ug/L         330         8.85         ug/kg $24$ -Dirchlyphenol         10         0.96         ug/L         330         45.60         ug/kg $24$ -Dirchlyphenol         50         2.22         ug/L         1600         59.06         ug/kg $26$ -Dirictoluene         10         0.45         ug/L         330         41.30         ug/kg $26$ -Dirictoluene         10         0.51         ug/L         330         5.9         ug/kg $2Chlorophenol$ 10         0.51         ug/L         330         5.9         ug/kg $2Methylphenol$ 10         0.51         ug/L         330         5.19         ug/kg $2Nitroaniline         50         0.50         ug/L         1600         533.26         ug/kg           2Nitroaniline         50         1.55         ug/L         1600         533.26         ug/kg           2Nitroaniline         50         1.55         ug/L         330         5.97         ug/kg        $	2 4 5-Trichlorophenol	10	0.99	ug/L	800	36.81	ug/kg
2.4-Dichlorophenol         10         0.79 $ug/L$ 330         8.85 $ug/kg$ 2.4-Dinterbylphenol         10         0.96 $ug/L$ 330         45.60 $ug/kg$ 2.4-Dintophenol         50         2.22 $ug/L$ 1600         59.06 $ug/kg$ 2.4-Dintophenol         10         0.45 $ug/L$ 330         26.13 $ug/kg$ 2.Chlorophenol         10         0.51 $ug/L$ 330         41.30 $ug/kg$ 2.Chlorophenol         10         0.08 $ug/L$ 330         2.04 $ug/kg$ 2.Methylphenol         10         0.23 $ug/L$ 330         5.19 $ug/kg$ 2.Nitrophenol         10         0.60 $ug/L$ 130         7.72 $ug/kg$ 3.Nitroanline         50         1.55 $ug/L$ 1600         58.29 $ug/kg$ 4.Chlorophenol         10         0.60 $ug/L$ 330         6.94 $ug/kg$ 4.Chlorophenyl phenol         50         1.55 $ug/L$ 1600         58.29 $ug/kg$	2.4.6-Trichlorophenol	10	0.99	ug/L	330	11 14	ug/kg
2-1-DimetryInternal         10         0.96 $ug/L$ 330         43.60 $ug/kg$ 2.4-Dimitrophenol         50         2.22 $ug/L$ 1600         59.06 $ug/kg$ 2.4-Dimitrobleme         10         0.45 $ug/L$ 330         26.13 $ug/kg$ 2.4-Dimitrobleme         10         0.51 $ug/L$ 330         41.30 $ug/kg$ 2.Chlorophenol         10         0.51 $ug/L$ 330         2.04 $ug/kg$ 2.Methylphenol         10         0.53 $ug/L$ 330         2.04 $ug/kg$ 2.Nitroanilne         50         0.50 $ug/L$ 330         7.72 $ug/kg$ 2.Nitroanilne         20         0.37 $ug/L$ 1600         53.26 $ug/kg$ 3.1biroberzdine         50         1.55 $ug/L$ 330         53.71 $ug/kg$ 4.4Choroanilne         10         0.90 $ug/L$ 330         53.71 $ug/kg$ 4.Nitrophenol         10         0.33 $ug/L$ 330         5.64 $ug/kg$	2 4-Dichlorophenol	10	0.79	110/L	330	8 85	119/kg
1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1	2 4-Dimethylphenol	10	0.96	11g/L	330	45.60	119/kg
2-4 Dinitrobluene $10$ $0.45$ $ug/L$ $330$ $24.13$ $2.4$ -Dinitrobluene         10 $0.45$ $ug/L$ $330$ $41.30$ $ug/kg$ $2.$ -Chloronaphthalene         10 $0.08$ $ug/L$ $330$ $11.33$ $ug/kg$ $2.$ -Chlorophenol         10 $0.08$ $ug/L$ $330$ $2.44$ $ug/kg$ $2.$ -Methylphenol         10 $0.23$ $ug/L$ $330$ $5.19$ $ug/kg$ $2.$ -Methylphenol         10 $0.23$ $ug/L$ $1600$ $54.15$ $ug/kg$ $2.$ -Nitroaniline $50$ $0.50$ $ug/L$ $1600$ $53.26$ $ug/kg$ $2.$ -Nitrobhenol         10 $0.60$ $ug/L$ $330$ $5.71$ $ug/kg$ $2.$ -Nitrobhenol         10 $0.90$ $ug/L$ $330$ $53.26$ $ug/kg$ $4.$ -Chlorophenyl phenyl ether         10 $0.33$ $ug/L$ $330$ $5.94$ $ug/kg$ $4.$ -Chlorophenyl phenyl ether         10 $0.33$ <	2 4-Dinitronhenol	50	2.22	110/L	1600	59.06	110/kg
Production         Product Product         Product         Product Product         Product Product         Product         Product         Product         Product         Product	2 4-Dinitrotoluene	10	0.45	110/L	330	26.13	110/kg
Definition         Definition <thdefinition< th="">         Definition         Definit</thdefinition<>	2 6-Dinitrotoluene	10	0.51	11g/L	330	41.30	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2-Chloronaphthalene	10	0.08	110/L	330	11.33	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2-Chlorophenol	10	0.51	11g/L	330	8 59	119/kg
Protection         Protec	2-Methylnaphthalene	10	0.08	11g/L	330	2.04	119/kg
Description         Description         Description         Description           2-Nitrophenol         10         0.60         ug/L         1600         54.15         ug/kg           3.3-Dichlorobenzidine         20         0.37         ug/L         1600         533.26         ug/kg           3.Nitroaniline         50         1.55         ug/L         1600         58.81         ug/kg           4.Bromophenyl phenyl ether         10         0.90         ug/L         330         53.71         ug/kg           4-Chloro-amethylphenol         10         0.60         ug/L         330         6.94         ug/kg           4-Chloro-amethylphenol         10         0.33         ug/L         330         49.55         ug/kg           4-Chloroaniline         10         0.35         ug/L         330         9.40         ug/kg           4-Nitroaniline         50         0.46         ug/L         1600         18.86         ug/kg           Acenaphthene         10         0.05         ug/L         330         1.38         ug/kg           Acetaphthylene         10         0.06         ug/L         330         1.38         ug/kg           Acenaphthylene         1	2-Methylphenol	10	0.23	11g/L	330	5 19	119/kg
Philodatha         Desc. $g_{2L}$ Philodatha $g_{2L}$ <th< td=""><td>2-Nitroaniline</td><td>50</td><td>0.50</td><td>11g/L</td><td>1600</td><td>54.15</td><td>119/kg</td></th<>	2-Nitroaniline	50	0.50	11g/L	1600	54.15	119/kg
Interpretation         Image: Im	2-Nitrophenol	10	0.60	11g/L	330	7 72	119/kg
Description         Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	3 3'-Dichlorobenzidine	2.0	0.37	11g/L	1600	533.26	119/kg
100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100         100 <td>3-Nitroaniline</td> <td>50</td> <td>1 55</td> <td>11g/L</td> <td>1600</td> <td>38.81</td> <td>119/kg</td>	3-Nitroaniline	50	1 55	11g/L	1600	38.81	119/kg
Intervention         Intervention         Intervention         Intervention         Intervention         Intervention           4-Bromophenyl phenyl ether         10         0.60         ug/L         330         6.94         ug/kg           4-Chloroaniline         10         0.33         ug/L         330         49.55         ug/kg           4-Chlorophenyl phenyl ether         10         0.17         ug/L         330         3.60         ug/kg           4-Methylphenol         10         0.35         ug/L         330         9.40         ug/kg           4-Nitroaniline         50         0.46         ug/L         1600         18.86         ug/kg           4-Nitrophenol         50         1.53         ug/L         1600         40.92         ug/kg           Acenaphthene         10         0.01         ug/L         330         1.98         ug/kg           Acetophenone         10         0.06         ug/L         330         4.32         ug/kg           Actophenone         10         0.06         ug/L         330         1.8.51         ug/kg           Benzol(a)nthracene         10         0.26         ug/L         330         1.8.51         ug/kg	4 6-Dinitro-2-methylphenol	50	2.27	11g/L	1600	58.29	119/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	4-Bromonhenvl phenvl ether	10	0.90	110/L	330	53 71	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	4-Chloro-3-methylphenol	10	0.50	110/L	330	6 94	110/kg
Periodalitie         10         0.17         ug/L         330         1.02         ug/kg           4-Chlorophenyl phenyl ether         10         0.17         ug/L         330         3.60         ug/kg           4-Methylphenol         10         0.35         ug/L         1600         18.86         ug/kg           4-Nitroaniline         50         0.46         ug/L         1600         18.86         ug/kg           Acenaphthene         10         0.11         ug/L         330         1.98         ug/kg           Acenaphthylene         10         0.05         ug/L         330         1.38         ug/kg           Acetophenone         10         0.06         ug/L         330         1.38         ug/kg           Acteraphthylene         10         0.06         ug/L         330         4.32         ug/kg           Anthracene         10         0.06         ug/L         330         4.32         ug/kg           Benzo(a)anthracene         10         0.26         ug/L         330         2.91         ug/kg           Benzo(a)pyrene         10         0.39         ug/L         330         3.28         ug/kg           Benzo(a)pyrene	4-Chloroaniline	10	0.33	110/L	330	49 55	110/kg
Perintopheny Pathy react100.170.122.0010.004-Methylphenol100.35ug/L3309.40ug/kg4-Nitroaniline500.46ug/L160018.86ug/kg4-Nitrophenol501.53ug/L160040.92ug/kgAcenaphthene100.11ug/L3301.98ug/kgAcenaphthylene100.05ug/L3301.38ug/kgAcetophenone100.10ug/L3304.32ug/kgAntracene100.06ug/L3304.32ug/kgAntracene100.06ug/L3301.8.51ug/kgBenzo(a)anthracene100.26ug/L33018.51ug/kgBenzo(a)pyrene100.39ug/L3302.91ug/kgBenzo(a)pyrene100.39ug/L3303.28ug/kgBenzo(a)pyrene100.37ug/L3301.86ug/kgBenzo(a)pyrene100.39ug/L3302.03ug/kgBenzo(a)pyrene100.39ug/L3301.86ug/kgBenzo(b)flooranthene100.07ug/L3301.86ug/kgBenzo(a)pyrene100.38ug/L3301.86ug/kgBenzo(b)flooranthene100.07ug/L3301.86ug/kgBenzo(b)flooranthene100.38ug/L3301	4-Chlorophenyl phenyl ether	10	0.55	110/L	330	3 60	110/kg
Principal	4-Methylphenol	10	0.35	110/L	330	9 40	110/kg
Prividulninc         50         6.40         ug/L         1000         10.50         12.53         ug/L           4-Nitrophenol         50         1.53         ug/L         1600         40.92         ug/kg           Acenaphthene         10         0.11         ug/L         330         1.98         ug/kg           Acenaphthylene         10         0.05         ug/L         330         1.38         ug/kg           Acetophenone         10         0.06         ug/L         330         8.66         ug/kg           Artazine         10         0.06         ug/L         330         4.32         ug/kg           Benzaldehyde         50         0.27         ug/L         330         1.85.1         ug/kg           Benzo(a)anthracene         10         0.26         ug/L         330         4.07         ug/kg           Benzo(a)apyrene         10         0.39         ug/L         330         3.0         2.03         ug/kg           Benzo(ghi)perylene         10         0.36         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         1.86         ug/kg	4-Nitroaniline	50	0.35	110/L	1600	18.86	110/kg
Accenaphthene         10         0.11         ug/L         330         1.98         ug/kg           Accenaphthylene         10         0.05         ug/L         330         1.38         ug/kg           Acenaphthylene         10         0.05         ug/L         330         1.38         ug/kg           Acenaphthylene         10         0.10         ug/L         330         8.66         ug/kg           Actazine         10         0.06         ug/L         330         4.32         ug/kg           Benzol(a)anthracene         10         1.09         ug/L         330         18.51         ug/kg           Benzo(a)pyrene         10         0.26         ug/L         330         4.07         ug/kg           Benzo(a)pyrene         10         0.39         ug/L         330         4.07         ug/kg           Benzo(b)fluoranthene         10         0.36         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         1.86         ug/kg           Benzo(k)fluora	4-Nitronhenol	50	1 53	110/L	1600	40.92	110/kg
Acenaphthylene100.01ug/L3301.38ug/kgAcetophenone100.05ug/L3301.38ug/kgAnthracene100.06ug/L3304.32ug/kgAtrazine101.09ug/L3307.51ug/kgBenzaldehyde500.27ug/L33018.51ug/kgBenzo(a)anthracene100.09ug/L3302.91ug/kgBenzo(a)anthracene100.09ug/L3304.07ug/kgBenzo(b)fluoranthene100.39ug/L3303.28ug/kgBenzo(b)fluoranthene100.36ug/L3302.03ug/kgBenzo(k)fluoranthene100.07ug/L3301.86ug/kgBenzo(k)fluoranthene100.07ug/L3301.86ug/kgBenzoic acid150100ug/L3301.86ug/kgBenzoic acid150100ug/L3301.86ug/kgBis(2-chloroethoxy) methane100.07ug/L33014.57ug/kgBis(2-chloroethyl) ether100.18ug/L3304.33ug/kgBis(2-chloroethyl) phthalate100.74ug/L3304.53ug/kgGarbazole100.77ug/L3301.457ug/kgBis(2-chloroethyl) ether100.18ug/L33014.57ug/kgBis(2-chloroethyl) ether<	Acenanhthene	10	0.11	110/L	330	1 98	110/kg
Acetophenone100.00ug/L3301.00ug/kgAnthracene100.10ug/L3308.66ug/kgAtrazine101.09ug/L3304.32ug/kgBenzaldehyde500.27ug/L3307.51ug/kgBenzo(a)anthracene100.09ug/L33018.51ug/kgBenzo(a)anthracene100.26ug/L3302.91ug/kgBenzo(a)pyrene100.09ug/L3303.28ug/kgBenzo(b)fluoranthene100.39ug/L3303.28ug/kgBenzo(k)fluoranthene100.36ug/L3302.03ug/kgBenzo(k)fluoranthene100.07ug/L3301.86ug/kgBenzoic acid150100ug/L3301.86ug/kgBenzoic acid150100ug/L3301.86ug/kgBenzoic acid150100ug/L33010.51ug/kgBis(2-chloroethoxy) methane100.38ug/L33014.57ug/kgBis(2-chloroethyl) ether100.18ug/L33044.53ug/kgBis(2-chloroethyl) ether100.18ug/L33014.57ug/kgBis(2-chloroethyl) ether100.18ug/L33045.33ug/kgCarbazole100.74ug/L33014.57ug/kgButyl benzyl phthalate10<	A cenanhthylene	10	0.05	110/L	330	1 38	110/kg
Anthracene100.10ug/L3304.32ug/kgAnthracene101.09ug/L3304.32ug/kgAtrazine101.09ug/L3307.51ug/kgBenzaldehyde500.27ug/L33018.51ug/kgBenzo(a)anthracene100.26ug/L3302.91ug/kgBenzo(a)pyrene100.09ug/L3304.07ug/kgBenzo(a)pyrene100.39ug/L3303.28ug/kgBenzo(b)fluoranthene100.36ug/L3302.03ug/kgBenzo(k)fluoranthene100.07ug/L3301.86ug/kgBenzo(a)pyrene100.07ug/L3301.86ug/kgBenzo(k)fluoranthene100.07ug/L3301.86ug/kgBenzoic acid150100ug/L4800242.61ug/kgBenzyl alcohol200.30ug/L33010.51ug/kgBis(2-chloroethxy) methane100.07ug/L33014.57ug/kgBis(2-chloroethyl) ether106.10ug/L33054.39ug/kgBuyl benzyl phthalate101.74ug/L33014.57ug/kgCarbazole100.09ug/L3301.95ug/kgDibenzo(a,h)anthracene100.27ug/L3301.99ug/kg	Acetonhenone	10	0.00	110/L	330	8 66	110/kg
Attrazine10 $0.00$ $ug/L$ $330$ $1.02$ $ug/kg$ Benzaldehyde50 $0.27$ $ug/L$ $330$ $18.51$ $ug/kg$ Benzo(a)anthracene10 $0.26$ $ug/L$ $330$ $2.91$ $ug/kg$ Benzo(a)pyrene10 $0.09$ $ug/L$ $330$ $2.91$ $ug/kg$ Benzo(a)pyrene10 $0.09$ $ug/L$ $330$ $2.91$ $ug/kg$ Benzo(b)fluoranthene10 $0.39$ $ug/L$ $330$ $3.28$ $ug/kg$ Benzo(ghi)perylene10 $0.36$ $ug/L$ $330$ $2.03$ $ug/kg$ Benzo(k)fluoranthene10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid150100 $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid150100 $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Bis(2-chloroethoxy) methane10 $0.07$ $ug/L$ $330$ $10.51$ $ug/kg$ Bis(2-chloroethyl) ether10 $0.18$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-chloroethyl) phthalate10 $1.74$ $ug/L$ $330$ $45.33$ $ug/kg$ Caprolactam10 $0.27$ $ug/L$ $330$ $1.99$ $ug/kg$ Carbazole10 $0.09$ $ug/L$ $330$ $1.99$ $ug/kg$ </td <td>Anthracene</td> <td>10</td> <td>0.06</td> <td>110/L</td> <td>330</td> <td>4 32</td> <td>110/kg</td>	Anthracene	10	0.06	110/L	330	4 32	110/kg
IntermIntermIntermIntermIntermIntermIntermIntermBenzaldehyde $50$ $0.27$ $ug/L$ $330$ $18.51$ $ug/kg$ Benzo(a)pyrene $10$ $0.26$ $ug/L$ $330$ $2.91$ $ug/kg$ Benzo(a)pyrene $10$ $0.09$ $ug/L$ $330$ $4.07$ $ug/kg$ Benzo(b)fluoranthene $10$ $0.39$ $ug/L$ $330$ $3.28$ $ug/kg$ Benzo(ghi)perylene $10$ $0.36$ $ug/L$ $330$ $2.03$ $ug/kg$ Benzo(k)fluoranthene $10$ $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzo(acid $150$ $100$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid $150$ $100$ $ug/L$ $330$ $242.61$ $ug/kg$ Benzoic acid $10$ $0.07$ $ug/L$ $330$ $8.07$ $ug/kg$ Biphenyl $10$ $0.07$ $ug/L$ $330$ $9.18$ $ug/kg$ Bis(2-chloroethoxy) methane $10$ $0.18$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-chloroethyl) ether $10$ $0.18$ $ug/L$ $330$ $54.39$ $ug/kg$ Butyl benzyl phthalate $10$ $1.74$ $ug/L$ $330$ $73.02$ $ug/kg$ Caprolactam $10$ $0.09$ $ug/L$ $330$ $1.95$ $ug/kg$ Chrysene $10$ $0.27$ $ug/L$ $330$ $1.99$ $ug/kg$	Atrazine	10	1 09	110/L	330	7 51	110/kg
Definition         3.0         0.27         ug/L         3.00         1.0.1         ug/kg           Benzo(a)anthracene         10         0.26         ug/L         330         2.91         ug/kg           Benzo(a)pyrene         10         0.09         ug/L         330         4.07         ug/kg           Benzo(b)fluoranthene         10         0.39         ug/L         330         3.28         ug/kg           Benzo(ghi)perylene         10         0.36         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         1.86         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         1.86         ug/kg           Benzo(aid         150         100         ug/L         4800         242.61         ug/kg           Benzyl alcohol         20         0.30         ug/L         330         10.51         ug/kg           Bis(2-chloroethoxy) methane         10         0.07         ug/L         330         10.51         ug/kg           Bis(2-chloroethyl) ether         10         0.18         ug/L         330         14.57         ug/kg	Renzaldehvde	50	0.27	110/L	330	18 51	110/kg
Denzo(a)pyrene         10         0.20         ug/L         330         4.07         ug/kg           Benzo(a)pyrene         10         0.39         ug/L         330         4.07         ug/kg           Benzo(b)fluoranthene         10         0.39         ug/L         330         3.28         ug/kg           Benzo(ghi)perylene         10         0.36         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.36         ug/L         330         2.03         ug/kg           Benzo(k)fluoranthene         10         0.07         ug/L         330         1.86         ug/kg           Benzoic acid         150         100         ug/L         4800         242.61         ug/kg           Benzyl alcohol         20         0.30         ug/L         330         8.07         ug/kg           Biphenyl         10         0.07         ug/L         330         10.51         ug/kg           Bis(2-chloroethoxy) methane         10         0.38         ug/L         330         14.57         ug/kg           Bis(2-chloroethyl) ether         10         6.10         ug/L         330         54.39         ug/kg	Renzo(a)anthracene	10	0.27	110/L	330	2 91	110/kg
Defizional/pyrene       10       0.07       ug/L       330       3.28       ug/kg         Benzo(b)fluoranthene       10       0.36       ug/L       330       2.03       ug/kg         Benzo(ghi)perylene       10       0.36       ug/L       330       2.03       ug/kg         Benzo(k)fluoranthene       10       0.07       ug/L       330       1.86       ug/kg         Benzoic acid       150       100       ug/L       4800       242.61       ug/kg         Benzyl alcohol       20       0.30       ug/L       330       8.07       ug/kg         Biphenyl       10       0.07       ug/L       330       10.51       ug/kg         Bis(2-chloroethoxy) methane       10       0.38       ug/L       330       14.57       ug/kg         Bis(2-chloroethyl) ether       10       0.18       ug/L       330       14.57       ug/kg         Bis(2-ethylhexyl) phthalate       10       1.74       ug/L       330       54.39       ug/kg         Caprolactam       10       4.59       ug/L       330       73.02       ug/kg         Carbazole       10       0.09       ug/L       330       1.95       ug/k	Benzo(a)nyrene	10	0.09	110/L	330	4 07	110/kg
Defize(o)Internation10 $0.36$ $ug/L$ $330$ $2.03$ $ug/kg$ Benzo(ghi)perylene10 $0.36$ $ug/L$ $330$ $2.03$ $ug/kg$ Benzo(k)fluoranthene10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid150100 $ug/L$ $4800$ $242.61$ $ug/kg$ Benzyl alcohol20 $0.30$ $ug/L$ $330$ $8.07$ $ug/kg$ Biphenyl10 $0.07$ $ug/L$ $330$ $10.51$ $ug/kg$ Bis(2-chloroethoxy) methane10 $0.38$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-chloroethyl) ether10 $0.18$ $ug/L$ $330$ $54.39$ $ug/kg$ Bis(2-chloroethyl) ether10 $1.74$ $ug/L$ $330$ $54.39$ $ug/kg$ Butyl benzyl phthalate10 $1.74$ $ug/L$ $330$ $73.02$ $ug/kg$ Caprolactam10 $0.09$ $ug/L$ $330$ $1.95$ $ug/kg$ Carbazole10 $0.27$ $ug/L$ $330$ $1.95$ $ug/kg$ Dibenzo(a,h)anthracene10 $0.20$ $ug/L$ $330$ $1.99$ $ug/kg$	Benzo(h)fluoranthene	10	0.39	110/L	330	3.28	110/kg
Benzo(k)fluoranthene10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzo(k)fluoranthene10 $0.07$ $ug/L$ $330$ $1.86$ $ug/kg$ Benzoic acid150100 $ug/L$ $4800$ $242.61$ $ug/kg$ Benzyl alcohol20 $0.30$ $ug/L$ $330$ $8.07$ $ug/kg$ Biphenyl10 $0.07$ $ug/L$ $330$ $10.51$ $ug/kg$ Bis(2-chloroethoxy) methane10 $0.38$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-chloroethyl) ether10 $0.18$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-ethylhexyl) phthalate10 $6.10$ $ug/L$ $330$ $54.39$ $ug/kg$ Butyl benzyl phthalate10 $1.74$ $ug/L$ $330$ $45.33$ $ug/kg$ Caprolactam10 $4.59$ $ug/L$ $330$ $1.95$ $ug/kg$ Chrysene10 $0.27$ $ug/L$ $330$ $1.69$ $ug/kg$ Dibenzo(a.h)anthracene10 $0.20$ $ug/L$ $330$ $1.99$ $ug/kg$	Benzo(ghi)pervlene	10	0.36	110/L	330	2.03	110/kg
Benzoic acid         10         10         ug/L         200         100         ug/L           Benzoic acid         150         100         ug/L         4800         242.61         ug/kg           Benzyl alcohol         20         0.30         ug/L         330         8.07         ug/kg           Biphenyl         10         0.07         ug/L         330         10.51         ug/kg           Bis(2-chloroethoxy) methane         10         0.38         ug/L         330         14.57         ug/kg           Bis(2-chloroethyl) ether         10         0.18         ug/L         330         14.57         ug/kg           Bis(2-ethylhexyl) phthalate         10         6.10         ug/L         330         45.33         ug/kg           Butyl benzyl phthalate         10         4.59         ug/L         330         73.02         ug/kg           Caprolactam         10         0.09         ug/L         330         1.95         ug/kg           Chrysene         10         0.27         ug/L         330         1.69         ug/kg           Dibenzo(a.h)anthracene         10         0.20         ug/L         330         1.99         ug/kg	Benzo(k)fluoranthene	10	0.07	110/L	330	1.86	110/kg
Benzyl alcohol20 $0.30$ $ug/L$ $330$ $2.1.201$ $ug/kg$ Benzyl alcohol20 $0.30$ $ug/L$ $330$ $8.07$ $ug/kg$ Biphenyl10 $0.07$ $ug/L$ $330$ $10.51$ $ug/kg$ Bis(2-chloroethoxy) methane10 $0.38$ $ug/L$ $330$ $9.18$ $ug/kg$ Bis(2-chloroethyl) ether10 $0.18$ $ug/L$ $330$ $14.57$ $ug/kg$ Bis(2-ethylhexyl) phthalate10 $6.10$ $ug/L$ $330$ $54.39$ $ug/kg$ Butyl benzyl phthalate10 $1.74$ $ug/L$ $330$ $45.33$ $ug/kg$ Caprolactam10 $4.59$ $ug/L$ $330$ $73.02$ $ug/kg$ Carbazole10 $0.09$ $ug/L$ $330$ $1.95$ $ug/kg$ Dibenzo(a,h)anthracene10 $0.20$ $ug/L$ $330$ $1.99$ $ug/kg$	Benzoic acid	150	100	110/L	4800	242.61	110/kg
Bin         Dom         Dom <td>Benzyl alcohol</td> <td>20</td> <td>0.30</td> <td>110/L</td> <td>330</td> <td>8 07</td> <td>110/kg</td>	Benzyl alcohol	20	0.30	110/L	330	8 07	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Rinhenvl	10	0.07	110/L	330	10.51	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Ris(2-chloroethoxy) methane	10	0.38	110/L	330	9.18	110/kg
Dist 2 endocury () end10 $0.10$ $0.10$ $0.20$ $0.20$ $1.01$ $0.20$ Bis(2-ethylhexyl) phthalate10 $6.10$ $ug/L$ $330$ $54.39$ $ug/kg$ Butyl benzyl phthalate10 $1.74$ $ug/L$ $330$ $45.33$ $ug/kg$ Caprolactam10 $4.59$ $ug/L$ $330$ $73.02$ $ug/kg$ Carbazole10 $0.09$ $ug/L$ $330$ $1.95$ $ug/kg$ Chrysene10 $0.27$ $ug/L$ $330$ $1.69$ $ug/kg$ Dibenzo(a,h)anthracene10 $0.20$ $ug/L$ $330$ $1.99$ $ug/kg$	Bis(2-chloroethyl) ether	10	0.18	110/L	330	14 57	110/kg
Disc endymetry planate         10         1.0         0.10         1.0         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10         0.10 <td>Bis(2-ethylbexyl) phthalate</td> <td>10</td> <td>610</td> <td>11g/L</td> <td>330</td> <td>54.39</td> <td>119/kg</td>	Bis(2-ethylbexyl) phthalate	10	610	11g/L	330	54.39	119/kg
Daty For Lyr philade         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10 <td>Butyl benzyl phthalate</td> <td>10</td> <td>1 74</td> <td>11g/L</td> <td>330</td> <td>45 33</td> <td>119/kg</td>	Butyl benzyl phthalate	10	1 74	11g/L	330	45 33	119/kg
Carbazole         10         0.09         ug/L         330         1.95         ug/kg           Chrysene         10         0.27         ug/L         330         1.69         ug/kg           Dibenzo(a,h)anthracene         10         0.20         ug/L         330         1.99         ug/kg	Caprolactam	10	4 59	11g/L	330	73.02	119/kg
Chrysene         10 $0.27$ $ug/L$ $330$ $1.69$ $ug/kg$ Dibenzo(a,h)anthracene         10 $0.20$ $ug/L$ $330$ $1.99$ $ug/kg$	Carbazole	10	0.09	110/L	330	1.95	110/kg
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Chrysene	10	0.27	110/L	330	1.69	110/kg
	Dibenzo(a,h)anthracene	10	0.20	ug/L	330	1.99	ug/kg

#### Table 1 Analytical Sampling Parameters for Water, Soil, and Sediment Matrices West Station Former MGP Site

	Water			Soil/Sediment ²		
Analyte	Laboratory RL	Laboratory MDL	Units	Laboratory RL	Laboratory MDL	Units
Dibenzofuran	10	0.10	ug/L	330	1.76	ug/kg
Diethyl phthalate	10	0.11	ug/L	330	5.10	ug/kg
Dimethyl phthalate	10	0.30	ug/L	330	4.40	ug/kg
Di-n-butyl phthalate	10	0.30	ug/L	330	58.35	ug/kg
Di-n-octyl phthalate	10	0.24	ug/L	330	3.95	ug/kg
Fluoranthene	10	0.10	ug/L	330	2.45	ug/kg
Fluorene	10	0.07	ug/L	330	3.89	ug/kg
Hexachlorobenzene	10	0.45	ug/L	330	8.39	ug/kg
Hexachlorobutadiene	10	2.60	ug/L	330	8.64	ug/kg
Hexachlorocyclopentadiene	45	6.06	ug/L	330	51.04	ug/kg
Hexachloroethane	10	2.82	ug/L	330	13.06	ug/kg
Indeno(1,2,3-cd)pyrene	10	0.15	ug/L	330	4.67	ug/kg
Isophorone	10	0.32	ug/L	330	8.44	ug/kg
Naphthalene	10	0.12	ug/L	330	2.81	ug/kg
Nitrobenzene	10	0.54	ug/L	330	7.48	ug/kg
N-Nitroso-Di-n-propylamine	10	0.45	ug/L	330	13.37	ug/kg
N-nitrosodiphenylamine	10	0.26	ug/L	330	9.23	ug/kg
Pentachlorophenol	50	5.14	ug/L	1600	57.90	ug/kg
Phenanthrene	10	0.11	ug/L	330	3.54	ug/kg
Phenol	10	0.45	ug/L	330	17.77	ug/kg
Pyrene	10	0.07	ug/L	330	1.09	ug/kg
Polychlorinated Biphenyls 8082	1					
Arochlor 1016	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1221	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1232	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1242	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1248	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1254	1	0.10	ug/L	0.44	0.02	mg/kg
Arochlor 1260	1	0.10	ug/L	0.44	0.02	mg/kg

Rochester, New York

# Table 1 Analytical Sampling Parameters for Water, Soil, and Sediment Matrices West Station Former MGP Site Rochester, New York

	Water		Water			Soil/Sediment ²		
Analyte	Laboratory RL	Laboratory MDL	Units	Laboratory RL	Laboratory MDL	Units		
Inorganics 6010 ¹								
Aluminum - Total	0.2	0.026	mg/L	10	1.2	mg/kg		
Antimony - Total	0.02	0.0056	mg/L	15	0.69	mg/kg		
Arsenic - Total	0.01	0.0042	mg/L	2	0.37	mg/kg		
Barium - Total	0.002	0.00017	mg/L	0.5	0.05	mg/kg		
Beryllium - Total	0.002	0.00025	mg/L	0.2	0.04	mg/kg		
Cadmium - Total	0.001	0.00036	mg/L	0.2	0.06	mg/kg		
Calcium - Total	0.5	0.021	mg/L	50	10	mg/kg		
Chromium - Total	0.004	0.00044	mg/L	0.5	0.09	mg/kg		
Cobalt - Total	0.004	0.00079	mg/L	0.5	0.11	mg/kg		
Copper - Total	0.01	0.002	mg/L	1	0.5	mg/kg		
Iron - Total	0.05	0.017	mg/L	10	3.1	mg/kg		
Lead - Total	0.005	0.0022	mg/L	1	0.19	mg/kg		
Magnesium - Total	0.2	0.0077	mg/L	20	0.7	mg/kg		
Manganese - Total	0.003	0.00016	mg/L	0.2	0.031	mg/kg		
Nickel - Total	0.01	0.0014	mg/L	0.5	0.14	mg/kg		
Potassium - Total	0.5	0.024	mg/L	30	8.4	mg/kg		
Selenium - Total	0.015	0.0061	mg/L	4	0.59	mg/kg		
Silver - Total	0.003	0.001	mg/L	0.5	0.15	mg/kg		
Sodium - Total	1	0.339	mg/L	140	29	mg/kg		
Thallium - Total	0.02	0.0067	mg/L	6	0.66	mg/kg		
Vanadium - Total	0.005	0.00098	mg/L	0.5	0.095	mg/kg		
Zinc - Total	0.01	0.0036	mg/L	1	0.16	mg/kg		
Inorganics 7470/7471 ¹								
Mercury	0.0002	0.00011	mg/L	0.02	0.0047	mg/kg		
Inorganics 9010/9012 ¹								
Total Cyanide	0.01	0.005	mg/L	1	0.17	mg/kg		

Notes:

RL - Reporting Limit

MDL - Method Detection Limit

1 USEPA. Office of Solid Waste and Emergency Response. *Test Methods for Evaluating Solid Waste SW-846 3rd ed. Washington, DC 1996* 

2 The target reporting limits are based on wet weight. The actual reporting limits will vary based on sample weight and moisture content.

## Table 2Analytical Program SummaryWest Station Former MGP SiteRochester, New York

Sample Locations/Matrix	Analytical Parameters	Analytical Method ¹	Estimated Number of Investigative Samples ²
	TCL VOCs	8260	7 minimum
	TCL SVOCs	8270	7 minimum
Subsurface Soil Samples from Test Pits	PCBs	8082	7 minimum
	TAL Metals	6010B/7000	7 minimum
	Total Cyanide	9012	7 minimum
	TCL VOCs	8260	86
Subsurface Sail Samples from Sail	TCL SVOCs	8270	86
Borings	PCBs	8082	86
Doningo	TAL Metals	6010B/7000	86
	Total Cyanide	9012	86
	TCL VOCs	8260	9
	TCL SVOCs	8270	9
Surface Soil Sampling	PCBs	8082	9
	TAL Metals	6010B/7000	9
	Total Cyanide	9012	9
	TCL VOCs	8260	44 minimum
	TCL SVOCs	8270	44 minimum
Groundwater Sampling (Overburden &	PCBs	8082	44 minimum
Bedrock)	TAL Metals	6010B/7000	44 minimum
	Total Cyanide	9012	44 minimum
	Field Measured Parameters	NA	44 minimum
	TCL VOCs	8260	12 minimum
	TCL SVOCs	8270	12 minimum
	MGP Chemical Fingerprint	GC/MS-SIM modified 8270	12 minimum
Sodimont Sompling (solid)	PCBs	8082	12 minimum
Sediment Sampling (solid)	TAL Metals	6010B/7000	12 minimum
	Total Cyanide	9012	12 minimum
	Total Organic Carbon	Lloyd-Kahn	12 minimum
	Grain Size	ASTM Method D-422	12 minimum
	TCL VOCs	8260	6 minimum
	TCL SVOCs	8270	6 minimum
Sediment Sampling (pore water)	PCBs	8082	6 minimum
	TAL Metals	6010B/7000	6 minimum
	Total Cyanide	9012	6 minimum
Investigation Derived Waste Material	*	*	1 soil/1 water

* Analytical parameter list and analytical methods dependant upon disposal facility requirements.

Notes:

1 USEPA. Office of Solid Waste and Emergency Response. *Test Methods for Evaluating* 

Solid Waste SW-846 3rd ed. Washington, DC 1996. Analytical protocols will be consistent with most recent NYS ASP requiremen 2 Number of samples subject to change and do not include field samples collected for quality control.

3 Dissolved oxygen, Specific conductance, Redox potential, pH, Turbidity.

## ATTACHMENT A TestAmerica Laboratories – Quality Assurance Manual



## **Cover Page:**

## **Quality Assurance Manual**

TestAmerica Buffalo 10 Hazelwood Drive Amherst, New York 14228 716.504.9800 716.691.7991

www.testamericainc.com

#### **Copyright Information:**

This documentation has been prepared by TestAmerica Laboratories, Inc. and its affiliates ("TestAmerica"), solely for their own use and the use of their customers in evaluating their qualifications and capabilities in connection with a particular project. The user of this document agrees by its acceptance to return it to TestAmerica upon request and not to reproduce, copy, lend, or otherwise disclose its contents, directly or indirectly, and not to use if for any other purpose other than that for which it was specifically provided. The user also agrees that where consultants or other outside parties are involved in the evaluation process, access to these documents shall not be given to said parties unless those parties also specifically agree to these conditions.

THIS DOCUMENT CONTAINS VALUABLE CONFIDENTIAL AND PROPRIETARY INFORMATION. DISCLOSURE, USE OR REPRODUCTION OF THESE MATERIALS WITHOUT THE WRITTEN AUTHORIZATION OF TESTAMERICA IS STRICTLY PROHIBITED. THIS UNPUBLISHED WORK BY TESTAMERICA IS PROTECTED BY STATE AND FEDERAL LAW OF THE UNITED STATES. IF PUBLICATION OF THIS WORK SHOULD OCCUR THE FOLLOWING NOTICE SHALL APPLY:

©COPYRIGHT 2007 TESTAMERICA LABORATORIES, INC. ALL RIGHTS RESERVED

Facility Distribution No. 001 Controlled Copy Distributed To: Waste Management

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Title Page 1 of 1

### Title Page:

### Quality Assurance Manual Approval Signatures

January 11, 2008

Date

Jaggy Gray-Eroman

Quality Assurance Manager - Peggy Gray-Erdmann

20 serel

Laboratory Director - Chris Spencer

Technical Director & EH&S - Kenneth Kasperek

Customer Service Manager - Mark Nemec

**Operations Manager – John Schove** 

January 11, 2008 Date January 11, 2008

Date

January 11, 2008

Date

January 11, 2008

Date

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 1 of 9

#### **SECTION 2**

#### TABLE OF CONTENTS

Section No.	Title	Page No.	Effective Date
-	COVER PAGE	COVER	02/01/2008
1.0	TITLE PAGE	1-1	02/01/2008
2.0	TABLE OF CONTENTS	2-1	02/01/2008
3.0	INTRODUCTION	3-1	02/01/2008
3.1	Introduction And Compliance References	3-1	02/01/2008
3.2	Terms And Definitions	3-2	02/01/2008
3.3	Scope / Fields Of Testing	3-2	02/01/2008
3.4	Management Of The Manual	3-2	02/01/2008
4.0	ORGANIZATION AND MANAGEMENT (NELAC 5.4.1)	4-1	02/01/2008
4.1	Overview	4-1	02/01/2008
4.2	Roles And Responsibilities	4-2	02/01/2008
4.3	Deputies	4-14	02/01/2008
5.0	QUALITY SYSTEM (NELAC 5.4.2)	5-1	02/01/2008
5.1	Quality Policy Statement	5-1	02/01/2008
5.2	Ethics And Data Integrity	5-1	02/01/2008
5.3	Quality System Supporting Documentation	5-2	02/01/2008
5.4	Qa/Qc Objectives For The Measurement Of Data	5-3	02/01/2008
5.5	Criteria For Quality Indicators	5-5	02/01/2008
5.6	Statistical Quality Control	5-5	02/01/2008
5.7	Quality System Metrics	5-6	02/01/2008
6.0	DOCUMENT CONTROL (NELAC 5.4.3)	6-1	02/01/2008
6.1	Overview	6-1	02/01/2008
6.2	Document Approval And Issue	6-1	02/01/2008
6.3	Procedures For Document Control Policy	6-2	02/01/2008
6.4	Obsolete Documents	6-2	02/01/2008
7.0	REVIEW OF WORK REQUEST	7-1	02/01/2008
7.1	Overview	7-1	02/01/2008
7.2	Review Sequence And Key Personnel	7-2	02/01/2008
7.3	Documentation	7-3	02/01/2008
8.0	SUBCONTRACTING OF TESTS (NELAC 5.4.5)	8-1	02/01/2008
8.1	Overview	8-1	02/01/2008
8.2	Qualifying And Monitoring Subcontractors	8-1	02/01/2008
8.3	Oversight And Reporting	8-5	02/01/2008
8.4	Contingency Planning	8-6	02/01/2008
9.0	PURCHASING SERVICES AND SUPPLIES (NELAC 5.4.6)	9-1	02/01/2008
9.1	Overview	9-1	02/01/2008
9.2	Glassware	9-1	02/01/2008
9.3	Reagents, Standards & Supplies	9-1	02/01/2008

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 2 of 9

Section	Title	Page No.	Effective
No.			Date
9.4	Purchase Of Equipment/Instruments/Software	9-3	02/01/2008
9.5	Services	9-4	02/01/2008
9.6	Suppliers	9-4	02/01/2008
10.0	SERVICE TO THE CLIENT (NELAC 5.4.7)	10-1	02/01/2008
10.1	Overview	10-1	02/01/2008
10.2	Special Services	10-1	02/01/2008
10.3	Client Communication	10-1	02/01/2008
10.4	Reporting	10-1	02/01/2008
10.5	Client Surveys	10-2	02/01/2008
11.0	COMPLAINTS (NELAC 5.4.8)	11-1	02/01/2008
11.1	Overview	11-1	02/01/2008
11.2	External Complaints	11-1	02/01/2008
11.3	Internal Complaints	11-2	02/01/2008
11.4	Management Review	11-2	02/01/2008
	CONTROL OF NON-CONFORMING WORK		
12.0	(NFLAC 5 4 9)	12-1	02/01/2008
12.1	Overview	12-1	02/01/2008
12.2	Responsibilities And Authorities	12-1	02/01/2008
12.3	Evaluation Of Significance And Actions Taken	12-2	02/01/2008
12.0	Prevention Of Nonconforming Work	12-2	02/01/2008
16.1	Method Suspension/Restriction (Stop Work		
12.5	Procedures)	12-3	02/01/2008
13.0	COBRECTIVE ACTION (NELAC 5.4.10)	13-1	02/01/2008
13.1	Overview	13-1	02/01/2008
13.2	Definitions	13-1	02/01/2008
13.3	General	13-1	02/01/2008
13.4	Closed Loop Corrective Action Process	13-2	02/01/2008
13.5	Technical Corrective Actions	13-3	02/01/2008
13.6	Basic Corrections	13-4	02/01/2008
14.0	PREVENTIVE ACTION (NEL AC 5.4.11)	14-1	02/01/2008
1/ 1	Overview	14-1	02/01/2008
14.1	Management Of Change - Reserved	14-2	Reserved
15.0	CONTROL OF RECORDS ( $NELAC 5.4.12$ )	15_1	02/01/2008
15.0		15-1	02/01/2008
15.1	Technical And Analytical Records	15-4	02/01/2008
15.2	Laboratory Support Activities	15-4	02/01/2008
15.5	Administrative Records	15-6	02/01/2008
15.4	Percende Management, Storage And Disposal	15-6	02/01/2008
10.0	ALIDITS ( $NELA \subset 5.4.13$ )	16_1	02/01/2008
10.0		16_1	02/01/2008
10.1	Technical And Analytical Pacarda	16_1	02/01/2008
10.2	External Audita	10-1	02/01/2008
10.3	External Audits	10-3	02/01/2008
10.4		10-0	02/01/2000
17.0	NIANAGENIENI KEVIEVVS (NELAC 3.4.14)		02/01/2000
17.1		1/-1	02/01/2008
17.2	Annual Management Review	1/-2	02/01/2008

17.3         Petential Integrity Related Managerial Reviews         17.3         02/01/2008           18.0         PERSONNEL ( <i>NELAC 5.5.2</i> )         18-1         02/01/2008           18.1         Overview         18-1         02/01/2008           18.2         Education And Experience Requirements For Technical Personnel         18-1         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-2         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS ( <i>NELAC 5.5.3</i> )         19-1         02/01/2008           19.1         Overview         19-2         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION ( <i>NELAC 5.5.4</i> )         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.3         Eaboratory Methods Manual         20-2         02/01/2008           20.4         Selection Of Methods         20-5         02/01/2008	Section No.	Title	Page No.	Effective Date
18.0         PERSONNEL ( <i>NELAC 5.5.2</i> )         18-1         02/01/2008           18.1         Overview         18-1         02/01/2008           18.2         Education And Experience Requirements For Technical Personnel         18-1         02/01/2008           18.3         Training         18-2         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS ( <i>NELAC 5.5.3</i> )         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION ( <i>NELAC 5.5.4</i> )         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.4         Selection Of Methods         20-5         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.1         Retention Time Windows         20-10         02/0	17.3	Potential Integrity Related Managerial Reviews	17-3	02/01/2008
18.1         Overview         18-1         02/01/2008           18.2         Education And Experience Requirements For Technical Personnel         18-1         02/01/2008           18.3         Training         18-2         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS ( <i>NELAC 5.5.3</i> )         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-2         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.7         (Lob)         Overification Of Methods         20-10         02/01/2008 </td <td>18.0</td> <td>PERSONNEL (NELAC 5.5.2)</td> <td>18-1</td> <td>02/01/2008</td>	18.0	PERSONNEL (NELAC 5.5.2)	18-1	02/01/2008
18.2         Education And Experience Requirements For Technical Personnel         18-1         02/01/2008           18.3         Training         18-2         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS (NELAC 5.5.3)         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008         20.2           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-10         02	18.1	Overview	18-1	02/01/2008
18.3         Training         18-2         02/01/2008           18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS ( <i>NELAC 5.5.3</i> )         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION ( <i>NELAC 5.5.4</i> )         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-5         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.7         Method Detection Limits (Idl)         20-9         02/01/2008         20.1           20.7         Method Detection And Reporting Limits         20-10         02/01/2008	18.2	Education And Experience Requirements For Technical Personnel	18-1	02/01/2008
18.4         Data Integrity And Ethics Training Program         18-3         02/01/2008           19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS (NELAC 5.5.3)         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.4         Selection Of Methods         20-5         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Method         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008         20.1           20.7         Method Detection Limits (IdI)         20-9         02/01/2008 <td>18.3</td> <td>Training</td> <td>18-2</td> <td>02/01/2008</td>	18.3	Training	18-2	02/01/2008
19.0         ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS ( <i>NELAC 5.5.3</i> )         19-1         02/01/2008           19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION ( <i>NELAC 5.5.4</i> )         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-10         02/01/2008         20.1         02/01/2008           20.7         Method Detection Limits (Idl)         20-9         02/01/2008         20/01/2008         20.10         02/01/2008           20.1         Retention Time Windows         20-10         02/01/2008         20.11         02/01/20	18.4	Data Integrity And Ethics Training Program	18-3	02/01/2008
19.1         Overview         19-1         02/01/2008           19.2         Environment         19-1         02/01/2008           19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-2         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Szlection Of Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-9         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Detection And Reporting Limits         20-10         02/01/2008           20.11         Evalu	19.0	ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS (NELAC 5.5.3)	19-1	02/01/2008
19.2       Environment       19-1       02/01/2008         19.3       Work Areas       19-2       02/01/2008         19.4       Floor Plan       19-2       02/01/2008         19.5       Building Security       19-3       02/01/2008         20.0       TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)       20-1       02/01/2008         20.1       Overview       20-1       02/01/2008         20.2       STANDARD OPERATING PROCEDURES (Sops)       20-1       02/01/2008         20.3       Laboratory Methods Manual       20-1       02/01/2008         20.4       Selection Of Methods       20-2       02/01/2008         20.5       Laboratory Developed Methods And Non-Standard Methods       20-5       02/01/2008         20.6       Validation Of Methods       20-5       02/01/2008         20.7       Method Detection Limits (Mdl)/ Limits Of Detection (Lod)       20-8       02/01/2008         20.9       Verification Of Detection And Reporting Limits       20-10       02/01/2008         20.10       Retention Time Windows       20-10       02/01/2008         20.11       Evaluation Of Uncertainty Of Measurement       20-11       02/01/2008         20.12       Estimation Of Uncertainty Of Measurement	19.1	Overview	19-1	02/01/2008
19.3         Work Areas         19-2         02/01/2008           19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-9         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Detection And Reporting Limits         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12	19.2	Environment	19-1	02/01/2008
19.4         Floor Plan         19-2         02/01/2008           19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (MdI)/ Limits Of Detection (Lod)         20-9         02/01/2008           20.8         Instrument Detection Limits (IdI)         20-9         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Uncertainty Of Measurement         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           21.1         Oz/01/2008         21-1	19.3	Work Areas	19-2	02/01/2008
19.5         Building Security         19-3         02/01/2008           20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.11         Evaluation Of Selectivity         20-10         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance <td< td=""><td>19.4</td><td>Floor Plan</td><td>19-2</td><td>02/01/2008</td></td<>	19.4	Floor Plan	19-2	02/01/2008
20.0         TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)         20-1         02/01/2008           20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (MdI)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (IdI)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           20.12         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventi	19.5	Building Security	19-3	02/01/2008
20.1         Overview         20-1         02/01/2008           20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection And Reporting Limits         20-10         02/01/2008           20.9         Verification Of Detectivity         20-11         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-2<	20.0	TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)	20-1	02/01/2008
20.2         STANDARD OPERATING PROCEDURES (Sops)         20-1         02/01/2008           20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.1         Overview         21-2         02/01/2008           21.2         Preventive Maintenance         21-1	20.1	Overview	20-1	02/01/2008
20.3         Laboratory Methods Manual         20-1         02/01/2008           20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-2         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         <	20.2	STANDARD OPERATING PROCEDURES (Sops)	20-1	02/01/2008
20.4         Selection Of Methods         20-2         02/01/2008           20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           21.1         Overview         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis	20.3	Laboratory Methods Manual	20-1	02/01/2008
20.5         Laboratory Developed Methods And Non-Standard Methods         20-5         02/01/2008           20.6         Validation Of Methods         20-5         02/01/2008           20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-12         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis         21-14         02/01/2008           21.6         Policy On Gc/Ms Tu	20.4	Selection Of Methods	20-2	02/01/2008
20.6       Validation Of Methods       20-5       02/01/2008         20.7       Method Detection Limits (MdI)/ Limits Of Detection (Lod)       20-8       02/01/2008         20.8       Instrument Detection Limits (IdI)       20-9       02/01/2008         20.9       Verification Of Detection And Reporting Limits       20-10       02/01/2008         20.10       Retention Time Windows       20-10       02/01/2008         20.11       Evaluation Of Selectivity       20-11       02/01/2008         20.12       Estimation Of Uncertainty Of Measurement       20-11       02/01/2008         20.13       Control Of Data       20-12       02/01/2008         21.0       EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)       21-1       02/01/2008         21.1       Overview       21-1       02/01/2008         21.2       Preventive Maintenance       21-1       02/01/2008         21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1	20.5	Laboratory Developed Methods And Non-Standard Methods	20-5	02/01/2008
20.7         Method Detection Limits (Mdl)/ Limits Of Detection (Lod)         20-8         02/01/2008           20.8         Instrument Detection Limits (Idl)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Gc/Ms Analysis         21-1         02/01/2008           21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02	20.6	Validation Of Methods	20-5	02/01/2008
20.8         Instrument Detection Limits (IdI)         20-9         02/01/2008           20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) –         21-13         02/01/2008           21.5         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008 </td <td>20.7</td> <td>Method Detection Limits (Mdl)/ Limits Of Detection (Lod)</td> <td>20-8</td> <td>02/01/2008</td>	20.7	Method Detection Limits (Mdl)/ Limits Of Detection (Lod)	20-8	02/01/2008
20.9         Verification Of Detection And Reporting Limits         20-10         02/01/2008           20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis         21-14         02/01/2008           21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008 </td <td>20.8</td> <td>Instrument Detection Limits (IdI)</td> <td>20-9</td> <td>02/01/2008</td>	20.8	Instrument Detection Limits (IdI)	20-9	02/01/2008
20.10         Retention Time Windows         20-10         02/01/2008           20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis         21-14         02/01/2008           21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2 <td< td=""><td>20.9</td><td>Verification Of Detection And Reporting Limits</td><td>20-10</td><td>02/01/2008</td></td<>	20.9	Verification Of Detection And Reporting Limits	20-10	02/01/2008
20.11         Evaluation Of Selectivity         20-11         02/01/2008           20.12         Estimation Of Uncertainty Of Measurement         20-11         02/01/2008           20.13         Control Of Data         20-12         02/01/2008           21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis         21-13         02/01/2008           21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008	20.10	Retention Time Windows	20-10	02/01/2008
20.12       Estimation Of Uncertainty Of Measurement       20-11       02/01/2008         20.13       Control Of Data       20-12       02/01/2008         21.0       EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)       21-1       02/01/2008         21.1       Overview       21-1       02/01/2008         21.2       Preventive Maintenance       21-1       02/01/2008         21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis       21-13       02/01/2008         21.6       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	20.11	Evaluation Of Selectivity	20-11	02/01/2008
20.13       Control Of Data       20-12       02/01/2008         21.0       EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)       21-1       02/01/2008         21.1       Overview       21-1       02/01/2008         21.2       Preventive Maintenance       21-1       02/01/2008         21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) –       21-13       02/01/2008         21.5       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	20.12	Estimation Of Uncertainty Of Measurement	20-11	02/01/2008
21.0         EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)         21-1         02/01/2008           21.1         Overview         21-1         02/01/2008           21.2         Preventive Maintenance         21-1         02/01/2008           21.3         Support Equipment         21-2         02/01/2008           21.4         Instrument Calibrations         21-5         02/01/2008           21.5         Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis         21-13         02/01/2008           21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008	20.13	Control Of Data	20-12	02/01/2008
21.1       Overview       21-1       02/01/2008         21.2       Preventive Maintenance       21-1       02/01/2008         21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis       21-13       02/01/2008         21.6       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	21.0	FQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)	21-1	02/01/2008
21.2       Preventive Maintenance       21-1       02/01/2008         21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis       21-13       02/01/2008         21.6       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	21.1	Overview	21-1	02/01/2008
21.3       Support Equipment       21-2       02/01/2008         21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis       21-13       02/01/2008         21.6       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	21.2	Preventive Maintenance	21-1	02/01/2008
21.4       Instrument Calibrations       21-5       02/01/2008         21.5       Policy On Tentatively Identified Compounds (Tics) –       21-13       02/01/2008         21.6       Policy On Gc/Ms Tuning       21-14       02/01/2008         22.0       MEASUREMENT TRACEABILITY (NELAC 5.5.6)       22-1       02/01/2008         22.1       Overview       22-1       02/01/2008         22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008	21.3	Support Equipment	21-2	02/01/2008
21.1Instrument of an analysis21.11.121.5Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis21-1302/01/200821.6Policy On Gc/Ms Tuning21-1402/01/200822.0MEASUREMENT TRACEABILITY (NELAC 5.5.6)22-102/01/200822.1Overview22-102/01/200822.2Nist-Traceable Weights And Thermometers22-202/01/200822.3Deference Standards (Materials22-202/01/2008	21.4	Instrument Calibrations	21-5	02/01/2008
21.6         Policy On Gc/Ms Tuning         21-14         02/01/2008           22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008           22.3         Deference Standards / Materials         22-2         02/01/2008	21.5	Policy On Tentatively Identified Compounds (Tics) – Gc/Ms Analysis	21-13	02/01/2008
22.0         MEASUREMENT TRACEABILITY (NELAC 5.5.6)         22-1         02/01/2008           22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008           22.3         Deference Standards / Materials         22-2         02/01/2008	21.6	Policy On Gc/Ms Tuning	21-14	02/01/2008
22.1         Overview         22-1         02/01/2008           22.2         Nist-Traceable Weights And Thermometers         22-2         02/01/2008           22.3         Deference Standards / Materials         22-2         02/01/2008	22.0	MEASUREMENT TRACEABILITY (NELAC 5.5.6)	22-1	02/01/2008
22.2       Nist-Traceable Weights And Thermometers       22-2       02/01/2008         22.2       Deference Standards / Materials       22.2       02/01/2008	22.1	Overview	22-1	02/01/2008
22.2 Deference Standarda / Materiala 22.2 02/01/2009	22.2	Nist-Traceable Weights And Thermometers	22-2	02/01/2008
ZZ.3 Reference Standards / Materials ZZ-Z UZ/01/2000	22.3	Reference Standards / Materials	22-2	02/01/2008
22.4 Documentation And Labeling Of Standards, Reagents, And Reference Materials 22-3 02/01/2008	22.4	Documentation And Labeling Of Standards, Reagents, And Reference Materials	22-3	02/01/2008
23.0 SAMPLING (NELAC 5.5.7) 23-1 02/01/2008	23.0	SAMPLING (NELAC 5.5.7)	23-1	02/01/2008

#### Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 4 of 9

Section No.	Title	Page No.	Effective Date
23.1	Overview	23-1	02/01/2008
23.2	Sampling Containers	23-1	02/01/2008
23.3	Field Quality Control (Qc)	23-2	02/01/2008
23.4	Definition Of Holding Time	23-3	02/01/2008
23.5	Sampling Containers, Preservation Requirements, Holding Times	23-3	02/01/2008
23.6	Sample Aliquots / Subsampling	23-3	02/01/2008
24.0	HANDLING OF SAMPLES (NELAC 5.5.8)	24-1	02/01/2008
24.1	Chain Of Custody (Coc)	24-1	02/01/2008
24.2	Sample Receipt	24-2	02/01/2008
24.3	Sample Acceptance Policy	24-4	02/01/2008
24.4	Sample Storage	24-5	02/01/2008
24.5	Hazardous Samples And Foreign Soils	24-5	02/01/2008
24.6	Sample Shipping	24-6	02/01/2008
24.7	Sample Disposal	24-6	02/01/2008
25.0	ASSURING THE QUALITY OF TEST RESULTS (NELAC 5.5.9)	25-1	02/01/2008
25.1	Overview	25-1	02/01/2008
25.2	Controls	25-1	02/01/2008
25.3	Negative Controls	25-1	02/01/2008
25.4	Positive Controls	25-2	02/01/2008
25.5	Sample Matrix Controls	25-4	02/01/2008
25.6	Acceptance Criteria (Control Limits)	25-6	02/01/2008
25.7	METHOD DETECTION LIMITS (Mdls)	25-8	02/01/2008
25.8	Additional Procedures To Assure Quality Control	25-9	02/01/2008
26.0	REPORTING RESULTS (NELAC 5.5.10)	26-1	02/01/2008
26.1	Overview	26-1	02/01/2008
26.2	Test Reports	26-1	02/01/2008
26.3	Reporting Level Or Report Type	26-3	02/01/2008
26.4	Supplemental Information For Test	26-4	02/01/2008
26.5	Environmental Testing Obtained From Subcontractors	26-5	02/01/2008
26.6	Client Confidentiality	26-5	02/01/2008
26.7	Format Of Reports	26-5	02/01/2008
26.8	Amendments To Test Reports	26-5	02/01/2008
26.9	Policies On Client Requests For Amendments	26-6	02/01/2008

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 5 of 9

#### LIST OF TABLES

Table No.	Title	Page	Effective Date
9-1	Storage of Reagents and Chemicals	9-6	02/01/2008
13-1	Example - General Corrective Action Procedures	13-6	02/01/2008
15-1	Record Index	15-1	02/01/2008
15-2	Special Record Retention Requirements	15-3	02/01/2008
16-1	Audit Types and Frequency	16-1	02/01/2008
21-1	Example - Laboratory Equipment & Instrumentation	21-16	02/01/2008
21-2	Example – Schedule of Routine Maintenance	21-19	02/01/2008
21-3	Example – Periodic Calibration	21-21	02/01/2008
22-1	Example – Standard Source & Preparation	22-1	02/01/2008
23-1	Holding Times, Preservation and Container Requirements - Drinking Water (SDWA)	23-6	02/01/2008
23-2	<u>Holding Times, Preservation and Container</u> <u>Requirements - NPDES – Bacteria, Protozoa, Toxicity</u> <u>Tests</u>	23-9	02/01/2008
23-3	Holding Times, Preservation and Container Requirements - NPDES – Inorganic	23-10	02/01/2008
23-4	Holding Times, Preservation and Container Requirements - NPDES – Organic	23-13	02/01/2008
23-5	Holding Times, Preservation and Container Requirements - NPDES - Radiological	23-15	02/01/2008
23-6	Holding Times, Preservation and Container Requirements - RCRA – Aqueous	23-16	02/01/2008
23-7	Holding Times, Preservation and Container Requirements - RCRA – Non-Aqueous	23-18	02/01/2008
23-8	Holding Times, Preservation and Container Requirements - Air Samples	23-20	02/01/2008
Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 6 of 9

# LIST OF FIGURES

Figure No.	Title	Page	Effective Date
3-1	Format for a QA/QC Policy Memorandum	3-4	02/01/2008
4-1	Corporate Organizational Chart	4-15	02/01/2008
8-1	Example - Subcontracting Laboratory Approval Form (Initial / Renewal)	8-7	02/01/2008
8-2	Example - Subcontracted Sample Form	8-8	02/01/2008
9-1	Example - JD Edwards Vendor Add Request Form	9-7	02/01/2008
13-1	Example - Corrective Action Report	13-5	02/01/2008
16-1	Example - Internal Audit Workbook	16-7	02/01/2008
16-2	Example – Internal Audit System Checklist	16-8	02/01/2008
17-1	Example - QA Monthly Report to Management	17-4	02/01/2008
17-2	Example – Laboratory Metrics Categories	17-6	02/01/2008
20-1	Example – Demonstration of Capability Documentation	20-20	02/01/2008
20-2	Example – New Method / Additional Analyte Checklist	20-21	02/01/2008
24-1	Example – Chain of Custody	24-7	02/01/2008
24-2	Example – Internal Chain of Custody Form	24-8	02/01/2008
24-3	Example – Sample Disposal Record	24-9	02/01/2008
24-4	Example – Sample Acceptance Policy	24-10	02/01/2008
24-5	Example – Cooler Receipt Form	24-13	02/01/2008

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 7 of 9

# LIST OF APPENDICES

Appendix No.	Title	Page	Effective Date
1	TestAmerica Ethics Policy No. CA-L-P-001	Appendix 1-1	02/01/2008
2	Laboratory Organization Chart	Appendix 2-1	02/01/2008
3	Laboratory Floor Plan	Appendix 3-1	02/01/2008
4	Summary of Calibration, QC Procedures and Corrective Action	Appendix 4-1	Reserved
5	Glossary / Acronyms	Appendix 5-1	02/01/2008
6	Laboratory Certifications, Accreditations, Validations	Appendix 6-1	02/01/2008
7	Data Qualifiers	Appendix 7-1	02/01/2008

# SOPS AND POLICIES REFERRED TO IN THE QA MANUAL

SOP/Policy Reference	Title
CA-Q-S-001	Solvent and Acid Lot Testing and Approval
CA-Q-S-002	Acceptable Manual Integration Practices
CA-Q-S-003	Management of Change Procedure
CA-Q-S-004	Method Compliance & Data Authenticity Audits
CA-Q-S-005	Calibration Curves (General)
CW-Q-S-001	Corporate Document Control and Archiving
CW-Q-S-002	Writing a Standard Operating Procedure (SOPs)
CA-L-S-001	Internal Investigation of Potential Data Discrepancies and Determination for Data Recall
CA-L-S-002	Subcontracting Procedures
CA-L-P-001	Ethics Policy
CA-L-P-002	Contract Compliance Policy
CW-L-P-001	Record Retention
CW-F-P-002	Authorization Matrix
CA-C-S-001	Work Sharing Process
CA-T-P-001	Qualified Products List
CW-F-S-004	Controlled Purchases Policy
BF-GP-001	Calibration of Autopipettes and Repipetters
BF-GP-002	Support Equipment: Maintenance, Record Keeping and Corrective Actions
BF-GP-005	Sample Homogenization and Subsampling
BF-GP-012	Technical Data Review
BF-GP-013	Manual Integration
BF-GP-015	Record Storage and Retention

#### Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Table of Contents Page 9 of 9

- BF-GP-018 Strict Internal Chain or Custody
- BF-GP-019 Standard Treaceability and Preparation
- BF-GP-020 Thermometer Calibration
- BF-OP-013 Solvent Purity
- BF-PM-001 Project Information Requirements
- BF-PM-003 Bottle Order Set-up
- BF-PM-005 Correctness of Analysis
- BF-QA-001 Determination of Method Detection Limits
- BF-QA-002 Quality Control Limits
- BF-QA-003 Procedure for Writing, Reviewing and Revising Controlled Documents
- BF-QA-004 Laboratory Personnel Training
- BF-QA-005 Preventative and Corrective Action
- BF-QA-006 Data Quality Review
- BF-SR-001 Cooler Shipping Bottle Kits and Samples
- BF-SR-002 Receipt of Analytical Samples
- BF-WM-001 Waste Management

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 3-1 of 3-4

#### **SECTION 3**

# INTRODUCTION (NELAC 5.1 - 5.3)

#### 3.1 INTRODUCTION AND COMPLIANCE REFERENCES

TestAmerica Buffalo's Quality Assurance Manual (QAM) is a document prepared to define the overall policies, organization objectives and functional responsibilities for achieving TestAmerica's data quality goals. Each TestAmerica laboratory maintains a local perspective in its scope of services and client relations and maintains a national perspective in terms of quality.

The QAM has been prepared to assure compliance with the 2003 National Environmental Laboratory Accreditation Conference (NELAC) standards and ISO/IEC Guide 17025 (1999). In addition, the policies and procedures outlined in this manual are compliant with the various accreditation and certification programs listed in Appendix 6. The relevant NELAC section is included in the heading of each QAM section.

The QAM has been prepared to be consistent with the requirements of the following documents:

- EPA 600/4-88/039, Methods for the Determination of Organic Compounds in Drinking Water, EPA, Revised July 1991.
- EPA 600/R-95/131, Methods for the Determination of Organic Compounds in Drinking Water, Supplement III, EPA, August 1995.
- EPA 600/4-79-019, Handbook for Analytical Quality Control in Water and Wastewater Laboratories, EPA, March 1979.
- EPA SW-846, Test Methods for the Evaluation of Solid Waste, 3rd Edition, September 1986; Update I, July 1992; Update II, September 1994; and Update III, December 1996.
- Federal Register, 40 CFR Parts 136, 141, 172, 173, 178, 179 and 261.
- USEPA Contract Laboratory Program. Statement of Work for Inorganics Analysis. Multi-Media, Multi-Concentration. Document ILM04.0/4.1/4.2
- USEPA Contract Laboratory Program. Statement of Work for Inorganics Analysis. Multi-Media, Multi-Concentration. Document ILM05.1/5.2/ 5.3.
- USEPA Contract Laboratory Program. Statement of Work for Organics Analysis. Multi-Media, Multi-Concentration. Document Number OLM03.1, August 1994.
- USEPA Contract Laboratory Program. Statement of Work for Organics Analysis. Multi-Media, Multi-Concentration. Document Number OLM04.0, August 1994 and updates
- New York State Analytical Services Protocol, July 2005
- APHA, Standard Methods for the Examination of Water and Wastewater, 18th Edition, 19th, 20th and 21st Edition.
- U.S. Department of Energy Order 414.1B, Quality Assurance, Approved April 29, 2004.
- U.S. Department of Energy, Quality Systems for Analytical Services, Revision 2.1, November 2005.
- Toxic Substances Control Act (TSCA).

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 3-2 of 3-4

# 3.2 TERMS AND DEFINITIONS

A Quality Assurance Program is a company-wide system designed to ensure that data produced by TestAmerica Buffalo conforms to the standards set by state and/or federal regulations. The program functions at the management level through company goals and management policies, and at the analytical level through Standard Operating Procedures (SOPs) and quality control. The TestAmerica program is designed to minimize systematic error, encourage constructive, documented problem solving, and provide a framework for continuous improvement within the organization.

Refer to Appendix 5 for the Glossary/Acronyms.

# 3.3 SCOPE / FIELDS OF TESTING

TestAmerica Buffalo analyzes thousands of environmental and industrial samples every month. Sample matrices vary among air, drinking water, effluent water, groundwater, hazardous waste, sludge and soils. The Quality Assurance Program contains specific procedures and methods to test samples of differing matrices for chemical and physical parameters. The Program also contains guidelines on maintaining documentation of analytical process, reviewing results, servicing clients and tracking samples through the laboratory. The technical and service requirements of all requests to provide analyses are thoroughly evaluated before commitments are made to accept the work. Measurements are made using published reference methods or methods developed and validated by the laboratory.

The methods covered by this manual include the most frequently requested water, air, industrial waste, and soil methodologies needed to provide analytical services in the United States, its territories and for foreign countries. The specific list of test methods used by the laboratory can be found in Appendix 4 of this QAM. The approach of this manual is to define the minimum level of quality assurance and quality control necessary to meet requirements. All methods performed by TestAmerica Buffalo shall meet these criteria as appropriate. In some instances, quality assurance project plans (QAPPs), project specific data quality objectives (DQOs) or local regulations may require criteria other than those contained in this manual. In these cases, the laboratory will abide by the requested criteria following review and acceptance of the requirements by the Laboratory Director, Technical Director, Quality Assurance (QA) Manager, Customer Service Manager or Operations Manager. In some cases, QAPPs and DQOs may specify less stringent requirements. The Laboratory Director and the QA Manager must determine if it is in the lab's best interest to follow the less stringent requirements.

# 3.4 MANAGEMENT OF THE MANUAL

# 3.4.1 <u>Review Process</u>

The manual is reviewed annually by the QA Manager and laboratory personnel to assure that it reflects current practices and meets the requirements of TestAmerica Buffalo's clients and regulators. Occasionally, the manual may need changes in order to meet new or changing regulations and operations. The QA Manager will review the changes in the normal course of business and incorporate changes into revised sections of the document. The updates will be reviewed by the QA Manager, Laboratory Director, Operations Manager, Technical Director, Department Managers, relevant operational staff and Corporate Quality Assurance (if a change is made to the Corporate template) and then formally incorporated into the document in periodic

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 3-3 of 3-4

updates. The QAM is based on a Corporate QAM Template that is prepared and approved by the Chief Operating Officers (COOs) and Corporate Quality Assurance. This template is reviewed annually by the COOs, Corporate Quality, and each laboratory. Necessary changes are coordinated by the Vice President of Quality and Environmental Health & Safety (EHS) and distributed to each laboratory for inclusion in the laboratory specific QA Manuals.

Policies in the QAM that require immediate attention may be addressed through the use of Corporate QA/QC Policy Memoranda. QA/QC Policy Memoranda are published from time to time to facilitate immediate changes to QA/QC Policy. QA/QC Policy Memoranda supersede the QAM and all other SOPs (refer to Section 5.3). All policy memoranda are dated, archived and distributed by their placement into the front of the QAM between the signature page and Section 2. At a minimum, each policy memorandum is approved by the same authorized signatories as shown on the cover page of the QA Manual. In addition, Corporate QA/QC Policy Memoranda are signed by the COOs and VP of Quality and EHS. The QA/QC Policy Memoranda are incorporated into the QAM during the periodic updates. Policy memorandum may also include an expiration date if appropriate. An example format can be found in Figure 3-1. A similar procedure is followed for local laboratory changes.

Laboratory-specific QAM changes are approved and documented through the Interim Change Procedure defined for controlled documents (SOP BF-QA-003, "Procedure for Writing, Reviewing and Revising Controlled Documents").

# 3.4.2 <u>Control</u>

This manual is considered confidential within TestAmerica and may not be altered in any manner by other than a duly appointed representative from TestAmerica. If the document has been provided to external users or regulators, it is for the exclusive purpose of reviewing **TestAmerica Buffalo**'s quality systems and shall not be used in any other way without the written permission of an appointed representative of TestAmerica. The procedure for control of distribution is incorporated by reference to SOP BF-QA-003, "Procedure for Writing, Reviewing and Revising Controlled Documents".

The order of precedence in the event of a conflict between policies is outlined in Section 5.3 of this Quality Assurance Manual.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 3-4 of 3-4

Figure 3-1.

# Format for a QA/QC Policy Memorandum

# Corporate QA/QC Policy Memorandum #_____

Effective Date: _____ Expiration Date: When Appropriate QAM Section is Revised

Corporate: (Only needed for C	Corporate N	lemoran	dum – Delete if Laboratory)	
COO - West	Date		Vice-President, QA and EHS	Date
COO - East	Date			
Local:				
Laboratory Director Approval		Date	Quality Assurance Approval	Date
Operations Manager Approval		Date	Technical Director Approval	Date
Customer Service Manager Appr	roval	Date		

# 1. Purpose

#### 2. Procedure

# 3. Attachments

# 4. <u>References/Cross References</u>

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-1 of 4-15

#### **SECTION 4**

#### ORGANIZATION AND MANAGEMENT (NELAC 5.4.1)

#### 4.1 <u>OVERVIEW</u>

**TestAmerica Buffalo** is part of a national network of laboratories known as TestAmerica. This Quality Assurance Manual (QAM) is applicable to the **TestAmerica Buffalo** laboratory only.

#### TestAmerica Buffalo 10 Hazelwood Drive Amherst, New York 14228 EPA ID #: NY00044

The Corporate organization chart can be found in Figure 4-1 and the laboratory's organization chart can be found in Appendix 2. The locations of other TestAmerica labs are as follows:

TestAmerica Anchorage **TestAmerica Austin TestAmerica Burlington** TestAmerica Cedar Falls **TestAmerica** Chicago TestAmerica Connecticut TestAmerica Corpus Christi TestAmerica Dayton **TestAmerica Denver** TestAmerica Edison **TestAmerica Honolulu TestAmerica Houston TestAmerica** Irvine **TestAmerica King of Prussia** TestAmerica Knoxville **TestAmerica Los Angeles TestAmerica Mobile** TestAmerica Morgan Hill **TestAmerica** Nashville **TestAmerica North Canton** TestAmerica Ontario TestAmerica Orlando TestAmerica Pensacola **TestAmerica Phoenix** TestAmerica Pittsburgh **TestAmerica Portland** TestAmerica Richland TestAmerica San Francisco TestAmerica Savannah **TestAmerica Seattle** 

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-2 of 4-15

TestAmerica Spokane TestAmerica St. Louis TestAmerica Tacoma TestAmerica Tallahassee TestAmerica Tampa TestAmerica Valparaiso TestAmerica Watertown TestAmerica West Sacramento TestAmerica West field

# 4.2 ROLES AND RESPONSIBILITIES

In order for the Quality Assurance Program to function properly, all members of the staff must clearly understand and meet their individual responsibilities as they relate to the quality program. The following descriptions define each role in its relationship to the Quality Assurance Program. More extensive job descriptions are maintained by laboratory management and TestAmerica Human Resources.

# 4.2.1 Quality Assurance Program

The responsibility for quality lies with every employee of **TestAmerica Buffalo**. All employees have access to the QAM and are responsible for knowing the content of this manual and upholding the standards therein. Each person carries out his/her daily tasks in a manner consistent with the goals and in accordance with the procedures in this manual and the laboratory's SOPs.

# 4.2.2 Chairman/Chief Executive Officer (CEO)

The Chairman/CEO is the Chairman of the Board of Directors and is ultimately responsible for the quality and performance of all TestAmerica facilities. Together with the President/CEO of the Analytical Division, the Chairman/CEO establishes the overall quality standard and data integrity program for the company, providing the necessary leadership and resources to assure that the standard and integrity program are met.

# 4.2.3 <u>President/Chief Executive Officer (CEO)</u>

The President/CEO is a member of the Board of Directors and is ultimately responsible for the quality and performance of all TestAmerica facilities. Together with the Chairman/CEO, the President/CEO establishes the overall quality standard and data integrity program for the Analytical Division, providing the necessary leadership and resources to assure that the standard and integrity program are met.

# 4.2.4 Chief Operating Officer (COO) – East and West

The COOs serve as the ranking executives for all respective analytical laboratory operational functions and report to the President/CEO of the Analytical Division. They are responsible for the daily management of all analytical laboratories, long-term planning and development of technical policies and management plans. They ensure the attainment of corporate objectives through the selection, development, motivation, and evaluation of top management personnel. The COOs approve all operating budgets and capital expenditures. The COOs sign-off on the final QAM template that contains company policies for implementing the Quality Program.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-3 of 4-15

# 4.2.5 <u>General Manager (GM)</u>

Each GM reports directly to a COO. Each GM has full responsibility for the overall administrative and operational management of their respective laboratories. The GM's responsibilities include allocation of personnel and resources, long-term planning, setting goals, and achieving the financial, business, and quality objectives of TestAmerica. The GM ensures timely compliance with corporate management directives, policies, and management systems reviews. The GM is also responsible for restricting any laboratory from performing analyses that cannot be consistently and successfully performed to meet the standards set forth in this manual.

# 4.2.6 Vice President of Quality and Environmental Health and Safety (VP-QA/EHS)

The Vice President of QA/EHS reports directly to the Chairman/CEO. With the aid of the Analytical Division and Non-Analytical Division Senior Management Teams, Laboratory Director/ Managers, Quality Directors, EHS Directors, QA Managers and EHS Coordinators, the VP-QA/EHS has the responsibility for the establishment, general overview and Corporate maintenance of the Quality Assurance and Environmental, Health and Safety Program within TestAmerica. Additional responsibilities include:

- Review of QA/QC aspects of Corporate SOPs, national projects and expansions or changes in services.
- Coordination/preparation of the Corporate QAM Template that is used by each laboratory to prepare its own laboratory-specific QAM.
- Maintenance of Corporate Policies, Quality Memorandums and SOPs. Maintenance of data investigation records that are reported to Corporate Management.
- Work with various organizations outside of TestAmerica to further the development of quality standards and represent TestAmerica at various trade meetings.
- Preparation of a monthly report that includes quality metrics across the Analytical Division and a summary of any quality related initiatives and issues.
- With the assistance of the Corporate Senior Management Teams and the EHS Directors, development and implementation of the TestAmerica Environmental, Health and Safety Program.

#### 4.2.7 Quality Directors (Corporate)

The Quality Directors report to the VP-QA/EHS. Together with the VP-QA/EHS, the Quality Directors have the responsibility for the establishment, general overview and maintenance of the Analytical Division's Quality Assurance Program within TestAmerica. The Quality Directors are responsible for:

- Oversight of the QA/QC programs within each laboratory. This includes a final review of each laboratory-specific QAM and receipt of each laboratory's QA monthly report.
- Review of QA/QC aspects of national projects.
- Assistance with certification activities.

# 4.2.8 Ethics and Compliance Officers (ECOs)

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-4 of 4-15

TestAmerica has designated two senior members of the Corporate staff to fulfill the role of Ethics and Compliance Officer (ECO) – VP-QA/EHS and VP-Client and Technical Services. Each ECO acts as a back-up to the other ECO and both are involved when data investigations occur. Each ECO has a direct line of communication to the entire senior Corporate and lab management staff.

The ECOs ensure that the organization distributes the data integrity and ethical practices policies to all employees and ensures annual trainings and orientation of new hires to the ethics program and its policies. The ECO is responsible for establishing a mechanism to foster employee reporting of incidents of illegal, unethical, or improper practices in a safe and confidential environment.

The ECOs monitor and audit procedures to determine compliance with policies and to make recommendations for policy enhancements to the CEOs, COOs, Laboratory Director or other appropriate individuals within the laboratory. The ECO will assist the laboratory QA Manager in the coordination of internal auditing of ethical policy related activities and processes within the laboratory, in conjunction with the laboratories regular internal auditing function.

The ECOs will also participate in investigations of alleged violations of policies and work with the appropriate internal departments to investigate misconduct, remedy the situation, and prevent recurrence of any such activity.

## 4.2.9 Vice President of Client and Technical Services

The Vise President (VP) of Client and Technical Services is responsible for offerings to clients including risk management, technical assistance, legal compliance and contract administration. The VP of Client and Technical Services provides support and direction to the Managers of these areas, and supports the COOs in decisions regarding long term planning, resource allocation and capital expenditures.

#### 4.2.10 Director of Technical Services

The Director of Technical Services is responsible for establishing, implementing and communicating TestAmerica's Analytical Division's Technical Policies, SOPs, and Manuals. Other responsibilities include conducting technical assessments as required, acting as a technical resource in national contracts review, coordinating new technologies, establishing best practices, advising staff on technology advances, innovations, and applications.

# 4.2.11 Chief Information Officer (CIO)

The CIO is responsible for establishing, implementing and communicating TestAmerica's Information Technology (IT) Policies, SOPs and Manuals. Other responsibilities include coordinating new technologies, development of electronic communication tools such as TestAmerica's intranet and internet sites, ensuring data security and documentation of software, ensuring compliance with the NELAC standard, and assistance in establishing, updating, and maintaining Laboratory Information Management Systems (LIMS) at the various TestAmerica facilities.

# 4.2.12 Environmental Health and Safety Directors (EHSDs) (Corporate)

The EHSDs report directly to the VP-QA/EHS. The EHSDs are responsible for the development and implementation of the TestAmerica Environmental, Health and Safety program. Responsibilities include:

- Consolidation and tracking all safety and health-related information and reports for the company, and managing compliance activities for TestAmerica locations.
- Coordination/preparation of the corporate Environmental, Health and Safety Manual Template that is used by each laboratory to prepare its own laboratory-specific Safety Manual/ CHP.
- Preparation of information and training materials for laboratory EHS Coordinators.
- Assistance in the internal and external coordination of employee exposure and medical monitoring programs to insure compliance with applicable safety and health regulations.
- Serving as Department of Transportation (D.O.T.) focal point and providing technical assistance to location management.
- Serving as Hazardous Waste Management main contact and providing technical assistance to location management.

# 4.2.13 Laboratory Director

**TestAmerica Buffalo**'s Laboratory Director is responsible for the overall quality, safety, financial, technical, human resource and service performance of the whole laboratory and reports to their respective GM. The Laboratory Director provides the resources necessary to implement and maintain an effective and comprehensive Quality Assurance and Data Integrity Program.

The Laboratory Director has the authority to affect those policies and procedures to ensure that only data of the highest level of excellence are produced. As such, the Laboratory Director is responsible for maintaining a working environment which encourages open, constructive problem solving and continuous improvement.

Specific responsibilities include, but are not limited to:

• Provides one or more department managers for the appropriate fields of testing. If the Department Manager is absent for a period of time exceeding 15 consecutive calendar days, the Laboratory Director must designate another full time staff member meeting the qualifications of the Department Manager to temporarily perform this function. If the absence

exceeds 65 consecutive calendar days, the primary NELAC accrediting authority must be notified in writing.

- Ensures that all analysts and supervisors have the appropriate education and training to properly carry out the duties assigned to them and ensures that this training has been documented.
- Ensures that personnel are free from any commercial, financial and other undue pressures which might adversely affect the quality of their work.
- Ensures TestAmerica's human resource policies are adhered to and maintained.
- Ensures that sufficient numbers of qualified personnel are employed to supervise and perform the work of the laboratory.
- Ensures that appropriate corrective actions are taken to address analyses identified as requiring such actions by internal and external performance or procedural audits. Procedures that do not meet the standards set forth in the QAM or laboratory SOPs may be temporarily suspended by the Laboratory Director.
- Reviews and approves all SOPs prior to their implementation and ensures all approved SOPs are implemented and adhered to.
- Pursues and maintains appropriate laboratory certification and contract approvals. Supports ISO 17025 requirements.
- Ensures client specific reporting and quality control requirements are met.
- Leads the management team, consisting of the QA Manager, the Technical Director, Customer Service Manager, and the Operations Manager as direct reports.

# 4.2.14 Quality Assurance (QA) Manager

The QA Manager has responsibility and authority to ensure the continuous implementation of the quality system based on ISO 17025.

The QA Manager reports directly to the Laboratory Director and has access to Corporate QA for advice and resources. This position is able to evaluate data objectively and perform assessments without outside (i.e., managerial) influence. Corporate QA may be used as a resource in dealing with regulatory requirements, certifications and other quality assurance related items. The QA Manager directs the activities of the QA department to accomplish specific responsibilities, which include, but are not limited to:

- Having functions independent from laboratory operations for which he/she has quality assurance oversight.
- Maintaining and updating the QAM.
- Monitoring and evaluating laboratory certifications; scheduling proficiency testing samples.
- Monitoring and communicating regulatory changes that may affect the laboratory to management.
- Training and advising the laboratory staff on quality assurance/quality control procedures that are pertinent to their daily activities.

- Having a general knowledge of the analytical test methods for which data audit/review is performed (and/or having the means of getting this information when needed).
- Arranging for or conducting internal audits on quality systems, data authenticity and the technical operation.
- The laboratory QA Manager will maintain records of all ethics-related training, including the type and proof of attendance.
- Maintain, improve, and evaluate the corrective action and preventive action systems.
- Notifying laboratory management of deficiencies in the quality system and ensuring corrective action is taken
- Monitoring standards of performance in quality control and quality assurance.
- Coordinating of document control of SOPs, MDLs, control limits, and miscellaneous forms and information.
- Review a subset of all final data reports for internal consistency.
- Review of external audit reports and data validation requests.
- Follow-up with audits to ensure client QAPP requirements are met.
- Establishment of reporting schedule and preparation of various quality reports for the Laboratory Director, clients and/or Corporate QA.
- Development of suggestions and recommendations to improve quality systems.
- Research of current state and federal requirements and guidelines.
- Leads the QA team to enable communication and to distribute duties and responsibilities.

The QAM shall have the final authority to accept or reject data, and to stop work in progress in the event that procedures or practices compromise the validity and integrity of analytical data. The QAM is available to any employee at the facility to resolve data quality or ethical issues.

#### 4.2.15 Quality Assurance Specialist

The QA Specialist is responsible for QA documentation and involvement in the following activities:

- Assist the QA Manager in performing the annual internal laboratory audits, compiling the evaluation, and coordinating the development of an action plan to address any deficiency identified.
- Facilitate external audits, coordinating with the QA Manager and Laboratory Staff to address any deficiencies noted at the time of the audit and subsequently presented in the final audit report.
- Assist the QA Manager in the preparation of new SOP's and in the maintenance of existing SOPs, coordinating annual reviews and updates.
- Manages the performance testing (PT) studies, coordinates follow up studies for failed analytes and works with QA Manager and Laboratory Staff to complete needed corrective action reports.
- Document control maintenance.

- Assists the Quality Manager and Project Management Group in the review of program plans for consistency with organizational and contractual requirements. Summarize and convey to appropriate personnel anomalies or inconsistencies observed in the review process.
- Manages certifications and accreditations.
- Monitors for compliance the following QA Metrics: Temperature Monitoring of refrigeration units and incubators; thermometer calibrations; balance calibrations; eppendorf/pipette calibrations; and proper standard/reagent storage.
- Periodic checks on the proper use and review of instrument logs.
- Assists with authenticity audits
- Assists in the technical review of data packages which require QA review.

# 4.2.16 <u>Technical Director</u>

The Technical Director reports directly to the Laboratory Director and is responsible for assessing the construction and management of the facility design, maintaining environmental conditions, technical and financial evaluation of capital equipment and capital budgeting and asset valuation.

In addition, the Technical Director solves day to day technical issues, provides technical training and guidance to staff, project managers and clients, investigates technical issues identified by operations personnel or QA, and directs evaluation of new methods. Specific responsibilities include but are not limited to:

- Reviewing and approving, with input from the QA Manager, proposals from marketing, in accordance with an established procedure for the review of requests and contracts. This procedure addresses the adequate definition of methods to be used for analysis and any limitations, the laboratory's capability and resources, the client's expectations. Differences are resolved before the contract is signed and work begins. A system documenting any significant changes is maintained, as well as pertinent discussions with the client regarding their requirements or the results of the analyses during the performance of the contract. All work subcontracted by the laboratory must be approved by the client. Any deviations from the contract must be disclosed to the client. Once the work has begun, any amendments to the contract must be discussed with the client and so documented.
- Monitoring the validity of the analyses performed and data generated in the laboratory. This
  activity begins with reviewing and supporting all new business contracts, insuring data
  quality, analyzing internal and external non-conformances to identify root cause issues and
  implementing the resulting corrective and preventive actions, facilitating the data review
  process (training, development, and accountability at the bench), and providing technical
  and troubleshooting expertise on routine and unusual or complex problems.
- Enhancing efficiency and improving quality through technical advances and improved LIMS utilization. Capital forecasting and instrument life cycle planning for second generation methods and instruments as well as asset inventory management.

# 4.2.17 **Operations Manager**

The Operations Manager reports to the Laboratory Director and oversees the daily operations of the analytical laboratory, maintaining a working environment that encourages open, constructive problem solving and continuous improvement.

The Operations Manager is responsible for supervision of laboratory staff, setting goals and objectives for the laboratory, ensuring compliance with project/client requirements and ensuring on-time performance, supervises maintenance of equipment and scheduling of repairs. Responsibilities also include implementation of the quality system in the laboratory and ensuring timely compliance with audit and QA corrective actions.

In addition, the Operations Manager works with the Technical Director in evaluating technical equipment, assessing capital budget needs and determining the most efficient instrument utilization. More specifically he:

- Evaluates the level of internal/external non-conformances for all departments.
- Continuously evaluates production capacity and improves capacity utilization.
- Continuously evaluates turnaround time and addresses any problems that may hinder meeting the required and committed turnaround time from the various departments.
- Develops and improves the training of all analysts in cooperation with the Technical Director and QA Manager and in compliance with regulatory requirements.
- Works with the Preventive Maintenance Coordinator to ensure that scheduled instrument maintenance is completed.
- Is responsible for efficient utilization of supplies.
- Constantly monitors and modifies the processing of samples through the departments.
- Fully supports the quality system and, if called upon in the absence of the QA Manager, serves as his substitute in the interim.

# 4.2.18 Department Managers

Department Managers report to the Operations Manager. The Department Managers serve as the technical experts on assigned projects, provide technical liaison, assist in resolving any technical issues within the area of their expertise; and implement established policies and procedures to assist the Operations Manager in achieving section goals. Each one is responsible to:

- Ensure that analysts in their department adhere to applicable SOPs and the QA Manual. They perform frequent SOP and QA Manual review to determine if analysts are in compliance and if new, modified, and optimized measures are feasible and should be added to these documents.
- With regard to analysts, participates in the selection, training, development of performance objectives and standards of performance, appraisal (measurement of objectives), scheduling, counseling, discipline, and motivation of analysts and documents these activities in accordance with systems developed by the QA and Human Resources Departments. They evaluate staffing sufficiency and overtime needs. Training consists of familiarization with SOP, QC, Safety, and computer systems.

- Encourage the development of analysts to become cross-trained in various methods and/or operate multiple instruments efficiently while performing maintenance and documentation, self-supervise, and function as a department team.
- Provide guidance to analysts in resolving problems encountered daily during sample prep/analysis in conjunction with the Technical Director, Operations Manager, and/or QA Manager. Each is responsible for 100% of the data review and documentation, nonconformance and CPAR issues, the timely and accurate completion of performance evaluation samples and MDLs, for his department.
- Ensure all logbooks are maintained, current, and properly labeled or archived.
- Report all non-conformance conditions to the QA Manager, Technical Director, Operations Manager, and/or Laboratory Director.
- Ensure that preventive maintenance is performed on instrumentation as detailed in the QA Manual or SOPs. He is responsible for developing and implementing a system for preventive maintenance, troubleshooting, and repairing or arranging for repair of instruments.
- Maintain adequate and valid inventory of reagents, standards, spare parts, and other relevant resources required to perform daily analysis.
- Achieve optimum turnaround time on analyses and compliance with holding times.
- Conduct efficiency and cost control evaluations on an ongoing basis to determine optimization of labor, supplies, overtime, first-run yield, capacity (designed vs. demonstrated), second- and third-generation production techniques/instruments, and longterm needs for budgetary planning.
- Develop, implement, and enhance calibration programs.
- Provide written responses to external and internal audit issues.

# 4.2.19 Environmental Health & Safety / Hazardous Waste Coordinator

The Health and Safety Coordinator is responsible for the safety and well-being of all employees while at the laboratory. This includes, but is not limited to, administering the Corporate Safety Manual that complies with federal regulations, MSDS training and review, conducting laboratory safety orientation and tours for all new employees, providing instructions on safety equipment, cleaning up laboratory spills, and instructing personnel of laboratory procedures for emergency situations. The Health and Safety Coordinator is on-call 24-hours a day, 7-days a week for all laboratory situations.

The Health and Safety Coordinator responsibilities additionally include waste management of laboratory generated hazardous waste in accordance with appropriate regulations. This includes maintenance of required documentation, such as waste manifests, segregation of waste in accordance with requirements, and training of personnel in proper segregation of waste and preparation of Safety related SOPs. The EHSC maintains overall EH&S program oversight, but may delegate specific day-to-day activities as necessary.

- Staying current with the hazardous waste regulations.
- Continuing training on hazardous waste issues.

- Reviewing and updating annually the Hazardous Waste Contingency Plan in the Environmental Health & Safety Manual.
- Auditing the staff with regard to compliance with the Hazardous Waste Contingency Plan.
- Contacting the hazardous waste subcontractors for review of procedures and opportunities for minimization of waste.
- Conduct ongoing, necessary safety training and conduct new employee safety orientation.
- Assist in developing and maintaining the Chemical Hygiene/Safety Manual.
- Administer dispersal of all Material Safety Data Sheet (MSDS) information.
- Perform regular chemical hygiene and housekeeping instruction.
- Give instruction on proper labeling and practice.
- Serve as chairman of the laboratory safety committee.
- Provide and train personnel on protective equipment.
- Oversee the inspection and maintenance of general safety equipment fire extinguishers, safety showers, eyewash fountains, etc. and ensure prompt repairs as needed.
- Supervise and schedule fire drills and emergency evacuation drills.
- Determine what initial and subsequent exposure monitoring, if necessary to determine potential employee exposure to chemicals used in the laboratory.
- When determined necessary, conduct exposure monitoring assessments.
- Determine when a complaint of possible over-exposure is "reasonable" and should be referred for medical consultation.
- Assist in the internal and external coordination of the medical consultation/monitoring program conducted by TestAmerica's medical consultants.

# 4.2.20 Laboratory Analysts

Laboratory analysts are responsible for conducting analysis and performing all tasks assigned to them by the group leader or supervisor. The responsibilities of the analysts are listed below:

- Perform analyses by adhering to analytical and quality control protocols prescribed by current SOPs, this QA Manual, and project-specific plans honestly, accurately, timely, safely, and in the most cost-effective manner.
- Document standard and sample preparation, instrument calibration and maintenance, data calculations, sample matrix effects, and any observed non-conformance on worklists, benchsheets, lab notebooks and/or the Non-Conformance Database.
- Report all non-conformance situations, instrument problems, matrix problems and QC failures, which might affect the reliability of the data, to their supervisor, the Technical Director, and/or the QA Manager or member of QA staff.
- Perform 100% review of the data generated prior to entering and submitting for secondary level review.
- Suggest method improvements to their supervisor, the Technical Director, and the QA Manager. These improvements, if approved, will be incorporated. Ideas for the optimum

performance of their assigned area, for example, through the proper cleaning and maintenance of the assigned instruments and equipment, are encouraged.

• Work cohesively as a team in their department to achieve the goals of accurate results, optimum turnaround time, cost effectiveness, cleanliness, complete documentation, and personal knowledge of environmental analysis.

# 4.2.21 Data Validation and Data Packaging Managers

The Data Validation and Data Packaging Specialists are responsible for coordinating receipt of all data from the various service groups within the laboratory, reviewing data for compliance to laboratory QC criteria and/or criteria in the LIMs Project Profile Specification, and ensuring that data are reported in a timely manner and in the proper format.

#### 4.2.22 Sample Management

The Sample Custodian is designated as the Sample Management Coordinator for any work performed internally and responsible for the receipt and login of client samples. The sample custodian confirms the samples received against the Chain of Custody, transports the samples to the proper storage unit within the facility and tracks the disposal of client samples after the required holding time has expired. The Sample Management Manager reports to the Technical Director. The responsibilities are outlined below:

- Direct the logging of incoming samples into the LIMS.
- Ensure the verification of data entry from login.
- Waste management and disposal

# 4.2.23 <u>Customer Service Manager</u>

The Customer Service Manager reports to the Laboratory Director and serves as the primary interface between the laboratory and the Sales and Marketing staff. Responsibilities include:

- Coordinates marketing efforts with General Manager, Laboratory Director, Project Managers, and laboratory marketing group
- Coordinates proposal and contract review and response process
- Compiles and interprets receipts forecast to show near term business trends.
- Prepares proposals for new business opportunities.
- Provides general sales support to Account Executives for business development activities started in the field.
- Develops and maintains business materials and organized information resource files that include project descriptions, resumes, original proposals, boilerplates, and company qualifications materials.

#### 4.2.24 Project Management Manager

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-13 of 4-15

The Project Management Manager reports to the Customer Service Manager and serves as the interface between the laboratory's technical departments and the laboratory's clients. The staff consists of the Project Management team. With the overall goal of total client satisfaction, the functions of this position are outlined below:

- Technical training and growth of the Project Management team.
- Technical liaison for the Project Management team.
- Human resource management of the Project Management team.
- Responsible to ensure that clients receive the proper sampling supplies.
- Accountable for response to client inquiries concerning sample status.
- Responsible for assistance to clients regarding the resolution of problems concerning COC.
- Ensuring that client specifications, when known, are met by communicating project and quality assurance requirements to the laboratory.
- Notifying the supervisors of incoming projects and sample delivery schedules.
- Accountable to clients for communicating sample progress in daily status meeting with agreed-upon due dates.
- Responsible for discussing with client any project-related problems, resolving service issues, and coordinating technical details with the laboratory staff.
- Responsible for staff familiarization with specific quotes, sample log-in review, and final report completeness.
- Monitor the status of all data package projects in-house to ensure timely and accurate delivery of reports.
- Inform clients of data package-related problems and resolve service issues.
- Coordinate requests for sample containers and other services (data packages).

# 4.2.25 <u>Project Manager</u>

The laboratory recognizes the importance of efficient project management. The laboratory Project Managers (PM) are responsible for preparing the LIMs project technical specifications which summarize QA/QC requirements for the project, maintaining the laboratory schedule, communicating technical requirements to the laboratory, and advising the Operations Manager, QA and Department Managers of all variances. The laboratory Project Manager will provide technical guidance and the necessary laboratory-related information to the preparer of project-specific QAPPs and provide peer review of the final document to ensure accuracy of the laboratory information.

#### 4.2.26 Bottle Kit Preparation and Shipping Manager

The Bottle Order Preparation Manager reports to the Customer Service Manager. He is responsible for ensuring the timely and correct shipment of sample containers, including proper preservatives and instructions, to clients. He maintains accurate records of sample container shipments.

# 4.3 <u>DEPUTIES</u>

The following table defines who assumes the responsibilities of key personnel in their absence:

Key Personnel	Deputy	Comment
Laboratory Director	Operations Manager (1)	
	Technical Director (2)	
QA Manager	QA Specialist (1)	
	Operations Manager (2)	
lechnical Director	Laboratory Director (1)	
	Operations Manager (2)	
Operations Manager	Department Manager (1)	Selected based on availability
	Department Manager (2)	
Customer Service Manager	Project Mng't Manager (1)	
	Laboratory Director (2)	
Project Management Manager	Customer Srv. Manager (1)	(2) Selected based on availability
	Project Manager (2)	
Project Manager	Project Manager (1)	(1) 2° team PM
	Project Management Asst. (2)	(2) Team PMA
Organic Department Manager	Analyst (1)	Selected based on department,
	Analyst (2)	experience and availability
Inorganic Department	Analyst (1)	Selected based on department,
Manager	Analyst (2)	experience and availability
Data Validation / Data	Data Validation Specialist	Selected based on department
Packaging Manager	Data Packaging Specialist	and availability
EHS Coordinator	Safety Officer (1)	
	Sample Mng't Manager (2)	
Sample Management	Sample Custodian (1)	
Manager	EHS Coordinator (2)	
Bottle Preparation / Shipping	Bottle Prep Technician (1)	
Manager	Sample Mng't Manager (2)	

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 4-15 of 4-15

Figure 4-1.

# **Corporate Organization Chart**



Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 5-1 of 5-6

#### **SECTION 5**

### QUALITY SYSTEM (NELAC 5.4.2)

#### 5.1 QUALITY POLICY STATEMENT

The management of TestAmerica and *TestAmerica Buffalo* are committed to providing data of known quality to its clients by adhering to approved methodologies, regulatory requirements and the QA/QC protocols described in this manual.

In all aspects of the laboratory and business operations, management is dedicated in maintaining the highest ethical standards. An Ethics Policy sign-off can be viewed in Appendix 1. Training on ethical and legal responsibilities is provided annually and each employee signs off annually on the policy as a condition of employment.

It is TestAmerica's Policy to continually improve systems and provide support to quality improvement efforts in laboratory, administrative and managerial activities. The company recognizes that the implementation of a quality assurance program requires management's commitment and support as well as the involvement of the entire staff.

*TestAmerica Buffalo* strives to provide clients with the highest level of professionalism and the best service practices in the industry.

Every staff member at **TestAmerica Buffalo** plays an integral part in quality assurance and is held responsible and accountable for the quality of their work. It is, therefore, required that all laboratory personnel are trained and agree to comply with applicable procedures and requirements established by this document.

#### 5.2 ETHICS AND DATA INTEGRITY

TestAmerica is committed to ensuring the integrity of its data and meeting the quality needs of its clients. The 7 elements of TestAmerica's Ethics and Data Integrity Program include:

- An Ethics Policy (Policy No. CA-L-P-001) and employee ethics statements (Appendix 1).
- An Ethics and Compliance Officer (ECO).
- A training program.
- Self-governance through disciplinary action for violations.
- A confidential mechanism for anonymously reporting alleged misconduct and a means for conducting internal investigations of all alleged misconduct. (SOP No. CA-L-S-001)
- Procedures and guidance for recalling data if necessary (SOP No. CA-L-S-001).
- An effective external and internal monitoring system that includes procedures for internal audits (Section 16).

As an American Council of Independent Laboratories (ACIL) member, all TestAmerica laboratories adhere to the following ACIL Code of Ethics:

- Produce results, which are accurate and include QA/QC information that meets client predefined Data Quality Objectives (DQOs).
- Present services in a confidential, honest and forthright manner.
- Provide employees with guidelines and an understanding of the ethical and quality standards of our industry.
- Operate our facilities in a manner that protects the environment and the health and safety of employees and the public.
- Obey all pertinent federal, state and local laws and regulations and encourage other members of our industry to do the same.
- Educate clients as the extent and kinds of services available.
- Assert competency only for work for which adequate personnel and equipment are available and for which adequate preparation has been made.
- Promote the status of environmental laboratories, their employees, and the value of services rendered by them.

# 5.3 QUALITY SYSTEM SUPPORTING DOCUMENTATION

The laboratory's Quality System is communicated through a variety of documents prepared by the laboratory and company management:

- Quality Assurance Manual (QAM) Template
- Quality Assurance Manual Each laboratory has a lab specific quality assurance manual.
- <u>Corporate SOPs and Policies</u> Corporate SOPs and Policies are developed for use by all relevant laboratories. They are incorporated into the laboratory's normal SOP distribution, training and tracking system. Corporate SOPs may be general or technical.
- <u>Work Instructions</u> A subset of procedural steps, tasks or forms associated with an operation of a management system (e.g., checklists, preformatted bench sheets, forms).
- Laboratory SOPs General and Technical
- Corporate TestAmerica QA/QC Policy Memorandums (Refer to Section 3.4).
- Laboratory QA/QC Policy Memorandums (Refer to Section 3.4).

# 5.3.1 Order of Precedence

In the event of a conflict or discrepancy between policies, the order of precedence is as follows:

- TestAmerica QA/QC Policy Memorandum Corporate
- Laboratory QA/QC Policy Memorandum
- Quality Assurance Manual
- Corporate SOPs and Policies
- Laboratory SOPs and Policies

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 5-3 of 5-6

• Other (Work Instructions (WI), memos, flow charts, etc.)

# 5.4 QA/QC OBJECTIVES FOR THE MEASUREMENT OF DATA

Quality Assurance (QA) and Quality Control (QC) are activities undertaken to achieve the goal of producing data that accurately characterize the sites or materials that have been sampled. Quality Assurance is generally understood to be more comprehensive than Quality Control. Quality Assurance can be defined as the integrated system of activities that ensures that a product or service meets defined standards.

Quality Control is generally understood to be limited to the analyses of samples and to be synonymous with the term *"analytical quality control"*. QC refers to the routine application of statistically based procedures to evaluate and control the accuracy of results from analytical measurements. The QC program includes procedures for estimating and controlling precision and bias and for determining reporting limits.

Request for Proposals (RFPs) and Quality Assurance Project Plans (QAPP) provide a mechanism for the client and the laboratory to discuss the data quality objectives in order to ensure that analytical services closely correspond to client needs. The client is responsible for developing the QAPP. In order to ensure the ability of the laboratory to meet the Data Quality Objectives (DQOs) specified in the QAPP, clients are advised to allow time for the laboratory to review the QAPP before being finalized. Additionally, the laboratory will provide support to the client for developing the sections of the QAPP that concern laboratory activities.

Historically, laboratories have described their QC objectives in terms of precision, accuracy, representativeness, comparability, completeness, selectivity and sensitivity (PARCCSS).

# 5.4.1 <u>Precision</u>

The laboratory objective for precision is to meet the performance for precision demonstrated for the methods on similar samples and to meet data quality objectives of the EPA and/or other regulatory programs. Precision is defined as the degree of reproducibility of measurements under a given set of analytical conditions (exclusive of field sampling variability). Precision is documented on the basis of replicate analysis, usually duplicate or matrix spike (MS) duplicate samples. The calculation of precision is described in Section 25.

# 5.4.2 <u>Accuracy</u>

The laboratory objective for accuracy is to meet the performance for accuracy demonstrated for the methods on similar samples and to meet data quality objectives of the EPA and/or other regulatory programs. Accuracy is defined as the degree of bias in a measurement system. Accuracy may be documented through the use of laboratory control samples (LCS) and/or MS. A statement of accuracy is expressed as an interval of acceptance recovery about the mean recovery. The calculation of accuracy is described in Section 25.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 5-4 of 5-6

### 5.4.3 <u>Representativeness</u>

The laboratory objective for representativeness is to provide data which is representative of the sampled medium. Representativeness is defined as the degree to which data represent a characteristic of a population or set of samples and is a measurement of both analytical and field sampling precision. The representativeness of the analytical data is a function of the procedures used in procuring and processing the samples. The representativeness can be documented by the relative percent difference between separately procured, but otherwise identical samples or sample aliguots.

The representativeness of the data from the sampling sites depends on both the sampling procedures and the analytical procedures. The laboratory may provide guidance to the client regarding proper sampling and handling methods in order to assure the integrity of the samples.

#### 5.4.4 <u>Comparability</u>

The comparability objective is to provide analytical data for which the accuracy, precision, representativeness and reporting limit statistics are similar to these quality indicators generated by other laboratories for similar samples, and data generated by the laboratory over time.

The comparability objective is documented by inter-laboratory studies carried out by regulatory agencies or carried out for specific projects or contracts, by comparison of periodically generated statements of accuracy, precision and reporting limits with those of other laboratories, and by the degree to which approval from the US EPA or other pertinent regulatory agencies is obtained for any procedure for which significant modifications have been made.

#### 5.4.5 <u>Completeness</u>

The completeness objective for data is 90% (or as specified by a particular project), expressed as the ratio of the valid data to the total data over the course of the project. Data will be considered valid if they are adequate for their intended use. Data usability will be defined in a QAPP, project scope or regulatory requirement. Data validation is the process for reviewing data to determine its usability and completeness. If the completeness objective is not met, actions will be taken internally and with the data user to improve performance. This may take the form of an audit to evaluate the methodology and procedures as possible sources for the difficulty or may result in a recommendation to use a different method.

# 5.4.6 <u>Selectivity</u>

Selectivity is defined as: The capability of a test method or instrument to respond to a target substance or constituent in the presence of non-target substances. Target analytes are separated from non-target constituents and subsequently identified/detected through one or more of the following, depending on the analytical method: extractions (separation), digestions (separation), interelement corrections (separation), use of matrix modifiers (separation), specific retention times (separation and identification), confirmations with different columns or detectors (separation and identification), specific wavelengths (identification), specific mass spectra (identification), specific electrodes (separation and identification), etc..

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 5-5 of 5-6

# 5.4.7 <u>Sensitivity</u>

Sensitivity refers to the amount of analyte necessary to produce a detector response that can be reliably detected (Method Detection Limit) or quantified (Reporting Limit).

# 5.5 CRITERIA FOR QUALITY INDICATORS

The laboratory prepares a *Quality Control Limit Summary that contains tables* that summarize the precision and accuracy acceptability limits for analyses performed at *TestAmerica Buffalo*. This summary includes an effective date, is updated each time new limits are generated and is located within the laboratory LIMs system. The quality control limits are method, protocol, matrix and date sensitive and may be retrieved as needed after expiration. Unless otherwise noted, limits within these tables are laboratory generated. Some acceptability limits are derived from US EPA methods when they are required. Where US EPA method limits are not required, *TestAmerica Buffalo* has developed limits from evaluation of data from similar matrices. Criteria for development of control limits is contained in Section 25.

# 5.6 STATISTICAL QUALITY CONTROL

Statistically-derived precision and accuracy limits are required by selected methods (such as SW-846) and programs [such as the Ohio Voluntary Action Plan (VAP)]. *TestAmerica Buffalo* routinely utilizes statistically-derived limits to evaluate method performance and determine when corrective action is appropriate. The procedure for determining the statistical limits may be found in SOP BF-QA-002, Quality Control Limits. The analysts are instructed to use the current limits in the laboratory (dated and approved the QA Manager) and entered into the Laboratory Information Management System (LIMS). The Quality Assurance department maintains an archive of all limits used within the laboratory through date sensitive tables within the LIMs System. If a method defines the QC limits, the method limits are used.

If a method requires the generation of historical limits, the lab develops such limits from recent data in the QC database of the LIMS following the guidelines described in Section 25. All calculations and limits are documented and dated when approved and effective. On occasion, a client requests contract-specified limits for a specific project.

Surrogate recoveries are determined for a specific time period as defined above. The resulting ranges are entered in LIMS.

Current QC limits are entered and maintained in the LIMS analyte database. As sample results and the related QC are entered into LIMS, the sample QC values are compared with the limits in LIMS to determine if they are within the acceptable range. The analyst then evaluates if the sample needs to be rerun or re-extracted/rerun or if a comment should be added to the report explaining the reason for the QC outlier.

# 5.6.1 <u>QC Charts</u>

As the QC limits are calculated, QC charts may be generated showing warning and control limits for the purpose of evaluating trends. The QA Manager periodically evaluates these to determine if adjustments need to be made or for corrective actions to methods. All findings are documented and kept on file.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 5-6 of 5-6

# 5.7 QUALITY SYSTEM METRICS

In addition to the QC parameters discussed above, the entire Quality System is evaluated on a monthly basis through the use of specific metrics (refer to Section 17). These metrics are used to drive continuous improvement in the laboratory's Quality System.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 6-1 of 6-2

#### **SECTION 6**

# DOCUMENT CONTROL (NELAC 5.4.3)

#### 6.1 <u>OVERVIEW</u>

The QA Department is responsible for the control of documents used in the laboratory to ensure that approved, up-to-date documents are in circulation and out-of-date (obsolete) documents are archived or destroyed. The following documents, at a minimum, must be controlled at each laboratory Facility:

- Laboratory Quality Assurance Manual
- Laboratory Standard Operating Procedures (SOP)
- Laboratory Policies
- Work Instructions and Forms
- Corporate Policies and Procedures distributed outside the intranet

The Corporate staff posts Corporate Manuals, SOPs, Policies, Work Instructions, White Papers and Training Materials on the company intranet site. These are collectively termed "Official Documents" and encompass the Policies and Procedures that all facilities are required to employ. These official documents are only considered controlled when they are read on the company intranet site. Printed copies are considered uncontrolled unless the laboratory physically distributes them as controlled documents. A detailed description of the procedure for issuing, authorizing, controlling, distributing, and archiving official documents is found in Corporate SOP No. CW-Q-S-001, Corporate Document Control and Archiving

The laboratory QA Department also maintains access to various references and document sources integral to the operation of the laboratory. This includes reference methods and regulations. Instrument manuals (hard or electronic copies) are also maintained by the laboratory.

The laboratory maintains control of records for raw analytical data and supporting records such as audit reports and responses, logbooks, standard logs, training files, MDL studies, Proficiency Testing (PT) studies, certifications and related correspondence, and corrective action notices. Raw analytical data consists of bound logbooks, instrument printouts, any other notes, magnetic media, electronic data and final reports. Discussion on records control is described in Section 15.

The maintenance of purchasing data is discussed in Section 9.

The maintenance of sales and marketing contracts is discussed in Section 7.

#### 6.2 DOCUMENT APPROVAL AND ISSUE

The pertinent elements of a control system for each document include a unique name and number, the number of pages of the item, the effective date, revision number and the laboratory's name. The Quality Assurance Specialist and Document Control Officer are

responsible for the maintenance of the system and maintain the items in the QA Storage Room, On-Site Archive Storage and Off-Site Archive Storage.

Controlled documents are authorized by the QA Department and other management. In order to develop a new document, a Department Manager submits an electronic draft to the QA Department for suggestions and approval before use. Upon approval, QA personnel add the identifying version information to the document and retain the official document on file. The official document is provided as needed to those using it. Controlled documents shall be available at all locations where the operational activity described in the document is performed (may include electronic access). Controlled documents are identified as such and records of their distribution are kept by the QA Department. Document control may be achieved by either electronic or hardcopy distribution.

The QA Department maintains a list of the official versions of controlled documents.

Quality System Policies and Procedures will be reviewed at a minimum of every two years for the majority of procedures, every 1 year for Drinking Water programs and revised as appropriate. Changes to documents occur when a procedural change warrants a revision of the document.

# 6.3 PROCEDURES FOR DOCUMENT CONTROL POLICY

For changes to the QA Manual, refer to SOP No. BF-QA-003, "Writing, Reviewing and Revising Controlled Documents". Uncontrolled copies must not be used within the laboratory. Previous revisions and back-up data are stored by the QA department. A controlled electronic copy of the current version is maintained on the laboratory IntraNet site (BufNet) and is available to all personnel.

For changes to SOPs, refer to SOP No. BF-QA-003, "Writing, Reviewing and Revising Controlled Documents".

Forms, worksheets, work instructions and information are organized by department in the QA office. Electronic versions are kept in a controlled access electronic folder in the QA department. As revisions are required, a new version number and revision date is assigned and the document placed on the laboratory IntraNet (BufNet) for use.

# 6.4 OBSOLETE DOCUMENTS

All invalid or obsolete documents are removed, or otherwise prevented from unintended use. The laboratory has specific procedures as described above to accomplish this. In general, obsolete documents are collected from employees according to distribution lists and are marked obsolete on the cover or destroyed. At least one copy of the obsolete document is archived as described in Section 15.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 7-1 of 7-4

#### **SECTION 7**

#### **REVIEW OF WORK REQUEST**

#### 7.1 OVERVIEW

**TestAmerica Buffalo** has established procedures for the review of work requests and contracts, oral or written. The procedures include evaluation of the laboratory's capability and resources to meet the contract's requirements within the requested time period. All requirements, including the methods to be used, must be adequately defined, documented and understood. For many environmental sampling and analysis programs, testing design is site or program specific and does not necessarily "fit" into a standard laboratory service or product. It is TestAmerica's intent to provide both standard and customized environmental laboratory services to our clients.

A thorough review of technical and QC requirements contained in contracts is performed to ensure project success. The appropriateness of requested methods, and the lab's capability to perform them must be established. Projects, proposals and contracts are reviewed for adequately defined requirements and TestAmerica's capability to meet those requirements. Alternate test methods that are capable of meeting the clients' requirements may be proposed by the lab. A review of the lab's capability to analyze non-routine analytes is also part of this review process.

All projects, proposals and contracts are reviewed for the client's requirements in terms of compound lists, test methodology requested, sensitivity (detection and reporting levels), accuracy, and precision requirements (% Recovery and RPD). The reviewer ensures that the laboratory's test methods are suitable to achieve these regulatory and client requirements and that the laboratory holds the appropriate certifications and approvals to perform the work. The laboratory and any potential subcontract laboratories must be certified, as required, for all proposed tests.

The laboratory must determine if it has the necessary physical, personnel and information resources to meet the contract, and if the personnel have the expertise needed to perform the testing requested. Each proposal is checked for its impact on the capacity of the laboratory's equipment and personnel. As part of the review, the proposed turnaround time will be checked for feasibility.

Electronic or hard copy deliverable requirements are evaluated against the lab's capacity for production of the documentation.

If the laboratory cannot provide all services but intends to subcontract such services, whether to another TestAmerica facility or to an outside firm, this will be documented and discussed with the client prior to contract approval. (Refer to Section 8 for Subcontracting Procedures.)

The laboratory informs the client of the results of the review if it indicates any potential conflict, deficiency, lack of accreditation, or inability of the lab to complete the work satisfactorily. Any discrepancy between the client's requirements and TestAmerica's capability to meet those requirements is resolved in writing before acceptance of the contract. It is necessary that the

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 7-2 of 7-4

contract be acceptable to both the laboratory and the client. Amendments initiated by the client and/or TestAmerica, are documented in writing.

All contracts, QAPPs, Sampling and Analysis Plans (SAPs), contract amendments, and documented communications become part of the project record.

The review process is repeated when there are amendments to the original contract by the client, and the participating personnel are informed of the changes.

#### 7.2 REVIEW SEQUENCE AND KEY PERSONNEL

Appropriate personnel will review the work request at each stage of evaluation.

For routine projects and other simple tasks, a review by the Project Manager (PM) is considered adequate. The PM confirms that the laboratory has any required certifications, that it can meet the clients' data quality and reporting requirements and that the lab has the capacity to meet the clients turn around needs. It is recommended that, where there is a sales person assigned to the account, an attempt should be made to contact that sales person to inform them of the incoming samples.

For new, complex or large projects, the proposed contract is given to the National Account Director, who will decide which lab will receive the work based on the scope of work and other requirements, including certification, testing methodology, and available capacity to perform the work. The contract review process is outlined in SOP No. CA-L-P-002, Contract Compliance Policy.

This review encompasses all facets of the operation. The scope of work is distributed to the appropriate personnel, as needed based on scope of contract, to evaluate all of the requirements shown above (not necessarily in the order below):

- Legal & Contracts Director
- General Manager
- Customer Service Manager
- Operations Manager
- Laboratory and/or Corporate Technical Directors
- Corporate Information Technology Managers/Directors
- Regional and/or National Account representatives
- Laboratory and/or Corporate Quality
- Laboratory and/or Corporate Environmental Health and Safety Managers/Directors
- The Laboratory Director reviews the formal laboratory quote and makes final acceptance for their facility.

The National Account Director, Legal Contracts Director, or local account representative then submits the final proposal to the client.

In the event that one of the above personnel is not available to review the contract, his or her back-up will fulfill the review requirements.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 7-3 of 7-4

The Legal & Contracts Director maintains copies of all signed contracts. The Customer Service Manager at the TestAmerica Buffalo facility also maintains copies of these documents.

# 7.3 DOCUMENTATION

Appropriate records are maintained for every contract or work request. All stages of the contract review process are documented and include records of any significant changes.

The contract will be distributed to and maintained by the appropriate sales/marketing personnel and the Regional Account Manager. A copy of the contract and formal quote will be filed with the laboratory PM and the Customer Service Manager.

Records are maintained of pertinent discussions with a client relating to the client's requirements or the results of the work during the period of execution of the contract. The PM keeps a phone log of conversations with the client.

# 7.3.1 <u>Project-Specific Quality Planning</u>

Communication of contract specific technical and QC criteria is an essential activity in ensuring the success of site specific testing programs. To achieve this goal, *TestAmerica Buffalo* assigns a PM to each client. The PM is the first point of contact for the client. It is the PM's responsibility to ensure that project specific technical and QC requirements are effectively evaluated and communicated to the laboratory personnel before and during the project. QA department involvement may be needed to assist in the evaluation of custom QC requirements. Specific information related to project planning may be found in SOP BF-PM-001, Project Information Requirements.

PM's are the direct client contact and they ensure resources are available to meet project requirements. Although PM's do not have direct reports or staff in production, they coordinate opportunities and work with laboratory management staff to ensure available resources are sufficient to perform work for the client's project. Project management is positioned between the client and laboratory resources.

Prior to work on a new project, the dissemination of project information and/or project opening meetings may occur to discuss schedules and unique aspects of the project. Items to be discussed may include the project technical profile, turnaround times, holding times, methods, analyte lists, reporting limits, deliverables, sample hazards, or other special requirements. The PM introduces new projects to the laboratory staff through project kick-off meetings or to the management staff during production meetings. These meetings provide direction to the laboratory staff in order to maximize production and client satisfaction, while maintaining quality. In addition, project notes may be associated with each sample batch as a reminder upon sample receipt and analytical processing.

During the project, any change that may occur within an active project is agreed upon between the client/regulatory agency and the PM/laboratory. These changes (e.g., use of a non-standard method or modification of a method) and approvals must be documented prior to implementation. Documentation pertains to any document, e.g., letter, e-mail, variance, contract addendum, which has been signed by both parties.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 7-4 of 7-4

Such changes are also communicated to the laboratory during production meetings. Such changes are updated to the project notes and are introduced to the managers at these meetings. The laboratory staff is then introduced to the modified requirements via the PM or the individual laboratory Department Manager.

TestAmerica strongly encourages client visits to the laboratory and for formal/informal information sharing session with employees in order to effectively communicate ongoing client needs as well as project specific details for customized testing programs.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-1 of 8-8

#### **SECTION 8**

# SUBCONTRACTING OF TESTS (NELAC 5.4.5)

#### 8.1 <u>OVERVIEW</u>

For the purpose of this quality manual, the phrase subcontract laboratory refers to a laboratory external to the corporate network. The phrase "work sharing" refers to internal transfers of samples between company laboratories. The term outsourcing refers to the act of subcontracting tests.

When contracting with our clients, the laboratory makes commitments regarding the services to be performed and the data quality for the results to be generated. When we must outsource testing for our clients because project scope, changes in laboratory capabilities, capacity or unforeseen circumstances, we must be assured that the subcontractors or work sharing laboratories understand the requirements and will meet the same commitments we have made to the client. Refer to the SOP on Subcontracting Procedures (CA-L-S-002) and the Work Sharing Process SOP (CA-C-S-001).

When outsourcing analytical services, the laboratory will assure, to the extent necessary, that the subcontract or work sharing laboratory maintains a program consistent with the requirements of this document, the requirements specified in NELAC/ISO 17025 and/or the client's Quality Assurance Project Plan (QAPP). All QC guidelines specific to the client's analytical program are transmitted to the subcontractor and agreed upon before sending the samples to the subcontract facility. Additionally, work requiring accreditation will be placed with an appropriately accredited laboratory. The laboratory performing the subcontracted work will be identified in the final report, as will non-NELAC accredited work where required.

Project Managers (PMs), Customer Service Managers (CSM), or Regional Account Executives (RAE) for the Export Lab are responsible for obtaining client approval prior to outsourcing any samples. The laboratory will advise the client of a subcontract or work sharing arrangement in writing and when possible approval from the client shall be retained in the project folder.

**Note:** In addition to the client, some regulating agencies, such as the Department of Energy and the USDA, require notification prior to placing such work.

Approval may be documented through reference in a quote / contract or e-mail correspondence.

#### 8.2 QUALIFYING AND MONITORING SUBCONTRACTORS

Whenever a PM, Regional Account Executive (RAE) or Customer Service Manager (CSM] becomes aware of a client requirement or laboratory need where samples must be outsourced to another laboratory, the other laboratory(s) shall be selected based on the following:

- The first priority is to attempt to place the work in a qualified network laboratory;
- Firms specified by the client for the task (Documentation that a subcontractor was designated by the client must be maintained with the project file. This documentation can be
as simple as placing a copy of an e-mail from the client in the project folder);

- Firms listed as pre-qualified and currently under a subcontract with the company (in JD Edwards): A listing of all approved subcontracting laboratories and supporting documentation is available on the TestAmerica intranet site. Verify necessary accreditation for the requested tests prior to sending samples.
- Firms identified in accordance with the company's Small Business Subcontracting program as small, women-owned, veteran-owned and/or minority-owned businesses;
- NELAC or A2LA accredited laboratories.
- In addition, the firm must hold the appropriate certification to perform the work required.

All intra-company laboratories are pre-qualified for work-sharing provided they hold the appropriate accreditations, can adhere to the project/program requirements, and the client approved sending samples to that laboratory. The client must provide acknowledgement that the samples can be sent to that facility (an e-mail is sufficient documentation or if acknowledgement is verbal, the date, time, and name of person providing acknowledgement must be documented). The originating laboratory is responsible for communicating all technical, quality, and deliverable requirements as well as other contract needs. Refer to SOP No. CA-C-S-001, Work Sharing Process.

When the potential sub-contract laboratory has not been previously approved, then to begin the process, Account Executives or PMs may nominate a laboratory as a subcontractor based on need. The decision to nominate a laboratory must be approved by the Laboratory Director. The Laboratory Director requests that the QA Manager begin the process of approving the subcontract laboratory. The client must provide acknowledgement that the samples can be sent to that facility (an e-mail is sufficient documentation or if acknowledgement is verbal, the date, time, and name of person providing acknowledgement must be documented).

**8.2.1** The QA Manager must ensure that the Subcontracting Approval Form (Figure 8-1) has been completed and have supporting documentation on file prior to initiation of any work. A letter or e-mail is sent to the lab requesting the following information:

- 8.2.1.1 If a lab is NELAC or A2LA accredited,
- **8.2.1.1.1** Copy of necessary certifications verifying that the required approvals are current. Ensure that all needed analytes are included; some may not be accredit-able (if so, document). Certificate and scope of International Standard accreditation are required, when applicable.
- 8.2.1.1.2 Insurance Certificate. This is required by TestAmerica's Chief Financial Officer
- 8.2.1.1.3 USDA soil permit if available**
- 8.2.1.2 For Laboratories accredited by other agencies with an auditing program:
- **8.2.1.2.1** Copy of necessary certifications verifying that the required approvals are current. Ensure that all needed analytes are included; some may not be accredit-able (if so, document). Certificate and scope of International Standard accreditation are

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-3 of 8-8

required, when applicable.

- 8.2.1.2.2 Insurance Certificate. This is required by TestAmerica's Chief Financial Officer
- 8.2.1.2.3 USDA soil permit if available**
- **8.2.1.2.4** Description of Ethics and Data Integrity Plan.
- **8.2.1.2.5** The most recent 2 sets of full proficiency testing (PT) results relevant to the analyses of interest and any associated corrective action.
- 8.2.1.2.6 State Audit with Corrective Action Response
- **8.2.1.2.7** Example final report to confirm format is compliant and provides the necessary information. (minimally, it must be determined that Batch QC results are included in the laboratory reports and data is appropriately qualified.
- **8.2.1.2.8** A copy of raw data associated with the first project is requested for internal review. The raw data is reviewed by the QA Manager and the PM to ensure that the results meet the client's needs. If the QA manager is unfamiliar with the analysis being performed, notify Corporate QA for guidance on the review (it may need to be sent elsewhere for evaluation). This requirement can be skipped if an on-site visit of the laboratory is planned. (This requirement is effective as of the effective date of this section. Laboratories worked with previously [minimum of 6 months] are grandfathered in.
- **8.2.1.3** For laboratories performing tests that are unaccredited or accredited by an agency without an audit program:
- **8.2.1.3.1** A copy of their Quality Assurance Manual (controlled if possible). Ensure data quality limits for relevant methods are acceptable and that training procedures are adequate.
- **8.2.1.3.2** Copy of necessary certifications (if available) verifying that the required approvals are current. Ensure that all needed analytes are included; some may not be accredit-able (if so, document). Certificate and scope of International Standard accreditation are required, when applicable.
- **8.2.1.3.3** Insurance Certificate. This is required by TestAmerica's Chief Financial Officer.
- 8.2.1.3.4 USDA soil permit if available**
- **8.2.1.3.5** Evidence of a current SOP per method. A copy of the first page and signature page of the SOP is acceptable. A table of contents including effective dates may also be acceptable. The SOP can be examined if an on-site audit is performed.
- **8.2.1.3.6** Description of Ethics and Data Integrity Plan.
- **8.2.1.3.7** The most recent 2 sets of full proficiency testing (PT) results relevant to the analyses

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-4 of 8-8

of interest and any associated corrective action.

- **8.2.1.3.8** Example final report to confirm format is compliant and provides the necessary information. (minimally, it must be determined that Batch QC results are included in the laboratory reports and data is appropriately qualified.
- **8.2.1.3.9** Statement of Qualification (SOQ) or summary list of Technical Staff and Qualifications position, education and years of experience.
- **8.2.1.3.10** A copy of raw data associated with the first project is requested for internal review. The raw data is reviewed by the QA Manager and the PM to ensure that the results meet the client's needs. If the QA manager is unfamiliar with the analysis being performed, notify Corporate QA for guidance on the review (it may need to be sent elsewhere for evaluation). This requirement can be skipped if an on-site visit of the laboratory is planned. (This requirement is effective as of the effective date of this section. Laboratories worked with previously [minimum of 6 months] are grandfathered in.)

**8.2.2** Once the information is received by the QA Manager, it is evaluated for acceptability and forwarded to Corporate Contracts for formal contracting with the laboratory. They will add the lab to the approved list on the intranet site along with the associate documentation and notify the finance group for JD Edwards.

**USDA permit is required if soils less than three feet deep from New York, North Carolina, South Carolina, Georgia, Florida, Tennessee, Alabama, Mississippi, Louisiana, Arkansas, Texas, Oklahoma, New Mexico, Arizona, California, Hawaii, or outside the continental U. S. are to be analyzed. These samples require special shipping measures; check with the EHS Department. It may be necessary to heat-treat the samples before shipping if the subcontract laboratory does not have a USDA permit; however, some analytes/tests may be irrelevant after heat treatment.

**8.2.3** The client will assume responsibility for the quality of the data generated from the use of a subcontractor they have requested the lab to use. The qualified subcontractors on the intranet site are known to meet minimal standards. The company does not certify laboratories. The subcontractor is on our approved list and can only be recommended to the extent that we would use them.

**8.2.4** The status and performance of qualified subcontractors will be monitored periodically by the Corporate Contract Department. Any problems identified will be brought to Corporate QA attention.

- Complaints shall be investigated. Documentation of the complaint, investigation and corrective action will be maintained in the subcontractor's file on the intranet site. Complaints must be posted using the Vendor Performance Report (Form No. CW-F-WI-009).
- Information must be updated on the intranet when new information is received from the subcontracted laboratories.
- Subcontractors in good standing will be retained on the intranet listing. The QA Manager will

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-5 of 8-8

notify all network laboratories and Corporate QA and Corporate Contracts if any laboratory requires removal from the intranet site. This notification will be posted on the intranet site and e-mailed to all Lab Directors/Managers, QA Managers and Sales Directors.

#### 8.3 OVERSIGHT AND REPORTING

The PM must request that the selected subcontractor be presented with a subcontract, if one is not already executed between the laboratory and the subcontractor. The subcontract must include terms which flow down the requirements of our clients, either in the subcontract itself or through the mechanism of work orders relating to individual projects. A standard subcontract and the Lab Subcontractor Vendor Package (posted on the intranet) can be used to accomplish this, and the Legal & Contracts Director can tailor the document or assist with negotiations, if needed. The PM (or RAE or CSM) responsible for the project must advise and obtain client consent to the subcontract as appropriate, and provide the scope of work to ensure that the proper requirements are made a part of the subcontract and are made known to the subcontractor.

Prior to sending samples to the subcontracted laboratory, the PM confirms their certification status to determine if it's current and scope-inclusive. The information is documented on a Subcontract Laboratory Certification Verification Form (Figure 8-2) and the form is retained in the project folder. For network laboratories, certifications can be viewed on the company website.

The Sample Control department is responsible for ensuring compliance with QA requirements and applicable shipping regulations when shipping samples to a subcontracted laboratory.

All subcontracted samples must be accompanied by a Chain of Custody (COC). A copy of the original COC sent by the client must be included with all samples subbed within the network.

The PM will communicate with the subcontracted laboratory to monitor the status of the analyses, facilitate successful execution of the work and ensure the timeliness and completeness of the analytical report.

Non-NELAC accredited work must be identified in the subcontractor's report as appropriate. If NELAC accreditation is not required, the report does not need to include this information.

Reports submitted from subcontractor laboratories are not altered and are included in their original form in the final project report. This clearly identifies the data as being produced by a subcontractor facility. If subcontract laboratory data are incorporated into the laboratories EDD (i.e., imported), the report must explicitly indicate which lab produced the data for which methods and samples.

**Note:** The results submitted by a network work sharing laboratory may be transferred electronically and the results reported by the network work sharing lab are identified on the final report. The report must explicitly indicate which lab produced the data for which methods and samples. The final report must include a copy of the completed COC for all work sharing reports.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-6 of 8-8

#### 8.4 CONTINGENCY PLANNING

The Laboratory Director may waive the full qualification of a subcontractor process temporarily to meet emergency needs. In the event this provision is utilized, Corporate QA must be informed, and the QA Manager will be required to verify adequacy of proficiency scores and certifications. The laboratory must also request a copy of the raw data to support the analytical results for the first project submitted to the subcontract laboratory unless the laboratory has NELAC accreditation. The raw data is reviewed by the QA Manager and the PM to ensure that the results meet the client's needs. The QA Manager will request full documentation and qualify the subcontractor under the provisions above. The approval process should be completed within 30 calendar days of subcontracting.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-7 of 8-8

_____

# Figure 8-1. Subcontracting Laboratory Approval Form (Initial / Renewal)

_____

#### SUBCONTRACTING LABORATORY APPROVAL

Reference: Section 8 – Quality Assurance Manual

Laboratory:

Date:

Address:

Fax _____

Requested Item ³	Date Received	Reviewed/ Accepted	Date
1. Copy of State Certification ¹			
2. Insurance Certificate			
3. USDA Soil Permit			
4. Description of Ethics Program ³			
5. QA Manual ³			
6. Most Recent (and relevant) 2 Sets of WP/WS Reports with Corrective Action Response ^{1,3}			
7. State Audit with Corrective Action Response (or NELAC or A2LA Audit) ³			
8. Sample Report ³			
9. SOQ or Summary list of Technical Staff and Qualifications ³			
10. SOPs for Methods to Be Loadshifted ^{2,3}			
11. For DoD Work: Statement that Lab quality system complies with QSM.			
12. For DoD Work: Approved by specific DoD Component laboratory approval process.			

1 - Required when emergency procedures are implemented.

2 - Some labs may not submit copies due to internal policies. In these cases, a copy of the first page and signature page of the SOP is acceptable. This requirement may also be fulfilled by supplying a table of SOPs with effective dates.
3 - If the laboratory has NELAC accreditation, <u>Item #s 4 through 10 are not required.</u>

On Site Audit Planned: YES	NO If yes, Date Cor	npleted: By Whom:	
Comments:			
Lab Acceptable for Subcontract	ing Work: YES NO	Limitations:	
QA Manager (Signature):	······	Date:	
(Prin	ited Name)		
Forwarded to Contract Coord	dinator, by:	Date:	

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 8-8 of 8-8

Figure 8-2.

# Subcontract Laboratory Certification Verification Form

TestAmerica Buffalo				
Certification Verification				
Subcontract Lab				
To be completed by Project Manager:				
Date:				
Project Manager: PROJECT #: TASK:				
Client: State: Client Notified Yes No				
Orinking Water     - Waste Water     - Solid     - Air Method : Parameters : List Attached for multi-analyte/project specified reporting limits Yes No Requested Subcontract Lab:				
To be completed by Q/A:				
APPROVED				
- Lab is Certified – forward samples				
- Analysis is not certifiable by State – forward samples				
Signature (QA Dept.) and Date				

# NOT APPROVED Appropriate certification not in place at this time. DO NOT SEND SAMPLES TO THIS SUBCONTRACT LAB Signature (QA Dept.) and Date

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 9-1 of 9-7

#### **SECTION 9**

# PURCHASING SERVICES AND SUPPLIES (NELAC 5.4.6)

#### 9.1 <u>OVERVIEW</u>

Evaluation and selection of suppliers and vendors is performed, in part, on the basis of the quality of their products, their ability to meet the demand for their products on a continuous and short term basis, the overall quality of their services, their past history, and competitive pricing. This is achieved through evaluation of objective evidence of quality furnished by the supplier, which can include certificates of analysis, recommendations, and proof of historical compliance with similar programs for other clients. To ensure that quality critical consumables and equipment conform to specified requirements, all purchases from specific vendors are approved by a member of the supervisory or management staff.

Capital expenditures are made in accordance with the Controlled Purchases Procedure, CW-F-S-004. Only one quote is required where the item being purchased is a sole source product, Examples of sole source capital expenditures are laboratory test equipment, client specified purchases and building leases. A minimum of two quotes is required where the opportunity exists to source from more than one vendor. All documentation related to the purchase of capital items will be maintained in the individual CapEx files located in Corporate Purchasing. Data will be held in accordance with the record retention policy.

TestAmerica will enter into formal contracts with vendors when it is advantageous to do so. Contracts will be signed in accordance with the Authorization Matrix Policy, CW-F-P-002. Examples of items that are purchased through vendor contracts are laboratory instruments, consumables, copiers and office supplies. Request for Proposals (RFP's) will be issued where more information is required from the potential vendors than just price. RFP's allow TestAmerica to determine if a vendor is capable of meeting requirements such as supplying all of the TestAmerica facilities, meeting required quality standards and adhering to necessary ethical and environmental standards. The RFP process also allows potential vendors to outline any additional capabilities they may offer.

Non-capital expenditure items are purchased through the requisition and approval process in JD Edwards or through other TestAmerica authorized methods (approved web-sites, purchasing cards). Labs have the ability to select from the approved vendors in JD Edwards.

#### 9.2 GLASSWARE

Glassware used for volumetric measurements must be Class A or verified for accuracy according to laboratory procedure. Pyrex (or equivalent) glass should be used where possible. For safety purposes, thick-wall glassware should be used where available.

#### 9.3 REAGENTS, STANDARDS & SUPPLIES

Chemical reagents, solvents, glassware, and general supplies are ordered as needed to maintain sufficient quantities on hand. Purchasing guidelines for equipment and reagents must meet with the requirements of the specific method and testing procedures for which they are

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 9-2 of 9-7

being purchased. Solvents and acids are pre-tested in accordance with Corporate SOP on Solvent & Acid Lot Testing & Approval, SOP No. CA-Q-S-001 and TestAmerica Buffalo SOP on Solvent Purity, SOP BF-OP-013.

#### 9.3.1 <u>Purchasing</u>

The nature of the analytical laboratory demands that all material used in any of the procedures is of a known quality. The wide variety of materials and reagents available makes it advisable to specify recommendations for the name, brand, and grade of materials to be used in any determination. This information is contained in the method SOP. Purchase requisitions are placed into the J.D. Edwards system by designated departmental personnel. The listing of items available in the J.D. Edwards system has been approved for use by the corporate purchasing staff. Each purchase requisition receives final approval by the laboratory Operations Manager or purchasing coordinator before the order is submitted.

The analyst may also check the item out of the on-site consignment system that contains items approved for laboratory use.

# 9.3.2 <u>Receiving</u>

It is the responsibility of the purchasing coordinator to receive the shipment. It is the responsibility of the department that ordered the materials to date the material when received. Once the ordered reagents or materials are received, the department that submitted the order compares the information on the label or packaging to the original order to ensure that the purchase meets the quality level specified. Material Safety Data Sheets (MSDSs) are kept online through the Company's intranet website. Anyone may review these for relevant information on the safe handling and emergency precautions of on-site chemicals.

#### 9.3.3 <u>Specifications</u>

There are many different grades of analytical reagents available to the analyst. All methods in use in the laboratory specify the grade of reagent that must be used in the procedure. If the quality of the reagent is not specified, it may be assumed that it is not significant in that procedure and, therefore, any grade reagent may be used. It is the responsibility of the analyst to check the procedure carefully for the suitability of grade of reagent.

Chemicals must not be used past the manufacturer's expiration date and must not be used past the expiration time noted in a method SOP. If dates are not provided, the laboratory may contact the manufacturer to determine an expiration date.

The laboratory assumes a five year expiration date on inorganic dry chemicals unless noted otherwise by the manufacturer or by the reference source method.

• An expiration date can not be extended if the dry chemical is discolored or appears otherwise physically degraded, the dry chemical must be discarded.

- Expiration dates can be extended if the dry chemical is found to be satisfactory based on acceptable performance of quality control samples (Continuing Calibration Verification (CCV), Blanks, Laboratory Control Sample (LCS), etc.).
- If the dry chemical is used for the preparation of standards, the expiration dates can be extended 6 months if the dry chemical is compared to an unexpired independent source in performing the method and the performance of the dry chemical is found to be satisfactory. The comparison must show that the dry chemical meets CCV limits. The comparison studies are maintained along with the calibration raw data for which the reagent was used.

Wherever possible, standards must be traceable to national or international standards of measurement or to national or international reference materials. Records to that effect are available to the user.

Compressed gases in use are checked for pressure and secure positioning daily. The minimum total pressure must be 200 psig or the tank must be replaced. The quality of the gases must meet method or manufacturer specification or be of a grade that does not cause any analytical interference.

Water used in the preparation of standards or reagents must have a conductivity of less than 1mmho/cm (or resistivity of greater than 1.0 megaohm-cm) at 25°C. The conductivity is checked and recorded daily. If the water's conductivity is outside the specified limit, the Technical Director, Operations Manager and/or QA Manager must be notified immediately in order to notify all departments, decide on cessation (based on intended use) of activities, and make arrangements for correction.

The laboratory may purchase reagent grade (or other similar quality) for use in the laboratory. This water must be certified "clean" by the supplier for all target analytes or otherwise verified by the laboratory prior to use. This verification is documented.

Standard lots are verified before first time use if the laboratory switches manufacturers or has historically had a problem with the type of standard.

Purchased VOA vials must be certified clean and the certificates must be maintained. If uncertified VOA vials are purchased, all lots must be verified clean prior to use. This verification must be maintained.

# 9.3.4 <u>Storage</u>

Reagent and chemical storage is important from the aspects of both integrity and safety. Lightsensitive reagents may be stored in brown-glass containers. Table 9-1 details specific storage instructions for reagents and chemicals. Section 22 discusses conditions for standard storage.

# 9.4 PURCHASE OF EQUIPMENT/INSTRUMENTS/SOFTWARE

When a new piece of equipment is needed, either for additional capacity or for replacing inoperable equipment, the analyst or supervisor makes a supply request to the Operations Manager and/or the Laboratory Director. If they agree with the request the procedures outlined in Policy No. CA-T-P-001, Qualified Products List, are followed. A decision is made as to which

piece of equipment can best satisfy the requirements. The appropriate written requests are completed and purchasing places the order.

Upon receipt of a new or used piece of equipment, it is given a short name, such as HP-20, added to the equipment list described in Section 21 that is maintained by the Technical Director and IT must be notified so that can be linked for back-ups. Its capability is assessed to determine if it is adequate or not for the specific application. For instruments, a calibration curve is generated, followed by MDLs, Demonstration of Capabilities (DOCs), and other relevant criteria (see Section 20). For software, its operation must be deemed reliable and evidence of instrument verification must be retained by the IT Department or QA Department as specified in the laboratory's procedure for software verification. Software certificates supplied by the vendors are filed with the LIMS Administrator. The manufacturer's operation manual is retained at the bench.

# 9.5 <u>SERVICES</u>

Service to analytical instruments (except analytical balances) is performed on an as needed basis. Routine preventative maintenance is discussed in Section 21. The need for service is determined by analysts and/or Department Managers. The service providers that perform the services are approved by the Department Managers, Operations Manager and/or Technical Director.

# 9.6 <u>SUPPLIERS</u>

TestAmerica selects vendors through a competitive proposal / bid process, strategic business alliances or negotiated vendor partnerships (contracts). The level of control used in the selection process is dependent on the anticipated spend and the potential impact on TestAmerica business. Vendors that provide test and measuring equipment, solvents, standards, certified containers, instrument related service contracts or subcontract laboratory services shall be subject to more rigorous controls than vendors that provide off-the-shelf items of defined quality that meet the end use requirements. The JD Edwards purchasing system includes all suppliers /vendors that have been approved for use.

Evaluation of suppliers is accomplished by ensuring the supplier ships the product or material ordered and that the material is of the appropriate quality. This is documented by signing off on packing slips or other supply receipt documents. The purchasing documents contain the data that adequately describe the services and supplies ordered.

Any issues of vendor performance are to be reported immediately by the laboratory staff to the Corporate Purchasing Group by completing a Vendor Performance Report (CW-F-WI-009).

The Corporate Purchasing Group will work through the appropriate channels to gather the information required to clearly identify the problem and will contact the vendor to report the problem and to make any necessary arrangements for exchange, return authorization, credit, etc.

As deemed appropriate, the Vendor Performance Reports will be summarized and reviewed to determine corrective action necessary, or service improvements required by vendors

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 9-5 of 9-7

The laboratory has access to a listing of all approved suppliers of critical consumables, supplies and services. This information is provided through the JD Edwards purchasing system.

#### 9.6.1 <u>New Vendor Procedure</u>

TestAmerica employees who wish to request the addition of a new vendor must complete a J.D. Edwards Vendor Add Request Form (CW-F-WI-007 – refer to Figure 9-2).

New vendors are evaluated based upon criteria appropriate to the products or services provided as well as their ability to provide those products and services at a competitive cost. Vendors are also evaluated to determine if there are ethical reasons or potential conflicts of interest with TestAmerica employees that would make it prohibitive to do business with them as well as their financial stability. The QA Department and/or the Technical Director are consulted with vendor and product selection that have an impact on quality.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 9-6 of 9-7

# Table 9-1.

# **Storage of Reagents and Chemicals**

Chemical	Storage Requirements
Concentrated Acids and Bases	Stored in the original containers at room temperature. All organic acids must be stored separately from inorganic acids. Acids should not be stored with bases.
Bulk Dry Chemicals	Stored in the original containers at room temperature. All organic acids must be stored separately from inorganic acids. Acids should not be stored with bases.
Working Solutions containing Organic Compounds	Stored as per method recommendation/ requirement. They are generally stored refrigerated at 4°C± 2°C.
Working Solutions containing only Inorganics	Stored at room temperature; refrigeration is optional.
Flammable Solvents	Stored in solvent cabinets at room temperature.
Non-Flammable Solvents	Stored separately from the flammable solvents in cabinets at room temperature.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 9-7 of 9-7

# Figure 9-1 Example – JD Edwards Vendor Add Request Form



THE LEADER IN ENVIRONMENTAL TESTING

#### JD Edwards Vendor Add Request Form

Vendor name:	Lab location and individual making request:
Vendor address (remit to):	Vendor phone:
Vendor address (remit to):	Vendor fax:
Contact name:	Product / service provided:
	,

#### Reason for Vendor Addition: Check all reasons that apply

Cost Reduction	Estimated Annual Savings \$	
Replace Current Vendor	Reason?	
	Vendor being Replaced?	
New Product / Service	Describe:	
ISO Approved (Required for Aerotec	h / P&K only)	

#### Small Business:

Does this vendor help us to meet our small business objectives:

If yes, which category:

#### Personal and Ethical Considerations:

Is there any personal conflict of interest with a TestAmerica employee and the vendor listed above?

Have ethical considerations been taken into account in your evaluation of this vendor?_

#### Can this product be sourced from another TestAmerica facility?_____

Please complete form and email to NCPurchasing@testamericainc.com or fax to (330) 966-9275.

I approve the addition of this vendor.

Purchasing Manager - Patrick Eckman Corporate Controller - Leslie Bowers

Form No. CW-F-WI-007

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 10-1 of 10-2

#### **SECTION 10**

# SERVICE TO THE CLIENT (NELAC 5.4.7)

#### 10.1 <u>OVERVIEW</u>

**TestAmerica Buffalo** cooperates with clients and their representatives to monitor the laboratory's performance in relation to work performed for the client. It is the laboratory's goal to meet all client requirements in addition to statutory and regulatory requirements discussed in Section 5. The laboratory has procedures to ensure confidentiality to clients (Section 16 and 26).

**Note:** ISO 17025/NELAC 2003 states that a laboratory "shall afford clients or their representatives cooperation to clarify the client's request". This topic is discussed in Section 7.

#### 10.2 SPECIAL SERVICES

The laboratory's standard procedures for reporting data are described in Section 26. When requested the following special services are provided:

- The laboratory will provide the client or the client's representative reasonable access to the relevant areas of the laboratory for the witnessing of tests performed for the client.
- The laboratory will work with client-specified third party data validators as specified in the client's contract.
- The laboratory will provide the client with all requested information pertaining to the analysis of their samples. An additional charge may apply for additional data/information that was not requested prior to the time of sample analysis or previously agreed upon.

#### 10.3 CLIENT COMMUNICATION

Project managers are an important communication link to the clients. The lab shall inform its clients of any delays in project completion as well as any non-conformances in either sample receipt (refer to Section 24) or sample analysis. Project management will maintain ongoing client communication throughout the entire client project.

Department Managers, the Operations Manager, the Technical Director and the Quality Manager are available to discuss any technical questions or concerns that the client may have.

#### 10.4 <u>REPORTING</u>

The laboratory will work with the client to produce any special communication reports required by the contract.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 10-2 of 10-2

#### 10.5 CLIENT SURVEYS

The laboratory assesses both positive and negative client feedback. The results are used to improve overall laboratory quality and client service.

TestAmerica's Sales and Marketing teams periodically develop lab and client specific surveys to assess client satisfaction.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 11-1 of 11-2

#### **SECTION 11**

# COMPLAINTS (NELAC 5.4.8)

#### 11.1 <u>OVERVIEW</u>

**TestAmerica Buffalo** believes that effective client complaint handling processes have important business and strategic value. Listening to and documenting client concerns captures 'client knowledge' that helps to continually improve processes and improving client satisfaction. An effective client complaint handling process also provides assurance to the data user that the laboratory will stand behind its data, service obligations and products.

A client complaint is any expression of dissatisfaction with any aspect of our business services, communications, responsiveness, data, reports, invoicing and other functions expressed by any party, whether received verbally or in written form. Client inquiries, complaints or noted discrepancies are documented, communicated to management, and addressed promptly and thoroughly.

The laboratory has procedures for dealing with both external and internal complaints.

The nature of the complaint is identified, documented and investigated, and an appropriate action is determined and taken. In cases where a client complaint indicates that an established policy or procedure was not followed, the QA Department must evaluate whether a special audit must be conducted to assist in resolving the issue. A written confirmation or letter to the client, outlining the issue and response taken is recommended as part of the overall action taken.

The process of complaint resolution and documentation utilizes the procedures outlined in Section 13 (Corrective Actions) and is documented following the laboratory SOPs related to Data Quality Review (BF-QA-006), Corrective Action (BF-QA-005) and also using the Webbased Customer Complaint database. It is the laboratory's goal to provide a satisfactory resolution to complaints in a timely and professional manner.

#### 11.2 EXTERNAL COMPLAINTS

An employee that receives a complaint initiates the complaint resolution process and the documentation of the complaint.

Complaints fall into two categories: correctable and non-correctable. An example of a correctable complaint would be one where a report re-issue would resolve the complaint. An example of a non-correctable complaint would be one where a client complains that their data was repeatedly late. Non-correctable complaints should be reviewed for preventive action measures to reduce the likely hood of future occurrence and mitigation of client impact.

The general steps in the complaint handling process are:

- Receiving Complaints
- Complaint Investigation and Service Recovery
- Process Improvement

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 11-2 of 11-2

The laboratory shall inform the initiator of the complaint of the results of the investigation and the corrective action taken, if any.

# 11.3 INTERNAL COMPLAINTS

Internal complaints include, but are not limited to: errors and non-conformances, training issues, internal audit findings, and deviations from methods. Corrective actions may be initiated by any staff member who observes a nonconformance and shall follow the procedures outlined in Section 13. In addition, Corporate Management, Sales and Marketing and Information Technology (IT) may initiate a complaint by contacting the laboratory or through the corrective action system described in Section 13.

# 11.4 MANAGEMENT REVIEW

The number and nature of client complaints is reported by the QA Manager to the laboratory and QA Director in the QA Monthly report. Monitoring and addressing the overall level and nature of client complaints and the effectiveness of the solutions is part of the Annual Management Review (Section 17)

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 12-1 of 12-3

#### **SECTION 12**

# CONTROL OF NON-CONFORMING WORK (NELAC 5.4.9)

#### 12.1 <u>OVERVIEW</u>

When data discrepancies are discovered or deviations and departures from laboratory standard procedures, policies and/or client requests have occurred, corrective action is taken immediately. First, the laboratory evaluates the significance of the nonconforming work. Then, a corrective action plan is initiated based on the outcome of the evaluation. If it is determined that the nonconforming work is an isolated incident, the plan could be as simple as adding a qualifier to the final results and/or making a notation in the case narrative. If it is determined that the nonconforming work is a systematic or improper practices issue, the corrective action plan could include a more in depth investigation and a possible suspension of an analytical method. In all cases, the actions taken are documented using the laboratory's corrective action system (refer to Section 13).

Due to the frequently unique nature of environmental samples, sometimes departures from documented policies and procedures are needed. When an analyst encounters such a situation, the problem is presented to the department manager for advice. The department manager may elect to discuss it with the Technical Director, QA Manager or have a representative contact the client to decide on a logical course of action. Once an approach is agreed upon, the analyst documents it using the laboratory's job exception and corrective action system described in Section 13. This information can then be supplied to the client in the form of a footnote or a case narrative with the report.

Project Management may encounter situations where a client may request that a special procedure be applied to a sample that is not standard lab practice. Based on a technical evaluation, the lab may accept or opt to reject the request based on technical or ethical merit. An example might be the need to report a compound that the lab does not normally report. The lab would not have validated the method for this compound following the procedures in Section 20. The client may request that the compound be reported based only on the calibration. Such a request would need to be approved by the Laboratory Director, Technical Director, Operations Manager or QA Manager, documented and included in the project folder. Deviations must also be noted on the final report with a statement that the compound is not reported in compliance with the analytical method requirements and the reason.

# 12.2 **RESPONSIBILITIES AND AUTHORITIES**

SOP No. CA-L-S-001, Internal Investigation of Potential Data Discrepancies and Determination for Data Recall, outlines the general procedures for the reporting and investigation of data discrepancies and alleged incidents of misconduct or violations of the company's data integrity policies as well as the policies and procedures related to the determination of the potential need to recall data.

Under certain circumstances the Laboratory Director, the Technical Director, the Operations Manager or the QA Manager may exceptionally authorize departures from documented procedures or policies. The departures may be a result of procedural changes due to the nature

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 12-2 of 12-3

of the sample; a one-time procedure for a client; QC failures with insufficient sample to reanalyze, etc.. In most cases, the client will be informed of the departure prior to the reporting of the data. Any departures must be well documented using the laboratory's job exception and corrective action procedures described in Section 13. This information may also need to be documented in logbooks and/or data review checklists as appropriate. Any impacted data must be referenced in a case narrative and/or flagged with an appropriate data qualifier.

Any misrepresentation or possible misrepresentation of analytical data discovered by any laboratory staff member must be reported to facility senior laboratory management within 24-hours. The Senior Management staff is comprised of the Laboratory Director, Technical Director, Operations Manager, QA Manager, Customer Service Manager, Human Resources Manager and Business Development Manager. Suspected misrepresentation issues may also be reported to any member of the Corporate staff as identified in Ethics Policy, CA-L-P-001. The data integrity hotline (1-800-736-9407) may also be used. The reporting of issues involving alleged violations of the company's Data Integrity or Manual Integration procedures <u>must</u> be conveyed to an Ethics and Compliance Officer (ECO) and Quality Director within 24 hours.

Whether an inaccurate result was reported due to calculation or quantitation errors, data entry errors, improper practices, or failure to follow SOPs, the data must be evaluated to determine the possible effect.

The Laboratory Director, QA Manager, ECOs, COO's – East and West, General Managers and the Quality Directors – East and West have the authority and responsibility to halt work, withhold final reports, or suspend an analysis for due cause as well as authorize the resumption of work.

# 12.3 EVALUATION OF SIGNIFICANCE AND ACTIONS TAKEN

For each nonconforming issue reported, an evaluation of its significance and the level of management involvement needed is made. This includes reviewing its impact on the final data, whether or not it is an isolated or systematic issue, and how it relates to any special client requirements.

SOP No. CA-L-S-001 distinguishes between situations when it would be appropriate for the laboratory QA Manager and Laboratory Director (or his/her designee) to make the decision on the need for client notification (written or verbal) and data recall (report revision) and when the decision must be made with the assistance of the ECO's and Corporate Management. Laboratory level decisions are documented and approved using the laboratory's standard nonconformance/corrective action reporting (Section 13) in lieu of the data recall determination form contained in SOP No. CA-L-S-001.

# 12.4 PREVENTION OF NONCONFORMING WORK

If it is determined that the nonconforming work could recur, further corrective actions must be made following the laboratory's corrective action system (Section 13).

On a monthly basis, the QA Department evaluates non-conformances to determine if any nonconforming work has been repeated multiple times. If so, the laboratory's corrective action process may be followed.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 12-3 of 12-3

#### 12.5 METHOD SUSPENSION/RESTRICTION (STOP WORK PROCEDURES)

In some cases it may be necessary to suspend/restrict the use of a method or target compound which constitutes significant risk and/or liability to the laboratory. Suspension/restriction procedures can be initiated by any of the persons noted in Section 12.2, Paragraph 3 above.

Prior to suspension/restriction, confidentiality will be respected, and the problem and the required corrective and preventive action will be stated in writing and presented to the Laboratory Director.

The Laboratory Director shall arrange for the appropriate personnel to meet with the QA Manager as needed. This meeting shall be held to confirm that there is a problem, that suspension/restriction of the method is required and will be concluded with a discussion of the steps necessary to bring the method/target or test fully back on line. In some cases that may not be necessary if all appropriate personnel have already agreed there is a problem and there is agreement on the steps needed to bring the method, target or test fully back on line.

The QA Manager will also initiate a corrective action report as described in Section 13 if one has not already been started. A copy of any meeting notes and agreed upon steps should be faxed or e-mailed by the laboratory to the appropriate General Manager and member of Corporate QA. This fax/e-mail acts as notification of the incident.

After suspension/restriction, the lab will hold all reports to clients pending review. No faxing, mailing or distributing through electronic means may occur. The report must not be posted for viewing on the internet. It is the responsibility of the Laboratory Director to hold all reporting and to notify all relevant laboratory personnel regarding the suspension/restriction (i.e., Project Management, Log-in, etc...). Clients will NOT generally be notified at this time. Analysis may proceed in some instances depending on the non-conformance issue.

Within 72 hours, the QA Manager will determine if compliance is now met and reports can be released, OR determine the plan of action to bring work into compliance, and release work. A team, with all principals involved (Laboratory Director, Technical Director, Operations Manager, QA Manager, Department Manager) can devise a start-up plan to cover all steps from client notification through compliance and release of reports. Project Management, the Customer Service Manager and Sales and Marketing should be notified if clients must be notified or if the suspension/restriction affects the laboratory's ability to accept work. The QA Manager must approve start-up or elimination of any restrictions after all corrective action is complete. This approval is given by final signature on the completed corrective action report as described in Section 13.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 13-1 of 13-8

#### **SECTION 13**

# CORRECTIVE ACTION (NELAC 5.4.10)

#### 13.1 <u>OVERVIEW</u>

A major component of TestAmerica's Quality Assurance (QA) Program is the problem investigation and feedback mechanism designed to keep the laboratory staff informed on quality related issues and to provide insight to problem resolution. When nonconforming work or departures from policies and procedures in the quality system or technical operations are identified, the corrective action procedure provides a systematic approach to assess the issues, restore the laboratory's system integrity, and prevent reoccurrence. Corrective actions are documented using Job Exception Reports (JER) and Corrective Action Reports (CAR) (refer to Figure 13-1).

# 13.2 **DEFINITIONS**

- Correction: Actions necessary to correct or repair analysis specific non-conformances. The acceptance criteria for method specific QC and protocols as well as the associated corrective actions are contained in the method specific SOPs. The analyst will most frequently be the one to identify the need for this action as a result of calibration checks and QC sample analysis. No significant action is taken to change behavior, process or procedure.
- **Corrective Action**: The action taken is not only a correction made to the immediate event, but a change in process, procedure or behavior that is required to eliminate the causes of an existing nonconformity, defect, or other undesirable situation in order to prevent recurrence.

# 13.3 <u>GENERAL</u>

Problems within the quality system or within analytical operations may be discovered in a variety of ways, such as QC sample failures, internal or external audits, proficiency testing (PT) performance, client complaints, staff observation, etc..

The purpose of a corrective action system is to:

- Identify non-conformance events and assign responsibility for investigation.
- Resolve non-conformance events and assign responsibility for any required corrective action.
- Identify Systematic Problems before they become serious.
- Identify and track Client complaints and provide resolution (see more on client complaints in Section 11).

**13.3.1** Job Exception Report (JER) - is used to document the following types of corrective actions:

- Deviations from an established procedure or SOP
- QC outside of limits (non matrix related)

• Isolated Reporting / Calculation Errors

**13.3.2 Data Quality Review (DQR)** - is used to document the following types of corrective actions:

- Client Concerns regarding analytical results
- Project Management concerns regarding specific analytical results

**13.3.3** <u>Corrective Action Report (CAR)</u> - is used to document the following types of corrective actions:

- Questionable trends that are found in the monthly review of JERs.
- Issues found while reviewing JERs that warrant further investigation.
- Questionable trends that are found in the monthly review of DQRs or client complaints
- Internal and External Audit Findings Failed or Unacceptable PT results.
- Corrective actions that cross multiple departments in the laboratory.
- Systematic Reporting / Calculation Errors

# 13.4 CLOSED LOOP CORRECTIVE ACTION PROCESS

Any employee in the company can initiate a corrective action. There are four main components to a closed-loop corrective action process once an issue has been identified: Cause Analysis, Selection and Implementation of Corrective Actions (both short and long term), Monitoring of the Corrective Actions, and Follow-up.

#### 13.4.1 <u>Cause Analysis</u>

- Upon discovery of a non-conformance event, the event must be defined and documented. A JER or CAR must be initiated, someone is assigned to investigate the issue and the event is investigated for cause. Table 13-1 provides some general guidelines on determining responsibility for assessment.
- The cause analysis step is the key to the process as a long term corrective action cannot be determined until the cause is determined.
- If the cause is not readily obvious, the Department Manager, Operations Manager, Technical Director, or QA Manager (or QA designee) is consulted.

#### 13.4.2 Selection and Implementation of Corrective Actions

- Where corrective action is needed, the laboratory shall identify potential corrective actions. The action(s) most likely to eliminate the problem and prevent recurrence are selected and implemented. Responsibility for implementation is assigned.
- Corrective actions shall be to a degree appropriate to the magnitude of the problem identified through the cause analysis.
- Whatever corrective action is determined to be appropriate, the laboratory shall document and implement the changes. The JER or CAR is used for this documentation.

#### 13.4.3 <u>Monitoring of the Corrective Actions</u>

- The Department Manager, Operations Manager and QA Manager are responsible to ensure that the corrective action taken was effective.
- Ineffective actions will be documented and re-evaluated until acceptable resolution is achieved. Department Managers and the Operations Manager are accountable to the Laboratory Director to ensure final acceptable resolution is achieved and documented appropriately.
- Each JER and DQR are entered into a database and each CAR is entered into a spreadsheet for tracking purposes and a monthly summary of all corrective actions is printed out for review to aid in ensuring that the corrective actions have taken effect.
- The QA Manager reviews monthly JERs, DQRs and CARs for trends. Highlights are included in the QA monthly report (refer to Section 17). If a significant trend develops that adversely affects quality, an audit of the area is performed and corrective action implemented.
- Any out-of-control situations that are not addressed acceptably at the laboratory level may be reported to the Corporate Quality Director by the QA Manager, indicating the nature of the outof-control situation and problems encountered in solving the situation.

# 13.4.4 Follow-up Audits

- Follow-up audits may be initiated by the QA Manager and shall be performed as soon as possible when the identification of a nonconformance casts doubt on the laboratory's compliance with its own policies and procedures, or on its compliance with state or federal requirements. (Section 16 includes additional information regarding internal audit procedures.)
- These audits often follow the implementation of the corrective actions to verify effectiveness. An additional audit would only be necessary when a critical issue or risk to business is discovered.

# 13.5 TECHNICAL CORRECTIVE ACTIONS

In addition to providing acceptance criteria and specific protocols for technical corrective actions in the method SOPs the laboratory has general procedures to be followed to determine when departures from the documented policies and procedures and quality control have occurred (refer to Section 12 for information regarding the control of non-conforming work). The documentation of these procedures is through the use of a JER or CAR.

Table 13-1 includes examples of general technical corrective actions. For specific criteria and corrective actions refer to the analytical methods or specific method SOPs.

Table 13-1 provides some general guidelines for identifying the individual(s) responsible for assessing each QC type and initiating corrective action. The table also provides general guidance on how a data set should be treated if associated QC measurements are unacceptable. Specific procedures are included in Method SOPs, QAM Sections 20 and 21, and SOP CA-L-S-001 (Internal Investigation of Potential Data Discrepancies and Determination for Data Recall). All corrective actions are reviewed at a minimum monthly by the QA Manager and highlights are included in the QA monthly report.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 13-4 of 13-8

To the extent possible, samples shall be reported only if all quality control measures are acceptable. If the deficiency does not impair the usability of the results, data will be reported with an appropriate data qualifier and/or the deficiency will be noted in the case narrative. Where sample results may be impaired, the Project Manager is notified by a written JER and appropriate corrective action (e.g., reanalysis) is taken and documented.

#### 13.6 BASIC CORRECTIONS

When mistakes occur in records, each mistake shall be crossed-out, and not erased, deleted, made illegible, or otherwise obliterated (e.g. no white-out), and the correct value entered alongside. All such corrections shall be initialed (or signed) and dated by the person making the correction. In the case of records stored electronically, the original "uncorrected" file must be maintained intact and a second "corrected" file is created.

This same process applies to adding additional information to a record. All additions made later than the initial must also be initialed (or signed) and dated.

When corrections are due to reasons other than obvious transcription errors, the reason for the corrections (or additions) shall also be documented.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 13-5 of 13-8

# Figure 13-1. Example – Corrective Action Notice



THE LEADER IN ENVIRONMENTAL TESTING

CAN Statement Revision 2 August 5, 2005

#### **CORRECTIVE ACTION NOTICE**

Date Issued:	Issued By:	
Date Required:	Responsible Party:	
Source of Issue:		
Explanation of Issue:		
Investigation Summary:		
Root Cause:		
Impact on Client Data:		
Corrective Action or Resolution:		
Timetable for Action:	Antion	
interns to Document Corrective A	<b></b>	
Completed By:		Date:
Approved By:		Date:
Follow-Up Comments:		
Follow-Up By:		Date:

# Table 13-1.

# **Example – General Corrective Action Procedures**

QC Activity (Individual Responsible for Initiation/Assessment)	Acceptance Criteria	Recommended Corrective Action
Initial Instrument Blank	- Instrument response < MDL.	<ul> <li>Prepare another blank.</li> <li>If same response, determine cause of contamination: reagents, environment, instrument equipment failure. etc.</li> </ul>
(Analysi)		instrument equipment landre, etc
Initial Calibration Standards (Analyst, Department Manager)	<ul> <li>Correlation coefficient &gt; 0.99 or standard concentration value.</li> <li>% Recovery within acceptance range.</li> <li>See details in Method SOP. (or refer to Appendix 4 or Section 21)</li> </ul>	- Reanalyze standards. - If still unacceptable, remake standards and recalibrate instrument.
Independent Calibration Verification (Second Source) (Analyst, Department Manager)	- % Recovery within control limits.	<ul> <li>Remake and reanalyze standard.</li> <li>If still unacceptable, then remake calibration standards or use new primary standards and recalibrate instrument.</li> </ul>
Continuing Calibration Standards (Analyst, Data Reviewer)	% Recovery within control limits.	<ul> <li>Reanalyze standard.</li> <li>If still unacceptable, then recalibrate and rerun affected samples.</li> </ul>
Matrix Spike / Matrix Spike Duplicate (MS/MSD) (Analyst, Data Reviewer)	- % Recovery within limits documented in LIMs.	<ul> <li>If the acceptance criteria for duplicates or matrix spikes are not met because of matrix interferences, the acceptance of the analytical batch is determined by the validity of the LCS.</li> <li>If the LCS is within acceptable limits the batch is acceptable.</li> <li>The results of the duplicates, matrix spikes and the LCS are reported with the data set.</li> </ul>
Laboratory Control Sample (LCS) (Analyst, Data Reviewer)	- % Recovery within limits specified in LIMs.	<ul> <li>Batch must be re-prepared and re- analyzed.</li> <li>Note: If there is insufficient sample or the holding time cannot be met, contact client and report with flags.</li> </ul>

QC Activity (Individual Responsible for Initiation/Assessment)	Acceptance Criteria	Recommended Corrective Action	
Surrogates (Analyst, Data Reviewer)	<ul> <li>% Recovery within limits of method or within three standard deviations of the historical mean.</li> </ul>	- Individual sample must be repeated. Place comment in LIMS.	
Method Blank (MB) (Analyst, Data Reviewer)	< Reporting Limit ¹	<ul> <li>Reanalyze blank.</li> <li>If still positive, determine source of contamination. If necessary, reprocess (i.e. digest or extract) entire sample batch. Report blank results.</li> </ul>	
Proficiency Testing (PT) Samples (QA Manager, Department Manager)	- Criteria supplied by PT Supplier.	- Any failures or warnings must be investigated for cause. Failures may result in the need to repeat a PT sample to show the problem is corrected.	
Internal / External Audits (QA Manager, Department Manager, Operations Manager, Technical Director, Laboratory Director)	- Defined in Quality System documentation such as SOPs, QAM, etc	- Non-conformances must be investigated through CAR system and necessary corrections must be made.	
Reporting / Calculation Errors (Depends on issue – possible individuals include: Analysts, Data Reviewers, Project Managers, Department Manager, QA Manager, Corporate QA, Corporate Management)	- SOP CA-L-S-001, Internal Investigation of Potential Data Discrepancies and Determination for Data Recall.	- Corrective action is determined by type of error. Follow the procedures in SOP CA-L-S-001.	
Client Complaints (Project Managers, Lab Director, Sales and Marketing, QA Manager)	-	- Corrective action is determined by the type of complaint. For example, a complaint regarding an incorrect address on a report will result in the report being corrected and then follow- up must be performed on the reasons the address was incorrect (e.g., database needs to be updated).	

-

QC Activity (Individual Responsible for Initiation/Assessment)	Acceptance Criteria	Recommended Corrective Action
QA Monthly Report (Refer to Section 17 for an example) (QA Manager, Lab Director, Operations Manager Department Managers)	- QAM, SOPs.	- Corrective action is determined by the type of issue. For example, CARs for the month are reviewed and possible trends are investigated.
Health and Safety Violation (EH&S Coordinator, Lab Director, Operations Manager, Department Manager)	- Environmental Health and Safety (EHS) Manual.	- Non-conformance is investigated and corrected through EH&S office.

#### Note:

1. Except as noted below for certain compounds, the method blank should be below the reporting limit. Concentrations up to five times the reporting limit will be allowed for the ubiquitous laboratory and reagent contaminants: methylene chloride, acetone, 2-butanone and phthalates. This allowance presumes that the reporting limit is significantly below any regulatory limit to which the data are to be compared and that blank subtraction will not occur.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 14-1 of 14-2

#### **SECTION 14.0**

#### PREVENTIVE ACTION (NELAC 5.4.11)

#### 14.1 <u>OVERVIEW</u>

The laboratory's preventive action programs improve, or eliminate potential causes of nonconforming product and/or nonconformance to the quality system. This preventive action process is a proactive continuous process improvement activity that can be initiated through feedback from clients, employees, business providers, and affiliates. The QA Department has the overall responsibility to ensure that the preventive action process is in place, and that relevant information on actions is submitted for management review.

Dedicating resources to an effective preventive action system emphasizes **TestAmerica Buffalo**'s commitment to its Quality Assurance (QA) program. It is beneficial to identify and address negative trends before they develop into complaints, problems and corrective actions. Additionally, customer service and satisfaction can be improved through continuous improvements to laboratory systems.

Opportunities for improvement may be discovered during management reviews, the QA Metrics Report, internal or external audits, proficiency testing performance, client complaints, staff observation, etc..

The monthly Quality Assurance Metrics Report shows performance indicators in all areas of the quality system. These areas include revised reports, corrective actions, audit findings, internal auditing and data authenticity audits, client complaints, PT samples, holding time violations, SOPs, ethics training, etc. These metrics are used to help evaluate quality system performance on an ongoing basis and provide a tool for identifying areas for improvement.

The laboratory's Corrective Action process (Section 13) is integral to implementation of preventive actions. A critical piece of the corrective action process is the implementation of actions to prevent further occurrence of a non-compliance event. Historical review of corrective action provides a valuable mechanism for identifying preventive action opportunities.

**14.1.1** The following elements are part of a preventive action system:

- <u>Identification</u> of an opportunity for preventive action.
- <u>Process</u> for the preventive action.
- <u>Define the measurements of the effectiveness of the process once undertaken.</u>
- Execution of the preventive action.
- Evaluation of the plan using the defined measurements.
- <u>Verification</u> of the effectiveness of the preventive action.
- <u>Close-Out</u> by documenting any permanent changes to the Quality System as a result of the Preventive Action. Documentation of Preventive Action is incorporated into the monthly QA reports, corrective action process and management review

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 14-2 of 14-2

• Note: There may be varying levels of formality and documentation during the preventive action process due to the simplicity/complexity of the action taken.

**14.1.2** Any Preventive Actions undertaken or attempted shall be taken into account during the Annual Management Review (Section 17). A highly detailed recap is not required; a simple recount of success and failure within the preventive action program will provide management a measure for evaluation.

# 14.2 MANAGEMENT OF CHANGE

**Reserved**

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 15-1 of 15-7

#### **SECTION 15.0**

# CONTROL OF RECORDS (NELAC 5.4.12)

**TestAmerica Buffalo** maintains a record system appropriate to its needs and that complies with applicable standards or regulations as required. The system produces unequivocal, accurate records that document all laboratory activities. The laboratory retains all original observations, calculations and derived data, calibration records and a copy of the analytical report for a minimum of five years after it has been issued. TestAmerica Buffalo SOP BF-GP-015, Record Storage and Retention specifies additional storage, archiving and retention procedures.

#### 15.1 <u>OVERVIEW</u>

The laboratory has established procedures for identification, collection, indexing, access, filing, storage, maintenance and disposal of quality and technical records. A record index is listed in Table 15-1. Quality records are maintained by the Quality Assurance (QA) Manager in both hardcopy and electronic form. Hardcopy records are retained in the QA archive room and electronic records are retained in network folders which are backed up as part of the regular network backup. Records are of two types; either electronic or hard copy paper formats depending on whether the record is computer or hand generated (some records may be in both formats). Hardcopy technical records are maintained by the Data Deliverables Manager while electronic technical records are maintained by the IT Adminstrator.

Technical Records	Official Documents	QA Records	Project Records	Administrative Records
Retention: 5 Years from analytical report issue*	5 Years from document retirement date*	5 Years from archival* Data Investigation: 5years or the life of the affected raw data storage whichever is greater (beyond 5 years if ongoing project or pending investigation)	5 Years from analytical report issue*	Personnel: 7 Years (HR Records must be maintained as per Policy CW-L-P-001) Finance: See Accounting and Control Procedures Manual
Raw Data	Quality	Internal and External	Sample receipt and	Finance and Accounting
Logbooks ²	Manual (QAM)	Addits/ Responses	Documentation	
Standards	Work Instructions	Certifications	Contracts and Amendments	EH&S Manual, Permits, Disposal Records
Certificates	SOPs	Corrective/Preventive Action	Correspondence	Employee Handbook
Analytical	Manuals	Management Reviews	QAPP	Personnel files,

#### Table 15-1. Record Index¹

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 15-2 of 15-7

Technical Records	Official Documents	QA Records	Project Records	Administrative Records
Retention: 5 Years from analytical report issue*	5 Years from document retirement date*	5 Years from archival* Data Investigation: 5years or the life of the affected raw data storage whichever is greater (beyond 5 years if ongoing project or pending investigation)	5 Years from analytical report issue*	Personnel: 7 Years (HR Records must be maintained as per Policy CW-L-P-001) Finance: See Accounting and Control Procedures Manual
Records Lab Reports		Method & Software Validation, Verification data	SAP	Employee Signature & Initials, Administrative Training Records (e.g., Ethics)
		Data Investigation	Telephone Logbooks	Administrative Policies
	Policies	Performance Evaluations	Lab Reports	Technical Training Records

¹ Record Types encompass hardcopy and electronic records.

² Examples of Logbook types: Maintenance, Instrument Run, Preparation (standard and samples), Standard and Reagent Receipt, Archiving, Balance Calibration, Temperature (hardcopy or electronic records).

* Exceptions listed in Table 15-2.

All records are legible and stored and retained in such a way that they are secure and readily retrievable at the laboratory facility or an offsite location that provides a suitable environment to prevent damage or deterioration and to prevent loss. Retention of records is maintained on-site at the laboratory for at least 3 months after their generation and moved offsite for the remainder of the required storage time. Records are maintained for a minimum of five years unless other wise specified by a client or regulatory requirement.

For raw data and project records, record retention shall be calculated from the date the project report is issued. For other records, such as Controlled Documents, QA, or Administrative Records, the retention time is calculated from the date the record is formally retired. Records related to the programs listed in Table 15-2 have lengthier retention requirements and are subject to the requirements in Section 15.1.3. Policy CW-L-P-001 (Record Retention) provides additional information on record retention requirements.

#### 15.1.1 Programs with Longer Retention Requirements

Some regulatory programs have longer record retention requirements than the standard record retention time. These are detailed in Table 15-2 with their retention requirements. In these cases, the longer retention requirement is enacted. If special instructions exist such that client data cannot be destroyed prior to notification of the client, the container or box containing that data is marked as to who to contact for authorization prior to destroying the data. Specific Information related to archival of data for greater than 5 years may be found in TestAmerica Buffalo SOP BF-GP-015.

Program	¹ Retention Requirement
Drinking Water – All States	10 years (project records)
Drinking Water Lead and Copper Rule	12 years (project records)
Commonwealth of MA – All environmental data 310 CMR 42.14	10 years
FIFRA – 40 CFR Part 160	Retain for life of research or marketing permit for pesticides regulated by EPA
Housing and Urban Development (HUD) Environmental Lead Testing	10 years
Alaska	10 years
Louisiana – All	10 years
Michigan Department of Environmental Quality – all environmental data	10 years
Navy Facilities Engineering Service Center (NFESC)	5 years
NY Potable Water NYCRR Part 55-2	10 years
TSCA - 40 CFR Part 792	10 years after publication of final test rule or negotiated test agreement

# Table 15-2. Special Record Retention Requirements

¹Note: Extended retention requirements are noted with the archive documents or addressed in TestAmerica Buffalo facility-specific records retention procedure BF-GP-015.

**15.1.2** All records are held secure and in confidence. Records maintained at the laboratory are located in the locked on-site storage room. Records archived off-site are stored in a secure location. Access to the off-site storage facility is controlled and logs are maintained for the documented removal/return of records

**15.1.3** The laboratory has procedures to protect and back-up records stored electronically and to prevent unauthorized access to or amendment of these records. All analytical data is maintained as hard copy or in a secure readable electronic format. For analytical reports that are maintained as copies in PDF format, see section 20.12.1 'Computer and Electronic Data Related Requirements' for more information. TestAmerica Buffalo SOP BF-GP-015 also contains specific information for archival of scanned data.

**15.1.4** The record keeping system allows for historical reconstruction of all laboratory activities that produced the analytical data, as well as rapid recovery of historical data (records stored off site should be accessible within 2 business days of a request for such records). The history of the sample from when the laboratory took possession of the samples must be readily understood through the documentation. This shall include inter-laboratory transfers of samples and/or extracts.

• The records include the identity of personnel involved in sampling, sample receipt, preparation, or testing. All analytical work contains the initials (at least) of the personnel involved. The laboratory's copy of the chain of custody is stored with the project file and the

Job Number analytical service request form (ASRF) generated by the LIMS. The chain of custody would indicate the name of the sampler. If any sampling notes are provided with a Job Number, they are kept with this package.

- All information relating to the laboratory facilities equipment, analytical test methods, and related laboratory activities, such as sample receipt, sample preparation, or data verification are documented.
- The record keeping system facilitates the retrieval of all working files and archived records for inspection and verification purposes (e.g., set format for naming electronic files, set format for what is included with a given analytical data set. Instrument data is stored sequentially by instrument. Calibration data for a given sequence are maintained in the order of the analysis. Sample data are stored on a job number basis in the project file or as part of the daily batch or sequence. Run logs are maintained for each instrument or method; a copy of each day's run log or instrument sequence is stored with the data to aid in reconstructing an analytical sequence. Where an analysis is performed without an instrument, bound logbooks, bench sheets or excel spreadsheets are used to record and file data. Standard and reagent information is recorded in logbooks or on the raw data for each method as required.
- Changes to hardcopy records shall follow the procedures outlined in Section 13 and 20. Changes to electronic records in LIMS or instrument data are recorded in audit trails.
- The reason for a signature or initials on a document is clearly indicated in the records such as "sampled by," "prepared by," "reviewed by", or "analyzed by".
- All generated data except those that are generated by automated data collection systems, are recorded directly, promptly and legibly in permanent dark ink.
- Hard copy data may be scanned into PDF format for record storage as long as the scanning
  process can be verified in order to ensure that no data is lost and the data files and storage
  media must be tested to verify the laboratory's ability to retrieve the information prior to the
  destruction of the hard copy that was scanned. The procedure for this verification can be
  found in TestAmerica SOP BF-GP-015.
- Also refer to Section 20.13.1 'Computer and Electronic Data Related Requirements'.

# 15.2 TECHNICAL AND ANALYTICAL RECORDS

**15.2.1** The laboratory retains records of original observations, derived data and sufficient information to establish an audit trail, calibration records, staff records and a copy of each analytical report issued, for a minimum of five years unless otherwise specified by a client or regulatory requirement (refer to Section 15.1). The records for each analysis shall contain sufficient information to enable the analysis to be repeated under conditions as close as possible to the original. The records shall include the identity of laboratory personnel responsible for the sampling, performance of each analysis and checking of results.

**15.2.2** Observations, data and calculations are recorded at the time they are made and are identifiable to the specific task.

**15.2.3** Changes to hardcopy records shall follow the procedures outlined in Section 13 and 20. Changes to electronic records in LIMS or instrument data are recorded in audit trails. The essential information to be associated with analysis, such as strip charts, tabular printouts, computer data files, analytical notebooks, and run logs, include:

- laboratory sample ID code;
- Date of analysis and time of analysis is required if the holding time is seventy-two (72) hours
  or less, or when time critical steps are included in the analysis (e.g., drying times,
  incubations, etc.); instrumental analyses have the date and time of analysis recorded as part
  of their general operations. Where a time critical step exists in an analysis, location for such
  a time is included as part of the documentation in a specific logbook or on a benchsheet.
- Instrumentation identification and instrument operating conditions/parameters. Operating
  conditions/parameters are typically recorded in the method specific SOPs, in the instrument
  method detail records or the instrument maintenance logs where available.
- analysis type;
- all manual calculations and manual integrations;
- analyst's or operator's initials/signature;
- sample preparation including cleanup, separation protocols, incubation periods, ID codes, volumes, weights, instrument printouts, meter readings, temperatures, calculations, reagents;
- test results;
- standard and reagent origin, receipt, preparation, and use;
- calibration criteria, frequency and acceptance criteria;
- data and statistical calculations, review, confirmation, interpretation, assessment and reporting conventions;
- quality control protocols and assessment;
- electronic data security, software documentation and verification, software and hardware audits, backups, and records of any changes to automated data entries; and
- Method performance criteria including expected quality control requirements. These are indicated both in the LIMS and on specific analytical report formats.
Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 15-6 of 15-7

## 15.3 LABORATORY SUPPORT ACTIVITIES

In addition to documenting all the above-mentioned activities, the following are retained QA records and project records (previous discussions in this section relate where and how these data are stored):

- all original raw data, whether hard copy or electronic, for calibrations, samples and quality control measures, including analysts' work sheets and data output records (chromatograms, strip charts, and other instrument response readout records);
- a written description or reference to the specific test method used which includes a
  description of the specific computational steps used to translate parametric observations into
  a reportable analytical value;
- copies of final reports;
- archived SOPs;
- correspondence relating to laboratory activities for a specific project;
- all corrective action reports, audits and audit responses;
- proficiency test results and raw data; and
- results of data review, verification, and crosschecking procedures

## 15.3.1 Sample Handling Records

Sample handling and tracking is discussed in Section 24. Records of all procedures to which a sample is subjected while in the possession of the laboratory are maintained. These include but are not limited to records pertaining to:

- sample preservation including appropriateness of sample container and compliance with holding time requirement;
- sample identification, receipt, acceptance or rejection and login;
- sample storage and tracking including shipping receipts, sample transmittal / COC forms; and
- procedures for the receipt and retention of samples, including all provisions necessary to protect the integrity of samples.

## 15.4 ADMINISTRATIVE RECORDS

The laboratory also maintains the administrative records in either electronic or hard copy form. See Table 15-1.

## 15.5 RECORDS MANAGEMENT, STORAGE AND DISPOSAL

**15.5.1** All records (including those pertaining to test equipment), certificates and reports are safely stored, held secure and in confidence to the client. Certification related records are available to the accrediting body upon request.

**15.5.2** All information necessary for the historical reconstruction of data is maintained by the laboratory. Records that are stored only on electronic media must be supported by the hardware and software necessary for their retrieval.

**15.5.3** Records that are stored or generated by computers or personal computers have hard copy, write-protected backup copies, or an electronic audit trail controlling access.

**15.5.4 TestAmerica Buffalo** has a record management system for control of laboratory notebooks, instrument logbooks, standards logbooks, and records for data reduction, validation, storage and reporting. Laboratory notebooks are issued on a per instrument or analysis basis, and are numbered sequentially as they are issued. No instrument or analysis has more than one active notebook at a time, so all data are recorded sequentially within a series of sequential notebooks. Bench sheets and raw data sequence files are filed sequentially by date. Standard and reagent information is maintained in logbooks which are maintained on a departmental basis and are numbered sequentially as they are issued or as they are archived by QA.

**15.5.5** Records are considered archived when moved off-site. Access to archived hardcopy information is documented with an access log and in/out records is used to note data that is removed and returned. All records shall be protected against fire, theft, loss, environmental deterioration, and vermin. In the case of electronic records, electronic or magnetic sources, storage media are protected from deterioration caused by magnetic fields and/or electronic deterioration. Access to the data is limited to laboratory and company employees.

**15.5.6** In the event that the laboratory transfers ownership or goes out of business, **TestAmerica Buffalo** shall ensure that the records are maintained or transferred according to client's instructions. Upon ownership transfer, record retention requirements shall be addressed in the ownership transfer agreement and the responsibility for maintaining archives is clearly established. In addition, in cases of bankruptcy, appropriate regulatory and state legal requirements concerning laboratory records must be followed. In the event of the closure of the laboratory, all records will revert to the control of the corporate headquarters. Should the entire company cease to exist, as much notice as possible will be given to clients and the accrediting bodies who have worked with the laboratory during the previous 5 years of such action.

## 15.5.7 <u>Records Disposal</u>

- **15.5.7.1** Records are removed from the archive and disposed after 5 years unless otherwise specified by a client or regulatory requirement. On a project specific or program basis, clients may need to be notified prior to record destruction. Records are destroyed in a manner that ensures their confidentiality such as shredding, mutilation or incineration.
- **15.5.7.2** Electronic copies of records must be destroyed by erasure or physically damaging off-line storage media so no records can be read.
- **15.5.7.3** If a third party records management company is hired to dispose of records, a "Certificate of Destruction" is required. [Refer to Policy No. CW-L-P-001 (Records Retention).]

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 16-1 of 16-8

#### SECTION 16

## AUDITS (NELAC 5.4.13)

#### 16.1 <u>OVERVIEW</u>

Audits measure laboratory performance and insure compliance with accreditation/certification and project requirements. Audits specifically provide management with an on-going assessment of the quality of results produced by the laboratory, including how well the policies and procedures of the QA system and the Ethics and Data Integrity Program are being executed. They are also instrumental in identifying areas where improvement in the QA system will increase the reliability of data. There are two principle types of audits: Internal and External. Internal audits are performed by laboratory or corporate personnel. External audits are conducted by regulators, clients or third-party auditing firms. In either case, the assessment to program requirements is the focus.

Internal Audits	Description	Performed by	Frequency		
	Analyst & Method Compliance	QA Department or Designee	<ul> <li>100% of all methods over a two year period.</li> <li>100% of all analysts annually.</li> </ul>		
	Instrument	QA Department or Designee	100% of all organic instruments and any inorganic chromatography instruments over a two year period.		
	Work Order/ Final Report	QA Department or Designee	- 1 complete report each month.		
	Support Systems	QA Department or Designee	- Annual for entire labs support departments & equipment (e.g., thermometers, balances), can be divided into sub-sections over the course of the year.		
	Performance Audits (Double-Blind PTs)	Corporate QA, Laboratory QA Department or Designee	- As needed.		
	Special	QA Department or Designee	- As Needed		
External Audits	Description	Performed by	Frequency		
	Program / Method Compliance	Regulatory Agencies, Clients, accreditation organizations	<ul> <li>As required by program and/or clients needs</li> </ul>		
	Performance Audits	Provided by a third party.	<ul> <li>As required by a client or regulatory agency. Generally provided semi-annually through the analysis of PT samples.</li> </ul>		

#### Table 16-1. Audit Types and Frequency

## 16.2 INTERNAL AUDITS

Annually, the laboratory prepares a schedule of internal audits to be performed throughout the year. As previously stated, these audits verify and monitor that operations continue to comply with the requirements of the laboratory's QA Manual and the Corporate Ethics Program. A

#### Company Confidential & Proprietary

schedule of the internal audits is maintained by the QA Manager in the *Internal Audit Workbook*. An example can be found in Attachment 1.

It is the responsibility of the QA Manager to plan and organize audits in consideration of the laboratory work load and the department personnel schedules so that all pertinent personnel and operations are thoroughly reviewed. When designees (other than QA department personnel & approved by the QA Manager), perform audits, the QA Manager shall insure that these persons do not audit their own activities except when it can be demonstrated that an effective audit will be carried out. In general, the auditor:

- is neither the person responsible for the process being audited nor the immediate supervisor of the person responsible for the project/process.
- Is free of any conflicts of interest.
- Is free from bias and influences that could affect objectivity.

Laboratory personnel (e.g., supervisors and analysts) may assist with both method and support system audits as long as the items listed in the above paragraph are observed. These audits are conducted according to defined criteria listed in the checklists of the *Internal Audit Workbook*. These personnel must be approved by the QA Manager; and must complete the audit checklists in their entirety. This process introduces analyst experience and insight into the laboratory's auditing program.

The auditor must review the previous audit report and identify all items for verification of corrective actions. A primary focus will be dedicated to the ability of the laboratory to correct root-cause deficiencies and that the corrective action has been implemented and sustained as documented.

## 16.2.1 <u>Systems</u>

An annual systems audit is required to ensure compliance to analytical methods and SOPs, the laboratory's Data Integrity and Ethics Policies, NELAC quality systems and client and State requirements. This audit is performed in portions throughout the year through method, analyst, instrument, work order/final report and support system audits. Audits are documented and reported to management within 1 week of their performance. Systems audits cover all departments of the facility, both operational and support. When performed individually, the multiple audits are compiled into one systems audit package at the end of the year (*Internal Audit Workbook*). The systems audits may also be combined and performed as a single comprehensive internal systems audit during a 1-2 week time frame.

## 16.2.1.1 Method, Analyst, Instrument and Work Order/Final Report Audits

Procedures for the method compliance, analyst, instrument and work order/final report audits are incorporated by reference to SOP No. CA-Q-S-004, Method Compliance and Data Authenticity Audits. These audits are not mutually exclusive. For example, the performance of a method audit will also cover multiple analysts and instruments. The laboratory's goal is to review all analysts and instruments as described in SOP No. CA-Q-S-004. The laboratory will also audit all methods within a two year time period and audit a minimum of one Work Order/Final Report from receiving through reporting on a monthly basis.

## 16.2.1.2 Support Systems

Support system audits are performed to ensure that all departments & ancillary equipment are operating according to prescribed criteria. Support system audits include the review of both non-analytical and operational departments. Support equipment audits (e.g., metrology items) include the review of balance calibrations, weight calibrations; water quality testing, etc.. Non-analytical may include sample receiving and bottle preparation. These types of support audits ensure that the operations are being performed to support ethical data as well as ensuring the accuracy & precision of the utilized equipment.

These audits can be performed in portions throughout the year or in one scheduled session. However, the audit schedule must document that these aspects are reviewed annually. Many of the metrology systems are considered to be surveillance activities that can be monitored by QA personnel or delegated to specified department personnel. These surveillance activities are performed on a semi-annual basis unless issues warrant a greater frequency or previous audits continually showing no deficiencies allow the frequency to be reduced to once a year.

An example audit checklist can be found in Attachment 2. Instructions for reporting findings are included in the *Internal Audit Workbook*. In general, findings are reported to management within 1 week of the audit and a response is due from management within 30 days.

## 16.2.2 <u>Performance Audits</u>

Corporate QA may arrange for double blind PT studies to be performed in the laboratories. Results are given to Management and Corrective actions of any findings are coordinated at each facility by the QA Managers and Laboratory Directors/Managers. These studies are performed on an as needed basis. They may be performed when concerns are raised regarding the performance of a particular method in specific laboratories, periodically to evaluate methods that may not normally be covered in the external PT program or may be used in the process of developing best practices. The local QA Manager may also arrange for PT studies on an as needed basis. (Refer to Section 16.3.2 for additional information on Performance Audits.)

## 16.2.3 Special Audits

Special audits are conducted on an as needed basis, generally as a follow up to specific issues such as client complaints, corrective actions, PT results, data audits, system audits, validation comments, regulatory audits or suspected ethical improprieties. Special audits are focused on a specific issue, and report format, distribution, and timeframes are designed to address the nature of the issue.

## 16.3 EXTERNAL AUDITS

TestAmerica facilities are routinely audited by clients and external regulatory authorities. External audits are performed when certifying agencies or clients conduct on-site inspections or submit performance testing samples for analysis. It is TestAmerica's policy to cooperate fully with regulatory authorities and clients. The laboratory makes every effort to provide the auditors with access to personnel, documentation, and assistance. The laboratory department managers are responsible for providing corrective actions to the QA Manager who coordinates the response for any deficiencies discovered during an external audit. Audit responses are due

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 16-4 of 16-8

in the time allotted by the client or agency performing the audit. This time frame is generally 30 days.

**TestAmerica Buffalo** cooperates with clients and their representatives to monitor the laboratory's performance in relation to work performed for the client. The client may only view data and systems related directly to the client's work. All efforts are made to keep other client information confidential.

## 16.3.1 Confidential Business Information (CBI) Considerations

During on-site audits, on-site auditors may come into possession of information claimed as business confidential. A business confidentiality claim is defined as "a claim or allegation that business information is entitled to confidential treatment for reasons of business confidentiality or a request for a determination that such information is entitled to such treatment." When information is claimed as business confidential, the laboratory must place on (or attach to) the information at the time it is submitted to the auditor, a cover sheet, stamped or typed legend or other suitable form of notice, employing language such as "trade secret", "proprietary" or "company confidential". Confidential portions of documents otherwise non-confidential must be clearly identified. CBI may be purged of references to client identity by the responsible laboratory official at the time of removal from the laboratory. However, sample identifiers may not be obscured from the information. Additional information regarding CBI can be found in within the 2003 NELAC standards.

### 16.3.2 <u>Performance Audits</u>

The laboratory is involved in performance audits conducted semi-annually through the analysis of PT samples provided by a third party. The laboratory generally participates in the following types of PT studies:

Water Supply: 2 times per year Water Pollution: 2 times per year Soil: 2 times per year DMR-QA: once per year Client specific: as requested

- It is TestAmerica's policy that PT samples be treated as typical samples in the production process. Further, where PT samples present special or unique problems in the regular production process they may need to be treated differently, as would any special or unique request submitted by any client. The QA Manager must be consulted and in agreement with any decisions made to treat a PT sample differently due to some special circumstance.
- PTs generally do not have holding times associated with them. In the absence of any holding time requirement, it is recommended that the holding time begin when the PT sample is prepared according to the manufacturers instructions. Holding times should apply to full volume PT samples only if the provider gives a meaningful "sampling date". If this is not provided, it is recommended that the date/time of opening of the full volume sample be considered the beginning of holding time.

- Login will obtain the COC information from the documentation provided with the PTs with review by QA or other designated staff.
- Vials will be prepared as required in the instruction set provided with the samples. After preparation to full volume the sample may be spiked, digested, concentrated, etc., as would be done for any normal sample requiring similar analysis.
- PT samples will not undergo multiple preps, multiple runs, multiple methods (unless being used to evaluate multiple methods), multiple dilutions, UNLESS this is what would be done to a normal client sample (e.g. if a client requests, as PT clients do, that we split VOA coeluters, then dual analysis IS normal practice).
- The type, composition, concentration and frequency of quality control samples analyzed with the PT samples shall be the same as with routine environmental samples.
- Instructions may be included in the laboratory's SOPs for how low level samples are analyzed, including concentration of the sample or adjustment of the normality of titrant. When a PT sample falls below the range of the routine analytical method, the low-level procedure may be used.
- No special reviews shall be performed by operation and QA, UNLESS this is what would be done to a normal client sample. To the degree that special report forms or login procedures are required by the PT supplier, it is reasonable that the laboratory WOULD apply special review procedures, as would be done for any client requesting unusual reporting or login processes.
- Written responses to unacceptable PT results are required. In some cases it may be necessary for blind QC samples to be submitted to the laboratory to show a return to control.

## 16.4 AUDIT FINDINGS

Internal or External Audit findings should be documented using the corrective action process. (refer to Section 13). The laboratory is expected to prepare a response to audit findings within 30 days of receipt of an audit report unless the report specifies a different time frame. The response may include action plans that could not be completed within the 30 day timeframe. In these instances, a completion date must be set and agreed to by operations management and the QA Manager.

Responsibility for developing and implementing corrective actions to findings is the responsibility of the Department Manager where the finding originated. Findings that are not corrected by specified due dates are reported monthly to management in the QA monthly report.

If any audit finding casts doubt on the effectiveness of the operations or on the correctness or validity of the laboratory's test results, the laboratory shall take timely corrective action, and shall notify clients in writing if the investigations show that the laboratory results have been affected. Once corrective action is implemented, a follow-up audit is scheduled to ensure that the problem has been corrected.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 16-6 of 16-8

The procedures must be in accordance to SOP No. CA-L-S-001, Internal Investigations of Data Discrepancies and Determination of Data Recall.

Clients must be notified promptly in writing, of any event such as the identification of defective measuring or test equipment that casts doubt on the validity of results given in any test report or amendment to a test report. The investigation must begin within 24-hours of discovery of the problem and all efforts are made to notify the client within two weeks after the completion of the investigation.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 16-7 of 16-8

Figure 16-1.

## Example - Internal Audit Workbook

## Laboratory: TestAmerica Buffalo

Last Updated: 7/16/2007

## **Internal Audit Schedule 2008**

ltem No	Area Audited	Audit	Audit	Scheduled	Date	Date Closed	Tab No.	Comments
								and the second
1	Balances	System	6 mo				3	
2	Temperature Logs/ Thermometers	System	6 mo				4	
3	Sample Storage and Disposal	System	1 yr				5	
4	Maintenance Logs	System	6 mo				6	
5	Holding Blanks for Volatile Ref/Freezers (where required)	System	6 mo				7	
6	Lab Water Quality Testing	System	6 mo				8	
7	Sample Control (Log In)	System	1 yr				9	
8	Shipping Procedures	System	1 yr				10	
9	Computer Operations (LIMS)	System	1 yr				11	
10	SOP Distribution System	System	1 yr				12	
11	Archiving of Paper Records	System	1 yr				13	
12	Statistical Process Control	System	1 yr			r .	14	-
13	Electronic Archiving	System	1 yr				15	
14	Data Review System	System	1 yr				16	
15	Final Report Generation	System	1 yr				17	
16	Standards/Reagents	System	6 mo				18	
17	Manual Integration	System	1 yr				19	
18	Corrective Action System	System	1 yr				20	
19	Training Records	System	6 mo				21	
20	MDLs	System	1 yr				22	
21	SOPs – Prep/Review/Update Process	System	1 yr				23	
22	Purchasing/Procurement	System	1 yr				24	
23	Pipette/Diluter/Dispenser Calibration Check	System	6 mo				25	
24	Subcontract Lab Approval	System	1 yr				26	
25	Customer Complaint System	System	1 yr				27	
26	Methods	Method	2 yr				28	
	Checklist Pending							

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 16-8 of 16-8

Figure 16-2.

## Example – Internal Audit System Checklist: Corrective Actions



(Summary Page)

TestAmerica <Location>

#### **INTERNAL AUDIT - Corrective Actions**

	[ Printed Name(s) or Date(s) ]
Area Audited:	
Auditor:	
Date:	
Persons Contacted During Audit:	
Date Reported to Department Manager:	
Reported To:	
Date Reported to Lab Director/Manager:	
Reported To:	
Date Response Due:	
	· · · · · · · · · · · · · · · · · · ·
Response Received and Accepted by QA Manager:	
Associated Corrective Action Report Number(s):	
Scheduled Follow-up:	

Item	Requirement	Ref.	Y	N	NA	Evidence/Comments	Up
1	Does the laboratory have a corrective action program in place?	5,4.10.1	i i e				L
2	Does the laboratory have a current corrective action SOP or is this	5.4.10.1					
	information in the QA Manual?						
3	Do all laboratory personnel have documented training and access to	5.4.10.1			1.5		
	initiate corrective actions?						
4	Are causes clearly identified by department, staff name, scope of	5.4.10.6		da da A	1		
	issue (how many reports affected)?						
5	Is a root cause for the issue identified?	5.4.10.2					
6	Is a corrective action (plan) clearly described?						
7	Was the corrective action fully implemented?				1995. 1		
8	Is documentation (if applicable) completed as specifed by the						
	corrective action (training, revised SOP, etc)	1					
9	Has a follow-up assessment been conducted to verify the corrective						i I
	action was successful?						
10	Are corrective actions reviewed on a regular basis by management?	5.4.10.6a 5	4		- I		i I
11	Is there a defined distribution flow for corrective action notification,	5.4.10.6a					i I
	review, closure, and follow-up?		· .				<u> </u>
12	Are non-conformances reviewed on a regular basis and used, if				· ·		i i
	necessary, to initiate root cause corrective actions?			1			1
13	Does the lab have a documented procedure for QC corrective action (i.e.,	4.10.1		1			1
	documented within each method / parameter SOP or in the QA Manual)?						1
		1			ļ		<b></b>
14	Verify Corrective Actions from previous systems audits. L	<u>ist Items:</u>					L
15					Ì.		l
16			i in the s	14	1		1
17	······································		i i i i i			· · · · · · · · · · · · · · · · · · ·	
		1	1.1.1.1	1.1.1	1	· · · · · · · · · · · · · · · · · · ·	

Auditor Signature:

Primary Reference(s); Corporate SOP CA-Q-S-002, Acceptable Manual Integration Practices NELAC Standard, June 2003 DoD Quality Systems Manual, Version 3, January 2006

EPA Manual for the Certification of Laboratories Analyzing Drinking Water

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 17-1 of 17-7

#### **SECTION 17**

## MANAGEMENT REVIEWS (NELAC 5.4.14)

## 17.1 QUALITY ASSURANCE REPORT

A comprehensive QA Report shall be prepared each month by the laboratory's QA Department and forwarded to the Laboratory Director for review and comments. The final report shall be submitted to the Operation Manager as well as the appropriate Quality Director and General Manager. All aspects of the QA system are reviewed to evaluate the suitability of policies and procedures. At a minimum, the report content will contain the items listed below. During the course of the year, the Laboratory Director, General Manager or Corporate QA may request that additional information be added to the report.

The TestAmerica QA Report template is comprised of a discussion of three key QA issues facing the laboratory and ten specific sections (Figure 17-1):

- **Metrics:** Describe actions or improvement activities underway to address any outlying quality metrics that have been reported in the monthly Quality Systems Metrics Table.
- SOPs: Report SOPs that have been finalized and report status of any outstanding SOP reviews.
- **Corrective Actions:** Describe highlights and the most frequent cause for report revisions and corrective/preventive action measures underway. Include a discussion of any recalls handled at the lab level as per Section 6.2.2 in the Investigation/Recall SOP (SOP: CA-L-S-001). Include a section for client feedback and complaints. Include both positive and negative feedback. Describe the most serious client complaints and resolutions in progress.
- **MDLs and Control Limits:** Report which MDLs/ MDL verifications are due. Report the same for Control Limits.
- Audits: Report Internal and External Audits that were conducted. Include all relevant information such as which methods, by whom, corrective actions needed by when and discuss unresolved audit findings.
- **Performance Testing (PT) Samples:** Report the PT tests that are currently being tested with their due dates, report recent PT results by study, acceptable, total reported and the month and year.
- **Certifications:** Report on any certification programs being worked on by due date, packages completed. Describe any issues, lapses, or potential revocations.
- **Regulatory Updates:** Include information on new state or federal regulations that may impact the laboratory. Report new methods that require new instrumentation, deletion of methods, changes in sampling requirements and frequencies etc...
- Miscellaneous: Include any issues that may impact quality within the laboratory.
- Next Month: Report on plans for the upcoming month.
- Lab Director Comments Section: This section gives the Laboratory Director the opportunity to comment on issues discussed in the report and to document plans to resolve

these issues. Unresolved issues that reappear in subsequent monthly reports must be commented on by the Laboratory Director.

• Quality Systems Metrics Table: The report also includes statistical results that are used to assess the effectiveness of the quality system. Effective quality systems are the responsibility of the entire laboratory staff. Each laboratory provides their results in a template provided by Corporate QA (Figure 17-2).

On a monthly basis, Corporate QA compiles information from all the monthly laboratory reports. The VP-QA/EHS prepares a report that includes a compilation of all metrics and notable information and concerns regarding the QA programs within the laboratories. The report also includes a listing of new regulations that may potentially impact the laboratories. This report is presented to the Analytical Division Senior Management Team and General Managers.

#### 17.2 ANNUAL MANAGEMENT REVIEW

The senior lab management team (Laboratory Director, Technical Director, Operations Manager, Customer Service Manager, QA Manager) conducts an annual review of its quality systems and LIMS to ensure its continuing suitability and effectiveness in meeting client and regulatory requirements and to introduce any necessary changes or improvements. Corporate Operations and Corporate QA personnel may be included in this meeting at the discretion of the Laboratory Director. The LIMS review consists of examining any audits, complaints or concerns that have been raised through the year that are related to the LIMS. The laboratory will summarize any critical findings that can not be solved by the lab and report them to Corporate IT.

This review uses information generated during the preceding year to assess the "big picture" by ensuring that routine quality actions taken and reviewed on a monthly basis are not components of larger systematic concerns. The monthly review (refer to Section 17.1) should keep the quality systems current and effective, therefore, the annual review is a formal senior management process to review specific existing documentation. Significant issues from the following documentation are compiled or summarized by the QA Manager prior to the review meeting:

- Matters arising from the previous annual review.
- Prior Monthly QA Reports issues.
- Laboratory QA Metrics.
- Review of report reissue requests.
- Review of client feedback and complaints.
- Issues arising from any prior management or staff meetings.
- Minutes from prior Senior Management team meetings. Issues that may be raised from these meetings include:
  - Adequacy of staff, equipment and facility resources.
  - Adequacy of policies and procedures.
  - Future plans for resources and testing capability and capacity.

#### Company Confidential & Proprietary

- The annual internal double blind PT program sample performance (if performed),
- Compliance to the Ethics Policy and Data Integrity Plan. Including any evidence/incidents of inappropriate actions or vulnerabilities related to data Integrity.

The annual review includes the previous 12 months. Based on the annual review, a report is generated by the QA Manager and management. The report is distributed to the appropriate General Manager and the Quality Director. The report includes, but is not limited to:

- The date of the review and the names and titles of participants.
- A reference to the existing data quality related documents and topics that were reviewed.
- Quality system or operational changes or improvements that will be made as a result of the review [e.g., an implementation schedule including assigned responsibilities for the changes (Action Table)].

The QA Manual is also reviewed at this time and revised to reflect any significant changes made to the quality systems.

## 17.3 POTENTIAL INTEGRITY RELATED MANAGERIAL REVIEWS

Potential integrity issues (data or business related) must be handled and reviewed in a confidential manner until such time as a follow-up evaluation, full investigation, or other appropriate actions have been completed and issues clarified. The Corporate Data Investigation/ Recall SOP shall be followed (SOP No. CA-L-S-001). All investigations that result in finding of inappropriate activity are documented and include any disciplinary actions involved, corrective actions taken, and all appropriate notifications of clients.

The Chairman/CEO, President/CEO, COOs and Quality Directors receive a monthly report from the VP of Quality and EHS summarizing any current data integrity or data recall investigations as described in SOP No. CA-L-S-001. The General Manager's are also made aware of progress on these issues for their specific labs.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 17-4 of 17-7

#### Figure 17-1.

#### **Example - QA Monthly Report to Management**

LABORATORY: x PERIOD COVERED: Month/Year PREPARED BY: x DATE: Month Day, Year DISTRIBUTED TO: xx (Include LD, GM, QA Director, etc...)

THREE KEY ISSUES FOR MONTH: Include a discussion of three key issues that were focused in on this month. 1. x 2. x 3. x

#### **1. METRICS**

Describe actions or improvement activities underway to address any outlying quality metrics.

#### 2. SOPs

See Tab for SOP specifics.

The following SOPs were finalized (or reviewed for accuracy): (See Tab)

The following SOPs are due to QA: xx

In QA to complete: xx

#### **3. CORRECTIVE ACTION**

Highlights: xx

Revised Reports: Describe the most frequent cause for report revisions and corrective/preventive action measures underway.

Data Investigations/Recalls (Corporate Data Investigation/Recall SOP): Include a discussion of any recalls handled at the lab level as Corp SOP.

Client Feedback and Complaints: Include both positive and negative feedback.

Describe the most serious client complaints) and resolutions in progress.

#### 4. MDLs AND CONTROL LIMITS

MDLs Due:

Control Limits Due:

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 17-5 of 17-7

#### 5. AUDITS

#### INTERNAL AUDITS

Discuss Any Outstanding Issues (or Attach Summary):

EXTERNAL AUDITS Discuss Any Outstanding Issues (or Attach Summary):

#### 6. PT SAMPLES

The following PT samples are now in house (Due Dates): xx

#### 7. CERTIFICATIONS

Certification Packages Being Worked On (Include Due Date): x

Describe any issues, lapses, or potential revocations.

#### 8. REGULATORY UPDATE

Include information on new state or federal regulations that may impact the laboratory – new methods that require new instrumentation, deletion of methods, changes in sampling requirements or frequencies, ...

#### 9. MISCELLANEOUS

Include any issues that may impact quality within the laboratory.

#### 10. NEXT MONTH

Items planned for next month.

LAB DIRECTOR COMMENTS AND PLANNED CORRECTIVE ACTIONS:

LAB DIRECTOR REVIEW:

DATE:

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 17-6 of 17-7

Figure 17-2.

## **Example - Laboratory Metrics Categories**

# Reports for month
# Reports revised due to lab error
% Revised Reports
# of Data Recall Investigations
# of Reports Actually Recalled
# Corrective Action Reports
# Corrective Action Reports still open
Total Number of Unresolved Open Corrective Action Reports
% of Unresolved Open Corrective Action Reports
# Reports independent QA reviewed
% QA Data Review: Reports
# Technical staff (Analysts/technicians, including Temps)
# of Analyst work product reviewed year-to-date
# of Analytical instruments w/electronic data file storage capability
# of Analytical instruments reviewed for data authenticity year-to-date
% Analyst/Instrument Data Authenticity Audits
# Client Complaints
# Client Compliments
# of planned internal audits
# of planned internal method audits performed year-to-date
% Annual Internal Audits Complete
# of Open Internal Audit Findings Past Due
Total Number of External Audit Findings
# of Open External Audit Findings Past Due
% External Audit Findings Past Due
# of PT analytes participated and received scores
# of PT analytes not acceptable
% PT Cumulative Score
(analyte failed more than once in 4 consecutive studies by PT Type) (only applies to failed analytes)
# SOPs

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 17-7 of 17-7

# SOPs Reviewed/revised within 24 months

# Methods or Administrative procedures without approved SOPs

SOP Status

Method certification Losses due to performance/audit issues

Hold Time Violations due to lab error

Date of Last Comprehensive Ethics Training Session

# Staff that haven't Received Comprehensive Ethics Training (>30 Days From Employment Date)

MDL Status (Good, Fair, or Poor) >90%, >70%, <70%

Training Documentation Records (Good, Fair, or Poor)

LQM Revision/review Date

QAM Updated to New Integrated Template

Last Annual Internal Audit Date (Opened, Closed)

Last Management QS Review Date

#SOPs required for 12 month review cycle (DOD or drinking water)

#SOPs for 12 month cycle/revised within 12 months (Includes QS and Methods Listed in QSM)

12 month % SOP Status (Includes QS and Methods Listed in QSM)

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 18-1 of 18-4

#### **SECTION 18**

### PERSONNEL (NELAC 5.5.2)

#### 18.1 <u>OVERVIEW</u>

TestAmerica's management believes that its highly qualified and professional staff is the single most important aspect in assuring a high level of data quality and service. The staff consists of professionals and support personnel as outlined in the organization chart in Appendix 2.

All personnel must demonstrate competence in the areas where they have responsibility. Any staff that is undergoing training shall have appropriate supervision until they have demonstrated their ability to perform their job function on their own. Staff shall be qualified for their tasks based on appropriate education, training, experience and/or demonstrated skills as required.

The laboratory employs sufficient personnel with the necessary education, training, technical knowledge and experience for their assigned responsibilities.

All personnel are responsible for complying with all QA/QC requirements that pertain to the laboratory and their area of responsibility. Each staff member must have a combination of experience and education to adequately demonstrate a specific knowledge of their particular area of responsibility. Technical staff must also have a general knowledge of lab operations, test methods, QA/QC procedures and records management.

Laboratory management is responsible for formulating goals for lab staff with respect to education, training and skills and ensuring that the laboratory has a policy and procedures for identifying training needs and providing training of personnel. The training shall be relevant to the present and anticipated responsibilities of the lab staff.

The laboratory only uses personnel that are employed by or under contract to, the laboratory. Contracted personnel, when used, must meet competency standards of the laboratory and work in accordance to the laboratory's quality system.

## 18.2 EDUCATION AND EXPERIENCE REQUIREMENTS FOR TECHNICAL PERSONNEL

TestAmerica makes every effort to hire analytical staff that possesses a college degree (AA, BA, BS) in an applied science with some chemistry in the curriculum. Exceptions can be made based upon the individual's experience and ability to learn. There are competent analysts and technicians in the industry who have not earned a college degree. Selection of qualified candidates for laboratory employment begins with documentation of minimum education, training, and experience prerequisites needed to perform the prescribed task. Minimum education and training requirements for TestAmerica employees are outlined in job descriptions and are generally summarized for analytical staff in the table below.

The laboratory maintains job descriptions for all personnel who manage, perform or verify work affecting the quality of the environmental testing the laboratory performs. Job Descriptions are

located in the TestAmerica Buffalo Human Resource office (Also see Section 4 for position descriptions/responsibilities).

Experience and specialized training are occasionally accepted in lieu of a college degree (basic lab skills such as using a balance, pipette, quantitation techniques, etc. are also considered).

Specialty	Education	Experience
Extractions, Digestions, some electrode methods (pH, DO, Redox, etc.), or Titrimetric and Gravimetric Analyses	H.S. Diploma	On the job training (OJT)
CVAA, Single component or short list Chromatography (e.g., Fuels, BTEX-GC, IC)	A college degree in an applied science or 2 years of college and at least 1 year of college chemistry	Or 2 years prior analytical experience is required
ICP, ICPMS, Long List or complex chromatography (e.g., Pesticides, PCB, Herbicides, HPLC, etc.), GCMS	A college degree in an applied science or 2 years of college chemistry	or 5 years of prior analytical experience
Spectra Interpretation	A college degree in an applied science or 2 years of college chemistry	And 2 years relevant experience Or 5 years of prior analytical experience
Technical Directors/Department Managers – General	Bachelors Degree in an applied science or engineering with 24 semester hours in chemistry An advanced (MS, PhD.) degree may substitute for one year of experience	And 2 years experience in environmental analysis of representative analytes for which they will oversee

As a general rule for analytical staff:

When an analyst does not meet these requirements, they can perform a task under the direct supervision of a qualified analyst, peer reviewer or Department Manager, and are considered an analyst in training. The person supervising an analyst in training is accountable for the quality of the analytical data and must review and approve data and associated corrective actions.

## 18.3 TRAINING

TestAmerica is committed to furthering the professional and technical development of employees at all levels.

Orientation to the laboratory's policies and procedures, in-house method training, and employee attendance at outside training courses and conferences all contribute toward employee proficiency. Below are examples of various areas of required employee training:

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 18-3 of 18-4

Required Training	Time Frame	Employee Type		
Environmental Health & Safety	Refer to EH&S Manual	All		
Ethics – New Hires	1 week of hire	All		
Ethics - Comprehensive	90 days of hire	All		
Data Integrity	30 days of hire	Technical and PMs		
Quality Assurance	90 days of hire	All		
Ethics – Refresher	Annually	All		
Initial Demonstration of Capability (DOC)	Prior to unsupervised method performance	Technical		

The laboratory maintains records of relevant authorization/competence, education, professional qualifications, training, skills and experience of technical personnel (including contracted personnel) as well as the date that approval/authorization was given. These records are kept on file at the laboratory. Also refer to "Demonstration of Capability" in Section 20.

The training of technical staff is kept up to date by:

- Each employee must have documentation in their training file that they have read, understood and agreed to follow the most recent version of the laboratory QA Manual and SOPs in their area of responsibility. This documentation is updated as SOPs are updated.
- Documentation from any training courses or workshops on specific equipment, analytical techniques or other relevant topics are maintained in their training file.
- Documentation of proficiency (refer to Section 20).
- An Ethics Agreement signed by each staff member (renewed each year) and evidence of annual ethics training.
- A Confidentiality Agreement signed by each staff member signed at the time of employment.
- The Human Resource office maintains documentation and attestation forms on employment status & records; benefit programs; timekeeping/payroll; and employee conduct (e.g., ethics). This information is maintained in the employee's secured personnel file.

Further details of the laboratory's training program are described in TestAmerica Buffalo SOP BF-QA-004, Laboratory Personnel Training.

#### 18.4 DATA INTEGRITY AND ETHICS TRAINING PROGRAM

Establishing and maintaining a high ethical standard is an important element of a Quality System. Ethics and data integrity training is integral to the success of TestAmerica and is provided for each employee at TestAmerica. It is a formal part of the initial employee orientation within 1 week of hire, comprehensive training within 90 days, and an annual refresher for all employees. Senior management at each facility performs the ethics training for their staff.

In order to ensure that all personnel understand the importance TestAmerica places on maintaining high ethical standards at all times; TestAmerica has established an Ethics Policy No. CA-L-P-001 and an Ethics Statement/Agreement (Appendix 1). All initial and annual training is documented by signature on the signed Ethics Policy and Code of Ethical Conduct

#### **Company Confidential & Proprietary**

demonstrating that the employee has participated in the training and understands their obligations related to ethical behavior and data integrity.

Violations of this Ethics Policy will not be tolerated. Employees who violate this policy will be subject to disciplinary actions up to and including termination. Criminal violations may also be referred to the Government for prosecution. In addition, such actions could jeopardize TestAmerica's ability to do work on Government contracts, and for that reason, TestAmerica has a Zero Tolerance approach to such violations.

Employees are trained as to the legal and environmental repercussions that result from data misrepresentation. Key topics covered in the presentation include:

- Organizational mission and its relationship to the critical need for honesty and full disclosure in all analytical reporting.
- Ethics Policy (Appendix 1)
- How and when to report ethical/data integrity issues. Confidential reporting.
- Record keeping.
- Discussion regarding data integrity procedures.
- Specific examples of breaches of ethical behavior (e.g. peak shaving, altering data or computer clocks, improper macros, etc., accepting/offering kickbacks, illegal accounting practices, unfair competition/collusion)
- Internal monitoring. Investigations and data recalls.
- Consequences for infractions including potential for immediate termination, debarment, or criminal prosecution.
- Importance of proper written narration / data qualification by the analyst and project manager with respect to those cases where the data may still be usable but are in one sense or another partially deficient.

Additionally, a data integrity hotline (1-800-736-9407) is maintained by TestAmerica and administered by the Corporate Quality Department.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 19-1 of 19-3

#### **SECTION 19**

# ACCOMMODATIONS AND ENVIRONMENTAL CONDITIONS (NELAC 5.5.3)

#### 19.1 <u>OVERVIEW</u>

TestAmerica Buffalo is a 32,000 ft² secure laboratory facility with controlled access and designed to accommodate an efficient workflow and to provide a safe and comfortable work environment for employees. All visitors sign in and are escorted by laboratory personnel. Access is controlled by various measures.

The laboratory is equipped with structural safety features. Each employee is familiar with the location, use, and capabilities of general and specialized safety features associated with their workplace. The laboratory provides and requires the use of protective equipment including safety glasses, protective clothing, gloves, etc.. OSHA and other regulatory agency guidelines regarding required amounts of bench and fume hood space, lighting, ventilation (temperature and humidity controlled), access, and safety equipment are met or exceeded.

Traffic flow through sample preparation and analysis areas is minimized to reduce the likelihood of contamination. Adequate floor space and bench top area is provided to allow unencumbered sample preparation and analysis space. Sufficient space is also provided for storage of reagents and media, glassware, and portable equipment. Ample space is also provided for refrigerated sample storage before analysis and archival storage of samples after analysis. Laboratory HVAC and deionized water systems are designed to minimize potential trace contaminants.

The laboratory is separated into specific areas for field operations, bottle kit preparation, sample receiving, sample preparation, volatile organic sample analysis, non-volatile organic sample analysis and administrative functions.

#### 19.2 ENVIRONMENT

Laboratory accommodation, test areas, energy sources, lighting are adequate to facilitate proper performance of tests. The facility is equipped with heating, ventilation, and air conditioning (HVAC) systems appropriate to the needs of environmental testing performed at this laboratory.

The environment in which these activities are undertaken does not invalidate the results or adversely affect the required accuracy of any measurements.

The laboratory provides for the effective monitoring, control and recording of environmental conditions that may effect the results of environmental tests as required by the relevant specifications, methods, and procedures. Such environmental conditions include humidity, voltage, temperature, and vibration levels in the laboratory. Key equipment has been provided with back-up power supply in the event of a power outage.

When any of the method or regulatory required environmental conditions change to a point where they may adversely affect test results, analytical testing will be discontinued until the environmental conditions are returned to the required levels (refer to Section 12).

Environmental conditions of the facility housing the computer network and LIMS are regulated to protect against raw data loss.

## 19.3 WORK AREAS

There is effective separation between neighboring areas when the activities therein are incompatible with each other. Examples include:

• Volatile organic chemical handling areas, including sample preparation and waste disposal, and volatile organic chemical analysis areas.

Access to and use of all areas affecting the quality of analytical testing is defined and controlled by secure access to the laboratory building as described below in the Building Security section.

Adequate measures are taken to ensure good housekeeping in the laboratory and to ensure that any contamination does not adversely affect data quality. These measures include regular cleaning to control dirt and dust within the laboratory.

Work areas are available to ensure an unencumbered work area. Work areas include:

- Access and entryways to the laboratory.
- Sample receipt areas.
- Sample storage areas.
- Chemical and waste storage areas.
- Data handling and storage areas.
- Sample processing areas.
- Sample analysis areas.

## 19.4 FLOOR PLAN

A floor plan can be found in Appendix 3.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 19-3 of 19-3

#### 19.5 BUILDING SECURITY

Building pass cards and alarm codes are distributed to all facility employees.

Visitors to the laboratory sign in and out in a visitor's logbook. A visitor is defined as any person who visits the laboratory who is not an employee of TestAmerica Buffalo. [The reason for this is that it is important to know who is in the building in case of a safety emergency. The visitors logbook is used to ensure that everyone got out of the building safely.] In addition to signing into the laboratory, the Environmental, Health and Safety Manual contains requirements for visitors and vendors. There are specific safety forms that must be reviewed and signed.

Visitors (with the exception of company employees) are escorted by laboratory personnel at all times, or the location of the visitor is noted in the visitor's logbook.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-1 of 20-21

#### SECTION 20.0

# TEST METHODS AND METHOD VALIDATION (NELAC 5.5.4)

#### 20.1 <u>OVERVIEW</u>

**TestAmerica Buffalo** uses methods that are appropriate to meet our clients' requirements and that are within the scope of the laboratory's capabilities. These include sampling, handling, transport, storage and preparation of samples, and, where appropriate, an estimation of the measurement of uncertainty as well as statistical techniques for analysis of environmental data.

Instructions are available in the laboratory for the operation of equipment as well as for the handling and preparation of samples. All instructions, Standard Operating Procedures (SOPs), reference methods and manuals relevant to the working of the laboratory are readily available to all staff. Deviations from published methods are documented (with justification) in the laboratory's approved SOPs. SOPs are submitted to clients for review at their request. Significant deviations from published methods require client approval and regulatory approval where applicable.

#### 20.2 STANDARD OPERATING PROCEDURES (SOPs)

**TestAmerica Buffalo** maintains SOPs that accurately reflect all phases of the laboratory such as assessing data integrity, corrective actions, handling customer complaints as well as all analytical methods and sampling procedures. The method SOPs are derived from the most recently promulgated/approved, published methods and are specifically adapted to the laboratory facility. Modifications or clarifications to published methods are clearly noted in the SOPs. All SOPs are controlled in the laboratory (refer to Section 6 on Document Control):

- All SOPs contain a revision number, effective date, and appropriate approval signatures. Controlled copies are available to all staff.
- Procedures for preparation, review, revision and control are incorporated by reference to Corporate SOP CW-Q-S-002, Writing a Standard Operating Procedure (SOP) and Laboratory SOP BF-QA-003, Procedure for Writing, Reviewing and Revising Controlled Quality Documents (QAM, SOP, etc)
- SOPs are reviewed at a minimum of every 2 years (annually for Drinking Water), and where necessary, revised to ensure continuing suitability and compliance with applicable requirements.

#### 20.3 LABORATORY METHODS MANUAL

For each test method, the laboratory shall have available the published referenced method as well as the laboratory developed SOP. Refer to the corporate SOP CW-Q-S-002 "Writing a Standard Operating Procedure" for content and requirements of technical and non-technical SOPs.

**Note:** If more stringent standards or requirements are included in a mandated test method or regulation than those specified in this manual, the laboratory shall demonstrate that such requirements are met. If it is not clear which requirements are more stringent, the standard from

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-2 of 20-21

the method or regulation is to be followed. Any exceptions or deviations from the referenced methods or regulations are noted in the specific analytical SOP.

## 20.4 SELECTION OF METHODS

Since numerous methods and analytical techniques are available, continued communication between the client and laboratory is imperative to assure the correct methods are utilized. Once client methodology requirements are established, this and other pertinent information is summarized by the Project Manager. These mechanisms ensure that the proper analytical methods are applied when the samples arrive for log-in. For non-routine analytical services (e.g., special matrices, non-routine compound lists, etc.), the method of choice is selected based on client needs and available technology. The methods selected should be capable of measuring the specific parameter of interest, in the concentration range of interest, and with the required precision and accuracy.

## 20.4.1 <u>Sources of Methods</u>

Routine analytical services are performed using standard EPA-approved methodology. In some cases, modification of standard approved methods may be necessary to provide accurate analyses of particularly complex matrices. When the use of specific methods for sample analysis is mandated through project or regulatory requirements, only those methods shall be used.

In general, TestAmerica Buffalo follows procedures from the referenced methods shown below in 20.4.1.1.

When clients do not specify the method to be used or methods are not required, the methods used will be clearly validated and documented in an SOP and available to clients and/or the end user of the data.

**20.4.1.1** The analytical methods used by the laboratory are those currently accepted and approved by the U. S. EPA and the state or territory from which the samples were collected. Reference methods include:

- Method 1664, Revision A: N-Hexane Extractable Material (HEM; Oil and Grease) and Silica Gel <u>Treated N-Hexane Extractable Material (SGT-HEM); Non-polar Material) by Extraction and</u> <u>Gravimetry</u>, EPA-821-R-98-002, February 1999
- <u>Compendium of Methods for the Determination of Toxic Organic Compounds in Ambient Air</u>, US EPA, January 1996.
- <u>Guidelines Establishing Test Procedures for the Analysis of Pollutants Under the Clean Water Act</u>, and Appendix A-C; 40 CFR Part 136, USEPA Office of Water. <u>Revised as of July 1, 1995, Appendix</u> <u>A to Part 136 - Methods for Organic Chemical Analysis of Municipal and Industrial Wastewater (EPA 600 Series)</u>
- Methods for Chemical Analysis of Water and Wastes, EPA 600 (4-79-020), 1983.
- <u>Methods for the Determination of Inorganic Substances in Environmental Samples</u>, EPA-600/R-93/100, August 1993.
- <u>Methods for the Determination of Metals in Environmental Samples</u>, EPA/600/4-91/010, June 1991. Supplement I: EPA-600/R-94/111, May 1994.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-3 of 20-21

- <u>Methods for the Determination of Organic Compounds in Drinking Water</u>, EPA-600/4-88-039, December 1988, Revised, July 1991, Supplement I, EPA-600-4-90-020, July 1990, Supplement II, EPA-600/R-92-129, August 1992. <u>Supplement III EPA/600/R-95/131 - August 1995 (EPA 500 Series)</u> (EPA 500 Series methods)
- Technical Notes on Drinking Water Methods, EPA-600/R94-173, October 1994
- NIOSH Manual of Analytical Methods, 4th ed., August 1994.
- <u>Statement of Work for Inorganics Analysis</u>, ILM04.1, USEPA Contract Laboratory Program Multimedia, Multi-concentration.
- <u>Statement of Work for Inorganics Analysis</u>, ILM05.2/5.3, USEPA Contract Laboratory Program Multimedia, Multi-concentration
- <u>Statement of Work for Organics Analysis</u>, OLM04.2/4.3, USEPA Contract Laboratory Program, Multimedia, Multi-concentration.
- <u>Standard Methods for the Examination of Water and Wastewater</u>, 18th/19th /20th edition; Eaton, A.D. Clesceri, L.S. Greenberg, A.E. Eds; American Water Works Association, Water Pollution Control Federation, American Public Health Association: Washington, D.C.
- <u>Test Methods for Evaluating Solid Waste Physical/Chemical Methods (SW846)</u>, Third Edition, September 1986, Final Update I, July 1992, Final Update IIA, August 1993, Final Update II, September 1994; Final Update IIB, January 1995; Final Update III, December 1996.
- <u>Annual Book of ASTM Standards</u>, American Society for Testing & Materials (ASTM), Philadelphia, PA.
- <u>National Status and Trends Program</u>, National Oceanographic and Atmospheric Administration, Volume I-IV, 1985-1994.
- <u>Manual for the Certification of Laboratories Analyzing Drinking Water (EPA 815-R-05-004, January</u> 2005) (DW labs only)
- Code of Federal Regulations (CFR) 40, Parts 136, 141, 172, 173, 178, 179 and 261
- <u>New York State DEC Analytical Services Protocol</u>, 2005
- <u>New York State DOH Methods Manual</u>

The laboratory reviews updated versions to all the aforementioned references for adaptation based upon capabilities, instrumentation, etc., and implements them as appropriate. As such, the laboratory strives to perform only the latest versions of each approved method as regulations allow or require.

Other reference procedures for non-routine analyses may include methods established by specific states (e.g., Underground Storage Tank methods), ASTM or equipment manufacturers. Sample type, source, and the governing regulatory agency requiring the analysis will determine the method utilized.

The laboratory shall inform the client when a method proposed by the client may be inappropriate or out of date. After the client has been informed, and they wish to proceed contrary to the laboratory's recommendation, it will be documented.

#### 20.4.2 Demonstration of Capability

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-4 of 20-21

Before the laboratory may institute a new method and begin reporting results, the laboratory shall confirm that it can properly operate the method. In general, this demonstration does not test the performance of the method in real world samples, but in an applicable and available clean matrix sample. If the method is for the testing of analytes that are not conducive to spiking, demonstration of capability may be performed on quality control samples.

- **20.4.2.1** A demonstration of capability is performed whenever there is a significant change in instrument type, method or personnel.
- **20.4.2.2** The initial demonstration of capability must be thoroughly documented and approved by the Department Manager and QA Manager prior to independently analyzing client samples. All associated documentation must be retained in accordance with the laboratories archiving procedures (refer to Section 15, Control of Records).
- **20.4.2.3** The laboratory must have an approved SOP, demonstrate satisfactory performance, and conduct a method detection limit study (when applicable). There may be other requirements as stated within the published method or regulations (i.e., retention time window study).

**Note:** In some instances, a situation may arise where a client requests that an unusual analyte be reported using a method where this analyte is not normally reported. If the analyte is being reported for regulatory purposes, the method must meet all procedures outlined within this QA Manual (SOP, MDL, and Demonstration of Capability). If the client states that the information is not for regulatory purposes, the result may be reported as long as the following criteria are met:

- The instrument is calibrated for the analyte to be reported using the criteria for the method and ICV/CCV criteria are met (unless an ICV/CCV is not required by the method).
- The reporting limit is set at or above the first standard of the curve for the analyte.
- The client request is documented and the lab informs the client of its procedure for working with unusual compounds. The final report must be footnoted: *Reporting Limit based on the low standard of the calibration curve.*
- Refer to Section 12 (Control of Non-Conforming Work).

## 20.4.3 Initial Demonstration of Capability (IDOC) Procedures

Procedures for generation of IDOCs are detailed below and in laboratory SOP BF-QA-004, Laboratory Personnel Training.

- **20.4.3.1** The spiking standard used must be prepared independently from those used in instrument calibration.
- **20.4.3.2** The analyte(s) shall be diluted in a volume of clean matrix sufficient to prepare four aliquots at the concentration specified by a method or the laboratory SOP.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-5 of 20-21

- **20.4.3.3** At least four aliquots shall be prepared (including any applicable clean-up procedures) and analyzed according to the test method (either concurrently or over a period of days).
- **20.4.3.4** Using all of the results, calculate the mean recovery in the appropriate reporting units and the standard deviations for each parameter of interest.
- **20.4.3.5** When it is not possible to determine the mean and standard deviations, such as for presence, absence and logarithmic values, the laboratory will assess performance against criteria described in the Method SOP.
- **20.4.3.6** Compare the information obtained above to the corresponding acceptance criteria for precision and accuracy in the test method (if applicable) or in laboratory generated acceptance criteria (LCS or interim criteria) if there is no mandatory criteria established. If any one of the parameters do not meet the acceptance criteria, the performance is unacceptable for that parameter.
- **20.4.3.7** When one or more of the tested parameters fail at least one of the acceptance criteria, the analyst must proceed according to either option listed below:
  - Locate and correct the source of the problem and repeat the test for all parameters of interest beginning with 20.4.3.3 above.
  - Beginning with 20.4.3.3 above, repeat the test for all parameters that failed to meet criteria. Repeated failure, however, will confirm a general problem with the measurement system. If this occurs, locate and correct the source of the problem and repeat the test for all compounds of interest beginning with 20.4.3.1 above.

A certification statement (see Figure 20-1) shall be used to document the completion of each initial demonstration of capability. A copy of the certification is archived in the analyst's training folder.

## 20.5 LABORATORY DEVELOPED METHODS AND NON-STANDARD METHODS

Any new method developed by the laboratory must be fully defined in an SOP/Methods Manual (Section 20.2) and validated by qualified personnel with adequate resources to perform the method. Method specifications and the relation to client requirements must be clearly conveyed to the client if the method is a non-standard method (not a published or routinely accepted method). The client must also be in agreement to the use of the non-standard method. The information included in the checklist below (Figure 20-2) is needed before samples are accepted for analysis by a new method.

## 20.6 VALIDATION OF METHODS

Validation is the confirmation by examination and the provision of objective evidence that the particular requirements for a specific intended use are fulfilled. (From 2003 NELAC Standard)

All non-standard methods, laboratory designed/developed methods, standard methods used outside of their scope, and major modifications to published methods must be validated to

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-6 of 20-21

confirm they are fit for their intended use. The validation will be as extensive as necessary to meet the needs of the given application. The results are documented with the validation procedure used and contain a statement as to the fitness for use.

## 20.6.1 Method Validation and Verification Activities for All New Methods

While method validation can take various courses, the following activities can be required as part of method validation. Method validation records are designated QC records and are archived accordingly.

#### 20.6.1.1 Determination of Method Selectivity

Method selectivity is the demonstrated ability to discriminate the analyte(s) of interest from other compounds in the specific matrix or matrices from other analytes or interference. In some cases to achieve the required selectivity for an analyte, a confirmation analysis is required as part of the method.

#### 20.6.1.2 Determination of Method Sensitivity

Sensitivity can be both estimated and demonstrated. Whether a study is required to estimate sensitivity depends on the level of method development required when applying a particular measurement system to a specific set of samples. Where estimations and/or demonstrations of sensitivity are required by regulation or client agreement, such as the procedure in 40 CFR Part 136 Appendix B, under the Clean Water Act, these shall be followed. The laboratory determinations of MDLs are described in Section 20.7.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-7 of 20-21

#### 20.6.1.3 Relationship of Limit of Detection (LOD) to the Quantitation Limit (QL)

An important characteristic of expression of sensitivity is the difference in the LOD and the QL. The LOD is the minimum level at which the presence of an analyte can be reliably concluded. The QL is the minimum level at which both the presence of an analyte and its concentration can be reliably determined. For most instrumental measurement systems, there is a region where semi-quantitative data is generated around the LOD (both above and below the estimated MDL or LOD) and below the QL. In this region, detection of an analyte may be confirmed but quantification of the analyte is unreliable within the accuracy and precision guidelines of the measurement system. When an analyte is detected below the QL, and the presence of the analyte is confirmed by meeting the qualitative identification criteria for the analyte, the analyte can be reliably reported, but the amount of the analyte can only be estimated. If data is to be reported in this region, it must be done so with a qualification that denotes the semi-quantitative nature of the result.

#### 20.6.1.4 Determination of Interferences

A determination that the method is free from interferences in a blank matrix is performed.

#### 20.6.1.5 Determination of Range

Where appropriate, a determination of the applicable range of the method may be performed. In most cases, range is determined and demonstrated by comparison of the response of an analyte in a curve to established or targeted criteria. The curve is used to establish the range of quantitation and the lower and upper values of the curve represent upper and lower quantitation limits. Curves are not limited to linear relationships.

#### 20.6.1.6 Determination of Accuracy and Precision

Accuracy and precision studies are generally performed using replicate analyses, with a resulting percent recovery and measure of reproducibility (standard deviation, relative standard deviation) calculated and measured against a set of target criteria.

#### 20.6.1.7 Documentation of Method

The method is formally documented in an SOP. If the method is a minor modification of a standard laboratory method that is already documented in an SOP, an SOP Attachment describing the specific differences in the new method is acceptable in place of a separate SOP.

#### 20.6.1.8 Continued Demonstration of Method Performance

Continued demonstration of Method Performance is addressed in the SOP. Continued demonstration of method performance is generally accomplished by batch specific QC samples such as LCS, method blanks or PT samples.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-8 of 20-21

## 20.7 METHOD DETECTION LIMITS (MDL)/ LIMITS OF DETECTION (LOD)

Method detection limits (MDL) are initially determined in accordance with <u>40 CFR Part 136</u>, <u>Appendix B</u> or alternatively by other technically acceptable practices that have been accepted by regulators. MDL is also sometimes referred to as Limit of Detection (LOD). The MDL theoretically represents the concentration level for each analyte within a method at which the Analyst is 99% confident that the true value is not zero. The MDL is determined for each analyte initially during the method validation process and updated as required in the analytical methods, whenever there is a significant change in the procedure or equipment, or based on project specific requirements (refer to 20.7.10). The analyst prepares at least seven replicates of solution spiked at one to five times the estimated method detection limit (most often at the lowest standard in the calibration curve) into the applicable matrix with all the analytes of interest. Each of these aliquots is extracted (including any applicable clean-up procedures) and analyzed in the same manner as the samples. Where possible, the seven replicates should be analyzed over 2-4 days to provide a more realistic MDL. To allow for some flexibility, this low level standard may be analyzed every batch or every week or some other frequency rather than doing the study all at once. In addition, a larger number of data points may be used if the appropriate t-value multiplier is used.

**20.7.1** MDL's are initially performed for each individual instrument and non-microbiological method analysis. Unless there are requirements to the contrary, the laboratory will use the highest calculated MDL for all instruments used for a given method as the MDL for reporting purposes. This MDL is not required for methods that are not readily spiked (e.g. pH, turbidity, etc.) or where the lab does not report values to the MDL. For titration and gravimetric methods where there is no additional preparation involved, the MDL is based on the lowest discernable unit of measure that can be observed.

**20.7.2** MDL's must be run against acceptable instrument QC, including ICV's and Tunes. This is to insure that the instrument is in proper working condition and falsely high or low MDL's are not calculated.

**20.7.3** Use only clean matrix which is free of target analytes (e.g.: Laboratory reagent water, Ottawa Sand) unless a project specific MDL is required in a field sample matrix.

**20.7.4** The Reporting Limit (also may be referred to as Limit of Quantitation or LOQ) should generally be between 2 and 5 times the MDL (or 10X the standard deviation of the MDL for Wisconsin projects). If the MDL is being performed during method development, use this guideline to determine the Reporting Limit for the analysis. If a sample is diluted, the reported MDL is adjusted according to the dilution factor.

20.7.5 The calculated MDL cannot be greater than the spike amount.

**20.7.6** If the most recent calculated MDL does not permit qualitative identification of the analyte then the laboratory may use technical judgment for establishing the MDL (e.g., calculate what level would give a qualitative ID, compare with IDL (20.8), spike at a level where qualitative ID is determined and assign that value as MDL, minimum sensitivity requirements, standard deviation of method blanks over time, etc.). Alternate verification procedures have been included in laboratory SOP BF-QA-001, Determination of Method Detection Limits.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-9 of 20-21

**20.7.7** Each of the 7 spikes must be qualitatively identifiable (e.g., appear in both columns for dual column methods, characteristic ions for GCMS mass spectra, etc). Manual integrations to force the baseline for detection are not allowed.

20.7.8 The initial MDL is calculated as follows:

 $MDL = t_{(n-1, 1-a=0.99)} x$  (Standard Deviation of replicates)

where  $t_{(n-1, 1-a=0.99)} = 3.143$  for seven replicates.

**20.7.9** Subsequent to the initial MDL determination, periodic MDL verification, confirmation or determinations may be performed by the procedure in <u>40 CFR Part 136</u>, <u>Appendix B</u> or alternatively by other technically acceptable practices (e.g., method blanks over time, single standard spikes that have been subjected to applicable sample prep processes, etc.). The alternate procedures utilized at TestAmerica Buffalo are detailed in SOP BF-QA-001, Determination of Method Detection Limits.

**20.7.10** Because of the inherent variability in results outside of the calibration range, TestAmerica does not recommend the reporting of results below the lowest calibration point in a curve; however, it is recognized that some projects and agencies require the reporting of results below the RL. Any result that falls between the MDL and the Reporting limit, when reported, will be qualified as an estimated value.

20.7.11 Detections reported down to the MDL must be qualitatively identified.

**20.7.12** MDLs and Reporting limits are adjusted in LIMs based on moisture content and sample aliquot size. The formatting of the MDL (adjusted or not adjusted) on the final report is based on the project specifications.

## 20.8 INSTRUMENT DETECTION LIMITS (IDL)

**20.8.1** The IDL is sometimes used to assess the reasonableness of the MDLs or in some cases required by the analytical method or program requirements. IDLs are most used in metals analyses but may be useful in demonstration of instrument performance in other areas.

**20.8.2** IDLs are calculated to determine an instrument's sensitivity independent of any preparation method. IDLs are calculated either using 7 replicate spike analyses, like MDL but without sample preparation, or by the analysis of 10 instrument blanks and calculating 3 x the absolute value of the standard deviation. (For CLP procedures, the IDL is determined using the standard deviation of 7 replicate spike analyses on each of 3 non-consecutive days.)

**20.8.3** If IDL is > than the MDL, it may be used as the reported MDL.

## 20.9 VERIFICATION OF DETECTION AND REPORTING LIMITS

**20.9.1** Once an MDL is established, it must be verified, on each instrument, by analyzing a quality control sample (prepared as a sample) at approximately 2-3 times the calculated MDL for single analyte analyses (e.g. most wet chemistry methods, CVAA, etc.) and 1-4 times the calculated MDL for multiple analyte methods (e.g. GC, GCMS, ICP, etc.). The analytes must be qualitatively identified or see section 20.7.9 for other options. This verification does not apply to methods that are not readily spiked (e.g. pH, turbidity, etc.) or where the lab does not report to the MDL. If the MDL does not verify, then the lab will not report to the MDL, or redevelop their MDL or use the level where qualitative identification is established (See 20.7.9). MDLs must be verified at least annually

**20.9.2** When a Reporting limit is established, it must be initially verified by the analysis of a low level standard or QC sample (LCS at 1-2 the reporting limit) and annually thereafter. Unless there are requirements to the contrary the acceptance criteria is  $\pm$  50%. The annual requirement is waved for methods that have an annually verified MDL.

#### 20.10 RETENTION TIME WINDOWS

Most organic analyses and some inorganic analyses use chromatography techniques for qualitative and quantitative determinations. For every chromatography analysis each analyte will have a specific time of elution from the column to the detector. This is known as the analyte's retention time. The variance in the expected time of elution is defined as the retention time window. As the key to analyte identification in chromatography, retention time windows must be established on every column for every analyte used for that method. These records are kept with the files associated with an instrument for later quantitation of the analytes.

For GC, HPLC and IC methods, there must be sufficient separation between analyte peaks so as to not misidentify analytes. In the mid-level standard, the distance between the valley and peak height cannot be any less than 25% of the sum of the peak heights of the analytes. This also applies to GCMS in the case where the two compounds share the same quantitation ion.

**Note:** Some analytes do not separate sufficiently to be able to identify or quantitate them as separate analytes (e.g. m-xylene and p-xylene) and are quantitated and reported as a single analyte (e.g. m,p-xylenes).

Once the analyst has determined that the instrument is in optimum working condition through calibration and calibration verification procedures, he or she uses a mid-range calibration or calibration verification standard to establish the retention times for each of the individual analytes in a method. The analyst makes three injections of the same standard over a 72-hour (24 hr period for method 300.0) period, tabulating the retention times for each analyte for each of the three injections. The width of retention time window is normally the average absolute retention time  $\pm$  3 Standard Deviations. A peak outside the retention time window will not be identified by the computer as a positive match of the analyte of interest.

It is possible for the statistically calculated RT window to be too tight and need to be adjusted based on analyst experience. In these instances method default retention time windows may be

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-11 of 20-21

used (e.g., for 8000 series methods a default of 0.03 minutes may be used, and EPA CLP 0.05 minutes is used). The same concept is applied when any peak outside of that window will not be identified by the computer as a positive match.

The calibration verification standard at the beginning of a run may be used to adjust the RT for an analyte. This is essentially re-centering the window but the size of the window remains the same. The RTs are verified when all analytes are within their RT windows and are properly identified.

## 20.11 EVALUATION OF SELECTIVITY

The laboratory evaluates selectivity by following the checks within the applicable analytical methods, which include mass spectral tuning, second column confirmation, ICP interelement interference checks, chromatography retention time windows, sample blanks, spectrochemical or atomic absorption profiles and specific electrode response factors.

## 20.12 ESTIMATION OF UNCERTAINTY OF MEASUREMENT

**20.12.1** Uncertainty is "a parameter associated with the result of a measurement, that characterizes the dispersion of the values that could reasonably be attributed to the measurand" (as defined by the International Vocabulary of Basic and General Terms in Metrology, ISO Geneva, 1993, ISBN 92-67-10175-1). Knowledge of the uncertainty of a measurement provides additional confidence in a result's validity. Its value accounts for all the factors which could possibly affect the result, such as adequacy of analyte definition, sampling, matrix effects and interferences, climatic conditions, variances in weights, volumes, and standards, analytical procedure, and random variation. Some national accreditation organizations require the use of an "expanded uncertainty": the range within which the value of the measurand is believed to lie within at least a 95% confidence level with the coverage factor k=2.

**20.12.2** Uncertainty is not error. Error is a single value, the difference between the true result and the measured result. On environmental samples, the true result is never known. The measurement is the sum of the unknown true value and the unknown error. Unknown error is a combination of systematic error, or bias, and random error. Bias varies predictably, constantly, and independently from the number of measurements. Random error is unpredictable, assumed to be Gaussian in distribution, and reducible by increasing the number of measurements.

**20.12.3** The uncertainty associated with results generated by the laboratory can be determined by using the Laboratory Control Sample (LCS) accuracy range for a given analyte. The LCS limits are used to assess the performance of the measurement system since they take into consideration all of the laboratory variables associated with a given test over time (except for variability associated with the sampling). The percent recovery of the LCS is compared either to the method-required LCS accuracy limits or to the statistical, historical, in-house LCS accuracy limits.

**20.12.4** To calculate the uncertainty for the specific result reported, multiply the result by the decimal of the lower end of the LCS range percent value for the lower end of the uncertainty range, and multiply the result by the decimal of the upper end of the LCS range percent value for the upper end of the uncertainty range. These calculated values represent a 99%-certain range for the reported result. As an example, suppose that the result reported is 1.0 mg/l, and

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-12 of 20-21

the LCS percent recovery range is 50 to 150%. The uncertainty range would be 0.5 to 1.5 mg/l, which could also be written as 1.0 +/- 0.5 mg/l.

**20.12.5** In the case where a well recognized test method specifies limits to the values of major sources of uncertainty of measurement (e.g. 524.2, 525, etc) and specifies the form of presentation of calculated results, no further discussion of uncertainty is required.

## 20.13 CONTROL OF DATA

The laboratory has policies and procedures in place to ensure the authenticity, integrity, and accuracy of the analytical data generated by the laboratory.

#### 20.13.1 Computer and Electronic Data Related Requirements

The three basic objectives of our computer security procedures and policies are shown below. The laboratory is currently running the 'Analytical Information Management System (AIMS)' which is a custom in-house developed LIMS system that has been highly customized to meet the needs of the laboratory. It is referred to as LIMS for the remainder of this section. The LIMS utilizes FoxPro-2.6x which is an industry standard relational database platform. It is referred to as Database for the remainder of this section.

- **20.13.1.1** <u>Maintain the Database Integrity:</u> Assurance that data is reliable and accurate through data verification (review) procedures, password-protecting access, anti-virus protection, data change requirements, as well as an internal LIMS permissions procedure.
  - LIMS Database Integrity is achieved through data input validation, internal user controls, and data change requirements.
  - Spreadsheets and other software developed in-house must be verified with documentation through hand calculations prior to use.

**Note:** "Commercial off-the-shelf software in use within the designed application range is considered to be sufficiently validated." *From NELAC 2003 Standard.* However, laboratory specific configurations or modifications are validated prior to use.

- In order to assure accuracy, all data entered or transferred into the LIMS data system goes through a minimum of two levels of review.
- The QA department performs random data audits to ensure the correct information has been reported.
- Changes to reports are documented within the LIMs through audit trails, revision tracking and imposed case narrative comments.
- Analytical data file security is provided through three policies.
  - The first policy forbids unauthorized personnel from using laboratory data acquisition computers.
  - The second policy is the implementation of network passwords and login names that restrict directory access.
  - The third layer is maintained through the LIMS and includes the use of username/password combinations to gain access to the LIMS system, the fact that
all data in the LIMS is associated with the user to added/reviewed the data, and the restriction of review authority of data.

- All software installations will be in accordance with any relevant copyright licensing regulations.
- All software installed on any computer within the laboratory must be approved by the Information Technology Department regional support technician assigned to the laboratory Shrink-wrapped or otherwise sealed OEM software that is directly related to instrument usage does not need approval but the Information Technology department must be notified of the installation.
- Anti-virus software shall be installed on all servers and workstations. The anti-virus software shall be configured to check for virus signature file and program updates on a daily basis and these updates will be pushed to all servers and workstations. The antivirus software will be configured to clean any virus-infected file if possible, otherwise the file will be deleted. Disks and CDs brought from any outside source that are not OEM software must be scanned for viruses before being accessed.

## • Interlab LIMS Permissions Policy

- <u>PURPOSE</u> The purpose of this policy is to provide a mechanism for maintaining the integrity of information contained in each laboratory's LIMS while providing the necessary access for information sharing to staff at other laboratory facilities.
- <u>DEFINITIONS</u> Host Laboratory: The laboratory facility that 'owns' the LIMS system or 'hosts' a project/job.
- POLICIES

(a) All permissions for the laboratory's LIMS system must only be granted by a representative of that laboratory.

- If someone outside of the host lab needs permissions for Project Management or other uses, they must go through the Lab Director or his/her designated representative.
- Permissions must never be granted without the knowledge of the host laboratory.

(b) Only laboratory analytical or QA staff from the home laboratory may have edit permissions for laboratory analysis data.

(c) Any changes made in laboratory's LIMS system:

- Must be documented and traceable.
- If made by staff of an affiliate lab, written permission from the home lab to make the changes (email approval is sufficient) is required.
- No corrections may be made in another laboratories system without their knowledge.

(d) Data qualifiers in laboratory reports must only be corrected, edited, etc. by the staff at the host laboratory.

(e) Full analytical data "View" only permissions may be granted to outside Project Management and Sales staff. Search permissions may also be granted so status may be checked.

(f) All qualifiers must be approved by QA staff before adding to standard reference tables.

(g) Please contact Corporate QA or IT staff if you have any questions regarding implementation or interpretation of this policy.

- **20.13.1.2** <u>Ensure Information Availability:</u> Protection against loss of information or service through scheduled back-ups, secure storage of media, line filter, Uninterruptible Power Supply (UPS), and maintaining older versions of software as revisions are implemented.
  - Insured by timely backup procedures on reliable backup media, stable file server network architecture, and UPS protection
  - UPS Protection:
    - Each fileserver is protected by an appropriate power protection/backup unit. In the event of a power outage, there is approximately 15-30 minutes of up-time for the servers prior to shutdown. This allows for proper shutdown procedures to be followed with the fileservers.
  - File Server Architecture
    - All files are maintained on multiple Windows 2000 or newer servers which are secured physically in the Information Technology server room. Access to these servers is limited to members of the Information Technology staff.
    - All supporting software is maintained for at least 5 years from the last raw data generated using that software. [Length of time is dependent on local regulations or client requirements, e.g., NYS CLP requires 6 years
  - System Back-up Overview and Procedures
    - Data from both servers and instrument attached PC's are backed up and purged in compliance with the corporate back-up policy.
    - A Maintenance Plan has been defined to create a daily archive of all data within the LIMS database to a backup location. This backup is initiated automatically by either the database or back-up system.
    - Backup tapes will be stored in compliance with the corporate Data Backup Policy.
      Backup verifications are carried out in accordance with the corporate Data Backup Policy.
    - Instrument data back-ups are verified on a periodic basis by the QA department when performing electronic data audits. The audit takes place on data that has been moved to a back-up location ensuring that it has been moved.
- **20.13.1.3** <u>Maintain Confidentiality:</u> Ensure data confidentiality through physical access controls, and encryption of when electronically transmitting data.
  - All servers are located in a secure area of the IT department offices. Access to the servers is limited to IT staff members, Lab Director, Technical Director, the President and Vice President of Operations.
  - The reporting portion of the LIMS system requires a project manager to enter their unique password anytime they create a report that displays a signature on it (.PDF).
  - If electronic documents are made available outside of the web site, the customer must sign an agreement in advance that states they will not alter the data in any way.

#### 20.13.2 Data Reduction

The complexity of the data reduction depends on the analytical method and the number of discrete operations involved (e.g., extractions, dilutions, instrument readings and concentrations). The

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-15 of 20-21

analyst calculates the final results from the raw data or uses appropriate computer programs to assist in the calculation of final reportable values.

For manual data entry, e.g., Wet Chemistry, the data is reduced by the analyst and then verified by the Department Manager or alternate analyst prior to updating the data in LIMS. The data review sheets, or any other type of applicable documents, are signed by both the analyst and alternate reviewer to confirm the accuracy of the manual entry(s).

Manual integration of peaks will be documented and reviewed and the raw data will be flagged in accordance with the TestAmerica Corporate SOP CA-Q-S-002, *Acceptable Manual Integration Practices*.

Analytical results are reduced to appropriate concentration units specified by the analytical method, taking into account factors such as dilution, sample weight or volume, etc. Blank correction will be applied only when required by the method or per manufacturer's indication; otherwise, it should not be performed. Calculations are independently verified by appropriate laboratory staff. Calculations and data reduction steps for various methods are summarized in the respective analytical SOPs or program requirements.

- **20.13.2.1** All raw data must be retained in the project job folder, computer file, and/or runlog. All criteria pertinent to the method must be recorded. The documentation is recorded at the time observations or calculations are made and must be signed or initialed/dated (month/day/year). It must be easily identifiable who performed which tasks if multiple people were involved.
- 20.13.2.2 In general, concentration results are reported in milligrams per liter (mg/l) or micrograms per liter (μg/l) for liquids and milligrams per kilogram (mg/kg) or micrograms per kilogram (μg/kg) for solids. The units "mg/l" and "mg/kg" are the same as "parts per million (ppm)". The units "μg/l" and "μg/kg" are the same as "parts per billion (ppb)." For values greater than 10,000 mg/l, results can be reported in percent, i.e., 10,000 mg/l = 1%.
  - Several environmental methods, such as color, turbidity, conductivity, use very specific, non-concentration units to report results (e.g., NTU, umhos/cm etc).
  - Occasionally, the client requests that results be reported in units which take into account the measured flow of water or air during the collection of the sample. Results for wipe sampling may be reported based on the area wiped. When the client provides this information, the calculations can be performed and reported.
- **20.13.2.3** In reporting, the analyst or the instrument output records the raw data result using values of known certainty plus one uncertain digit. If final calculations are performed external to LIMS, the results should be entered in LIMS with at least three significant figures. In general, final inorganic results are reported to 2 significant figures for values less than 10 and 3 significant figures for values greater than 10 on the final report. Organic results are generally reported to 1 significant figure for values less than 10 and 2 significant figures for values greater than 10 on the final report. The number of significant figures may be adjusted based on client or project requirements.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-16 of 20-21

- **20.13.2.4** For those methods that do not have an instrument printout, an instrumental output or a calculation spreadsheet upload compatible with the LIMS System, the final results and dilution factors are entered directly into LIMS by the analyst, and the software formats the final result for the analytical report. LIMS has a defined significant figure criterion for each analyte.
- **20.13.2.5** The laboratory strives to import data directly from instruments or calculation spreadsheets to ensure that the reported data are free from transcription and calculation errors. For those analyses with an instrumental output compatible with the LIMS, the raw results and dilution factors are transferred into LIMS electronically after reviewing the quantitation report, and removing unrequested or poor spectrally-matched compounds. The analyst prints a copy of what has been entered to check for errors. This printout and the instrument's printout of calibrations, concentrations, retention times, chromatograms, and mass spectra, if applicable, are retained with the data file. The data file is automatically transferred to the network server and, eventually, to a back-up tape file.

## 20.13.3 Logbook / Worksheet Use Guidelines

Logbooks and worksheets are filled out 'real time' and have enough information on them to trace the events of the applicable analysis/task. (e.g. calibrations, standards, analyst, sample ID, date, time on short holding time tests, temperatures when applicable, calculations are traceable, etc.)

- Corrections are made following the procedures outlined in Section 13.
- Logbooks are controlled by the QA department. A record is maintained of all logbooks in the lab.
- Unused portions of pages must be "Z"'d out, signed and dated.
- Worksheets are created with the approval of the Technical Director/QA Manager at the facility. The QA Manager controls all worksheets following the procedures in Section 6.

#### 20.13.4 <u>Review / Verification Procedures</u>

Review procedures are out lined in several laboratory SOPs (e.g. BF-SR-002, "Receipt of Analytical Samples", BF-GP-012, "Technical Data Review", and BF-PM-001, "Project Information Requirements") to ensure that reported data are free from calculation and transcription errors, that QC parameters have been reviewed and evaluated before data is reported. The laboratory also has an SOP discussing Manual Integrations to ensure the authenticity of the data (BF-GP-013, Manual Integration). The general review concepts are discussed below, more specific information can be found in the SOPs.

**20.13.4.1** The data review process at **TestAmerica Buffalo** starts at the Sample Control level. Sample Control personnel review chain-of-custody forms and input the sample information and required analyses into a computer LIMS. The Project Managers perform review of the chain-of-custody forms and inputted information and approve the input in LIMs to make the samples available to the laboratory departments for batching and processing.

- **20.13.4.2** The next level of data review occurs with the Analysts. As results are generated, analysts review their work to ensure that the results generated meet QC requirements and relevant EPA methodologies. The Analysts transfer the data into the LIMS and add any manual data qualifiers or dilution codes if applicable (see Appendix 7 for list of common data qualifiers). To ensure data compliance, a different analyst performs a second level of review. Second level review is accomplished by checking reported results against raw data and evaluating the results for accuracy. During the second level review, blank runs, QA/QC check results, continuing calibration results, laboratory control samples, sample data, qualifiers and spike information are evaluated. Approximately 10% of all sample data from manual methods and from automated methods, all GC/MS spectra and all manual integrations are reviewed. Issues that deem further review include the following:
  - QC data are outside the specified control limits for accuracy and precision
  - Reviewed sample data does not match with reported results
  - Unusual detection limit changes are observed
  - Samples having unusually high results
  - Samples exceeding a known regulatory limit
  - Raw data indicating some type of contamination or poor technique
  - Inconsistent peak integration
  - Transcription errors
  - Results outside of calibration range
  - Results deviate from historical trends (if history available)
- **20.13.4.3** Unacceptable analytical results may require reanalysis of the samples. Any unusual or uncharacteristic circumstances are brought to the attention of the Department Manager. The Department Manager may involve the Project Manager, the Technical Director and/or the QA Manager for further investigation depending on the issue. Corrective action is initiated whenever necessary.
- **20.13.4.4** The results are then entered or directly transferred into the computer database and a hard copy (or .pdf) is printed for the client.
- **20.13.4.5** As a final review prior to the release of the report, the Project Manager reviews the results for appropriateness and completeness. This review and approval ensures that client requirements have been met and that the final report has been properly completed. The process includes, but is not limited to, verifying that chemical relationships are evaluated, COC is followed, cover letters/ narratives are present, flags are appropriate, and project specific requirements are met. The following are some examples of chemical relationships that are reviewed (if data are available):

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-18 of 20-21

- Total Results are > Dissolved results (e.g. metals)
- Total Solids (TS) ≥ TDS or TSS
- TKN > Ammonia
- Total Phosphorus > Orthophosphate
- $COD \ge TOC$
- Total cyanide > Amenable Cyanide
- TDS  $\geq$  individual anions
- **20.13.4.6** Any project that requires a data package is subject to a tertiary data review for transcription errors and acceptable quality control requirements. The Project Manager then signs the final report and creates the invoice. (*Also see section 26 on Reporting Results*). When complete, the report is issued to the client.

#### 20.13.5 Manual Integrations

Computerized data systems provide the analyst with the ability to re-integrate raw instrument data in order to optimize the interpretation of the data. Though manual integration of data is an invaluable tool for resolving variations in instrument performance and some sample matrix problems, when used improperly, this technique would make unacceptable data appear to meet quality control acceptance limits. Improper re-integrations lead to legally indefensible data, a poor reputation, or possible laboratory decertification. Because guidelines for re-integration of data are not provided in the methods and most methods were written prior to widespread implementation of computerized data systems, the laboratory trains all analytical staff on proper manual integration techniques using SOP CA-Q-S-002 as the guidelines.

- **20.13.5.1** The analyst must adjust baseline or the area of a peak in some situations, for example when two compounds are not adequately resolved or when a peak shoulder needs to be separated from the peak of interest. The analyst must use professional judgment and common sense to determine when manual integrating is required. Analysts are encouraged to ask for assistance from a senior analyst or manager when in doubt.
- **20.13.5.2** Analysts shall not increase or decrease peak areas for the sole purpose of achieving acceptable QC recoveries that would have otherwise been unacceptable. The intentional recording or reporting of incorrect information (or the intentional omission of correct information) is against company principals and policy and is grounds for immediate termination.
- **20.13.5.3** Client samples, performance evaluation samples, and quality control samples are all treated equally when determining whether or not a peak area or baseline should be manually adjusted.
- 20.13.5.4 All manual integrations receive a second level review. Manual integrations must be indicated on an expanded scale "after" chromatograms such that the integration performed can be easily evaluated during data review. Expanded scale "before" chromatograms are also required for all manual integrations on QC parameters

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-19 of 20-21

(calibrations, calibration verifications, laboratory control samples, internal standards, surrogates, etc.) unless the laboratory has another documented corporate approved procedure in place that can demonstrate an active process for detection and deterrence of improper integration practices.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-20 of 20-21

## Figure 20-1. Example - Demonstration of Capability Documentation

# TestAmerica

THE LEADER IN ENVIRONMENTAL TESTING

Quality Assurance Manager

DOC Cert. Statement Revision 7 Dec. 14, 2007

#### TESTAMERICA LABORATORIES, INC.

TRAINING & DEMONSTRATION OF CAPABILITY CERTIFICATION STATEMENT

Employee:			Page of			
Method Number:			Date:			
Parameters or Analytes:						
Initial Demonstration of Canability:	Ē					
minar Demonstration of Capability.	_		Date Band			
SOP Number:		Revision #	Date Read			
Trained By:						
Date training began:	<u> </u>	Date training comp	oleted:			
Continued Demonstration of Capability:						
SOP Number:		Revision #	Date Read			
I CERTIFY that I have read and understand the SOP identified above. I have also submitted data associated with the demonstration of capability.						
	Empl	oyee Signature	Date			
We, the undersigned, CERTIFY that:		<i></i>				
1. The analyst identified above, using the cited te the National Environmental Laboratory Accreditation	st method( ation Progr	s), which is in use at thi am, have met the Demo	s facility for the analyses of samples under nstration of Capability.			
2. The test method(s) was performed by the analy	yst(s) identi	fied on this certification	1.			
3. A copy of the test method(s) and the laboratory-specific Sops are available for all personnel on-site.						
4. The data associated with the demonstration cap	pability are	true, accurate, complet	e and self-explanatory.			
5. All raw data (including a copy of this certifical retained at this facility, and that the associated in	tion form) i formation i	necessary to reconstruct s well organized and av	and validate these analyses have been ailable for review by authorized assessors.			
John Schove	<b></b>					
Operations Manager	Signat	ture	Date			
Peggy Gray-Erdmann						
	<b>O</b> ¹		Data			

Signature

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 20-21 of 20-21

Figure 20-2.

#### Example - New Method / Additional Analyte Checklist

# New Method / Additional Analyte Checklist

The following items are <u>required</u> to be completed prior to the acceptance of client samples. Fill in any blanks that do not apply with "NA". Provide associated instrument QC when samples or QC samples are analyzed (includes run log).

New Method

Added Analytes _____

1_____ Standard Operating Procedure

- Note: For additional analytes, a ROMD [or whatever an internal communication memo is named in your lab] can be used to add the analytes, include RL and matrix.
  - _____ Analysis SOP
    - Preparation SOP

SOP for any other relevant process

Pages from any applicable logbooks (instrument, standards, etc)

2____Evaluation of Selectivity. As applicable: e.g. Retention Time Window Study, second column confirmation, Interelement correction checks, spectral or fluorescence profiles, etc.

3_____ Initial Calibration Curve (Include Tune verification or similar (e.g. degradation checks) if applicable)

4 Method Detection Limit (MDL) Study (summary and raw data)

Water Soil Other

5_____ Reporting Limit Verification standard

- Spike a blank matrix at the RL and process through the entire method. MDL study should be able to be used if recovery is good. Note the spike level(s) and recovery(yies)
- 6 Demonstration of Capability (DOC) per analyst (Precision and Accuracy (P&A) verification)
  - 4 LCS for each matrix most acceptance criteria are in the methods. The MDL study may be used if DOC criteria are met.
  - Non-Standard methods 3 x (1 LCS at LOQ-25%, 50%, 75% of the calibration range + Blank) prepared each day. (see NELAC Chpt 5, appendix C.3.3 (b))

7 Acceptable PT sample(s) if available

Notes: PT sample required for all new methods PT sample required for all new analytes under NELAP

Submitted by _____ Date _____

8_____ Certification/Approval from Regulatory Agency where available.

QA Review / Acceptance _____ Date _____

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-1 of 21-22

#### **SECTION 21**

#### EQUIPMENT (AND CALIBRATIONS) (NELAC 5.5.5)

#### 21.1 <u>OVERVIEW</u>

TestAmerica purchases the most technically advanced analytical instrumentation for sample analyses. Instrumentation is purchased on the basis of accuracy, dependability, efficiency and sensitivity. Each laboratory is furnished with all items of sampling, preparation, analytical testing and measurement equipment necessary to correctly perform the tests for which the laboratory has capabilities. Each piece of equipment is capable of achieving the required accuracy and complies with specifications relevant to the method being performed. Before being placed into use, the equipment (including sampling equipment) is calibrated and checked to establish that it meets its intended specification. The calibration routines for analytical instruments establish the range of quantitation. Calibration procedures are specified in laboratory SOPs. A list of laboratory equipment and instrumentation is presented in Table 21-1.

Equipment is only operated by authorized and trained personnel. Manufacturer's instructions for equipment use are readily accessible to all appropriate laboratory personnel.

## 21.2 PREVENTIVE MAINTENANCE

**21.2.1** *TestAmerica Buffalo* follows a well-defined program to ensure proper equipment operation and to prevent the failure of laboratory equipment or instrumentation during use. This program of preventive maintenance helps to avoid delays due to instrument failure.

**21.2.2** Routine preventive maintenance procedures and frequency, such as lubrication, cleaning, and replacements, should be performed according to the procedures outlined in the manufacturer's manual. Qualified personnel must also perform maintenance when there is evidence of degradation of peak resolution, a shift in the calibration curve, loss of sensitivity, or failure to continually meet one of the quality control criteria.

- **21.2.2.1** Calibrations, routine maintenance, and adjustments are part of the analysts' and Department Managers' responsibilities. However, service contracts may be in place for some instruments to cover any major repairs.
- **21.2.2.2** High purity gases, reagents, and spare parts are kept on hand to minimize repair time and optimize instrument performance.

**21.2.3** Table 21-2 summarizes the schedule for routine maintenance. It is the responsibility of each Department Manager to ensure that instrument maintenance logs are kept for all equipment in his/her department. Preventative maintenance procedures may also be outlined in analytical SOPs or instrument manuals. (Note: for some equipment, the log used to monitor performance is also the maintenance log. Multiple pieces of equipment may share the same log as long as it is clear as to which instrument is associated with an entry.)

**21.2.4** Instrument maintenance logs are controlled and are used to document instrument problems, instrument repair and maintenance activities. Maintenance logs shall be kept for all

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-2 of 21-22

major pieces of equipment. Instrument maintenance logs may also be used to specify instrument parameters.

- **21.2.4.1** Documentation must include all major maintenance activities such as contracted preventive maintenance and service and in-house activities such as the replacement of electrical components, lamps, tubing, valves, columns, detectors, cleaning and adjustments.
- **21.2.4.2** Each entry in the instrument log includes the Analyst's initials, the date, a detailed description of the problem (or maintenance needed/scheduled), a detailed explanation of the solution or maintenance performed, and a verification that the equipment is functioning properly (state what was used to determine a return to control. e.g. CCV run on 'date' was acceptable, or instrument recalibrated on 'date' with acceptable verification, etc.).
- **21.2.4.3** When maintenance or repair is performed by an outside agency, service receipts detailing the service performed can be affixed into the logbooks adjacent to pages describing the maintenance performed. This stapled in page must be signed across the page entered and the logbook so that it is clear that a page is missing if only half a signature is found in the logbook.
- **21.2.5** In addition, the maintenance records contain:
- The identification of the instrument/equipment (instrument's Serial Number and Model Number)
- The date the instrument/equipment was put into use.
- If available, the condition when the instrument was received (e.g. new, used, reconditioned).

**21.2.6** If an instrument requires repair (subjected to overloading or mishandling, gives suspect results, or otherwise has shown to be defective or outside of specified limits) it shall be taken out of operation and tagged as out of service or otherwise isolated until such a time as the repairs have been made and the instrument can be demonstrated as operational by calibration and/or verification or other test to demonstrate acceptable performance. The laboratory shall examine the effect of this defect on previous analyses (refer to Sections 12 and 13).

**21.2.7** In the event of equipment malfunction that cannot be resolved, service shall be obtained from the instrument vendor manufacturer, or qualified service technician, if such a service can be tendered. If on-site service is unavailable, arrangements shall be made to have the instrument shipped back to the manufacturer for repair. Back up instruments, which have been approved, for the analysis shall perform the analysis normally carried out by the malfunctioning instrument. If the back up is not available and the analysis cannot be carried out within the needed timeframe, the samples shall be subcontracted using the procedures outlined in Section 8.

If an instrument is sent out for service or transferred to another facility, it must be recalibrated and verified (including new initial MDL study) prior to return to lab operations.

#### 21.3 SUPPORT EQUIPMENT

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-3 of 21-22

This section applies to all devices that may not be the actual test instrument, but are necessary to support laboratory operations. These include but are not limited to: balances, ovens, refrigerators, freezers, incubators, water baths, field sampling devices, temperature measuring devices and volumetric dispensing devices if quantitative results are dependent on their accuracy, as in standard preparation and dispensing or dilution into a specified volume. All raw data records associated with the support equipment are retained to document instrument performance. Laboratory SOPs BF-GP-001, "Calibration of Autopipettes and Repipetters" and BF-GP-002, "Support Equipment: Maintenance, Record Keeping and Corrective Actions of Analytical Balances, Temperature Control Devises and Reagent Water" provide additional detail on the monitoring and record keeping for support equipment.

## 21.3.1 Weights and Balances

The accuracy of the balances used in the laboratory is checked every working day, before use. All balances are placed on stable counter tops.

Each balance is checked prior to use with at least two certified ASTM type 1 weights spanning its range of use (weights that have been calibrated to ASTM type 1 weights may also be used for daily verification). ASTM type 1 weights used only for calibration of other weights (and no other purpose) are inspected for corrosion, damage or nicks at least annually and if no damage is observed, they are calibrated at least every 5 years by an outside calibration laboratory. Any weights (including ASTM Type 1) used for daily balance checks or other purposes are recalibrated/recertified annually to NIST standards (this may be done internally if laboratory maintains "calibration only" ASTM type 1 weights).

All balances are serviced annually by a qualified service representative, who supplies the laboratory with a certificate that identifies traceability of the calibration to the NIST standards.

All of this information is recorded in logs, and the recalibration/recertification certificates are kept on file.

## 21.3.2 pH, Conductivity, and Turbidity Meters

The pH meters used in the laboratory are accurate to  $\pm$  0.1 pH units, and have a scale readability of at least 0.05 pH units. The meters automatically compensate for the temperature, and are calibrated with at least two working range buffer solutions before each use.

Conductivity meters are also calibrated before each use with a known standard to demonstrate the meters do not exceed an error of 1% or one umhos/cm.

Turbidity meters are also calibrated before each use. All of this information is documented in logs.

Consult pH and Conductivity, and Turbidity SOPs for further information.

## 21.3.3 <u>Thermometers</u>

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-4 of 21-22

All reusable thermometers are calibrated on an annual basis with a NIST-traceable thermometer. Disposable glycol thermometers are discarded upon expiration and replaced with newly purchased thermometers. IR thermometers are verified daily and calibrated annually. Digital probes and thermocouples are calibrated guarterly.

The NIST thermometer is recalibrated every five years (unless thermometer has been exposed to temperature extremes or apparent separation of internal liquid) by an approved outside service and the provided certificate of traceability is kept on file. The NIST thermometer has increments of 0.2 °C, and has a range applicable to all method and certification requirements. The NIST traceable thermometer is used for no other purpose than to calibrate other thermometers.

All of this information is documented in logbooks. Monitoring method-specific temperatures, including incubators, heating blocks, water baths, and ovens, is documented in method-specific logbooks. More information on this subject can be found in the laboratory SOP BF-GP-020, "Thermometer Calibration".

#### 21.3.4 <u>Refrigerators/Freezer Units, Waterbaths, Ovens and Incubators</u>

The temperatures of all refrigerator units and freezers used for sample and standard storage are monitored each working day.

Ovens, waterbaths and incubators are monitored on days of use.

All of this equipment has a unique identification number, and is assigned a unique thermometer for monitoring.

Sample storage refrigerator temperatures are kept between > 0°C and  $\leq$  6 °C.

Specific temperature settings/ranges for other refrigerators, ovens waterbaths, and incubators can be found in method specific SOPs.

All of this information is documented in Daily Temperature Logbooks and method-specific logbooks.

#### 21.3.5 <u>Autopipettors, Dilutors, and Syringes</u>

Mechanical volumetric dispensing devices including burettes (except Class A Glassware) are checked for accuracy at least quarterly. Glass micro-syringes are considered the same as Class A glassware.

The laboratory maintains a sufficient inventory of autopipettors, and dilutors of differing capacities that fulfill all method requirements.

These devices are given unique identification numbers, and the delivery volumes are verified each day of use and calibrated gravimetrically, at a minimum, on a quarterly basis.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-5 of 21-22

For those dispensers that are not used for analytical measurements, a label is applied to the device stating that it is not calibrated. Any device not regularly verified can not be used for any quantitative measurements.

Micro-syringes are purchased from Hamilton Company. Each syringe is traceable to NIST. The laboratory keeps on file an "Accuracy and Precision Statement of Conformance" from Hamilton attesting established accuracy.

## 21.3.6 Field Sampling Devices (Isco Auto Samplers)

Each Auto Sampler (ISCO) is assigned a unique identification number in order to keep track of the calibration. This number is also recorded on the sampling documentation.

The Auto Sampler is calibrated monthly (or if not utilized monthly, immediately prior to its usage) by setting the sample volume to 100ml and recording the volume received. The results are filed in a logbook/binder. The Auto Sampler is programmed to run three (3) cycles and each of the three cycles is measured into a graduated cylinder to verify 100ml are received.

If the RSD (Relative Standard Deviation) between the 3 cycles is greater than 10%, the procedure is repeated and if the result is still greater than 10%, then the Auto Sampler is taken out of service until it is repaired and calibration verification criteria can be met. The results of this check are kept in a logbook/binder.

Additional calibration and use information is detailed in laboratory SOP BF-FS-006, "Calibration of Field Meter".

## 21.4 INSTRUMENT CALIBRATIONS

Calibration of analytical instrumentation is essential to the production of quality data. Strict calibration procedures are followed for each method. These procedures are designed to determine and document the method detection limits, the working range of the analytical instrumentation and any fluctuations that may occur from day to day.

Sufficient raw data records are retained to allow an outside party to reconstruct all facets of the initial calibration. Records contain, but are not limited to, the following: calibration date, method, instrument, analyst(s) initials or signatures, analysis date, analytes, concentration, response, type of calibration (Avg RF, curve, or other calculations that may be used to reduce instrument responses to concentration.)

Sample results must be quantitated from the initial calibration and may not be quantitated from any continuing instrument calibration verification unless otherwise required by regulation, method or program.

If the initial calibration results are outside of the acceptance criteria, corrective action is performed and any affected samples are reanalyzed if possible. If the reanalysis is not possible, any data associated with an unacceptable initial calibration will be reported with appropriate data qualifiers (refer to Section 13).

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-6 of 21-22

**Note:** Instruments are calibrated initially and as needed after that and at least annually.

#### 21.4.1 CALIBRATION STANDARDS

Calibration standards are prepared using the procedures indicated in the Reagents and Standards section of the determinative method SOP. However, the general procedures are described below.

- **21.4.1.1** For each analyte and surrogate (if applicable) of interest, prepare calibration standards at the minimum number of concentrations as stated in the analytical methods. If a reference or mandated method does not specify the number of calibration standards, the minimum number is three, not including blanks or a zero standard. All of the standard solutions are prepared using Class A volumetric glassware, calibrated pipettes, and/or microsyringes and appropriate laboratory quality solvents and stock standards.
- **21.4.1.2** Standards for instrument calibration are obtained from a variety of sources. All standards are traceable to NIST whenever possible. Dilution standards are prepared from stock standards purchased from commercial suppliers. A standard preparation log is maintained for each department, containing concentration, date of receipt, date of standard preparation, any dilutions made, lot number, supplier, type of solvent and a unique code number to identify the standard.
- **21.4.1.3** The lowest concentration calibration standard that is analyzed during an initial calibration must be at or below the stated reporting limit for the method based on the final volume of extract (or sample).
- **21.4.1.4** The other concentrations define the working range of the instrument/method or correspond to the expected range of concentrations found in actual samples that are also within the working range of the instrument/method. Results of samples not bracketed by initial instrument calibration standards (within calibration range to 3 significant figures) must be reported as having less certainty, e.g., defined qualifiers or flags (additional information may be included in the case narrative). The lowest calibration standard must be at or below the reporting limit. The exception to these rules is ICP methods or other methods where the referenced method does not specify two or more standards.
- **21.4.1.5** Given the number of target compounds addressed by some of the organic methods, it may be necessary to prepare several sets of calibration standards, each set consisting of the appropriate number of solutions at different concentrations. The initial calibration will then involve the analysis of each of these sets of the appropriate number of standards.
- **21.4.1.6** All initial calibrations are verified with a standard obtained from a second source and traceable to a national standard, when available (or vendor certified different lot if a second source is not available). For unique situations, such as air analysis where no other source or lot is available, a standard made by a different analyst would be

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-7 of 21-22

considered a second source. This verification occurs immediately after the calibration curve has been analyzed, and before the analysis of any samples.

#### 21.4.2 CALIBRATION FOR ORGANIC METHODS (GC, HPLC, GC/MS)

- **21.4.2.1** Many of the organic analytical methods utilize an internal standard calibration (GCMS and some GC). Because of the complex nature of the multipeak chromatograms produced by the method, some instruments necessitate the use of external standard calibration (most GC and HPLC). Surrogate compounds are included in the calibration processes for all appropriate organic analyses. For more details on the calibration types listed below, refer to SOP No. CA-Q-S-005, Calibration Curves.
- **21.4.2.2** Once the operating parameters have been established according to the method, each instrument is calibrated for the appropriate method. The analyst prepares five or more standard solutions at various concentrations containing all of the analytes of interest, internal standards, and surrogates that are appropriate for the method. Note: There are several EPA methods that have different requirements and are exceptions (e.g. EPA 625) where a minimum of 3 calibration standards are prepared and analyzed.
- **21.4.2.3** The standard solutions are introduced into the instrument in the same manner as samples are; whether by direct injection, by headspace analysis, or by purge and trap. The calibration factor (CF) for methods that use external standards, and the response factor (RF) for methods that use internal standards are calculated for the five standards.
  - External standard calibration involves comparison of instrument responses from the sample to the responses from the target compounds in the calibration standards. Sample peak areas (or peak heights) are compared to peak areas (or heights) of the standards. The ratio of the response to the amount of analyte in the calibration standard is defined as the Calibration factor (CF).
  - Internal standard calibration involves the comparison of instrument responses from the target compounds in the sample to the responses of specific standards added to the sample or sample extract prior to injection. The ratio of the peak area (or height) of the target compound in the sample or sample extract to the peak area (or height) of the internal standard in the sample or sample extract is compared to a similar ratio derived for each calibration standard. The ratio is termed the response factor (RF), and may also be known as a relative response factor in other methods.

In many cases, internal standards are recommended. These recommended internal standards are often brominated, fluorinated, or stable isotopically labeled analogs of specific target compounds, or are closely related compounds whose presence in environmental samples is highly unlikely. The use of specific internal standards is available in the method SOP.

Whichever internal standards are employed, the analyst needs to demonstrate that the measurement of the internal standard is not affected by method analytes and surrogates or by matrix interferences. In general, internal standard calibration is not as useful for GC and HPLC

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-8 of 21-22

methods with non-MS detectors because of the inability to chromatographically resolve many internal standards from the target compounds. The use of MS detectors makes internal standard calibration practical because the masses of the internal standards can be resolved from those of the target compounds even when chromatographic resolution cannot be achieved.

When preparing calibration standards for use with internal standard calibration, add the same amount of the internal standard solution to each calibration standard, such that the concentration of each internal standard is constant across all of the calibration standards, whereas the concentrations of the target analytes will vary. The internal standard solution will contain one or more internal standards and the concentration of the individual internal standards may differ within the spiking solution (e.g., not all internal standards need to be at the same concentration in this solution). The mass of each internal standard added to each sample extract immediately prior to injection into the instrument or to each sample prior to purging must be the same as the mass of the internal standard in each calibration standard. The volume of the solution spiked into sample extracts should be such that minimal dilution of the extract occurs (e.g., 10 uL of solution added to a 1 mL final extract results in only a negligible 1% change in the final extract volume which can be ignored in the calculations).

An ideal internal standard concentration would yield a response factor of 1 for each analyte. However, this is not practical when dealing with more than a few target analytes. Therefore, as a general rule, the amount of internal standard should produce an instrument response (e.g., area counts) that is no more than 100 times that produced by the lowest concentration of the least responsive target analyte associated with the internal standard. This should result in a minimum response factor of approximately 0.01 for the least responsive target compound. Refer to SOP No. CA-Q-S-005, Calibration Curves, for specific calculations.

- **21.4.2.4** Policies regarding the use of calibration standard results for creating the calibration curve are as follows:
  - A low calibration standard may be excluded from the calibration if the signal-to-noise ratio or spectral criteria are not suitable. The reporting level must be elevated to be the lowest calibration standard used for calibration.
  - The upper calibration standard may be excluded if it saturates the detector or is obviously becoming non-linear. Any sample exceeding the upper standard used in the calibration must be diluted and re-analyzed.
  - Mid-calibration standards may not be excluded unless an obvious reason is found, i.e., cracked vial, incorrectly made, etc. The failed standard should be re-run immediately and inserted into the initial calibration. If not useful, recalibration is required.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-9 of 21-22

#### 21.4.2.5 Percent RSD Corrective Action

Given the potentially large numbers of analytes that may be analyzed in some methods, it is likely that some analytes may exceed the acceptance limit for the RSD for a given calibration. In those instances, the following steps are recommended, but not required.

- **21.4.2.5.1** The first step is generally to check the instrument operating conditions. This option will apply in those instances where a linear instrument response is expected. It may involve some trade-offs to optimize performance across all target analytes. For instance, changes to the operating conditions necessary to achieve linearity for problem compounds may cause the RSD for other compounds to increase, but as long as all analytes meet the RSD limits for linearity, the calibration is acceptable.
- **21.4.2.5.2** If the RSD for any analyte is greater than the applicable acceptance criteria in the applicable analytical method SOP, the analyst may wish to review the results (area counts, calibration or response factors, and RSD) for those analytes to ensure that the problem is not associated with just one of the initial calibration standards. If the problem appears to be associated with a single standard, that one standard may be reanalyzed and the RSD recalculated. Replacing the standard may be necessary in some cases.
- **21.4.2.5.3** A third alternative is to narrow the calibration range by replacing one or more of the calibration standards with standards that cover a narrower range. If linearity can be achieved using a narrower calibration range, document the calibration linearity, and proceed with analyses. The changes to the upper end of the calibration range will affect the need to dilute samples above the range, while changes to the lower end will affect the overall sensitivity of the method. Consider the regulatory limits or action levels associated with the target analytes when adjusting the lower end of the range.

**Note:** When the purpose of the analysis is to demonstrate compliance with a specific regulatory limit or action level, the laboratory must ensure that the method quantitation limit is at least as low as the regulatory limit or action level.

- **21.4.2.6** Alternatively, the least squares regression may be used to determine linearity. A five point line must result in a correlation coefficient (r) of 0.990 or better using the least squares method to be considered acceptable. In many cases it may be preferred that the curves be forced through zero (not to be confused with including the origin as an additional data point, which is not allowed). **Note:** EPA method 8000B does not allow forcing through zero however the agency has revaluated this position and has since changed this stance to allow forcing through zero. In addition, from EPA Method 8000C: "However, the use of a linear regression or forcing the regression through zero may NOT be used as a rationale for reporting results below the calibration range demonstrated by the analysis of the standards.").
- **21.4.2.7** Instead of a linear curve model (either Average RF or least squares regression), a second order curve (Quadratic) may be used (and preferred) as long as it contains at least six data points. As a rule of thumb, if there is a consistent trend

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-10 of 21-22

in RFs (or CFs) in the calibration curve, either up or down, then quadratic curve fit may be indicated as the preferred calibration routine for that analyte. The coefficient of determination (COD or  $r^2$ ) for the quadratic curve must be at least 0.99 for it to be considered acceptable. For more details on the calculations see Calibration Curve SOP CA-Q-S-005. Some limitations on the use of Quadratic Curve fits:

- **21.4.2.7.1** Care MUST be exercised to assure that the results from this equation are real, positive, and fit the range of the initial calibration.
- **21.4.2.7.2** They **may not** be used to mask instrument problems that can be corrected by maintenance. (Not to be used where the analyte is normally found to be linear in a properly maintained instrument).
- **21.4.2.7.3** They **may not** be used to compensate for detector saturation. If it is suspected that the detector is being saturated at the high end of the curve, remove the higher concentration standards from the curve and try a 1st order fit or average RF.

## 21.4.3 <u>Calibration for Inorganic Analyses</u>

EPA Method 7000 from EPA SW-846 is a general introduction to the quality control requirements for metals analysis. For inorganic methods, quality control measures set out in the individual methods and in the *Standard Methods for the Examination of Water and Wastewater* (20th Edition) may also be included. Standard Operating Procedures for the analysis and the quality control documentation measures are maintained on the laboratory intranet (BufNet).

In general, inorganic instrumentation is calibrated with external standards. Some exceptions would be Inductively Coupled Plasma (ICP) and Inductively Coupled Plasma Mass Spec (ICPMS). These analyses may use an internal standard to compensate for viscosity or other matrix effects. While the calibration procedures are much the same for inorganics as they are for organics, CF's or RF's are not used. The calibration model in 21.4.2.6 is generally used for most methods, however in some instances the model from section 21.4.2.7 may be used. A correlation coefficient (r) of 0.995 or greater must be used to accept a calibration curve generated for an inorganic procedure. Correlation coefficients are determined by hand-held scientific calculators, excel spreadsheets or by computer programs and documented as part of the calibration raw data. Coefficients of calibration curves used for quantitation must be documented as part of the raw data. Curves are not allowed to be stored in calculator memories and must be written on the raw data for the purposes of data validation.

- **21.4.3.1** "Calibrations" for titrimetric analyses are performed by standardizing the titrants against a primary standard solution. See specific methods in *Standard Methods for the Examination of Water and Wastewater* (20th Edition) for more information.
- **21.4.3.2** Spreadsheets that are used for general chemistry calculations must have all cells containing calculations locked to prevent accidental changes to the calculations.

## 21.4.4 <u>Calibration Verification</u>

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-11 of 21-22

The calibration relationship established during the initial calibration must be verified at periodic intervals as specified in the laboratory method SOPs in accordance with the referenced analytical methods and NELAC (2003) standard, Section 5.5.5.10. The process of calibration verification applies to both external standard and internal standard calibration techniques, as well as to linear and non-linear calibration models.

**Note:** The process of calibration verification referred to is fundamentally different from the approach called "calibration" in some methods. As described in those methods, the calibration factors or response factors calculated during calibration are used to update the calibration factors or response factors used for sample quantitation. This approach, while employed in other EPA programs, amounts to a daily single-point calibration, and is not appropriate nor permitted in SW-846 chromatographic procedures for trace environmental analyses.

- **21.4.4.1** Generally, the initial calibrations must be verified at the beginning of each 12-hour analytical shift during which samples are analyzed. (Some methods may specify more or less frequent verifications). The 12-hour analytical shift begins with the injection of the calibration verification standard (or the MS tuning standard in MS methods). The shift ends after the completion of the analysis of the last sample or standard that can be injected within 12 hours of the beginning of the shift.
- **21.4.4.2** A continuing instrument calibration verification (CCV) must be repeated at the beginning and, for methods that have quantitation by external calibration models, at the end of each analytical batch. Some methods have more frequent CCV requirements (see specific SOPs). Most Inorganic methods require the CCV to be analyzed after ever 10 samples.
- **21.4.4.3** The acceptance limits for calibration verifications can be found in each method SOP. As a rule of thumb: GCMS  $\pm$  20%, GC and HPLC  $\pm$  15%, Inorganics:  $\pm$  10 or 15%. Actual methods may have wider or tighter limits; see the method SOP for specifics.
- **21.4.4.4** If the response (or calculated concentration) for an analyte is within the acceptance limits of the response obtained during the initial calibration, then the initial calibration is considered still valid, and the analyst may continue to use the CF, RF or % drift values from the initial calibration to quantitate sample results.
- **21.4.4.5** If the response (or calculated concentration) for any analyte varies from the mean response obtained during the initial calibration by more than the acceptance criteria, then the initial calibration relationship may no longer be valid. If routine corrective action procedures fail to produce a second consecutive (immediate) calibration verification within acceptance criteria, then either the laboratory has to demonstrate performance after corrective action with two consecutive successful calibration verifications, or a new initial instrument calibration must be performed. However, sample data associated with an unacceptable calibration verification may be reported as qualified data under the following special conditions:
  - **21.4.4.5.1** When the acceptance criteria for the calibration verification are exceeded high, i.e., high bias, and there are associated samples that are non-detects, then those non-detects may be reported. Otherwise, the samples affected by the unacceptable calibration verification shall be reanalyzed after a new calibration curve has been established, evaluated and accepted.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-12 of 21-22

21.4.4.5.2 When the acceptance criteria for the calibration verification are exceeded low, i.e., low bias, those sample results may be reported if they exceed a maximum regulatory limit/decision level. Otherwise, the samples affected by the unacceptable verification shall be reanalyzed after a new calibration curve has been established, evaluated and accepted. Alternatively, a reporting limit standard may be analyzed to demonstrate that the laboratory can still support non-detects at their reporting limit.

#### 21.4.4.6 Verification of Linear Calibrations

Calibration verification for linear calibrations involves the calculation of the percent drift or the percent difference of the instrument response between the initial calibration and each subsequent analysis of the verification standard. Use the equations below to calculate % Drift or % Difference, depending on the procedure specified in the method SOP. Verification standards are evaluated based on the % Difference from the average CF or RF of the initial calibration or based on % Drift or % Recovery if a linear or quadratic curve is used.

The Percent Difference is calculated as follows:

% Difference = 
$$(CF(v) \text{ or } RF(v)) - (Avg. CF \text{ or } RF) X$$
 100  
(Avg. CF or RF)

Where:

CF(v) or RF(v) = CF or RF from verification standard Avg. CF or RF = Average CF or RF from Initial Calibration.

The Percent Drift is calculated as follows:

The Percent Recovery is calculated as follows:

% Recovery = <u>Result</u> X 100 True Value

## 21.4.4.7 Verification of a Non-Linear Calibration

Calibration verification of a non-linear calibration is performed using the percent drift or percent recovery calculations described in 21.4.4.6 above.

Regardless of whether a linear or non-linear calibration model is used, if initial verification criterion is not met, then no sample analyses may take place until the calibration has been verified or a new initial calibration is performed that meets the specifications listed in the method SOPs. If the calibration cannot be verified after the analysis of a single verification standard,

## Company Confidential & Proprietary

then adjust the instrument operating conditions and/or perform instrument maintenance, and analyze another aliquot of the verification standard. If the calibration cannot be verified with the second standard, then a new initial calibration is performed.

All target analytes and surrogates, including those reported as non-detects, must be included in periodic calibration verifications for purposes of retention time confirmation and to demonstrate that calibration verification criteria are being met.

All samples must be bracketed by periodic analyses of standards that meet the QC acceptance criteria (e.g., calibration and retention time). The frequency is found in the determinative methods or SOPs.

**Note:** If an internal standard calibration is being used (basically GCMS) then bracketing standards are not required, only daily verifications are needed. The results from these verification standards must meet the calibration verification criteria and the retention time criteria (if applicable).

## 21.5 POLICY ON TENTATIVELY IDENTIFIED COMPOUNDS (TICS) – GC/MS ANALYSIS

For samples containing components not associated with the calibration standards, a library search may be made for the purpose of tentative identification. The necessity to perform this type of identification will be determined by the purpose of the analyses being conducted. Data system library search routines should not use normalization routines that would misrepresent the library or unknown spectra when compared to each other.

**Note:** If the TIC compound is not part of the client target analyte list but is calibrated by the laboratory and is both qualitatively and/or quantitatively identifiable, it will not be reported as a TIC. If the compound is reported on the same form as true TICs, it must be qualified and/or narrated that the reported compound is qualitatively and quantitatively (if verification in control) reported compared to a known standard that is in control (where applicable).

For example, the RCRA permit or waste delisting requirements may require the reporting of non-target analytes. Only after visual comparison of sample spectra with the nearest library searches may the analyst assign a tentative identification. See laboratory SOP's BF-MB-005 and BF-MV-007 for guidelines for making tentative identifications

#### Note:

For general reporting if TICs are requested, the ten (10), largest non-target analyte peaks whose area count exceeds 10% of the nearest internal standard will be termed "Tentatively Identified Compounds" (TICs). More or fewer TICs may be identified based on client requirements.

# 21.5.1 <u>TIC Reporting Limits</u>

In general Reporting limits cannot be specified because of the unknown nature of the TIC. Any reporting limit that is reported can only be evaluated as an estimate as the quantitation is based on the assumption that the TIC responds exactly as the IS responds which is most likely not the case. In general, it is not recommended to set a Reporting limit at too low of a concentration as it gives a false impression.

TICs that meet the required identification criteria at 10% area of the IS: The RL would be 10% of the concentration of the internal standard used for quantitation. (e.g. 2.5 ug/L for 8260B, 4.0 ug/L for 8270C). In general, if the 10% area criterion is not met, the TIC RLs should be set at a level approximately 5x the level of the poorest performer in the analysis.

If a compound meets the TIC criteria, the reporting limit will reflect the ratio between the TIC and the IS or 5x the level of the poorest performer whichever is lower.

## 21.6 POLICY ON GC/MS TUNING

Prior to any GCMS analytical sequence, including calibration, the instrument parameters for the tune and subsequent sample analyses within that sequence must be set.

Prior to tuning/auto-tuning the mass spec, the parameters may be adjusted within the specifications set by the manufacturer or the analytical method. These generally don't need any adjustment but it may be required based on the current instrument performance. If the tune verification does not pass it may be necessary to clean the source or perform additional maintenance. Any maintenance is documented in the maintenance log.

**21.6.1** The concentration of the BFB or DFTPP must be at or below the concentrations that are referenced in the analytical methods. Part of the purpose of the tune is to demonstrate sensitivity and analyzing solutions at higher concentrations does not support this purpose. Tune failures may be due to saturation and a lower BFB/DFTPP concentration may be warranted.

**21.6.2** Tune evaluations usually utilize the "Autofind" function and are set up to look at the apex +/- 1 scan and average the three scans. Background correction is required prior to the start of the peak but no more than 20 scans before. Background correction cannot include any part of the target peak.

## 21.6.3 Other Options or if Auto Tune Fails:

- **21.6.3.1** Sometimes the instrument does not always correctly identify the apex on some peaks when the peak is not perfectly shaped. In this case, manually identify and average the apex peak +/- 1 scan and background correct as in 21.6.2 above. This is consistent with EPA 8260 and 8270.
- **21.6.3.2** Or the scan across the peak at one half peak height may be averaged and background corrected. This is consistent with Standard Methods 6200, EPA 624 and EPA 625.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-15 of 21-22

- **21.6.3.3** Adjustments such as adjustments to the repeller and ion focus lenses, adjusting the EM Voltage, etc. may be made prior to tune verification as long as <u>all</u> of the subsequent injections in the 12 hour tune cycle are analyzed under the same MS tune settings and it is documented in the run sequence log and/or maintenance log that an adjustment was made. Excessive adjusting (more than 2 tries) without clear documentation is not allowed. Necessary maintenance is performed and documented in instrument log.
- **21.6.3.4** A single scan at the Apex (only) may also be used for the evaluation of the tune. For SW 846 and EPA 600 series methods, background correction is still required.
- **21.6.3.5** Cleaning the source or other maintenance may be performed and then follow steps for tune evaluation above. Note: If significant maintenance was performed, see methods 8000B or 8000C then the instrument may require recalibration prior to proceeding.

**21.6.4** Tune evaluation printouts must include the chromatogram and spectra as well as the Tune evaluation information. In addition, the verifications must be sent directly to the printer or pdf file (no screen prints for DFTPP or BFB tunes). This ability should be built into the instrument software.

**21.6.5** Since the limits are expressed in whole percentages, the results may be rounded to whole percentage before comparing to criteria when assessing the tune verification against the tune requirements. However, the comparison to the criteria is usually done automatically by the software and if the printout says "Fail" then there would have to be documentation of the hand calculation on the raw data and comparison to the criteria if the lab intends to still accept the tune. In most cases the analyst is better off performing an adjustment and rerunning the tune standard.

**21.6.6** All MS tune settings must remain constant between running the tune check and all other samples. It is recommended that a separate tune method not be used, however a separate method may be used as long as the MS conditions between the methods are the same as the sample analysis method and tracked so any changes that are made to the analysis method are also made to the tune method.

## Table 21-1.

# Laboratory Equipment and Instrumentation – TestAmerica Buffalo

Instrument Type	Manufacturer	Model Number	Serial Number	Year Put into Service	Condition When Received
GC/MS	Hewlett Packard	5973	US44621446	2005	new
	Hewlett Packard	5973	US52420646	2005	new
	Hewlett Packard	5973	US41720721	2004	new
	Hewlett Packard	5973	US35120354	2004	new
	Hewlett Packard	5973	US41720707	2004	new
	Hewlett Packard	5973	US10241053	2003	new
	Hewlett Packard	5973	US30965634	2003	new
	Hewlett Packard	5973	US03965692	2003	new
	Hewlett Packard	5973	US05060076	2001	new
	Hewlett Packard	5973	US05060084	2001	new
	Hewlett Packard	5973	US03950346	2001	new
	Hewlett Packard	5973	US82321636	2001	new
GC	Hewlett Packard	6890 dual uECD	CN10520009	2005	new
	Hewlett Packard	6890 dual uECD	CN10520010	2005	new
	Hewlett Packard	6890 dual uECD	CN10448015	2005	new
	Hewlett Packard	5890II dual ECD	3336A53126	1994	new
	Hewlett Packard	5890II dual ECD	3336A63465	1994	new
	Hewlett Packard	5890II dual ECD	3336A53464	1994	new
	Hewlett Packard	5890II dual ECD	3336A53463	1994	new
	Hewlett Packard	5890II dual ECD	3336A54409	1994	new
	Hewlett Packard	5890II dual ECD	3336A54408	1994	new
	Hewlett Packard	5890II FID/FID	3115A34892	1994	new
	Hewlett Packard	5890II PID/FID	3336A60622	1994	new
	Hewlett Packard	5890II Hall/PID	3235A54089	1994	new
	Hewlett Packard	5890II PID/FID	3336A53465	1994	new
	Hewlett Packard	5890II dual FID	3336A53727	1994	new
	Hewlett Packard	5890II dual ECD	3310A47661	1993	new
	Hewlett Packard	5890II dual ECD	3336A53325	1993	new
	Hewlett Packard	5890II PID/FID	3133A37157	1993	new
	Hewlett Packard	5890II dual ECD	3203A42206	1992	new
	Hewlett Packard	5890II dual FID	3019A28433	1991	new
	Hewlett Packard	5890II Hall/PID	3121A35782	1990	new
LC	Hewlett Packard	1100 HPLC Fluor./DAD	DE92001578	2000	new
Metals	Perkin Elmer	Elan 9000 ICP-MS	P0230202	2002	new
	Leeman	PS200 II	HG9045	2000	new
	Leeman	PS200 II	HG0033	2000	new
	Thermo Jarrell Ash	ICP61E Trace	334490	1995	new
	Thermo Jarrell Ash	ICP61E Trace	382590	1995	new

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-17 of 21-22

Instrument Type	Manufacturer	Model Number	Serial Number	Year Put into Service	Condition When Received
Water Quality	Konelab	20XT	E3719731	2005	new
-	Thermo ECA	1200 TOX/AOX	2004.901	2004	new
	Dionex Ion				new
	Chromatograph	DX-120	20126	2004	
	Konelab	20	S5019455	2004	new
		CN Midi-			new
	Glastron	distillation	2502	2003	
		Phenol Midi-			new
	Glastron	distillation	2069	2003	
		Phenol Midi-			new
	Glastron	distillation	2053	2003	
		BOD Magic -			new
	Labtronics	Autoanalyzer	270H3XB531	2004	
		BOD Magic -			new
	Labtronics	Autoanalyzer	270J2XB669	2003	
	ManTech	PC Titrator	MS-OK2-607	2003	new
	HACH				new
	Spectrophotometer	DR/2500	30200004886	2003	
	Dionex Ion				new
	Chromatograph	DX-120	2060196	2002	
	OI Carbon Analyzer	1010 #2	H014710903	2000	new
	Spectronic Genesis	4001/4	3SGC199091	2000	new
		8000			new
	Lachat Quickchem	Autoanalyzer	A83000-1527	2000	
	OI Carbon Analyzer	1010 #1	H92170411	1999	new
		8000			new
	Lachat Quickchem	Autoanalyzer	A83000-1439	1999	
	Dionex Ion				new
	Chromatograph	DX-120	99010157	1999	
	Orion	Ion Meter 230A	2229	1999	new
	VWR Ion	Meter 2100	1063	1997	new
	YSI	Oxygen Meter 57	93J09826	1995	new
		Hi-Lo BOD			new
	Lab-Line	chamber	391-010	1994	
		Accumet Ion		4004	new
	Fischer	Meter 925	860	1991	
Sample	U2 ACCUPREP GPC		03F-10723	2003	new
Preparation	I urboVap II	TurboVap II	IV9445N5816	1996	new
	TurboVap II	TurboVap II	1V9427N4133	1996	new
	TurboVap II	TurboVap II	TV944N5819	1996	new
	TurboVap II	TurboVap II	TV944N5820	1996	new
	TurboVap II	TurboVap II	TV0024N9623	2000	new
	TurboVap II	TurboVap II	TV0022N9604	2000	new

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-18 of 21-22

Instrument Type	Manufacturer	Model Number	Serial Number	Year Put into Service	Condition When Received
		TurboVap II	TV0312N1159		new
	TurboVap II		2	2003	
		TurboVap II	TV0312N1159		new
	TurboVap II		1	2003	
	Organomation	Rot-X-Tractor	16902	1999	new
	Organomation	Rot-X-Tractor	16907	1999	new
	Organomation	Rot-X-Tractor	16913	1999	new
	Heat Systems				new
	Sonicator	#XL-2020	G1647/C5659	1994	
	Heat Systems				new
Sample	Sonicator	#XL-2020	G2665/C5674	1994	
Preparation	Heat Systems				new
(Continued)	Sonicator	#XL-2020	G2620/C5660	1994	
	Heat Systems				new
	Sonicator	#XL-2020	G2245/C6328	1995	
	Heat Systems				new
	Sonicator	#XL-2020	G2621/C6733	1995	
	Heat Systems				new
	Sonicator	#XL-2020	G2713/C6732	1995	
	Heat Systems				new
	Sonicator	#XL-2020	G1643/C6837	1995	
	Heat Systems				new
	Sonicator	#XL-2020	G2742/C6842	1995	

Table 21-2.

#### Schedule of Routine Maintenance

Instrument	Procedure	Frequency
Leeman Mercury Analyzer	Check tubing for wear Fill rinse tank with 10% HCl Change dryer tube Fill reductant bottle with 10% Stannous Chloride	Daily Daily As Needed Daily
ICP & ICP/MS	Check pump tubing Check liquid argon supply Check fluid level in waste container Check re-circulator levels Clean or replace filters Check torch Check sample spray chamber for debris Clean and align nebulizer Change pump oil Change Cones Change printer cartridge Replace pump tubing	Daily Daily Daily Monthly As required Daily Monthly Monthly Monthly As required As required As required
UV-Vis Spectrophotometer	Clean ambient flow cell Precision check/alignment of flow cell Wavelength verification check	As required As required Annually
Auto Analyzers	Clean sampler Check all tubing Clean inside of colorimeter Clean pump well and pump rollers Clean wash fluid receptacle Oil rollers/chains/side rails Clean optics and cells	Daily Daily Daily Quarterly Weekly Weekly Quarterly
Agilent GC/MS	Pump oil-level check Pump oil changing Analyzer bake-out Analyzer cleaning Resolution adjustment COMPUTER SYSTEM AND PRINTER: Air filter cleaning Change data system air filter Printer head carriage lubrication Paper sprocket cleaning	Monthly Annually As required As required As required As required As required As required

Instrument	Procedure	Frequency
Gas Chromatograph	Compare standard response to previous day or since last initial calibration	Daily
······	Check carrier gas flow rate in column	Daily via use of known compound retention
	Check temp. of detector, inlet, column oven	Daily
	Septum replacement	As required
	Check system for gas leaks with SNOOP	W/cylinder change as required
	Check for loose/frayed power wires and	As Required
	Bake injector/column	As Required
	Change/remove sections of guard column	As Required
	Replace connectors/liners Change/replace column(s)	As Required
Electron Capture Detector (ECD)	Detector wipe test (Ni-63) Detector cleaning	Semi-annually As required
Flame Ionization Detector (FID)	Detector cleaning	As required
Photoionization Detector (PID)	Change O-rings Clean lamp window	As required As required
HPLC	Change guard columns	As required
	Change pump seals	Semi-annually or as
	Replace tubing	As required
	Change fuses in power supply	As required
	Filter all samples and solvents	Daily
	Change autosampler rotor/stator	As required
Air Compressor	Drained Belts checked	VVEEKIY
	Lubricated	Semi-annually
Centrifuge	Check brushes and bearings	Every 6 months or as needed

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-21 of 21-22

## Table 21-3.

# **Periodic Calibration**

Instrument	Type of Calibration/ Number of Standards	Frequency	Acceptance Limits	Corrective Action
Analytical Balance	Accuracy determined using "S" NIST traceable weights. Minimum of 2 standards bracketing the weight of interest.	Daily, when used	± 0.2%	Clean, check level, insure lack of drafts, and that unit is warmed up, recheck. If fails, call service.
	Inspected and calibrated by A2LA accredited person annually.	Annual		
Top Loading Balance	Accuracy determined using "S" NIST traceable. Minimum of 2 standards bracketing the weight of interest.	Daily, when used	± 0.5%	Clean. Replace.
	Inspected and calibrated by A2LA accredited person annually.	Annual		
NIST Certified Weights	Accuracy determined by accredited weights and measurement laboratory.	1 year	As per certificate.	Replace.
NIST- Traceable Thermometer	Accuracy determined by accredited measurement laboratory.	5 years	As per certificate.	Replace.
Thermometer	Against NIST-traceable thermometer	Yearly at appropriate temperature range for intended use	± 1.2°C	Replace
Minimum- Maximum Thermometers	Against NIST-traceable thermometer	Yearly	± 1.5°C	Replace
InfraRed Temperature Guns	Against NIST-traceable thermometer	Daily at appropriate temperature range for intended use.	± 1.5°C	Repair/replace
	Accuracy determined by accredited measurement laboratory.	Annual		
Dial-type Thermometers	Against NIST-traceable thermometer	Quarterly at appropriate temperature range for intended use.	± 1.5°C	Replace

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 21-22 of 21-22

Instrument	Type of Calibration/ Number of Standards	Frequency	Acceptance Limits	Corrective Action
Refrigerator	Temperature checked using NIST-traceable thermometer.	Daily. If out of range, check again in two hours.	0-6°C	Adjust. Repair. While waiting for repair, seal door, attach "Out of Service" sign, move items to functional unit. Notify supervisor.
Freezer	Temperature checked using NIST-traceable thermometer	Daily. If out of range, check again in two hours.	(-10)-(-20)°C	Adjust. Repair. While waiting for repair, seal door, attach "Out of Service" sign, move items to functional unit. Notify supervisor.
Oven	Temperature checked using NIST-traceable thermometer.	When in use.	104 ± 1°C (drying) 180 ± 2°C (TDS)	Adjust. Replace.
Water Bath	Temperature checked using NIST-traceable thermometer.	When in use.	± 2°C	Adjust. Replace.
Volumetric Dispensing Devices (Eppendorf ® pipette, automatic dilutor or dispensing devices)	One delivery by weight. Using DI water or solvent of use, dispense into tared vessel. Record weight with device ID number. Calibrate using 4 replicate gravimetric measurements	Each day of use Quarterly	± 2% Calculate accuracy by dividing weight by stated volume times 100 for percent.	Adjust. Replace.
Glass Microliter Syringes	None	Accuracy must be initially demonstrated if syringe was not received with a certificate attesting to established accuracy.	± 1%	Not applicable.
Deionized Water	Check in-line conductivity meter on system with conductivity meter in Inorganics Department.	Daily	<1.0 µmho at 25°C	Record on log. Report discrepancies to QA Manager, Operations Manager or Technical Director.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 22-1 of 22-6

#### **SECTION 22**

#### MEASUREMENT TRACEABILITY (NELAC 5.5.6)

#### 22.1 OVERVIEW

Traceability of measurements shall be assured using a system of documentation, calibration, and analysis of reference standards. Laboratory equipment that are peripheral to analysis and whose calibration is not necessarily documented in a test method analysis or by analysis of a reference standard shall be subject to ongoing certifications of accuracy. At a minimum, these must include procedures for checking specifications of ancillary equipment: balances, thermometers, temperature, Deionized (DI) and Reverse Osmosis (RO) water systems, automatic pipettes and other volumetric measuring devices. With the exception of Class A Glassware (including glass microliter syringes that have a certificate of accuracy), quarterly accuracy checks are performed for all mechanical volumetric devices. Wherever possible, subsidiary or peripheral equipment is checked against standard equipment or standards that are traceable to national or international standards. The following definitions are provided by the American Association for Laboratory Accreditation (A2LA):

"Traceability is the property of a measurement result whereby it can be related to stated references, usually national or international standards, through an unbroken chain of comparisons, each step in the chain having stated uncertainties." There are six essential elements:

- An unbroken chain of comparison
- A calculated measurement uncertainty for each step in the chain to allow for an overall uncertainty calculation
- Documentation of each step in each calibration report
- All steps in the chain are performed by individuals with evidence of technical competence and accredited by a recognized accreditation body
- Reference to International Standard (SI) units
- Recalibration at appropriate intervals to preserve traceability

Calibration is defined as "determining and documenting the deviation of the indication of a measuring instrument (or the stated value of a material measure) from the conventional 'true' value of the measurand."

Uncertainty is defined as "a parameter associated with the result of a measurement that characterizes the dispersion of the value that could reasonably be attributed to the measurand." Measurement of Uncertainty is discussed is Section 20 of this QA Manual.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 22-2 of 22-6

#### 22.2 NIST-TRACEABLE WEIGHTS AND THERMOMETERS

Reference standards of measurement shall be used for calibration only and for no other purpose, unless it can be shown that their performance as reference standards would not be invalidated.

For NIST-traceable weights and thermometers, the laboratory requires that all calibrations be conducted by a calibration laboratory accredited by A2LA, NVLAP (National Voluntary Laboratory Accreditation Program), (APLAC (Asia-Pacific Laboratory Accreditation Cooperation), or EA (European Cooperation for Accreditation). A certificate and scope of accreditation is kept on file at the laboratory. Refer to Section 21 for calibration of weights and thermometers.

The calibration report or certificate submitted to **TestAmerica Buffalo** contains, in a well designed format, a traceability statement, the conditions under which the calibrations were made in the context of any potential influence, a compliance statement with an identified metrological specification and the pertinent clauses, a clearly identified record of the quantities and functional test results before and after re-calibration, and no recommendation on the calibration interval. Opinions and interpretations of results are presented along with the basis upon which they were made and identified as such. The report may be submitted by facsimile or other electronic means as long as the requirements of the International Standard are achieved. If significant amendments are made to a calibration certificate, a supplemental certificate for the serial-number-specified piece of equipment is so identified. When a new certificate is offered, it uniquely identifies and references the one it replaces. All calibration reports are filed in the QA Office.

An external certified service engineer services laboratory balances on an annual basis. This service is documented on each balance with a signed and dated certification sticker. Balance calibrations are checked each day of use. All mercury thermometers are calibrated annually against a traceable reference thermometer. Temperature readings of ovens, refrigerators, and incubators are checked on each day of use.

## 22.3 REFERENCE STANDARDS / MATERIALS

Reference standards/materials, where commercially available, are traceable to certified reference materials. Commercially prepared standard materials are purchased from vendors accredited by A2LA or NVLAP with an accompanying Certificate of Analysis that documents the standard purity. If a standard cannot be purchased from a vendor that supplies a Certificate of Analysis, the purity of the standard is documented by analysis. (Refer to Section 9 for additional information on purchasing). The receipt of all reference standards must be documented. Reference standards are labeled with a unique Standard Identification Number and expiration date. All documentation received with the reference standard is retained as a QC record and references the Standard Identification Number.

All reference, primary and working standards/materials, whether commercially purchased or laboratory prepared, must be checked regularly to ensure that the variability of the standard or material from the 'true' value does not exceed method requirements. The accuracy of calibration standards is checked by comparison with a standard from a second source. In cases where a second standard manufacturer is not available, a vendor certified different lot is acceptable for

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 22-3 of 22-6

use as a second source. For unique situations, such as air analysis where no other source or lot is available, a standard made by a different analyst would be considered a second source. The appropriate Quality Control (QC) criteria for specific standards are defined in laboratory SOPs. In most cases, the analysis of an Initial Calibration Verification (ICV) or LCS (where there is no sample preparation) is used as the second source confirmation. These checks are generally performed as an integral part of the analysis method (e.g. calibration checks, laboratory control samples).

All standards and materials must be stored and handled according to method or manufacturer's requirements in order to prevent contamination or deterioration. Refer to Table 9-1 in Section 9 for general storage requirements and Table 22-1 for additional storage information. Method specific information may also be found in the laboratory method SOPs in the "Standards and Reagents" sections. For safety requirements, please refer to method SOPs and the laboratory Environmental Health and Safety Manual.

## 22.4 <u>DOCUMENTATION AND LABELING OF STANDARDS, REAGENTS, AND</u> <u>REFERENCE MATERIALS</u>

Reagents must be at a minimum the purity required in the test method. The date of reagent receipt and the expiration date are documented. The lots for most of the common solvents and acids are tested for acceptability prior to company wide purchase. Refer to SOP No. CA-Q-S-001, Solvent and Acid Lot Testing and Approval.

All manufacturer or vendor supplied Certificate of Analysis or Purity must be retained, stored appropriately, and readily available for use and inspection. These records are maintained by each department in bound or electronic folders. Records must be kept of the date of receipt and date of expiration of standards, reagents and reference materials. In addition, records of preparation of laboratory standards, reagents, and reference materials must be retained, stored appropriately, and be readily available for use and inspection. For detailed information on documentation and labeling, please refer laboratory SOP BF-GP-019, "Standard Traceability and Preparation" and also to the method specific SOPs.

Commercial materials purchased for preparation of calibration solutions, spike solutions, etc.., are usually accompanied with an assay certificate or the purity is noted on the label. If the assay purity is 96% or better, the weight provided by the vendor may be used without correction. If the assay purity is less than 96% a correction will be made to concentrations applied to solutions prepared from the stock commercial material.

**22.4.1** All standards, reagents, and reference materials must be labeled in an unambiguous manner. Standards are logged into the laboratory department's chemical history log and are assigned a unique identification number. Preparation of working standards or reagents prepared from the stock is documented in the laboratory Department's Standard Preparation Log. The following information is typically recorded:

- Standard ID
- Description of Standard
- Department
- Preparer's name

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 22-4 of 22-6

- Final volume and number of vials prepared
- Solvent type and lot number
- Preparation Date
- Expiration Date
- Standard source type (stock or daughter)
- Standard type (spike, surrogate, other)
- Parent standard ID (if applicable)
- Parent Standard Analyte Concentration (if applicable)
- Parent Standard Amount used (if applicable)
- Component Analytes
- Final concentration of each analyte
- Comment section

Records are maintained for standard and reference material preparation. These records show the traceability to purchased stocks or neat compounds. These records also include method of preparation, date of preparation, expiration date and preparer's name or initials. Preparation procedures are provided in the Method SOPs.

**22.4.2** All standards, reagents, and reference materials must be clearly labeled with a minimum of the following information:

- Expiration Date
- Standard ID
- Special Health/Safety warnings if applicable

**22.4.3** In addition, the following information may be helpful:

- Date of receipt for commercially purchased items or date of preparation for laboratory prepared items
- Date opened (for multi-use containers, if applicable)
- Description of standard (if different from manufacturer's label or if standard was prepared in the laboratory)
- Concentration (if applicable)
- Initials of analyst preparing standard or opening container

All containers of prepared reagents must include a preparation date, expiration date and an ID number to trace back to preparation.

Procedures for preparation of reagents can be found in the Method SOPs.

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 22-5 of 22-6

Standard ID numbers must be traceable through associated logbooks, worksheets and raw data.

All reagents and standards must be stored in accordance to the following priority: 1) with the manufacturer's recommendations; 2) with requirements in the specific analytical methods; and 3) according to Table 22-1.
Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 01/14/2008 Page 22-6 of 22-6

# Table 22-1.

# **Standard Sources and Preparation**

Instrument	Source	How Received	Stock Storage	Preparation	Intermediate & Working Standard Storage	Frequency
ICP	SPEX; Environmental Express: CPI	1000 ppm Solutions	Room Temperature	Working standards from stock	Room Temperature	Daily
CC	Supelco; Restek; Ultra	Solutions	Freezer (-10°C)	Working standards from stock	Refrigerate	Monthly
ТОХ	Chemservice; ERA	Solutions	Refrigerate	Working standards from stock	Refrigerate	Monthly
TOC	Ricca; ERA	Solutions	Refrigerate	As received	Refrigerate	N/A
Volatile Organics	Restek; Supelco; Ultra	Ampoule/ Solutions	Freezer (-10°C)	Working standards from stock	Refrigerate	Monthly; Gas, weekly
Semi-Volatile Organics	Restek; Supelco; Ultra	Ampoule/ Solutions	Refrigerate or Room temp.	Working standards from stock	Refrigerate	Monthly
Infrared Spec- trophotometry	Hach; Accustandard	Pure Reagent	Room Temperature	Working standards from stock	Refrigerate	Weekly
lon Chromatography	Ultra; Accustandard	Solutions	Refrigerate	Working standards from stock	Refrigerate	Monthly
Lachat; Konelab	ERA; Ricca;Accustan dard; Ultra	Solutions	Refrigerate	Working standards from stock	Refrigerate	Weekly, monthly
BOD Magic	Ricca, ERA	Solution	Refrigerate	As received	Refrigerate	Daily

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-1 of 23-20

#### SECTION 23.0

#### SAMPLING (NELAC 5.5.7)

#### 23.1 <u>OVERVIEW</u>

*TestAmerica Buffalo* provides sampling services. Sampling procedures are described in the following SOPs:

- BF-FS-001 Chain of Custody Documentation
- **BF-FS-002** Sample Packaging and Shipment Off-Site
- **BF-FS-003** Groundwater Sampling Field Data Collection
- BF-FS-004 Equipment Decontamination
- BF-FS-005 Groundwater/Surface Water Sampling
- BF-FS-006 Calibration of Field Meter
- BF-FS-007 Low Flow Sampling Procedures
- BF-FS-008 Surface and Subsurface Soil/Sediment Sampling

#### 23.2 SAMPLING CONTAINERS

The laboratory offers clean sampling containers for use by clients. These containers are obtained from reputable container manufacturers and meet EPA specifications as required. Any certificates of cleanliness that are provided by the supplier are maintained at the laboratory.

#### 23.2.1 <u>Preservatives</u>

Upon request, preservatives are provided to the client in pre-cleaned sampling containers. In some cases containers may be purchased pre-preserved from the container supplier. Whether prepared by the laboratory or bought pre-preserved, the grades of the preservatives are at a minimum:

- Hydrochloric Acid Reagent ACS (Certified VOA Free) or equivalent
- Methanol Purge and Trap grade
- Nitric Acid Instra-Analyzed or equivalent
- Sodium Bisulfate ACS Grade or equivalent
- Sodium Hydroxide Instra-Analyzed or equivalent
- Sulfuric Acid Instra-Analyzed or equivalent
- Sodium Thiosulfate ACS Grade or equivalent

#### 23.2.2 Preparing Container Orders

When new containers arrive at the laboratory, the date of receipt is recorded on the packing list received with them for retained documentation. Upon request or based on scheduled monitoring events, the containers are then sent to clients for use in collecting samples. The

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-2 of 23-20

shipping date, type and number of containers are maintained on file by the lab. Shipping personnel insure that container stock is rotated so that "first in" is "first out." When a client requests containers, a project management representative creates a container request in LIMS; it is then stored permanently in LIMS with a unique container order number. Copies of the container request are printed by the bottle kit preparation department. One copy goes to the client with the containers; one copy is filed in the bottle kit preparation department. For additional detail refer to laboratory SOPs BF-SR-001, "Cooler Shipping – Bottle Kits and Samples" and BF-PM-003, "Bottle Order Set-up".

The laboratory also provides EnCore, TerraCore or other soil sampling devices when requested.

If containers are provided directly to the client from the manufacturer or from other sources, the laboratory will not be responsible for any of the above records.

#### 23.3 FIELD QUALITY CONTROL (QC)

Common field quality control samples are defined in the following paragraphs. The frequency of field quality control samples should be specified in the site specific Quality Assurance Project Plan (QAPP) or by the client. TestAmerica provides trip blanks for VOC analysis with the sample containers for all volatile organic analyses. Blanks generated in the field will be analyzed along with the field samples (exception soil samples where the blank is aqueous).

**23.3.1** Equipment Blank / Rinsate Blank - The equipment blank, sometimes referred to as a rinsate blank, is a sample of the water used to decontaminate sampling equipment. The source water should be as free of target analytes as possible. An aliquot of this water is poured over or through the sample collection device after decontamination, collected in a sample container, preserved with appropriate reagents, and returned to the laboratory. This serves as a check on sampling device cleanliness, and will also be affected by the site and sample handling conditions evaluated by the other types of blanks. The sampling time for the equipment blank should begin when the equipment is rinsed and the water is collected.

**23.3.2** <u>Field Blank</u> - The field blank is water that is as free of target analytes as possible and from the same source as the equipment blank. The water is poured into a sampling container at the sampling site, preserved with the appropriate reagents, and returned to the laboratory. This serves as a check on reagent and environmental contamination. The sampling time for the field blank should be when the blank is prepared in the field.

**23.3.3** <u>Trip Blank</u> - The trip blank pertains to volatile analysis only. This serves as a check on sample contamination originating from sample transport, sample container contamination, shipping and storage, or from certain site conditions. Trip blanks are often referred to as travel blanks. They are prepared using pre-cleaned sample containers. They are filled with organic-free water (the source of the organic free water is the same source of water used to prepare volatile standards, method blanks, LCS and sample dilutions), sealed and taken into the field with the empty containers which will be used for sampling. The recommended frequency is one trip blank per cooler (in duplicate or triplicate), per volatiles method. Unless otherwise specified, the sampling time for the trip blank is the time of receipt at the laboratory (When the "Trip" ends).

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-3 of 23-20

**23.3.4** <u>Field Duplicates</u> - Field duplicates are replicate samples collected from the same sampling point or location during a field collection event. This control sample is used to demonstrate the ability of both the sampling and analytical process to generate data of acceptable precision.

#### 23.4 DEFINITION OF HOLDING TIME

The date and time of sampling documented on the chain-of-custody (COC) form establishes the day and time zero. As a general rule, when the maximum allowable holding time is expressed in "days" (e.g 14 days, 28 days), the holding time is based on calendar day measured. Holding times expressed in "hours" (e.g. 6 hours, 24 hours, etc.) are measured from date and time zero. The first day of holding time for time critical parameters ends twenty-four hours after sampling. Holding times for analysis include any necessary reanalysis. However there are some programs that determine holding time compliance based on the date and specific time of analysis compared to the time of sampling regardless of how long the holding time is. These programs will be addressed on a case-by-case basis.

**23.4.1** <u>Semi-Volatile</u> - Holding times for sample preparation for semi-volatile organics are measured from the sampling date until the day solvent contacts the sample. Holding times for analysis are measured from the date of initiation of extraction to the date/time of injection into the gas chromatograph.

**23.4.2** <u>Volatiles</u> - Holding times for volatile organics are measured from the date (and time where applicable) of sampling to the date and time of injection into the gas chromatograph. The time of initiation of purging is considered the injection time, but data systems record the start of the chromatographic run rather than the start of purging. Hence, if a sample is so near expiration that the start-of-purging time rather than the chromatographic run time is needed to document the integrity of the sample; the analyst must observe and record the start-of-purging time in the instrument log. Medium-level extractions, e.g. for high level soils, must be completed in time to allow for analysis to be initiated within the maximum allowable holding time.

**23.4.3** <u>Inorganics</u> - For inorganic and metals analysis, the preparation/digestion/distillation must be started within the maximum holding time as measured from the sampling date (and time where applicable).

#### 23.5 SAMPLING CONTAINERS, PRESERVATION REQUIREMENTS, HOLDING TIMES

The preservation and holding time criteria specified in the following tables are derived from the source documents for the methods. If method required holding times (refer to Tables 23-1 to 23-7) or preservation requirements are not met, the reports will be qualified using a flag, footnote or case narrative. As soon as possible or "ASAP" is an EPA designation for tests for which rapid analysis is advised, but for which neither EPA nor the laboratory have a basis for a holding time.

#### 23.6 SAMPLE ALIQUOTS / SUBSAMPLING

Taking a representative sub-sample from a container is necessary to ensure that the analytical results are representative of the sample collected in the field. The size of the sample container, the quantity of sample fitted within the container, and the homogeneity of the sample need consideration when sub-sampling for sample preparation. It is the laboratory's responsibility to

take a representative subsample or aliquot of the sample provided for analysis. In that regard the following guidelines apply to analysts:

Analysts should handle each sample as if it is potentially dangerous. At a minimum, safety glasses, gloves, and lab coats must be worn when preparing aliquots for analysis.

The following information provides general guidance for homogenization and subsampling. For laboratory specific procedures refer to SOP BF-GP-005, "Sample Homogenization and Subsampling".

**23.6.1** For water samples, before taking each aliquot for analysis, invert the sample container end-over-end three times and immediately pour off the aliquot. Especially when suspended solids are present, adequate mixing of the sample is extremely important.

**23.6.2** For solid samples, when volatile organics are not requested, if the solid can be mixed, stir before removing the aliquot. Mix more than is needed for the analysis to be performed (e.g. if 30 g are needed, mix 50-100 g, if 1 g is needed, mix 20 g, etc...).

- If the solid cannot be easily mixed: After thoroughly mixing the sample within the sample container or, for non-organic methods, the sample can be transferred to a clean glass container, Teflon boat or other suitable container for manual mixing, a sub-sample from various quadrants and depths of the sample are taken to acquire the required sample weight.
- For soil samples, avoid debris in the subsample aliquot as much as possible (e.g. gravel, sticks, roots and grass); note this information in the sample preparation record.
- If the solid is extremely heterogeneous, and the client has given no instructions, utilize the following technique: separate the like materials into groups on a clean surface and take portions of masses from each group, proportional to their contribution to the original sample, to make a composite. Record in detail exactly how the composite was created. For very unusual samples, consult with the Technical Director or Department Manager.

**23.6.3** For solid samples, when volatile organics analysis is requested, the sample should be manipulated as little as possible. In most cases, the sample will arrive already preserved or in an EnCoreTM sampler of the correct mass (requiring quick preservation of the entire amount). If the client requests volatiles on a solid sample which has been collected in a jar and is in a common container from which aliquots for other test methods must be taken, sample management personnel should deliver the container to the volatiles department for preparing a proper aliquot prior to any other aliquots being taken out.

**23.6.4** For multiphasic samples, the client should instruct the laboratory as to the intent of the testing and how to handle the sample. If the entire sample is to be accounted for, and the phases do not mix easily with inversion/stirring, such that a representative aliquot can be taken, the analyst should record the percent by volume of each phase. The analysis must be conducted on each phase separately; the final results are combined mathematically, weighting the individual phase results by volume. One exception to this procedure is the situation addressed in the TCLP and SPLP methods for wastes containing free liquids. However, if the leachate and final filtrate are not miscible, it is necessary to combine mathematically the concentrations of the two (or more) solutions by volume.

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-5 of 23-20

Tables 23-1 to 23-7 detail holding times, preservation and container requirements, and sample volumes for SDWA and NPDES methods. The sample volumes are intended to be a minimal amount to perform the method, the containers that are used may be of larger size. TestAmerica Buffalo does not currently retain capability for all listed methods or parameters however the additional details are included for reference purposes. *Please note: the holding times are program specific and different programs may have different holding times for equivalent methods (e.g., there are difference in Holding times for many Organic analytes between SDWA and NPDES. RCRA methods may also be different.)* 

 Table 23-1.

 Holding Times, Preservation and Container Requirements: Drinking Water (SDWA)

PARAMETER	CONTAINER	PRE Temp. ²	SERVATION ^{1,2} ³ Chemical	HOLDING TIME ³	SAMPLE VOLUME
Asbestos	Plastic/Glass	4°C	None	48 hours⁵	1 L
Coliforms (Total and Fecal)	Plastic/Glass ²⁰	10ºC	$Na_2S_2O_3$	30 hours ²¹	120 mL
Cyanide	Plastic/Glass	4°C	NaOH to pH >12 Ascorbic acid ⁹ or Sodium arsenite ⁹	14 days	500 mL
Fluoride	Plastic/Glass	None	None	None	250 mL
Heterotrophic Plate Count	Plastic/Glass ²⁰	10°C	$Na_2S_2O_3$	8 hours (24 hours ²² )	120 mL
Mercury	Plastic/Glass	None	HNO₃ to pH<2	28 days	250 mL
Metals⁴	Plastic/Glass	None	HNO ₃ to $pH<2^{24}$	6 months	250 mL
Nitrate	Plastic/Glass	4ºC	None	48 hours ⁶	250 mL
Nitrate-Nitrite	Plastic/Glass	None	H₂SO₄ to pH<2	28 days	250 mL
Nitrite	Plastic/Glass	4°C	None	48 hours	250 mL
THMs Only	Glass ⁸	4ºC	Na ₂ S ₂ O ₃ ⁹ HCl to pH <2 may also be used	14 days	3 X 40 mL
Volatile Organic Compounds	Glass ⁸	4ºC	HCl to pH <2 Na ₂ S ₂ O ₃ or Ascorbic acid ⁹	14 days	3 X 40 mL
EDB, DBCP, 1,2,3- TCP (EPA 504.1)	Glass ⁸	4°C	$Na_2S_2O_3$	14 days	3 X 40 mL
Organochlorine Pesticides/PCBs (EPA 505) ¹⁰	Glass ⁸	4ºC	$Na_2S_2O_3$	14 days ¹¹	3 X 40 mL
Nitrogen and Phos. Pesticides (EPA 507)	Glass-Amber ⁸	4ºC	Na ₂ S ₂ O ₃	14 days ¹²	1 L
Total PCBs (EPA 508A)	Glass-Amber ⁸	4ºC	None	14 days ¹³	1 L
Pesticides and PCBs (EPA 508.1) ¹⁴	Glass-Amber ⁸	4°C	HCI to pH <2 Na ₂ S ₂ O ₃ ⁹	14 days ¹³	1 L

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-7 of 23-20

PARAMETER	CONTAINER	PRES Temp. ²	SERVATION ^{1,2} ³ Chemical	HOLDING TIME ³	SAMPLE VOLUME
Chlorinated Acids (EPA 515.1)	Glass-Amber ⁸	4°C	$Na_2S_2O_3$	14 days ¹²	1 L
Semivolatiles (EPA 525.2)	Glass-Amber ⁸	4ºC	HCl to pH <2 Na ₂ S ₂ O ₃ ⁹	14 days ¹³	1 L
N-Methylcarbamoyloxamines and N-Methcarbamates (EPA 531.1)	Glass ⁸	4ºC	Na ₂ S ₂ O ₃ , Monochloroacetic Acid buffer to pH<3	28 days	3 X 60 mL
Glyphosate (EPA 547)	Glass ⁸	4°C	$Na_2S_2O_3$	14 days	3 X 60 mL
Endothall (EPA 548)	$Na_2S_2O_3$	4°C	None	7 days ¹⁵	1 L
Diquat/Parquat (EPA 549.1)	Glass-Amber ⁸ (Silanized or PVC amber)	4°C	H₂SO₄ to PH <2 Na₂S₂O₃ ⁹	7 days ¹⁶	1 L
Chlorinated Disinfection Byproducts, Chlorinated Solvents, and Halogenated Pesticides/Herbicides (EPA 551)	Glass ⁸	4°C	Phosphate Buffer and Ammonium Chloride ¹⁹	14 days ¹⁷	3 X 60 mL
Haloacetic Acids (EPA 552.1)	Glass-Amber ⁸	4°C	Ammonium Chloride	28 days ¹⁸	250 mL

#### Key to Table

- Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at 4°C until compositing and sample splitting is completed.
- 2. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Hydrochloric acid, (HCl) in water, solutions at concentrations of 0.04% by weight or less (pH about 1.96 or greater); Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less pH about 1.15 or greater); and Sodium hydroxide (NaOH) in water solutions at concentrations at concentrations of 0.080% by weight or less (pH about 12.30 or less).
- 3. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 4. All metals except Hg.
- 5. Instructions for containers, preservation procedures and holding times as specified in Method 100.2 must be adhered to for all compliance analysis including those conducted with Method 100.1.
- 6. If the sample is chlorinated, the holding time for an un-acidified sample kept at 4°C is extended to 14 days.
- 7. Nitrate-Nitrite refers to a measurement of total nitrite.
- 8. With Teflon lined septum.
- 9. If chlorinated add reagent prior to acidification (for Cyanide, add before NaOH).

#### **Company Confidential & Proprietary**

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-8 of 23-20

- 10. Heptaclor has a 7 day hold time
- 11. 14 days until extraction. 24 hours after extraction.
- 12. 14 days until extraction. 28 days after extraction.
- 13. 14 days until extraction. 30 days after extraction.
- 14. For cyanazine, cool to 4°C only.
- 15. 7 days until derivitization. 1 day after derivitization.
- 16. 7 days until extraction. 21 days after extraction.
- 17. 14 days until extraction. 14 days after extraction.
- 18. 28 days until extraction. 48 hours after extraction.
- 19. Sodium Sulfite may be used as a dechlorinating agent in some instances. Verify with laboratory prior to sampling.
- 20. Sterilized. Plastic must be Polypropylene.
- 21. 40 CFR part 141.74 regulations to avoid filtration or disinfection state 8 hours (DW compliance testing). Most facilities are using either disinfection or filtration so the 8 would not apply in most cases.
- 22. 40 CFR part 141.74 regulations for Disinfection By-Product rule state 8 hours (DW compliance testing) where SM 9215 allows up to 24 hours if sample is stored between > 0 and  $\leq 4^{\circ}$  C
- 23. For samples with a temperature requirement of 4°C, a sample temperature of just above the water freezing temperature to < 6°C is acceptable.
- 24. Acid preservation may be omitted for shipping and laboratory will acidify at least 24 hours prior to analysis.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-9 of 23-20

# Table 23-2Holding Times, Preservation and Container Requirements:NPDES – Bacteria, Protozoa,Toxicity Tests

PARAMETER	CONTAINER ¹	PRES Temp.	SERVATION ^{2,3} Chemical	HOLDING TIME⁴	SAMPLE VOLUME
Total, Fecal, and E.coli Coliforms	Plastic/Glass	10ºC	0.0008 % Na ₂ S ₂ O ₃ ⁶	6 hours	100 mL
Fecal Streptococci	Plastic/Glass	10ºC	0.0008 % Na ₂ S ₂ O ₃ ⁶	6 hours	100 mL
Enterococci	Plastic/Glass	10ºC	0.0008 % Na₂S₂O₃ ⁶	6 hours	100 mL
Cryptosporidium	LPDE Plastic	0-8°C	None	96 Hours	500 mL
Giardia	LPDE Plastic	0-8°C	None	96 Hours	500 mL
Toxicity – Acute/Chronic	Plastic/Glass	≤ 6°C ⁵	None	36 Hours	2 L

- 1. Plastic should be Polypropylene or other sterilizable plastic.
- 2. Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at 4°C until compositing and sample splitting is completed.
- 3. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Hydrochloric acid, (HCI) in water, solutions at concentrations of 0.04% by weight or less (pH about 1.96 or greater); Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less pH about 1.15 or greater); and Sodium hydroxide (NaOH) in water solutions at concentrations at concentrations of 0.080% by weight or less (pH about 12.30 or less).
- 4. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 5. Samples must not be frozen. Sufficient ice should be placed with the samples in the shipping container to ensure that ice is still present when the samples arrive at the laboratory. However, even if ice is present, when samples arrive, it is necessary to measure the temperature of the samples and confirm that the < 6°C temperature has not been exceeded.</p>
- 6. Should only be used in the presence of residual chlorine.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-10 of 23-20

		PRE	SERVATION ^{2,3}	HOLDING	SAMPLE
PARAMETER	CONTAINER ¹	Temp ¹⁴	. Chemical	TIME⁴	VOLUME
Acidity	Plastic/Glass	<u>≤</u> 6°C	None	14 days	100 mL
Alkalinity	Plastic/Glass	<u>≤</u> 6°C	None	14 days	100 mL
Ammonia	Plastic/Glass	<u>≤</u> 6°C	H₂SO₄ to pH<2	28 days	400 mL
BOD 5 Day	Plastic/Glass	≤ 6°C	None	48 hours	1000 mL
Boron	Plastic ⁵	None	HNO₃ to pH<2	6 months	200 mL
Bromide	Plastic/Glass	None	None	28 days	100 mL
CBOD 5 Day	Plastic/Glass	≤ 6°C	None	48 hours	1000 mL
COD	Plastic/Glass	≤ 6ºC	H₂SO₄ to pH<2	28 days	100 mL
Chloride	Plastic/Glass	None	None	28 days	50 mL
Chlorine, Residual	Plastic/Glass	None	None	15 min. ⁶	200 mL
Color	Plastic/Glass	≤ 6ºC	None	48 hours	50 mL
Cvanide -Total ^{16,17}	Plastic/Glass	< 6%	NaOH to pH >12,	14 days	100 ml
Cyanide – rotai	1 12310/01233	200	0.6 g ascorbic Acid ⁷	14 days	100 mL
Cyanide –	Plastic/Glass	< 6%	NaOH to pH >12,	14 days	100 ml
Amenable ^{16,17}	1 105110/ 01030		0.6 g ascorbic Acid ⁷		
Fluoride	Plastic	None	None	28 days	300 mL
Hardness	Plastic/Glass	None	HNO₃ to pH<2 ⁸	6 months	100 mL
Hexavalent, Chromium	Plastic/Glass	≤ 6°C	Ammonium sulfate buffer pH = 9.3 - 9.7	28 dys / 24 hrs ¹⁵	200 mL
Hydrogen Ion (pH)	Plastic/Glass	None	None	15 min. ⁶	200 mL
Kjeldahl and organic Nitrogen	Plastic/Glass	≤ 6°C	H₂SO₄ to pH <2	28 days	500 mL
Mercury ¹¹	Plastic/Glass	None	HNO ₃ to pH<2	28 days	200 mL
Metals ^{9,10}	Plastic/Glass	None	HNO ₃ to pH<2 ¹⁸	6 months	200 mL
Nitrate	Plastic/Glass	<u>≤</u> 6ºC	None	48 hours	100 mL
Nitrate-Nitrite	Plastic/Glass	≤ 6°C	$H_2SO_4$ to pH <2	28 days	100 mL
Nitrite	Plastic/Glass	≤ 6°C	None	48 hours	100 mL
Oil and Grease	Glass	≤ 6ºC	H₂SO₄ or HCl to pH <2	28 days	1 L

 Table 23-3

 Holding Times, Preservation and Container Requirements:
 NPDES - Inorganic

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-11 of 23-20

PARAMETER	CONTAINER ¹	PRE Temn ¹⁴	SERVATION ^{2,3} Chemical	HOLDING TIMF ⁴	SAMPLE
	CONTRACTO	remp	H-SO, or HCI to		
(TOC)	Plastic/Glass	≤ 6°C	pH <2 ¹²	28 days	250 mL
Orthophosphate	Plastic/Glass	<u>≤</u> 6ºC	Filter within 15 min.	48 hours	250 mL
Oxygen, Dissolved Probe	Glass ¹³	None	None	15 min. ⁶	200 mL
Oxygen, Winkler	Glass ¹³	None	Fix on site and store in dark.	8 hours	300 mL
Phenols	Glass	<u>≤</u> 6°C	H₂SO₄ to pH <2	28 days	500 mL
Phosphorus, Elemental	Glass	<u>≤</u> 6ºC	None	48 hours	250 mL
Phosphorus, Total	Plastic/Glass	<u>≤</u> 6ºC	H₂SO₄ to pH <2	28 days	250 mL
Residue, Total	Plastic/Glass	≤ 6°C	None	7 days	1 L
Residue, Filterable	Plastic/Glass	≤ 6°C	None	7 days	1 L
Residue, Non- Filterable	Plastic/Glass	<u>≤</u> 6ºC	None	7 days	1 L
Residue, Settleable	Plastic/Glass	≤ 6°C	None	48 hours	1 L
Residue, Volatile	Plastic/Glass	<u>≤</u> 6°C	None	7 days	1 L
Silica	Plastic ⁵	≤ 6°C	None	28 days	250 mL
Specific Conductance	Plastic/Glass	<u>≤</u> 6°C	None	28 days	250 mL
Sulfate	Plastic/Glass	≤ 6°C	None	28 days	250 mL
Sulfide	Plastic/Glass	<u>≤</u> 6ºC	Zinc acetate plus NaOH to pH>9	7 days	500 mL
Sulfite	Plastic/Glass	None	None	15 min. ⁶	200 mL
Surfactants	Plastic/Glass	≤ 6°C	None	48 hours	1 L
Temperature	Plastic/Glass	None	None	N/A	100 mL
Turbidity	Plastic/Glass	≤ 6°C	None	48 hours	1 L

#### Key to Table

1. Plastic should be Polyethylene.

Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at <<u>6</u>°C until compositing and sample splitting is completed.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-12 of 23-20

- 3. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Hydrochloric acid, (HCl) in water, solutions at concentrations of 0.04% by weight or less (pH about 1.96 or greater); Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less pH about 1.15 or greater); and Sodium hydroxide (NaOH) in water solutions at concentrations at concentrations of 0.080% by weight or less (pH about 12.30 or less).
- 4. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 5. May also be collected in quartz or PFTE Plastic.
- 6. For compliance testing, the analysis must be performed in the field at the time of analysis. If transported to the laboratory for analysis, the analysis will be performed as soon as practical and reported qualified.
- 7. Should only be used in the presence of residual chlorine. (Alternatively, sodium arsenite may be used)
- 8.  $H_2SO_4$  to a pH <2 is also acceptable.
- 9. Except Mercury and Hexavalent Chromium.
- 10. For dissolved metals, samples must be filtered on site before adding HNO₃ preservative (or before shipping to laboratory).
- 11. Samples collected for determination of trace level mercury (100 ng/L) using EPA 1631 must be collected in tightly capped fluoropolymer or glad bottles and preserved with BrCl or HCl solution within 48 hours of sample collection. The time to preservation may be extended to 28 days if a sample is oxidized in the sample bottle. Samples collected for dissolved trace level mercury should be filtered in the laboratory. However, if circumstances prevent overnight shipping, samples should be filtered in a designated clean area in the field in accordance with procedures given in Method 1669. Samples that been collected for determination of total or dissolved trace level mercury must be analyzed within 90 days of sample collection.
- 12. Phosphoric acid  $(H_3PO_4)$  may also be used.
- 13. Should have glass lid or top.
- 14. Aqueous samples must be preserved at ≤6 °C unless otherwise indicated, and should not be frozen unless data demonstrating that sample freezing does not adversely impact sample integrity is maintained on file and accepted as valid by the regulatory authority. Also, for purposes of NPDES monitoring, the specification of "≤ °C" is used in place of the "4 °C" and "<4 °C" sample temperature requirements listed in some methods. It is not necessary to measure the sample temperature to three significant figures (1/100th of 1 degree); rather, three significant figures are specified so that rounding down to 6 °C may not be used to meet the ≤6 °C requirement. The preservation temperature does not apply to samples that are analyzed immediately (less than 15 minutes).
- 15. Holding time is 24 hours if pH adjustment is not performed.
- 16 In the Field: Samples are to be tested for Sulfide using lead acetate paper prior to the addition of Sodium Hydroxide (NaOH). If sulfide is present, the sample must be treated with Cadmium Chloride and filtered prior to the addition of NaOH. If the sulfide test and treatment is not performed in the field, the lab will test the samples for sulfide using lead acetate paper at the time of receipt and if sulfide is present in the sample, the client will be notified and given the option of retaking the sample and treating in the field per the method requirements or the laboratory can analyze the samples as delivered (with sulfide treatment by laboratory) and qualify the results in the final report.
- 17 It is the responsibility of the client to notify the laboratory if thiosulfate, sulfite, or thiocyanate are known or suspected to be present in the sample. This notification may be on the chain of custody. The samples may need to be subcontracted to a laboratory that performs a UV digestion. If the lab does not perform the UV digestion on samples that contain these compounds, the results must be qualified in the final report.
- 18 Acid preservation may be omitted for shipping and laboratory will acidify at least 24 hours prior to analysis.

Table 23-4	
Holding Times, Preservation and Container Requirements:	NPDES - Organic

		PRE	SERVATION ^{1,2}	HOLDING	SAMPLE
PARAMETER	CONTAINER	lemp.'	Chemical	I IME*	VOLUME
Purgeable Halocarbons	Glass⁴	<u>≤</u> 6ºC	0.0008 % Na ₂ S ₂ O ₃ ⁵	14 days	40 mL
Purgeable Aromatic Hydrocarbons	Glass⁴	<u>≤</u> 6ºC	0.0008 % Na ₂ S ₂ O ₃ ⁵ , HCl to pH<2	14 days ⁶	40 mL
Acrolein and Acrylonitrile	Glass⁴	<u>≤</u> 6ºC	0.0008 % Na ₂ S ₂ O ₃ ⁵ , adjust pH to 4-5 ⁷	14 days	40 mL
Phenols ⁹	Glass⁴	≤ 6°C	0.0008 % Na ₂ S ₂ O ₃ ⁵	7 days ⁸	1 L
Benzidines ⁹	Glass⁴	<u>≤</u> 6°C	0.0008 % Na ₂ S ₂ O ₃ ⁵	7 days ^{8, 11}	1 L
Phthalate esters ⁹	Glass⁴	≤ 6°C	None	7 days ⁸	1 L
Nitosamines ^{9,12}	Glass⁴	≤ 6°C	0.0008 % Na ₂ S ₂ O ₃ ^{5,13}	7 days ⁸	1 L
PCBs ⁹	Glass⁴	<u>≤</u> 6°C	None	1 year ⁸	1 L
Nitroaromatics and Isophorone ⁹	Glass⁴	≤ 6°C	0.0008 % Na ₂ S ₂ O ₃ ^{5,13}	7 days ⁸	1 L
Polynuclear Aromatic Hydrocarbons ⁹	Glass⁴	<u>≤</u> 6ºC	0.0008 % Na ₂ S ₂ O ₃ ^{5,13}	7 days ⁸	1 L
Haloethers ⁹	Glass⁴	<u>≤</u> 6°C	0.0008 % Na ₂ S ₂ O ₃ ⁵	7 days ⁸	1 L
Chlorinated Hydrocarbons ⁹	Glass⁴	≤ 6°C	None	7 days ⁸	1 L
CDD/CDFs ⁹ – Aqueous: Field/Lab Preservation	Glass	<u>≤</u> 6ºC	рН <9, 0.0008 % Na₂S₂O₃ ⁵	1 year	1 L
CDD/CDFs ⁹ – Solids/Mixed Phase/ - Field Preservation	Glass	<u>≤</u> 6ºC	None	7 days	1 L
CDD/CDFs ^{9 -} Tissue – Field Preservation	Glass	≤ 6°C	None	24 hours	
CDD/CDFs ⁹ – Solids/Mixed Phase/Tissue - Lab Preservation	Glass	< -10ºC	None	1 year	1 L
Pesticides ⁹	Glass	≤ 6°C	pH 5-9 ¹⁴	7 days ⁸	1 L

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-14 of 23-20

- Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at < 6°C until compositing and sample splitting is completed.</li>
- 2. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Hydrochloric acid, (HCl) in water, solutions at concentrations of 0.04% by weight or less (pH about 1.96 or greater); Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less pH about 1.15 or greater); and Sodium hydroxide (NaOH) in water solutions at concentrations at concentrations of 0.080% by weight or less (pH about 12.30 or less).
- 3. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 4. With Teflon lined septum.
- 5. Should only be used in the presence of residual chlorine. (Ascorbic acid may be used instead)
- 6. Samples receiving no pH adjustments must be analyzed within 7 days. If 2-chlorovinylethylether is a target analyte, the sample should not be acidified.
- 7. The pH adjustment is not required if acrolein is not being measured. Samples for acrolein receiving no pH adjustment must be analyze within three days of sampling.
- 8. 7 days until extraction, 40 days after extraction. (PCB only 1 year after extraction)
- 9. When the extractable analytes of concern fall within a single chemical category, the specified preservative and maximum holding times should be observed for optimum safeguard of sample integrity. When the analytes of concern fall within two or more categories, the sample may be preserved by cooling to  $\leq 6^{\circ}$ C reducing residual chlorine with 0.0008 % sodium thiosulfate, storing in the dark, and adjusting the pH to 6-9. Samples preserved in this manner may be held for 7 days before extraction and for 40 days after extraction. Exceptions to this optional preservation and holding time procedure are noted in footnote 5 (re the requirement for thiosulfate reduction of residual chlorine) and footnotes 10 and 11(re the analysis of Benzidine).
- 10. If 1,2-diphenylhydrazine is likely to be present, adjust pH to of the sample to 4.0 ± 0.2 to prevent rearrangement to benzidine.
- 11. Extracts may be stored up to 30 days before analysis if storage temperature is  $< 0^{\circ}$ C.
- 12. For the analysis of diphenylnitrosamine, add 0.008 % Na₂S₂O₃ and ajust pH to 7-10 with NaOH within 24 hours of sampling.
- 13. Store in dark.
- 14. The pH adjustment may be performed upon receipt in the laboratory and may be omitted if the samples are extracted within 72 hours of collection. For the analysis of aldrin , add 0.0008 % Na₂S₂O_{3.}
- 15. Aqueous samples must be preserved at ≤6 °C unless otherwise indicated, and should not be frozen unless data demonstrating that sample freezing does not adversely impact sample integrity is maintained on file and accepted as valid by the regulatory authority. Also, for purposes of NPDES monitoring, the specification of "≤ °C" is used in place of the "4 °C" and "<4 °C" sample temperature requirements listed in some methods. It is not necessary to measure the sample temperature to three significant figures (1/100th of 1 degree); rather, three significant figures are specified so that rounding down to 6 °C may not be used to meet the ≤6 °C requirement. The preservation temperature does not apply to samples that are analyzed immediately (less than 15 minutes).</p>

# Table 23-5. Holding Times, Preservation and Container Requirements: NPDES - Radiological

		PRES	SERVATION ^{1,2}	HOLDING	SAMPLE
PARAMETER	CONTAINER	Temp.	Chemical	TIME ³	VOLUME
Alpha, Beta, Radium	Plastic/Glass	None	HNO₃ to pH<2	6 months	1 L

- Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at 4°C until compositing and sample splitting is completed.
- 2. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less (pH about 1.62 or greater).
- 3. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-16 of 23-20

# Table 23-6. Holding Times, Preservation and Container Requirements: RCRA - Aqueous

		PRE	SERVATION ^{2,3}	HOLDING	SAMPLE
PARAMETER	CONTAINER ¹	Temp. ¹²	Chemical	TIME ⁴	VOLUME
Chloride	Plastic/Glass	4°C	None	28 days	100 mL
Cyanide -Total	Plastic/Glass	4°C	NaOH to pH >12 ⁵	14 days	250 mL
Cyanide -Amenable	Plastic/Glass	4ºC	NaOH to pH >12 ⁵	14 days	250 mL
Hydrogen Ion (pH)	Plastic/Glass	4°C	None	24 hours ¹¹	100 mL
Nitrate	Plastic/Glass	4°C	None	48 hours	100 mL
Oil and Grease	Glass	4°C	HCI	28 days	1 L
Organic carbon (TOC)	Plastic/Glass	4ºC	pH to <2 ⁶ Store in dark	28 days	2 X 40 mL
Sulfate	Plastic/Glass	4°C	None	28 days	400 mL
Sulfide •	Plastic/Glass	4ºC	Add Zn Acetate	7 days	400 mL
Chromium VI	Plastic/Glass	4ºC	None	24 hours	250 mL
Mercury	Plastic/Glass	None	HNO ₃ to pH<2	28 days	250 mL
Other Metals	Plastic/Glass	None	HNO ₃ to pH<2	6 months	250 mL
Acrolein and Acrylonitrile	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷ Adjust pH to 4-5 ¹³	14 days	2-40 ml VOA
Benzidines	Glass ¹⁰	4°C	0.0008 % Na ₂ S ₂ O ₃ ⁷	7 days ⁸	1 L
Chlorinated Hydrocarbons	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷	7 days ⁸	1 L
Dioxins and Furans	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷	7 days ⁸	1 L
Haloethers	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷	7 days ⁸	1 L
Nitroaromatics and cyclic ketones	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷ , store in dark	7 days ⁸	1 L
Nitrosomines	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷ , store in dark	7 days ⁸	1 L
Organochlorine Pesticides	Glass ¹⁰	4ºC	None	7 days ⁸	1 L
Organophosphorus Pesticides	Glass ¹⁰	4ºC	Adjust pH ⁹	7 days ⁸	1 L
PCBs	Glass ¹⁰	4°C	None	7 days ⁸	1 L
Phenols	Glass ¹⁰	4°C	0.0008 % Na ₂ S ₂ O ₃ ⁷	7 days ⁸	1 L

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-17 of 23-20

PARAMETER	CONTAINER ¹	PRES Temp. ¹²	SERVATION ^{2,3} Chemical	HOLDING TIME⁴	SAMPLE VOLUME
Phthalate Esters	Glass ¹⁰	4ºC	None	7 days ⁸	1 L
Polynuclear Aromatic Hydrocarbons	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷ , store in dark	7 days ⁸	1 L
Purgeable Hydrocarbons	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷ Adjust pH <2 ²	14 days	40 mL
Purgeable Halocarbons	Glass ¹⁰	4ºC	0.0008 % Na ₂ S ₂ O ₃ ⁷	14 days	40 mL
Total Organic Halides (TOX)	Glass ¹⁰	4ºC	Adjust pH to <2 with H₂SO₄	28 days	2 X 250 mL
Radiological Tests (Alpha, Beta, Radium)	Plastic/Glass	None	HNO₃ to pH<2	6 months	250 mL

- 1. Plastic should be Polyethylene.
- 2. Sample preservation should be performed immediately upon sample collection. For composite chemical samples, each aliquot should be preserved at the time of collection. When use of an automated sampler makes it impossible to preserve each aliquot, then chemical samples may be preserved by maintaining at 4°C until compositing and sample splitting is completed.
- 3. When any sample is to be shipped by common carrier or sent through the United States mails, it must comply with the Department of Transportation Hazardous Materials Regulations (49 CFR Part 172). The person offering such material for transportation is responsible for ensuring compliance. For the preservation requirements of Table 6-8, the Office of Hazardous Materials, Materials Transportation Bureau, Department of Transportation has determined that the Hazardous Materials Regulations do not apply to the following materials: Hydrochloric acid, (HCI) in water, solutions at concentrations of 0.04% by weight or less (pH about 1.96 or greater); Nitric acid (HNO₃) in water solutions at concentrations of 0.15% by weight or less pH about 1.15 or greater); and Sodium hydroxide (NaOH) in water solutions at concentrations at concentrations of 0.080% by weight or less (pH about 12.30 or less).
- 4. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 5. If oxidizing agents are present, add 5 mL 0.1 N NaAsO₂ or 0.06 g of ascorbic acid per L. See Cyanide SOP for additional information about other interferences.
- 6. Adjust pH to <2 with  $H_2SO_4$ , HCl, or solid NaHSO₄. Free Chlorine must be removed prior to adjustment.
- 7. Free Chlorine must be removed by the appropriate addition of  $Na_2S_2O_3$ .
- 8. 7 days until extraction. 40 days after extraction.
- 9. Adjust pH to 5-8 using NaOH or  $H_2SO_4$ .
- 10. With Teflon lined septum.
- 11. Holding Time is listed as "As Soon as Possible" in SW 846. Per EPA MICE, the recommended maximum holding time for pH in water is 24 hours and pH in soil is 7 days. There are no mandated regulatory requirements.
- 12. For samples with a temperature requirement of  $4^{\circ}$ C, a sample temperature of just above the water freezing temperature to  $\leq 6^{\circ}$ C is acceptable.
- 13. Based on guidance from EPA MICE, if samples are received without pH adjustment, the holding time is 7 days.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-18 of 23-20

### Table 23-7. Holding Times, Preservation and Container Requirements: RCRA – Non-Aqueous

		PRE	SERVATION	HOLDING	SAMPLE
PARAMETER	CONTAINER ¹	Temp. ⁷	Chemical	TIME ²	WEIGHT
Chloride	Glass	4ºC	None	28 days	50 g
Cyanide -Total	Glass	4°C	None	14 days	50 g
Cyanide -Amenable	Glass	4°C	None	14 days	50 g
Hydrogen Ion (pH)	Glass	4°C	None	7 days ⁶	50 g
Nitrate	Glass	4°C	None	N/A	50 g
Oil and Grease	Glass	4ºC	None	28 days	50 g
Sulfide	Glass	4°C	Add Zn Acetate, zero headspace	7 days	50 g
Chromium VI	Glass	4°C	None	30 days ⁸	50 g
Mercury	Plastic/Glass	None	None	28 days	50 g
Other Metals	Plastic/Glass	None	None	6 months	50 g
Acrolein and Acrylonitrile	Glass⁴	4ºC	None	14 days	100 g
Benzidines	Glass⁴	4°C	None	14 days ³	100 g
Chlorinated Hydrocarbons	Glass⁴	4ºC	None	14 days ³	100 g
Dioxins and Furans	Glass⁴	4ºC	None	14 days ³	100 g
Haloethers	Glass⁴	4°C	None	14 days ³	100 g
Nitroaromatics and cyclic ketones	Glass⁴	4ºC	None	14 days ³	100 g
Nitrosomines	Glass⁴	4°C	None	14 days ³	100 g
Organochlorine Pesticides	Glass⁴	4ºC	None	14 days ³	100 g
Organophosphorus Pesticides	Glass⁴	4ºC	None	14 days ³	100 g
PCBs	Glass⁴	4°C	None	14 days ³	100 g
Phenols	Glass⁴	4°C	None	14 days ³	100 g
Phthalate Esters	Glass⁴	4ºC	None	14 days ³	100 g
Polynuclear Aromatic Hydrocarbons	Glass⁴	4°C	None	14 days ³	100 g

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 23-19 of 23-20

PARAMETER	CONTAINER ¹	PRE Temp. ⁷	SERVATION Chemical	HOLDING TIME ²	SAMPLE WEIGHT
Purgeable Hydrocarbons	Glass ⁴	4°C	None	14 days⁵	100 g
Purgeable Halocarbons	Glass⁴	4ºC	None	14 days⁵	100 g
Total Organic Halides (TOX)	Glass⁴	4ºC	None	28 days	50 g

- 1. Plastic should be Polyethylene.
- 2. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.
- 3. 14 days until extraction. 40 days after extraction.
- 4. With Teflon Lined Septum
- 5. See Volatile SOP for more detailed preservation requirements.
- 6. Holding Time is listed as "As Soon as Possible" in SW 846. Per EPA MICE, the recommended maximum holding time for pH in water is 24 hours and pH in soil is 7 days. There are no mandated regulatory requirements.
- 7. For samples with a temperature requirement of  $4^{\circ}$ C, a sample temperature of just above the water freezing temperature to  $\leq 6^{\circ}$ C is acceptable.
- 8. 30 days to digestion, 7 days from digestion to analysis.

Table 23-8.	
Holding Times, Preservation and Container Requirements:	Air Samples

		PRE	SERVATION	HOLDING	SAMPLE
PARAMETER	CONTAINER ¹	Temp.	Chemical	TIME ²	WEIGHT
Volatile Organics	Summa Cannister	None	None	30 days	6L or 1L
Volatile Organics	Tedlar Bag	None	None	72 hrs ^{3,4}	1 L

#### Key to Table

1. Plastic should be Polyethylene.

2. Samples should be analyzed as soon as possible after collection. The times listed are the maximum times that samples may be held before analysis and still be considered valid.

- 3. Holding Time is based on SW 846 Method 0040 "SAMPLING OF PRINCIPAL ORGANIC HAZARDOUS CONSTITUENTS FROM COMBUSTION SOURCES USING TEDLAR® BAGS". Some states specifically enforce this holding time (e.g. Florida, New Jersey) and others have not specified this information in their regulatory requirements.
- 4. The holding time is 72 hours unless the laboratory has a documented validation study that indicates a longer HT is acceptable for the analytes of interest.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-1 of 24-13

#### **SECTION 24**

#### HANDLING OF SAMPLES (NELAC 5.5.8)

Sample management procedures at **TestAmerica Buffalo** ensure that sample integrity and custody are maintained and documented from sampling/receipt through disposal.

#### 24.1 CHAIN OF CUSTODY (COC)

The COC form is the written documented history of any sample and can be initiated when bottles are sent to the field, or at the time of sampling. This form is completed by the sampling personnel and accompanies the samples to the laboratory where it is received and stored under the laboratory's custody. The purpose of the COC form is to provide a legal written record of the handling of samples from the time of collection until they are received at the laboratory. It also serves as the primary written request for analyses from the client to the laboratory. The COC form acts as a purchase order for analytical services when no other contractual agreement is in effect. An example of a COC form may be found in Figure 24-1.

#### 24.1.1 Field Documentation

The information the sampler needs to provide at the time of sampling on the container label is:

- Sample identification
- Date and time
- Preservative

During the sampling process, the COC form is completed and must be legible (see Figure 24-1). This form includes information such as:

- Client name, address, phone number and fax number (if available)
- Project name and/or number
- The sample identification
- Date, time and location of sampling
- Sample collectors name
- The matrix description
- The container description
- The total number of each type of container
- Preservatives used
- Analysis requested
- Requested turnaround time (TAT)
- Any special instructions
- Purchase Order number or billing information (e.g. quote number) if available
- The date and time that each person received or relinquished the sample(s), including their signed name.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-2 of 24-13

The samples are stored in a cooler with ice, as applicable, and remain solely in the possession of the client's field technician until the samples are delivered to the laboratory. The sample collector must assure that each container is in his/her physical possession or in his/her view at all times, or stored in such a place and manner to preclude tampering. The field technician relinquishes the samples in writing on the COC form to the sample control personnel at the laboratory or to a TestAmerica courier. Samples are only considered to be received by lab when personnel at the laboratory have physical contact with the samples.

**Note:** Independent couriers are not required to sign the COC form. The COC is usually kept in the sealed sample cooler. The shipping documents are retained with the project files.

#### 24.1.2 Legal / Evidentiary Chain-of-Custody

If samples are identified for legal/evidentiary purposes on the COC or in the project notes, sample management will initiate Strict Chain of Custody procedures as defined in SOP BF-GP-018, "Strict Internal Chain-of-Custody".

#### 24.2 SAMPLE RECEIPT

Samples are received at the laboratory by designated sample receiving personnel and a unique laboratory project identification number is assigned. Each sample container shall be assigned a unique sample identification number that is cross-referenced to the client identification number such that traceability of test samples is unambiguous and documented. Each sample container is affixed with a durable sample identification label. Sample acceptance, receipt, tracking and storage procedures are summarized in the following sections.

#### 24.2.1 Laboratory Receipt

When samples arrive at the laboratory, sample receiving personnel inspect the coolers and samples. The integrity of each sample must be determined by comparing sample labels or tags with the COC and by visual checks of the container for possible damage. Any non-conformance, irregularity, or compromised sample receipt must be documented on a Login – Analytical Receipt Resolution Form (Figure 24-6). and brought to the immediate attention of the client. The COC, shipping documents, documentation of any non-conformance, irregularity, or compromised sample receipt, record of client contact, and resulting instructions become part of the project record.

#### 24.2.1.1 Inspection of samples include a check for:

- Complete documentation to include sample identification, location, date and time of collection, collector's name, preservation type, sample type and any additional comments concerning the samples.
- Complete sample labels to include unique identification in indelible ink.
- Use of appropriate sample containers (see Section 23)
- Adherence to holding times as specified in the test method and/or summarized in Section 23.
- Adequate sample volume for required analyses (see Section 23).

- Damage or signs of contamination to sample container. Volatile vials are also inspected for headspace
- **24.2.1.2** Using an IR temperature gun, check and record the temperature of the samples that require thermal preservation. On a project specific basis, temperature blanks may also be required.
  - Samples shall be deemed acceptable if arrival temperature is just above freezing and less than or equal to 6.0° C. Samples that are hand-delivered immediately after collection may not be at the required temperatures; however, if there is evidence that the chilling process has begun, such as the arrival on ice, the samples shall be considered acceptable. This will be documented on the COC and in the LIMs job comments.
  - If the samples were shipped in ice and solid ice is still present and in direct contact with samples, report the samples as "received on ice." Direct contact means samples must be surrounded by ice cubes or crushed ice. Ice present in a plastic bottle or other container does not constitute direct contact. Samples shipped with only "blue ice" may not be reported as "received on ice".
- **24.2.1.3** Verify sample preservation as specified in the test method. Check for correct pH as specified in the test method. The results are documented in the LIMs sample preservation tables. In the case of volatiles (including TOC and TOX) the pH is recorded after analysis in the analysis logbook or on the raw data instrument printout. At the time of analysis, chlorine is checked on samples requiring extractable organics, BOD, TOX, cyanide, fluoride, ammonia, TKN, CBOD and Nitrate; presence or absence is recorded
- **24.2.1.4** After inspecting the samples, the sample receiving personnel sign and date the COC form, make any necessary notes of the samples' conditions and store them in appropriate refrigerators or storage locations.
- **24.2.1.5** If samples are received without a COC, TestAmerica will provide a generic COC form to be completed by the client when the samples are brought to the laboratory. The client is always provided with a copy of the completed COC form for their records.
- **24.2.1.6** If analyses with short holding times are requested, the dates and times are inspected to ensure that holding times have not already expired.
- **24.2.1.7** TestAmerica Buffalo maintains routine sample acceptance for both 1st and 2nd shift. However, samples received after normal working hours are left in their coolers and placed in a cold storage location. The person receiving the samples must record the date and time received and sign the CoC. Chains of custody are placed in the sample management office and associated samples are addressed by sample management personnel during the next working shift.
- **24.2.1.8** Any deviations from the checks described in Section 24.2.1 that question the suitability of the sample for analysis, or incomplete documentation as to the tests required will be resolved by consultation with the client. If the sample acceptance criteria (Section 24.3) are not met, the laboratory shall either:

• Retain all correspondence and/or records of communications with the client regarding the disposition of rejected samples

or

• Fully document any decision to proceed with sample analysis that does not meet sample acceptance criteria.

#### 24.2.2 Sample Log-in

All samples that are received by the laboratory are logged into the LIMS to allow the laboratory to track and evaluate sample progress. Each group of samples that are logged in together (typically one project from a given client/sampling event/sample receipt date) is assigned a unique job number. Within each job, each sampling point (or sample) receives a unique number. Sample numbers are generated sequentially over time, and are not re-assigned. A sample may be composed of more than one bottle since different preservatives may be required to perform all analyses requested. Even if multiple containers are received for a single sample, each container is uniquely identified with an alphabetic letter added to the sample number. The LIMS generates sample labels that are attached to each bottle for a given sample.

Each job/set of samples is logged into LIMS with a minimum of the following information:

- Client Name, Project Name, Address, Phone, Fax, Report to information, invoice to information (most of this information is "default information" that is stored in the LIMS).
- Date and time sampled;
- Date and time received;
- Job and/or project description, sample description;
- Sample matrix, special sample remarks;
- Reporting requirements (i.e., QC level, report format, invoicing format);
- Turn-around-time requirements;
- Parameters (methods and reporting limits or MDLs are default information for a given parameter)

#### 24.3 SAMPLE ACCEPTANCE POLICY

The laboratory has a written sample acceptance policy (Figure 24-5) that clearly outlines the circumstances under which samples shall be accepted or rejected. These include:

- a COC filled out completely;
- samples must be properly labeled;
- proper sample containers with adequate volume for the analysis and necessary QC;
- samples must be preserved according to the requirements of the requested analytical method;
- sample holding times must be adhered to;

#### Company Confidential & Proprietary

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-5 of 24-13

- all samples submitted for water/solid Volatile Organic analyses must have a Trip Blank submitted at the same time;
- the project manager will be notified if any sample is received in damaged condition.

Data from samples which do not meet these criteria are flagged and the nature of the variation from policy is defined. A copy of the sample acceptance policy is provided to each client prior to shipment of samples.

#### 24.4 SAMPLE STORAGE

In order to avoid deterioration, contamination or damage to a sample during storage and handling, from the time of receipt until all analyses are complete, samples are stored in refrigerators suitable for the sample matrix. Aqueous samples designated for metals analysis are stored at ambient temperature. In addition, samples to be analyzed for volatile organic parameters are stored in separate refrigerators designated for volatile organic parameters only. Samples are never to be stored with reagents, standards or materials that may create contamination.

To ensure the integrity of the samples during storage, refrigerator blanks are maintained in the volatile sample refrigerators and analyzed at a minimum of every two weeks.

Analysts and technicians provide a request form to the cooler custodian who then retrieves the requested samples. In the absence of the cooler custodian, the analysts may personally retrieve the sample containers allocated to their analysis from the designated refrigerator. The samples are placed on carts, transported the analytical area and analyzed. Following analysis the remaining sample is returned to the refrigerator from which it originally came. All unused portions of samples are returned to the secure sample control area. All samples are kept in the refrigerators for two to four weeks after analysis, which meets or exceeds most sample holding times. After two to four weeks the samples are moved to dry room temperature, sample archive area where they are retained a minimum of 2 weeks after the final report has been issued to the client at which time disposal occurs. Special arrangements may be made to store samples for longer periods of time. Extended archival periods allow additional metal analyses to be performed on the archived sample and assists clients in dealing with legal matters or regulatory issues.

Access to the laboratory is controlled such that sample storage need not be locked at all times unless a project specifically demands it. Samples are accessible to laboratory personnel only. Visitors to the laboratory are prohibited from entering the refrigerator and laboratory areas unless accompanied by an employee of TestAmerica.

#### 24.5 HAZARDOUS SAMPLES AND FOREIGN SOILS

To minimize exposure to personnel and to avoid potential accidents, samples which are known or suspected to be hazardous are segregated and a notification is issued to all laboratory personnel. All hazardous samples are either returned to the client or disposed of appropriately through a hazardous waste disposal firm. All soil samples, including foreign soil samples are heat treated or incinerated in accordance with USDA permit requirements and are transported / disposed by USEPA approved facilities.

Unused portions of samples found or suspected to be hazardous according to state or federal guidelines may be returned to the client upon completion of the analytical work.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-6 of 24-13

#### 24.6 SAMPLE SHIPPING

In the event that the laboratory needs to ship samples, the samples are placed in a cooler with enough ice to ensure the samples remain just above freezing and at or below 6.0°C during transit. The samples are carefully surrounded by packing material to avoid breakage (yet maintain appropriate temperature). For sample shipments which include water/solid volatile organic analyses, a trip blank is enclosed when required by method specifications or state or regulatory programs. The chain-of-custody form is signed by the sample control technician and attached to the shipping paperwork. Samples are generally shipped overnight express or hand-delivered by a TestAmerica courier to maintain sample integrity. All personnel involved with shipping and receiving samples must be trained to maintain the proper chain-of-custody documentation and to keep the samples intact and on ice. The Environmental, Health and Safety Manual contains additional shipping requirements.

#### 24.7 SAMPLE DISPOSAL

Samples should be retained for a minimum of 2 weeks after the project report is sent, however, provisions may be made for earlier disposal of samples once the holding time is exceeded. Some samples are required to be held for longer periods based on regulatory or client requirements (e.g., 60 days after project report is sent). The laboratory must follow the longer sample retention requirements where required by regulation or client agreement. Several possibilities for sample disposal exist: the sample may be consumed completely during analysis, the sample may be returned to the customer or location of sampling for disposal, or the sample may be disposed of in accordance with the laboratory's waste disposal procedures (SOP: BF-WM-001, "Waste Management".) All procedures in the laboratory Environmental, Health and Safety Manual are followed during disposal. Samples are normally maintained in the laboratory no longer than six weeks from receipt unless otherwise requested. Unused portions of samples found or suspected to be hazardous according to state or federal guidelines may be returned to the client upon completion of the analytical work.

If a sample is part of a known litigation, the affected legal authority, sample data user, and/or submitter of the sample may request to participate in the decision about the sample's disposal. All documentation and correspondence concerning the disposal decision process must be kept on file. Pertinent information includes the date of disposal and nature of disposal (such as sample depletion, hazardous waste facility disposal, return to client). All disposal of sample containers is accomplished through incineration. A Waste Disposal Record (Figure 24-4) should be completed.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-7 of 24-13

Figure 24-1.

#### Example: Chain of Custody (COC)

Chain of															Te	е	st	4	ł	Υ	Э	ŗ	ic	2	C					
Custody Record															TH	EL	AD	R IN	EN	VIRO	NME	NTA	L TE	STIP	١G					
TAL-4142 (0907)																														
Client				Project	Mar	agei	·														0	ale						Ch Ch	ain of Custody N	13
Address				Teleph	one	Num	ber (	Vea	Code	)/Fa	x Nu	mbe	ŕ							•	L	ed No	mbe	*				P		of
City	State	Zip Co	de	Site Co	witac	t				Lab	Con	tact	-							Ar mo	alys re sj	is (A bace	is n	h lis eed	t if ed)	·				
Project Name and Location (State)	4	·		Carrier	/Way	bill f	Vumt	er.									1												Special	Instructions/
Contract/Purchase Order/Duole No.				· · · · ·	Γ	1	Main	ir			;	Con Pres	taine erva	ers à tive	5 5														Condition	ns of Receipt
Sample I.D. No. and Descript (Containers for each sample may be combine	lon d on one	tine)	Date	Time	2	Aqueous	Set	Soil		Unpres.	H2SO4	DONH	Ę	NUMOH	Znaci Nach		1													
																	Γ							Τ						
		-														-														
		-+-				$\vdash$	⊢					-	-		Η	-	┝		-		_		+	-†		_			•	
······		-+-			-			-		_	-	-			-		┢			$\vdash$	-	-	-+	-+	-					
							-	-					-				$\vdash$	Н				-+	-	+	-	-				
							L			_						ŀ		_				_	-+		_					
																	L				_			_	_					
		1																												
																			i									$\square$		
																	[							-	-	-		$\square$		
····					$\vdash$	$\vdash$	$\vdash$	•				-			-					Η				+	-	-		$\square$		
Possible Hazard Identification					L Is	l	le Di	spos	, <u> </u>			L		L	L	I	L									<u> </u>	L	1		and a set
Non-Hazard Flammable S	kin Imtan	. 🗆	Poison B	Unknowr	, [t	<u>,</u> Э.я.	eturn	ToC	tient			Xispo	sal E	ly La	b		Arct	ive F	or			Mont	hs	iva Ke Kong	er ih	ay Di an 1	e ass mor	sesse nih)	iu ii sampies are	(19149)(K)
Turn Around Time Required											oc	Req	uiren	ients	s (Sp	ecity	,													
24 Hours     1 48 Hours     1 7 Day	<u>s Ц</u>	14 Days	21 Days	Date	ter_	-	TH	ne			1. R	lecer	vadi	9y										·				  '	Date	Time
2. Relinquished By				Date			Ta	ne			2. F	lecei	ved	3y														  '	Date	Time
3. Relinguished By	• • •			Date				ne		_	3. F	lecei	ved l	5y						•								 	Date	Типе
Comments			······································	. <b>L</b>			ł				l																			J

DISTRIBUTION: WHITE - Returned to Client with Report: CANARY - Stays with the Sample: PINK - Field Copy

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-8 of 24-13

#### Figure 24-2.

#### Example: Strict Internal Chain of Custody (COC)

JOB#		_ Received by	:	Date & Time	
Parameter	Samples	Analyst	Date & Time OUT	Date & Time IN	Disposal Date
Small Walk-in cooler	(SC#2)	<u> </u>			
	, , , , , , , , , , , , , , , , ,				
VOA applar (SC#1)		<u></u>			
VOA cooler (SC#1)					
[					I
VOA Freezer (SC#3)					
		· ·			
Extract cooler (SC#4)		Received by	y:	_Date & Time	
		Relinguish I	ру:	Date & Time	

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-9 of 24-13

#### Figure 24-3

#### Example: Sample Disposal Record

Date: Time:	12/26/2007 09:58:59		For	ASRF Dates	TestAmeric Sample Di 09/01/2007 -	a Laborato sposal Inv > 09/02/20	ories Inc. entory 07 AND Preserv	vation Code "00"	Page: 1 Rept: AN1468
Test	<u>Matrix:</u> SOILS	S							
	Samoled	Received	Client	Sample ID	Lab Smp ID	Bottles		Parameters	
	08/30/2007	09/01/2007	SB-07-01	(7.5-10)	A7983901	1-4ozGW	HSL PAH		
	08/30/2007	09/01/2007	SB-07-01	(7.5-10)	A7983901	1-4ozGW	T AS		
	08/30/2007	09/01/2007	SB-07-01	(7.5-10)	A7983901	3-ENCORE	BTEX		
	08/30/2007	09/01/2007	SB-07-02	(6-8)	A7983902	1-4ozGW	HSL PAH		
	08/30/2007	09/01/2007	SB-07-02	(6-8)	A7983902	1-4ozGW	T AS		
	08/30/2007	09/01/2007	SB-07-02	(6-8)	A7983902	3-ENCORE	BTEX		
	08/30/2007	09/01/2007	S8-07-03	(6-8)	A7983903	1-4ozGW	HSL PAH		
	09 (70 (2007	00/01/2007	69-07-07	14-91	A 7083003	1-60764	2A T		

- 02	\$75072007	09/01/2007	58-07-02 (0-0)	A1903902	1-402GW	ISL FAN
08	3/30/2007	09/01/2007	SB-07-02 (6-8)	A7983902	1-4ozGW	TAS
08	3/30/2007	09/01/2007	SB-07-02 (6-8)	A7983902	3-ENCORE	BTEX
08	3/30/2007	09/01/2007	\$8-07-03 (6-8)	A7983903	1-4ozGW	HSL PAH
08	3/30/2007	09/01/2007	SB-07-03 (6-8)	A7983903	1-4ozGW	TAS
08	3/30/2007	09/01/2007	SB-07-03 (6-8)	A7983903	3-ENCORE	BTEX
30	3/30/2007	09/01/2007	SB-07-04 (6-8)	A7983904	1-4ozGW	HSL PAH
08	3/30/2007	09/01/2007	SB-07-04 (6-8)	A7983904	1-4ozGW	T AS
08	3/30/2007	09/01/2007	SB-07-04 (6-8)	A7983904	3-ENCORE	BTEX
08	3/30/2007	09/01/2007	SB-07-05 (6-8)	A7983905	1-4ozGW	HSL PAH
08	3/30/2007	09/01/2007	\$8-07-05 (6-8)	A7983905	1-4ozGW	T AS
08	3/30/2007	09/01/2007	SB-07-05 (6-8)	A7983905	3-ENCORE	BTEX
08	3/30/2007	09/01/2007	DUP-01	A7983906	1-4ozG₩	HSL PAH
08	3/30/2007	09/01/2007	DUP-01	A7983906	1-4ozGW	T AS
08	3/30/2007	09/01/2007	DUP-01	A7983906	3-ENCORE	BTEX
08	3/27/2007	09/01/2007	\$B-38-15/16-03	A7984801	1-8ozGW	8260/8270 STARS
08	3/27/2007	09/01/2007	SB-36-15/16-03	A7984802	1-8ozGW	8260/8270 STARS
08	3/27/2007	09/01/2007	SB-36-25/30-03	A7984803	1-8ozGW	8260/8270 STARS
30	3/28/2007	09/01/2007	SB-35-19/20-03	A7984804	1-4ozGW	8260 STARS
08	3/28/2007	09/01/2007	SB-35-19/20-03	A7984804	1-4ozGW	8270 STARS
30	3/29/2007	09/01/2007	\$B-33-17/18-03	A7984805	1-4ozG₩	8260 STARS
08	3/29/2007	09/01/2007	SB-33-17/18-03	A7984805	1-4ozGW	8270 STARS
08	3/29/2007	09/01/2007	SB-32-20/21-03	A7984806	1-4ozGW	8260 STARS
08	3/29/2007	09/01/2007	SB-32-20/21-03	A7984806	1-4ozGW	8270 STARS
08	3/30/2007	09/01/2007	SW-20	A7984901	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-34	A7984902	1-8ozGW	8260;8270;LPH
90	3/30/2007	09/01/2007	SW-35	A7984903	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-36	A7984904	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-37	A7984905	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-38	A7984906	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-39	A7984907	1-8ozG₩	8260;8270;LPH
08	3/30/2007	09/01/2007	B-22	A7984908	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	B-23	A7984909	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	B-24	A7984910	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	8-25	A7984911	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	8-26	A7984912	1-8ozGW	8260;8270;LPH
08	3/29/2007	09/01/2007	SW-09	A7985001	1-8ozGW	8260;8270;LPH
08	3/29/2007	09/01/2007	SW-10	A7985002	1-8ozGW	8260;8270;LPH
08	3/29/2007	09/01/2007	SW-11	A7985003	1-8ozG₩	8260;8270;LPH
08	3/29/2007	09/01/2007	SW-12	A7985004	1-8ozGW	8260;8270;LPH
08	3/29/2007	09/01/2007	sw-13	A7985005	1-8ozGW	8260;8270;LPH
08	3/29/2007	09/01/2007	SW-14	A7985006	1-8ozG₩	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-15	A7985007	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	SW-16	A7985008	1-8ozG₩	8260;8270;LPH
08	3/30/2007	09/01/2007	S₩-17	A7985009	1-8ozGW	8260;82/0;LPN
08	3/30/2007	09/01/2007	SW-18	A7985010	1-8ozGW	8260;8270;LPN
08	3/30/2007	09/01/2007	S₩-19	A7985011	1-8ozGW	8260;8270;LPH
08	3/30/2007	09/01/2007	B-21	A7985012	1-8ozGW	8260;8270;LPH

**Company Confidential & Proprietary** 

#### Figure 24-4.

#### Example: Sample Acceptance Policy

All incoming work will be evaluated against the criteria listed below. Where applicable, data from any samples that do not meet the criteria listed below will be noted on the laboratory report defining the nature and substance of the variation. In addition the client will be notified either by telephone, fax or e-mail ASAP after the receipt of the samples.

- 1) Samples must arrive with labels intact with a Chain of Custody filled out completely. The following information must be recorded.
  - > Client name, address, phone number and fax number (if available)
  - > Project name and/or number
  - > The sample identification
  - > Date, time and location of sampling
  - > The collectors name
  - > The matrix description
  - > The container description
  - > The total number of each type of container
  - > Preservatives used
  - > Analysis requested
  - > Requested turnaround time (TAT)
  - > Any special instructions
  - > Purchase Order number or billing information (e.g. quote number) if available
  - > The date and time that each person received or relinquished the sample(s), including their signed name.
  - > The date and time of receipt must be recorded between the last person to relinquish the samples and the person who receives the samples in the lab, and they must be exactly the same.
  - > Information must be legible
- 2) Samples must be properly labeled.
  - > Use durable labels (labels provided by TestAmerica are preferred)
  - > Include a unique identification number
  - > Include sampling date and time & sampler ID
  - Include preservative used.
  - > Use indelible ink
  - > Information must be legible
- 3) Proper sample containers with adequate volume for the analysis and necessary QC are required for each analysis requested.
- 4) Samples must be preserved according to the requirements of the requested analytical method or in accordance with the instructions provided by TestAmerica. Note: Samples that are hand delivered to the laboratory immediately after collection may not have had time to cool sufficiently. In this case the samples will be considered acceptable as long as there is evidence that the chilling process has begun (arrival on ice).

- Chemical preservation (pH) will be verified prior to analysis and the project manager will be notified immediately if there is a discrepancy. If analyses will still be performed, all affected results will be flagged to indicate improper preservation.
- ➢ For Volatile Organic analyses in drinking water (Methods 502.2 or 524.2). Residual chlorine must be neutralized prior to preservation. If there is prior knowledge that the samples are not chlorinated, state it on the COC and use the VOA vials pre-preserved with HCI. The following are other options for a sampler and laboratory where the presence of chlorine is not known:
  - > 1. Test for residual chlorine in the field prior to sampling.
    - If no chlorine is present, the samples are to be preserved using HCI as usual.
    - > If chlorine is present, add either ascorbic acid or sodium thiosulfate prior to adding HCI.
  - 2. Use VOA vials pre-preserved with sodium thiosulfate or ascorbic acid and add HCI after filling the VOA vial with the sample.

#### **FOR WATER SAMPLES TESTED FOR CYANIDE – for NPDES samples**

- In the Field: Samples are to be tested for Sulfide using lead acetate paper prior to the addition of Sodium Hydroxide (NaOH). If sulfide is present, the sample must be treated with Cadmium Chloride and filtered prior to the addition of NaOH.
  - ➢ If the sulfide test and treatment is not performed in the field, the lab will test the samples for sulfide using lead acetate paper at the time of receipt and if sulfide is present in the sample, the client will be notified and given the option of retaking the sample and treating in the field per the method requirements or the laboratory can analyze the samples as delivered and qualify the results in the final report.
- It is the responsibility of the client to notify the laboratory if thiosulfate, sulfite, or thiocyanate are known or suspected to be present in the sample. This notification may be on the chain of custody. The samples may need to be subcontracted to a laboratory that performs a UV digestion. If the lab does not perform the UV digestion on samples that contain these compounds, the results must be qualified in the final report.
- The laboratory must test the sample for oxidizing agents (e.g. Chlorine) prior to analysis and treat according to the methods prior to distillation. (ascorbic acid or sodium arsenite are the preferred choice).
- 5) Sample Holding Times
  - TestAmerica will make every effort to analyze samples within the regulatory holding time. Samples must be received in the laboratory with enough time to perform the sample analysis. Except for short holding time samples (< 48hr HT) sample must be received with at least 48 hrs (2 working days) remaining on the holding time to ensure analysis.</p>
  - Analyses that are designated as "field" analyses (Odor, pH, Dissolved Oxygen, Disinfectant Residual; a.k.a. Residual Chlorine, and Redox Potential) should be analyzed ASAP by the field sampler prior to delivering to the lab (within 15 minutes). However, if the analyses are to be performed in the laboratory, TestAmerica will make every effort to analyze the samples within 24 hours from receipt of the samples in the testing laboratory. Samples for "field" analyses received after 4:00 pm on Friday or on the weekend will be analyzed no later than the next business day after receipt (Monday unless a holiday). Samples will remain refrigerated and sealed until the time of analysis.
- 6) All samples submitted for Volatile Organic analyses must have a Trip Blank submitted at the same time. TestAmerica will supply this blank with the bottle order.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-12 of 24-13

- 7) The project manager will be notified if any sample is received in damaged condition. TestAmerica will request that a sample be resubmitted for analysis.
- 8) Recommendations for packing samples for shipment.
  - > Pack samples in Ice rather than "Blue" ice packs.
  - Soil samples should be placed in plastic zip-lock bags. The containers often have dirt around the top and do not seal very well and are prone to intrusion from the water from melted ice.
  - Water samples would be best if wrapped with bubble-wrap or paper (newspaper, or paper towels work) and then placed in plastic zip-lock bags.
  - > Fill extra cooler space with bubble wrap.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 24-13 of 24-13

Figure 24-5.

#### Example: Cooler Receipt Form

TestAmerica Buffalo	Doc. Login/ARRF - Side A Rev 5 November 6 2007
SAMPLE LOGIN	JOB #
Shipment ID	Strict Internal COC: YES / NO
	Residual Chlorine Check:
	Radiation Check <0.02 mR/hr: YES / NO
ACProject / Task	/
TATBD/CD # OF SAMPL	ESTRIP BLANK Y/N #
SHIPPED BY	ATTACH SHIPPING TAGS
RECEIVED DATE / TIME:	
COOLER TEMP °C (<6 °C)	OK NO
Cooler Custody Seal intact? YES/NO NO	NE SEAL #
If NO to cooler temp or seal, PM notified? YES	(PM Name)
SUBCONTRACT YES/NO LAB	SM #
COMMENTS: SAMPLE TIME ACTUAL	+1HR +2 HR +3 HR NONE
Sample received outside hold time	
Headspace in VOA vials	
Problems with bottle labels	
OTHER SAMPLE RECEIPT COMMENTS (Fill out	ARRF, see reverse)
PRESERVATION CHECKED YES	NO NA Initials
ARE SAMPLE DATES AND TIMES CORRECT?	Initials
WERE ALL THE APPROPRIATE TESTS ASSIG	NED? Initials
Temp.Cert.Loss:	

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-1 of 25-9

#### SECTION 25.0

## ASSURING THE QUALITY OF TEST RESULTS (NELAC 5.5.9)

#### 25.1 <u>OVERVIEW</u>

In order to assure our clients of the validity of their data, the laboratory continuously evaluates the quality of the analytical process. The analytical process is controlled not only by instrument calibration as discussed in Section 21, but also by routine process quality control measurements (e.g. Blanks, Laboratory Control Samples (LCS), Matrix Spikes (MS), duplicates (DUP), surrogates, Internal Standards (IS)). These quality control checks are performed as required by the method or regulations to assess precision and accuracy. In addition to the routine process quality control samples, Proficiency Testing (PT) Samples (concentrations unknown to laboratory) are analyzed to help ensure laboratory performance.

#### 25.2 <u>CONTROLS</u>

Sample preparation or pre-treatment is commonly required before analysis. Typical preparation steps include homogenization, grinding, solvent extraction, sonication, acid digestion, distillation, reflux, evaporation, drying and ashing. During these pre-treatment steps, samples are arranged into discreet manageable groups referred to as preparation (prep) batches. Prep batches provide a means to control variability in sample treatment. Control samples are added to each prep batch to monitor method performance and are processed through the entire analytical procedure with investigative/field samples.

#### 25.3 NEGATIVE CONTROLS

**25.3.1** <u>Method Blanks</u> are used to assess preparation and analysis for possible contamination during the preparation and processing steps.

- **25.3.1.1** The method blank is prepared from a clean matrix similar to that of the associated samples that is free from target analytes (e.g., Reagent water, Ottawa sand, glass beads, etc.) and is processed along with and under the same conditions as the associated samples.
- **25.3.1.2** The method blank goes through all of the steps of the process (including as necessary: filtration, clean-ups, etc.).
- **25.3.1.3** The specific frequency of use for method blanks during the analytical sequence is defined in the specific standard operating procedure for each analysis. Generally it is 1 for each batch of samples; not to exceed 20 environmental samples.
- **25.3.1.4** Evaluation criteria and corrective action for method blanks is defined in the specific standard operating procedure for each analysis. Generally, corrective action is taken if the concentration of a target analyte in the blank is at or above the reporting limit as established by the method or regulation:
  - The source of contamination is investigated

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-2 of 25-9

- Measures are taken to minimize or eliminate the source of the contamination
- Affected samples are reprocessed or the results are qualified on the final report.

**25.3.2** <u>Calibration Blanks</u> are prepared and analyzed along with calibration standards where applicable. They are prepared using the same reagents that are used to prepare the standards. In some analyses the calibration blank may be included in the calibration curve.

**25.3.3 Instrument Blanks** are blank reagents or reagent water that may be processed during an analytical sequence in order to assess contamination in the analytical system. In general, instrument blanks are used to differentiate between contamination caused by the analytical system and that caused by the sample handling or sample prep process. Instrument blanks may also be inserted throughout the analytical sequence to minimize the effect of carryover from samples with high analyte content.

**25.3.4** <u>**Trip Blanks**</u> are required to be submitted by the client with each shipment of samples requiring aqueous and solid volatiles analyses. Additionally, trip blanks may be prepared and analyzed for volatile analysis of air samples, when required by the client. A trip blank may be purchased (certified clean) or is prepared by the laboratory by filling a clean container with pure deionized water that has been purged to remove any volatile compounds. Appropriate preservatives are also added to the container. The trip blank is sent with the bottle order and is intended to reflect the environment that the containers are subjected to throughout shipping and handling and help identify possible sources if contamination is found. The field sampler returns the trip blank in the cooler with the field samples. Trip Blanks are also sometimes referred to as Travel Blanks.

**25.3.5** <u>Field Blanks</u> are sometimes used for specific projects by the field samplers. A field blank is prepared in the field by filling a clean container with pure reagent water and appropriate preservative, if any, for the specific sampling activity being undertaken. (EPA OSWER)

**25.3.6** <u>Equipment Blanks</u> are also sometimes created in the field for specific projects. An equipment blank is a sample of analyte-free media which has been used to rinse common sampling equipment to check effectiveness of decontamination procedures. (NELAC)

**25.3.7** <u>Holding Blanks</u>, also referred to as refrigerator or freezer blanks, are used to monitor the sample storage units for volatile organic compounds during the storage of VOA samples in the laboratory (refer to section 24.4).

**25.3.8** <u>Field blanks, equipment blank and trip blanks</u>, when received, are analyzed in the same manner as other field samples. When known, blanks should not be selected for matrix QC, as it does not provide information on the behavior of the target compounds in the field samples. Usually, the client sample ID will provide information to identify the field blanks with labels such as "FB", "EB", or "TB".

#### 25.4 **POSITIVE CONTROLS**

Control samples (e.g., QC indicators) are analyzed with each batch of samples to evaluate data based upon (1) Method Performance (Laboratory Control Sample (LCS) or Blank Spike (BS)),
Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-3 of 25-9

which entails both the preparation and measurement steps; and (2) Matrix Effects (Matrix Spike (MS) (Matrix spikes are not applicable to air) or Sample Duplicate (MD, DUP), which evaluates field sampling accuracy, precision, representativeness, interferences, and the effect of the matrix on the method performed. Each regulatory program and each method within those programs specify the control samples that are prepared and/or analyzed with a specific batch

Note that frequency of control samples vary with specific regulatory, methodology and project specific criteria. Complete details on method control samples are as listed in each analytical SOP.

# 25.4.1 Method Performance Control - Laboratory Control Sample (LCS)

- **25.4.1.1** The LCS measures the accuracy of the method in a blank matrix and assesses method performance independent of potential field sample matrix affects in a laboratory batch.
- **25.4.1.2** The LCS is prepared from a clean matrix similar to that of the associated samples that is free from target analytes (for example: Reagent water, Ottawa sand, glass beads, etc.) and is processed along with and under the same conditions as the associated samples. The LCS is spiked with verified known amounts of analytes or is made of a material containing known and verified amounts of analytes, taken through all preparation and analysis steps along with the field samples. Where there is no preparation taken for an analysis (such as in aqueous volatiles), or when all samples and standards undergo the same preparation and analysis process (such as Phosphorus), a calibration verification standard may be reported as the LCS. In some instances where there is no practical clean solid matrix available, aqueous LCS's may be processed for solid matrices; final results may be calculated as mg/kg or ug/kg, assuming 100% solids and a weight equivalent to the aliquot used for the corresponding field samples, to facilitate comparison with the field samples.
- **25.4.1.3** Certified pre-made reference material purchased from a NIST/A2LA accredited vendor may also be used for the LCS when the material represents the sample matrix or the analyte is not easily spiked (e.g. solid matrix LCS for metals, TDS, etc.).
- **25.4.1.4** As stated in the opening of this section, the LCS goes through all of the steps of the process (including as necessary: filtration, clean-ups, etc.).
- **25.4.1.5** The specific frequency of use for LCS during the analytical sequence is defined in the specific standard operating procedure for each analysis. It is generally 1 for each batch of samples; not to exceed 20 environmental samples.
- 25.4.1.6 If the mandated or requested test method, or project requirements, do not specify the spiking components, the laboratory shall spike all reportable components to be reported in the Laboratory Control Sample (and Matrix Spike) where applicable (e.g. no spike of pH). However, in cases where the components interfere with accurate assessment (such as simultaneously spiking chlordane, toxaphene and PCBs in Method 608), the test method has an extremely long list of components or components are incompatible, at a minimum, a representative number of the listed components (see below) shall be used to control the test method. The selected

components of each spiking mix shall represent all chemistries, elution patterns and masses, permit specified analytes and other client requested components. However, the laboratory shall ensure that all reported components are used in the spike mixture within a two-year time period.

- **25.4.1.6.1** For methods that have 1-10 target analytes, spike all components.
- **25.4.1.6.2** For methods that include 11-20 target analytes, spike at least 10 or 80%, whichever is greater.
- 25.4.1.6.3 For methods with more than 20 target analytes, spike at least 16 components.
- **25.4.1.6.4** Exception: Due to analyte incompatibility in pesticides, Toxaphene and Chlordane are only spiked at client request based on specific project needs.
- **25.4.1.6.5** Exception: Due to analyte incompatibility between the various PCB aroclors, aroclors 1016 and 1260 are used for spiking as they cover the range of all of the aroclors. Specific aroclors may be used by request on a project specific basis.
- **25.4.1.7** <u>Accuracy Calculation</u>: Percent Recovery (%R) Calculation (applies to LCS, CCV, Surrogates, and Matrix Spikes.

$$\% R = \frac{AV}{TV} \times 100$$

Where: AV = Analyzed Value TV = True Value

# 25.5 <u>SAMPLE MATRIX CONTROLS</u>

# 25.5.1 <u>Matrix Spikes (MS)</u>

- **25.5.1.1** The Matrix spike is used to assess the effect sample matrix of the spiked sample has on the precision and accuracy of the results generated by the method used.
- **25.5.1.2** An MS is essentially a sample fortified with a known amount of the test analyte(s). At a minimum, with each matrix-specific batch of samples processed, an MS is carried through the complete analytical procedure. Unless specified by the client, samples used for spiking are randomly selected and rotated between different client projects.
- **25.5.1.3** If the mandated or requested test method does not specify the spiking components, the laboratory shall spike all reportable components to be reported in the Laboratory Control Sample and Matrix Spike. However, in cases where the components interfere with accurate assessment (such as simultaneously spiking chlordane, toxaphene and PCBs in Method 608), the test method has an extremely long list of components or components are incompatible, a representative number of the listed components (see LCS analytes 25.4.1.6 above) may be used to control the test method. The selected components of each spiking mix shall represent all

chemistries, elution patterns and masses, permit-specified analytes and other client requested components. However, the laboratory shall ensure that all reported components are used in the spike mixture within a two-year time period.

**25.5.1.4** The percent recovery calculation for matrix spikes is essentially the same as the calculation shown in 25.2.1.7 except that:

AV = Sp - Sa

Where: Sp = Spike result Sa = Sample result

# 25.5.2 <u>Surrogate Spikes</u>

- **25.5.2.1** Surrogate Spikes are similar to matrix spikes except the analytes are compounds with properties that mimic the analyte of interest and are unlikely to be found in environment samples.
- **25.5.2.2** Surrogate compounds are added to all samples, standards, and blanks, for all organic chromatography methods except when the matrix precludes its use or when a surrogate is not available. The recovery of the surrogates is compared to the acceptance limits for the specific method (also refer to Section 25.5). Poor surrogate recovery may indicate a problem with sample composition and shall be reported, with data qualifiers, to the client whose sample produced poor recovery.

# 25.5.3 <u>Duplicates</u>

- **25.5.3.1** For a measure of analytical precision, with each matrix-specific batch of samples processed, a matrix duplicate (MD or DUP) sample, matrix spike duplicate (MSD), or LCS duplicate (LCSD) is carried through the complete analytical procedure. Duplicate samples are usually analyzed with methods that do not require matrix spike analysis. LCSD's are normally not performed except when regulatory agencies or client specifications require them. The recoveries for the spiked duplicate samples must meet the same laboratory established recovery limits as the accuracy QC samples. If an LCSD is analyzed both the LCS and LCSD must meet the same recovery criteria and be included in the final report. The precision measurement is reported as "Relative Percent Difference" (RPD). Poor precision between duplicates (except LCS/LCSD) may indicate non-homogeneous matrix or sampling.
- 25.5.3.2 <u>Precision Calculation</u> (Relative Percent Difference RPD)

$$RPD = \frac{|S-D|}{(S+D)} \times 100$$

Where: S=Sample Concentration D=Duplicate Concentration

25.5.4 Internal Standards

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-6 of 25-9

- **25.5.4.1** In most organic analyses, internal standards are spiked into all environmental and quality control samples (including the initial calibration standards). An internal standard is also used with some metals analyses. It is added to sample extracts after the extraction (post-prep). The acceptance criteria in most methods are 50% to 200% of the responses in the mid-point of the corresponding calibration curve. Consult the method-specific SOPs for details on the internal standard compounds, calculations and acceptance criteria.
- **25.5.4.2** When the internal standard recoveries fall outside these limits, if there are not obvious chromatographic interferences, reanalyze the sample to confirm a possible matrix effect. If the recoveries confirm or there was obvious interference, results are reported from the original analysis and a qualifier is added. If the reanalysis meets internal standard recovery criteria, the second run is reported (or both are reported if requested by the client).

# 25.6 ACCEPTANCE CRITERIA (CONTROL LIMITS)

**25.6.1** Each individual analyte in the LCS, MS, or Surrogate Spike are evaluated against the control limits as published in the test method. Where there are no established acceptance criteria, the laboratory calculates control limits with the use of statistical evaluations or, in some cases, utilizes client project specific or regulatory mandated control limits. When this occurs, the regulatory or project limits will supersede the laboratory's in-house limits.

**Note:** For methods, analytes and matrices with very limited data (e.g., unusual matrices not analyzed often), interim limits are established using available data or by analogy to similar methods or matrices.

**25.6.2** Once control limits have been established, they are verified, reviewed, and updated if necessary on an annual basis unless the method requires more frequent updating (e.g. EPA SW846 8000 series methods). Control limits are established per method (as opposed to per instrument) regardless of the number of instruments utilized.

- **25.6.2.1** The lab should consider the effects of the spiking concentration control limits, and to avoid censoring of data. The acceptance criteria for recovery and precision are often a function of the spike concentration used. Therefore, caution must be used when pooling data to generate control limits.
- **25.6.2.2** Not only should the results all be from a similar matrix, but the spiking levels should also be approximately the same (within a factor of 2). Similarly, the matrix spike and surrogate results should all be generated using the same set of extraction, cleanup and analysis techniques.
- **25.6.2.3** The laboratory should try and avoid discarding data that do not meet a preconceived notion of acceptable performance. This results in a censored data set, which, when used to develop acceptance criteria, will lead to unrealistically narrow criteria. For a 99% confidence interval, 1 out of every 100 observations likely will still fall outside the limits. For methods with long analyte lists this may mean occasional failures every batch or two. While professional judgment is important in evaluating data to be

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-7 of 25-9

used to develop acceptance criteria, specific results are not discarded simply because they do not meet one's expectations. However, data points shall be discarded if they were the result of human or mechanical error or sample concentration exceeded spike level by > 4x.

**25.6.3** Laboratory generated % Recovery acceptance (control) limits are generally established by taking  $\pm$  3 Standard Deviations (99% confidence level) from the average recovery of a minimum of 20-30 data points (more points are preferred).

- **25.6.3.1** Regardless of the calculated limit, the limit should be no tighter than the Calibration Verification (ICV/CCV). (Unless the analytical method specifies a tighter limit).
- **25.6.3.2** In-house limits cannot be any wider than those mandated in a regulated analytical method.
- **25.6.3.3** The lowest acceptable recovery limit will be 10% (the analyte must be detectable). Exception: The lowest acceptable recovery limit for Benzidine will be 5% and the analyte must be detectable.
- **25.6.3.4** The maximum acceptable recovery limit will be 150%.
- **25.6.3.5** The maximum acceptable RPD limit will be 35% for waters and 40% for soils. The minimum RPD limit is 10%.
- **25.6.3.6** If either the high or low end of the control limit changes by  $\leq 5\%$  from previous, the data points are inspected and, using professional judgment, the limits may be left unchanged if there is no affect on laboratory ability to meet the existing limits.

**25.6.4** The lab must be able to generate a current listing of their control limits and track when the updates are performed. In addition, the laboratory must be able to recreate historical control limits.

**25.6.4.1** The control limits are maintained in the laboratory LIMs system. The limits for each analyte/method/matrix combination are assigned effective and expiration dates. The QA department is able to query the LIMs sytem and print an active list of control limits based on this database. The most current laboratory limits (based on the effective/expiration dates) are reflected on the laboratory worksheets and final reports unless superseded by project specific limits.

**25.6.5** A LCS that is within the acceptance criteria establishes that the analytical system is in control and is used to validate the process. Samples that are analyzed with an LCS with recoveries outside of the acceptance limits may be determined as out of control and should be reanalyzed if possible. If reanalysis is not possible, then the results for all affected analytes for samples within the same batch must be qualified when reported. The internal corrective action process (see Section 13) is also initiated if an LCS exceeds the acceptance limits. Sample results may be qualified and reported without reanalysis if:

- **25.6.5.1** The analyte results are below the reporting limit and the LCS is above the upper control limit.
- **25.6.5.2** If the analytical results are above the relevant regulatory limit and the LCS is below the lower control limit.
- **25.6.5.3** Or, for NELAC and Department of Defense (DOD) work, there are an allowable number of Marginal Exceedances (ME). This information may be found in the laboratory specific SOP BF-QA-005, "Preventative and Corrective Action".
  - <11 analytes 0 marginal exceedances are allowed.
  - 11 30 Analytes 1 marginal exceedance is allowed
  - 31-50 Analytes 2 marginal exceedances are allowed
  - 51-70 Analytes 3 marginal exceedances are allowed
  - 71-90 Analytes 4 marginal exceedances are allowed
  - > 90 Analytes 5 marginal exceedances are allowed
  - **25.6.5.3.1** Marginal exceedances are recovery exceedances between 3 SD and 4 SD from the mean recovery limit (NELAC).
  - **25.6.5.3.2** Marginal exceedances must be random. If the same analyte exceeds the LCS control limit repeatedly, it is an indication of a systematic problem. The source of the error must be located and corrective action taken.
  - **25.6.5.3.3** Though marginal excedences may be allowed, the data must still be qualified to indicate it is outside of the normal limits.

**25.6.6** If the MS/MSDs do not meet acceptance limits, the MS/MSD and the associated spiked sample is reported with a qualifier for those analytes that do not meet limits. If obvious preparation errors are suspected, or if requested by the client, unacceptable MS/MSDs are reprocessed and reanalyzed to prove matrix interference. A more detailed discussion of acceptance criteria and corrective action can be found in the method specific SOPs and in Section 13.

**25.6.7** If a surrogate standard falls outside the acceptance limits, if there is not obvious chromatographic matrix interference, reanalyze the sample to confirm a possible matrix effect. If the recoveries confirm or there was obvious chromatographic interference, results are reported from the original analysis and a qualifier is added. If the reanalysis meets surrogate recovery criteria, the second run is reported (or both are reported if requested by the client). Under certain circumstances, where all of the samples are from the same location and share similar chromatography, the reanalysis may be performed on a single sample rather than all of the samples and if the surrogate meets the recovery criteria in the reanalysis, all of the affected samples would require reanalysis.

# 25.7 METHOD DETECTION LIMITS (MDLs)

MDLs, calculated as described in Section 20.7, are updated or verified annually or more often if required by the method.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 25-9 of 25-9

# 25.8 ADDITIONAL PROCEDURES TO ASSURE QUALITY CONTROL

**25.8.1** The laboratory has written procedures to assure the accuracy of the test method including calibration (see Section 21), use of certified reference materials (see Section 22) and use of PT samples (see Section 16).

**25.8.2** A discussion regarding MDLs, Limit of Detection (LOD) and Limit of Quantitation (LOQ) can be found in Section 20.

**25.8.3** Use of formulae to reduce data is discussed in the method standard operating procedures and in Section 21.

**25.8.4** Selection of appropriate reagents and standards is included in Section 9 and 22.

**25.8.5** A discussion on selectivity of the test is included in Section 5.

**25.8.6** Constant and consistent test conditions are discussed in Section 19.

**25.8.7** The laboratories sample acceptance policy is included in Section 24.

**25.8.8** A listing of the type of test result correlations that are looked at during report review (e.g. Total Chromium should be greater or equal to Hexavalent Chromium) is included in Section 20.13.4.5 and in laboratory specific SOP BF-PM-005, "Correctness of Analysis".

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 26-1 of 26-7

# SECTION 26.0

# REPORTING RESULTS (NELAC 5.5.10)

# 26.1 <u>OVERVIEW</u>

The results of each test are reported accurately, clearly, unambiguously, and objectively in accordance with State and Federal regulations as well as client requirements. Analytical results are issued in a format that is intended to satisfy customer and laboratory accreditation requirements as well as provide the end user with the information needed to properly evaluate the results. Where there is a conflict between the client requirements and accreditation requirements or data usability information, accreditation requirements and data usability information will take precedence over client requests. A variety of report formats are available to meet specific needs.

In cases where a client asks for simplified reports, there must be a written request from the client. There still must be enough information that would show any analyses that were out of conformance (QC out of limits) and there should be a reference to a full report that is made available to the client.

Review of reported data is included in Section 20.

#### 26.2 <u>TEST REPORTS</u>

Analytical results are reported in a format that is satisfactory to the client and meets all requirements of applicable accrediting authorities and agencies. A variety of report formats are available to meet specific needs. The report is printed on laboratory letterhead, reviewed, and signed by the appropriate project manager. At a minimum, the standard laboratory report shall contain the following information:

**26.2.1** A report title (e.g. Analytical Report) with a "sample results" column header.

**26.2.2** The report cover page is printed on company letterhead which includes the laboratory name, address and telephone number.

**26.2.3** A unique identification of the report (e.g. job number) and on each page an identification in order to ensure the page is recognized as part of the report and a clear identification of the end.

**Note:** Page numbers of report are represented as # / ##. Where the first number is the page number and the second is the total number of pages.

**26.2.4** A copy of the chain of custody (COC).

- Any COCs involved with Subcontracting are included.
- In most cases, the applicable COC is paginated and is an integral part of the report.

• Any additional addenda to the report must be treated in a similar fashion so it is a recognizable part of the report and cannot accidentally get separated from the report (eg. Sampling information).

**26.2.5** The name and address of client and a project name/number, if applicable.

26.2.6 Client project manager or other contact

**26.2.7** Description and unambiguous identification of the tested sample(s) including the client identification code.

**26.2.8** Date of receipt of sample, date and time of collection, and date(s) of test preparation and performance, and time of preparation or analysis if the required holding time for either activity is less than or equal to 72 hours.

**26.2.9** Date reported or date of revision, if applicable.

26.2.10 Method of analysis including method code (EPA, Standard Methods, etc).

**26.2.11** Practical quantitation limits or client reporting limit.

26.2.12 Method detection limits (if requested)

**26.2.13** Definition of Data qualifiers and reporting acronyms (e.g. ND).

26.2.14 Sample results.

**26.2.15** QC data consisting of method blank, surrogate, LCS, and MS/MSD recoveries and control limits (if requested).

**26.2.16** Condition of samples at receipt including temperature. This may be accomplished in a narrative or by attaching sample login sheets (Refer to Sec. 26.2.4 – Item 3 regarding additional addenda). Sample temperatures are recorded in the report case narrative and on the CoC. Deviations from normal conditions (e.g., preservation, breakage) are recorded in the report case narrative.

**26.2.17** A statement to the effect that the results relate only to the items tested and the sample as received by the laboratory.

**26.2.18** A statement that the report shall not be reproduced except in full, without prior express written approval by the laboratory coordinator.

**26.2.19** A signature and title of the person(s) accepting responsibility for the content of the report and date of issue. Signatories are appointed by the Lab Director.

**26.2.20** When NELAC accreditation is required, the lab shall certify that the test results meet all requirements of NELAC or provide reasons and/or justification if they do not.

**26.2.21** The laboratory includes a cover letter.

**26.2.22** Where applicable, a narrative to the report that explains the issue(s) and corrective action(s) taken in the event that a specific accreditation or certification requirement was not met.

**26.2.23** When Soil samples are analyzed, a specific identification as to whether soils are reported on a "wet weight" or "dry weight" basis.

**26.2.24** Appropriate laboratory certification number for the state of origin of the sample if applicable.

**26.2.25** If only part of the report is provided to the client (client requests some results before all of it is complete), it must be clearly indicated on the report and that a complete report will follow once all of the work has been completed.

**26.2.26** Any out of network subcontracted analysis results are provided as an addendum to the report on the official letterhead of the subcontractor. All in-network subcontracting is clearly identified on the report as to which laboratory performed a specific analysis.

# 26.3 REPORTING LEVEL OR REPORT TYPE

**TestAmerica Buffalo** offers four levels of quality control reporting. Each level, in addition to its own specific requirements, contains all the information provided in the preceding level. The packages provide the following information in addition to the information described above:

- Level I is a report with the features described in Section 26.2 above.
- Level II is a Level I report plus summary information, including results for the method blank, percent recovery for laboratory control samples and matrix spike samples, and the RPD values for all MSD and sample duplicate analyses.
- Level III contains all the information supplied in Level II, but presented on CLP-like summary forms, and relevant calibration information. A Level II report is not included, unless specifically requested. No raw data is provided.
- Level IV is the same as Level III with the addition of all raw supporting data.

In addition to the various levels of QC packaging, the laboratory also provides reports in diskette deliverable form. Initial reports may be provided to clients by facsimile. All faxed reports are followed by hardcopy. Procedures used to ensure client confidentiality are outlined in Section 26.7.

# 26.3.1 <u>Electronic Data Deliverables (EDDs)</u>

EDDs are routinely offered as part of TestAmerica's services. *TestAmerica Buffalo* offers a variety of EDD formats including Environmental Restoration Information Management System (ERPIMS), Excel, Dbase, GISKEY, and Text Files.

EDD specifications are submitted to the IT department by the PM for review and undergo the contract review process. Once the facility has committed to providing data in a specific

electronic format, the coding of the format may need to be performed. This coding is documented and validated. The validation of the code is retained by the IT staff coding the EDD.

EDDs shall be subject to a review to ensure their accuracy and completeness. If EDD generation is automated, review may be reduced to periodic screening if the laboratory can demonstrate that it can routinely generate that EDD without errors. Any revisions to the EDD format must be reviewed until it is demonstrated that it can routinely be generated without errors. If the EDD can be reproduced accurately and if all subsequent EDDs can be produced error-free, each EDD does not necessarily require a review.

# 26.4 SUPPLEMENTAL INFORMATION FOR TEST

The lab identifies any unacceptable QC analyses or any other unusual circumstances or observations such as environmental conditions and any non-standard conditions that may have affected the quality of a result. This is typically in the form of a footnote or a qualifier and/or a narrative explaining the discrepancy in the front of the report. Refer to Appendix 7 for a list of the laboratory's standard footnotes and qualifiers.

**26.4.1** Numeric results with values outside of the calibration range, either high or low are qualified as 'estimated'.

**26.4.2** Where quality system requirements are not met, a statement of compliance/noncompliance with requirements and/or specifications, including identification of test results derived from any sample that did not meet NELAC sample acceptance requirements such as improper container, holding time, or temperature.

**26.4.3** Where applicable, a statement on the estimated uncertainty of measurements; information on uncertainty is needed when a client's instructions so require.

**26.4.4** Opinions and Interpretations - The test report contains objective information, and generally does not contain subjective information such as opinions and interpretations. If such information is required by the client, the Laboratory Director will determine if a response can be prepared. If so, the Laboratory Director will designate the appropriate member of the management team to prepare a response. The response will be fully documented, and reviewed by the Laboratory Director, before release to the client. There may be additional fees charged to the client at this time, as this is a non-routine function of the laboratory.

**Note:** Review of data deliverable packages for submittal to regulatory authorities requires responses to non-conforming data concerning potential impact on data quality. This necessitates a limited scope of interpretation, and this work is performed by the QA Department. This is the only form of "interpretation" of data that is routinely performed by the laboratory.

When opinions or interpretations are included in the report, the laboratory provides an explanation as to the basis upon which the opinions and interpretations have been made. Opinions and interpretations are clearly noted as such and where applicable, a comment should be added suggesting that the client verify the opinion or interpretation with their regulator.

# 26.5 ENVIRONMENTAL TESTING OBTAINED FROM SUBCONTRACTORS

If *TestAmerica Buffalo* is not able to provide the client the requested analysis, the samples would be subcontracted following the procedures outlined in Section 8.

Data reported from analyses performed by a subcontractor laboratory are clearly identified as such on the analytical report provided to the client. Results from a subcontract laboratory outside of the TestAmerica network are reported to the client on the subcontract laboratory's original report stationary and the report includes any accompanying documentation.

# 26.6 <u>CLIENT CONFIDENTIALITY</u>

In situations involving the transmission of environmental test results by telephone, facsimile or other electronic means, client confidentiality must be maintained.

TestAmerica will not intentionally divulge to any person (other than the Client or any other person designated by the Client in writing) any information regarding the services provided by TestAmerica or any information disclosed to TestAmerica by the Client. Furthermore, information <u>known</u> to be potentially endangering to national security or an entity's proprietary rights will not be released.

**Note:** This shall not apply to the extent that the information is required to be disclosed by TestAmerica under the compulsion of legal process. TestAmerica will, to the extent feasible, provide reasonable notice to the client before disclosing the information.

**Note:** Authorized representatives of an accrediting authority are permitted to make copies of any analyses or records relevant to the accreditation process, and copies may be removed from the laboratory for purposes of assessment.

**26.6.1** Report deliverable formats are discussed with each new client. If a client requests that reports be faxed or e-mailed, the reports are faxed with a cover sheet or e-mailed with the following note that includes a confidentiality statement similar to the following:

This material is intended only for the use of the individual(s) or entity to whom it is addressed, and may contain information that is privileged and confidential. It is our policy that facsimiles are intended for and should be used for business purposes only. If you are not the intended recipient, or the employee or agent responsible for delivering this material to the intended recipient, you are hereby notified that any dissemination, distribution or copying of this communication is strictly prohibited. If you have received this communication in error, please notify the sender.

# 26.7 FORMAT OF REPORTS

The format of reports are designed to accommodate each type of environmental test carried out and to minimize the possibility of misunderstanding or misuse.

# 26.8 AMENDMENTS TO TEST REPORTS

Corrections, additions, or deletions to reports are only made when justification arises through supplemental documentation. Justification is documented using the laboratory's corrective action system (refer to Section 13).

The revised report is retained on the Archive data server, as is the original report. The revised report is stored in the Archive data server under the sample number followed by "R". The revised report will have the word "revised" appended to the cover letter.

When the report is re-issued, a notation of "revised" is placed on the cover/signature page of the report. A brief explanation of reason for the re-issue is included in the report case narrative.

# 26.9 POLICIES ON CLIENT REQUESTS FOR AMENDMENTS

# 26.9.1 Sample Reanalysis Policy

Because there is a certain level of uncertainty with any analytical measurement a sample reanalysis may result in either a higher or lower value from an initial sample analysis. There are also variables that may be present (e.g. sample homogeneity, analyte precipitation over time, etc.) that may affect the results of a reanalysis. Based on the above comments, the laboratory will reanalyze samples at a client's request with the following caveats. Client specific arrangements for reanalysis protocols can be established.

- Homogenous samples: If a reanalysis agrees with the original result to within the RPD limits for MS/MSD or Duplicate analyses, or within <u>+</u> 1 reporting limit for samples <u><</u> 5x the reporting limit, the original analysis will be reported. At the client's request, both results may be reported on the same report but not on two separate reports.
- If the reanalysis does not agree (as defined above) with the original result, then the laboratory will investigate the discrepancy and reanalyze the sample a third time for confirmation if sufficient sample is available.
- Any potential charges related to reanalysis are discussed in the contract terms and conditions or discussed at the time of the request. The client will typically be charged for reanalysis unless it is determined that the lab was in error.
- Due to the potential for increased variability, reanalysis may not be applicable to Nonhomogenous, Encore, and Sodium Bisulfate preserved samples. See the Department Manager, Operations Manager, Technical Director, QA Manager or Laboratory Director if unsure.

# 26.9.2 Policy on Data Omissions or Reporting Limit Increases

Fundamentally, our policy is simply to not omit previously reported results (including data qualifiers) or to not raise reporting limits and report sample results as ND. This policy has few exceptions. Exceptions are:

- Laboratory error.
- Sample identification is indeterminate (confusion between COC and sample labels).

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Page 26-7 of 26-7

- An incorrect analysis (not analyte) was requested (e.g., COC lists 8315 but client wanted 8310). A written request for the change is required.
- Incorrect limits reported based on regulatory requirements.
- The requested change has absolutely <u>no possible</u> impact on the interpretation of the analytical results and there is <u>no possibility</u> of the change being interpreted as misrepresentation by anyone inside or outside of our company.

# 26.9.3 Multiple Reports

TestAmerica does not issue multiple reports for the same workorder where there is different information on each report (this does not refer to copies of the same report) unless required to meet regulatory needs and approved by QA.

Appendix 1.



Refer to CA-L-P-001 for complete policy.

# TestAmerica EMPLOYEE ETHICS STATEMENT

I understand that TestAmerica is committed to ensuring the highest standard of quality and integrity of the data and services provided to our clients. I have read the Ethics Policy of the Company.

- With regard to the duties I perform and the data I report in connection with my employment at the Company, I agree that:
- I will not intentionally report data values that are inconsistent with the actual values observed or measured.
- I will not intentionally report the dates, times, sample or QC identifications, or method citations of data analyses that are not the actual dates, times, sample or QC identifications, or method citations.
- I will not intentionally misrepresent another individual's work as my own or represent my own work as someone else's.
- I will not intentionally misrepresent any data where data does not meet Method or QC requirements. If it is to be reported, I will report it with all appropriate notes and/or qualifiers; I shall not modify data (either sample or QC data) unless the modification can be technically justified through a measurable analytical process, such as one deemed acceptable to the laboratory's Standard Operating Procedures, Quality Assurance Manual or Technical Director. All such modifications must be clearly and thoroughly documented in the appropriate laboratory notebooks/worksheets and/or raw data and include my initials or signature and date.
- I shall not make false statements to, or seek to otherwise deceive, members of Management or their representatives, agents, or clients/customers. I will not, through acts of commission, omission, erasure, or destruction, improperly report measurement standards, quality control data, test results or conclusions.
- I shall not compare or disclose results for any Performance Testing (PT) sample, or other similar QA or QC requirements, with any employee of any other laboratory, including any other TestAmerica laboratory, prior to the required submission date of the results to the person, organization, or entity supplying the PT sample.
- I shall immediately inform my supervisor or other member of management regarding any intentional or unintentional reporting of my own inauthentic data. Such report shall be given both orally and in writing to the supervisor or other member of management contacted and to the local Quality Assurance Manager. The Quality Assurance Manager will initial and date the information and return a copy to me. I shall not condone any accidental or intentional reporting of inauthentic data by other employees and will immediately report its occurrence. If I have actual knowledge of such acts committed by any other employees, and I do not report such information to designated members of Management, it shall be considered as serious as if I personally committed the offense. Accordingly, in that event, I understand that I may be subject to immediate termination of employment.
- I understand that if any supervisor, manager, or representative of TestAmerica management instructs, requests, or directs me to perform any of the aforementioned improper laboratory practices, or if I am in doubt or uncertain as to whether or not such laboratory practices are proper, I will not

comply. In fact, I must report such event to all appropriate members of Management including, but not limited to, the Lab Director, all supervisors and managers with direct line reporting relationship between me and the Lab Director, and the local Quality Assurance representative, excluding such individuals who participated in such perceived improper instruction, request, or directive. In addition, I may contact Corporate Quality Assurance / Ethics Compliance Officer(s) for assistance.

- I understand the critical importance of accurately reporting data, measurements, and results, whether initially requested by a client, or retained by TestAmerica and submitted to a client at a later date, or retained by TestAmerica for subsequent internal use;
- I will not share the pricing or cost data of Vendors or Suppliers with anyone outside of the TestAmerica family of companies.
- I shall not accept gifts of a value that would adversely influence judgment.
- I shall avoid conflicts of interest and report any potential conflicts to the management (e.g. employment or consulting with competitors, clients, or vendors).
- I shall not participate in unfair competition practices (e.g. slandering competitors, collusion with other labs to restrict others from bidding on projects).
- I shall not misrepresent certifications and status of certifications to clients or regulators.
- I shall not intentionally discharge wastes illegally down the drain or onto the ground.
- I understand that any attempt by management or an employee to circumvent these policies will be subject to disciplinary action.

As a TestAmerica employee, I understand that I have the responsibility to conduct myself with integrity in accordance with the ethical standards described in the Ethics Policy. I will also report any information relating to possible kickbacks or violations of the Procurement Integrity Act, or other questionable conduct in the course of sales or purchasing activities. I will not knowingly participate in any such activity and will report any actual or suspected violation of this policy to management.

I understand that if my job includes supervisory responsibilities, I shall not instruct, request, or direct any subordinate to perform any laboratory practice which is unethical or improper. Also, I shall not discourage, intimidate, or inhibit an employee who may choose to appropriately appeal my supervisory instruction, request, or directive which the employee perceives to be improper, nor retaliate against those who do.

The Ethics Policy has been explained to me by my supervisor or at a training session, and I have had the opportunity to ask questions if I did not understand any part of it. I understand that any violation of this policy subjects me to disciplinary action, which can include termination of my employment. In addition, I understand that any violation of this policy which relates to work under a government contract or subcontract could also subject me to the potential for prosecution under federal law.

EMPLOYEE SIGNATURE	Date
Supervisor/Trainer:	Date

Work Instruction No. CA-WI-005

# TestAmerica

# CONFIDENTIALITY AND PROPRIETARY INFORMATION AGREEMENT

TestAmerica and their predecessors, in their businesses, have developed and use commercially valuable technical and non-technical information and to guard the legitimate interests of TestAmerica and its clients, it is necessary to protect certain information as confidential and proprietary.

I, ______, understand and acknowledge that during the term of my employment by TestAmerica, I will be privy to and entrusted with certain confidential information and trade secrets of TestAmerica and its clients.

Confidential information and trade secrets include, but are not limited to: customer and client lists; price lists; marketing and sales strategies and procedures; operational and equipment techniques; standard operating procedures; business plans and systems; quality control procedures and systems; special projects and technological research, including projects, research and reports for any government entity or client; client's plans and processes; client's manner of operation; the trade secrets of clients; client's data; vendor or supplier pricing; employee lists and personal information, and any other records, data, files, drawings, inventions, discoveries, applications, or processes which are not in the public domain.

#### I agree as follows:

1. I will not in any way, during the term of my employment, or at any time thereafter, except as authorized in writing by the Legal Department of TestAmerica or the client where client data is involved, disclose to others, use for my own benefit, remove from TestAmerica's premises (except to the extent off-site work is approved by my supervisor), copy or make notes of any confidential information and/or trade secrets of TestAmerica or its clients, excepting only that information which may be public knowledge. Technical and business information of any previous employer or other third party which I may disclose to TestAmerica shall be limited to that which was acquired legitimately and disclosed to me without restriction as to secrecy.

2. I agree that all inventions (whether or not patentable) conceived or made by me during the period of my employment by TestAmerica shall belong to TestAmerica, provided such inventions grow out of my work for TestAmerica and are related to the business of TestAmerica. I agree to disclose and assign such inventions to TestAmerica. In California, this provision shall not apply to any invention which qualifies fully under Section 2870 of the California Labor Code.

3. On termination of my employment from TestAmerica, I will deliver to TestAmerica all documents, records, notes, data, memoranda, files, manuals, equipment and things of any nature which relate in any way to confidential information and/or trade secrets of TestAmerica or its clients and which are in my possession or under my control.

4. I agree that during the period of my employment and for one (1) year from and after the termination (for any reason) of my employment with TestAmerica, I shall not directly or indirectly (without first obtaining the written permission of TestAmerica), recruit for employment, or induce to terminate his or her employment with TestAmerica, any person who is an active employee of TestAmerica on the last day of my employment with TestAmerica.

5. I acknowledge that if I were to breach any provision of this Confidentiality Agreement, money damages will be inadequate, and I hereby agree that TestAmerica shall be entitled, where appropriate, to specific performance and/or injunctive relief (i.e. to require me to comply with this Agreement). I further acknowledge that the willingness of TestAmerica to hire me or to continue my employment constitutes full and adequate consideration for the agreements, and obligations to which I have agreed as set forth in this document.

I have executed this Agreement, intending to be legally bound.

Printed Name

Signature

Date Work Instruction No. CA-WI-006

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 2 Page 1 of 1

#### Appendix 2.

**TestAmerica Buffalo Laboratory Organization Chart** 

(The most current chart can be obtained from the QA Manager or Lab Director/Manager)



Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 3 Page 1 of 4

Appendix 3.





THE LEADER IN ENFRONMENTAL PESTING

# TAL BUFFALO HAZELWOOD DR. OFFICES, SUITE 106 CLIENT SERVICES/REPORT PREP FLOOR PLAN





Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 3 Page 3 of 4



Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 3 Page 4 of 4



Document No. BF-QAM Section Revision No.: 0 Section Effective Date: Reserved Appendix 4 Page 1 of 1

Appendix 4: Reserved

# Appendix 5. Glossary/Acronyms

#### Glossary:

Acceptance Criteria:

Specified limits placed on characteristics of an item, process, or service defined in requirement documents. (ASQC)

# Accreditation:

The process by which an agency or organization evaluates and recognizes a laboratory as meeting certain predetermined qualifications or standards, thereby accrediting the laboratory. In the context of the National Environmental Laboratory Accreditation Program (NELAP), this process is a voluntary one. (NELAC)

### Accrediting Authority:

The Territorial, State, or Federal Agency having responsibility and accountability for environmental laboratory accreditation and which grants accreditation (NELAC) [1.5.2.3]

### Accuracy:

The degree of agreement between an observed value and an accepted reference value. Accuracy includes a combination of random error (precision) and systematic error (bias) components which are due to sampling and analytical operations; a data quality indicator. (QAMS)

#### Analyst:

The designated individual who performs the "hands-on" analytical methods and associated techniques and who is the one responsible for applying required laboratory practices and other pertinent quality controls to meet the required level of quality. (NELAC)

#### Assessment:

The evaluation process used to measure or establish the performance, effectiveness, and conformance of an organization and/or its systems to defined criteria (to the standards and requirements of NELAC). (NELAC)

#### Assessment Criteria:

The measures established by NELAC and applied in establishing the extent to which an applicant is in conformance with NELAC requirements. (NELAC)

#### Assessment Team:

The group of people authorized to perform the on-site inspection and proficiency testing data evaluation required to establish whether an applicant meets the criteria for NELAP accreditation. (NELAC)

#### Assessor:

One who performs on-site assessments of accrediting authorities and laboratories' capability and capacity for meeting NELAC requirements by examining the records and other physical evidence for each one of the tests for which accreditation has been requested. (NELAC) Audit:

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 5 Page 2 of 14

A systematic evaluation to determine the conformance to quantitative and qualitative specifications of some operational function or activity. (EPA-QAD)

#### Batch:

Environmental samples which are prepared and/or analyzed together with the same process and personnel, using the same lot(s) of reagents. A preparation batch is composed of one to 20 environmental samples of the same matrix, meeting the above mentioned criteria and with a maximum time between the start of processing of the first and last sample in the batch to be 24 hours. An analytical batch is composed of prepared environmental samples (extracts, digestates or concentrates) and /or those samples not requiring preparation, which are analyzed together as a group using the same calibration curve or factor. An analytical batch can include samples originating from various environmental matrices and can exceed 20 samples. (NELAC Quality Systems Committee)

#### Blank:

A sample that has not been exposed to the analyzed sample stream in order to monitor contamination during sampling, transport, storage or analysis. The blank is subjected to the usual analytical and measurement process to establish a zero baseline or background value and is sometimes used to adjust or correct routine analytical results. (ASQC)

#### Blind Sample:

A sample for analysis with a composition known to the submitter. The analyst/laboratory may know the identity of the sample but not its composition. It is used to test the analyst's or laboratory's proficiency in the execution of the measurement process.

#### Calibration:

To determine, by measurement or comparison with a standard, the correct value of each scale reading on a meter, instrument, or other device. The levels of the applied calibration standard should bracket the range of planned or expected sample measurements. (NELAC)

#### Calibration Curve:

The graphical relationship between the known values, such as concentrations, of a series of calibration standards and their instrument response. (NELAC)

Calibration Method:

A defined technical procedure for performing a calibration. (NELAC)

Calibration Standard:

A substance or reference material used to calibrate an instrument (QAMS)

#### Certified Reference Material (CRM):

A reference material one or more of whose property values are certified by a technically valid procedure, accompanied by or traceable to a certificate or other documentation which is issued by a certifying body. (ISO Guide 30–2.2)

#### Chain of Custody:

An unbroken trail of accountability that ensures the physical security of samples and includes the signatures of all who handle the samples. (NELAC) [5.12.4]

Clean Air Act:

The enabling legislation in 42 U>S>C> 7401 et seq., Public Law 91-604, 84 Stat. 1676 Pub. L. 95-95, 91 Stat., 685 and Pub. L. 95-190, 91 Stat., 1399, as amended, empowering EPA to promulgate air quality standards, monitor and enforce them. (NELAC)

Comprehensive Environmental Response, Compensation and Liability Act (CERCLA/SUPERFUND):

The enabling legislation in 42 U.S.C. 9601-9675 et seq., as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA), 42 U.S.C. 9601 et seq., to eliminate the health and environmental threats posed by hazardous waste sites. (NELAC)

# Compromised Samples:

Those samples which are improperly sampled, insufficiently documented (chain of custody and other sample records and/or labels), improperly preserved, collected in improper containers, or exceeding holding times when delivered to a laboratory. Under normal conditions, compromised samples are not analyzed. If emergency situation require analysis, the results must be appropriately qualified. (NELAC)

# Confidential Business Information (CBI):

Information that an organization designates as having the potential of providing a competitor with inappropriate insight into its management, operation or products. NELAC and its representatives agree to safeguarding identified CBI and to maintain all information identified as such in full confidentiality.

#### Confirmation:

Verification of the identity of a component through the use of an approach with a different scientific principle from the original method. These may include, but are not limited to:

Second column confirmation Alternate wavelength Derivitization Mass spectral interpretation Alternative detectors or Additional Cleanup procedures

(NELAC)

Conformance:

An affirmative indication or judgment that a product or service has met the requirements of the relevant specifications, contract, or regulation; also the state of meeting the requirements. (ANSI/ASQC E4-1994)

#### Corrective Action:

The action taken to eliminate the causes of an existing nonconformity, defect or other undesirable situation in order to prevent recurrence. (ISO 8402)

#### Data Audit:

A qualitative and quantitative evaluation of the documentation and procedures associated with environmental measurements to verify that the resulting data re of acceptable quality (i.e., that they meet specified acceptance criteria). (NELAC)

### Data Reduction:

The process of transforming raw data by arithmetic or statistical calculations, standard curves, concentration factors, etc., and collation into a more useable form. (EPA-QAD)

### Deficiency:

An unauthorized deviation from acceptable procedures or practices, or a defect in an item. (ASQC)

### Detection Limit:

The lowest concentration or amount of the target analyte that can be identified, measured, and reported with confidence that the analyte concentration is not a false positive value. See Method Detection Limit. (NELAC)

### Document Control:

The act of ensuring that documents (and revisions thereto) are proposed, reviewed for accuracy, approved for release by authorized personnel, distributed properly, and controlled to ensure use of the correct version at the location where the prescribed activity if performed. (ASQC)

### Duplicate Analyses:

The analyses or measurements of the variable of interest performed identically on two subsamples of the same sample. The results from duplicate analyses are used to evaluate analytical or measurement precision but not the precision of sampling, preservation or storage internal to the laboratory. (EPA-QAD)

# Environmental Detection Limit (EDL):

The smallest level at which a radionuclide in an environmental medium can be unambiguously distinguished for a given confidence interval using a particular combination of sampling and measurement procedures, sample size, analytical detection limit, and processing procedure. The EDL shall be specified for the 0.95 or greater confidence interval. The EDL shall be established initially and verified annually for each test method and sample matrix. (NELAC Radioanalysis Subcommittee)

Equipment Blank:

Sample of analyte-free media which has been used to rinse common sampling equipment to check effectiveness of decontamination procedures. (NELAC)

#### External Standard Calibration:

Calibrations for methods that do not utilize internal standards to compensate for changes in instrument conditions.

Federal Insecticide, Fungicide and Rodenticide Act (FIFRA):

The enabling legislation under 7 U.S.C. 135 et seq., as amended, that empowers the EPA to register insecticides, fungicides, and rodenticides. (NELAC)

Federal Water Pollution Control Act (Clean Water Act, CWA):

The enabling legislation under 33 U.S.C. 1251 et seq., Public Law 92-50086 Stat 816, that empowers EPA to set discharge limitations, write discharge permits, monitor, and bring enforcement action for non-compliance. (NELAC)

### Field Blank:

Blank prepared in the field by filing a clean container with pure de-ionized water and appropriate preservative, if any, for the specific sampling activity being undertaken (EPA OSWER)

### Field of Testing:

NELAC's approach to accrediting laboratories by program, method and analyte. Laboratories requesting accreditation for a program-method-analyte combination or for an up-dated/improved method are required to submit to only that portion of the accreditation process not previously addressed (see NELAC, section 1.9ff). (NELAC)

#### Finding:

An assessment conclusion that identifies a condition having a significant effect on an item or activity. As assessment finding is normally a deficiency and is normally accompanied by specific examples of the observed condition. (NELAC)

### Holding Times (Maximum Allowable Holding Times):

The maximum times that samples may be held prior to analyses and still be considered valid or not compromised. (40 CFR Part 136)

### Inspection:

An activity such as measuring, examining, testing, or gauging one or more characteristics of an entity and comparing the results with specified requirements in order to establish whether conformance is achieved for each characteristic. (ANSI/ASQC E4-1994)

#### Internal Standard:

A known amount of standard added to a test portion of a sample and carried through the entire measurement process as a reference for evaluating and controlling the precision and bias of the applied analytical test method. (NELAC)

#### Internal Standard Calibration:

Calibrations for methods that utilize internal standards to compensate for changes in instrument conditions.

# Instrument Blank:

A clean sample (e.g., distilled water) processed through the instrumental steps of the measurement process; used to determine instrument contamination. (EPA-QAD)

#### Instrument Response:

Instrument response is normally expressed as either peak area or peak height however it may also reflect a numerical representation of some type of count on a detector (e.g. Photomultiplier tube, or Diode array detector) and is used in this document to represent all types.

#### Laboratory:

A defined facility performing environmental analyses in a controlled and scientific manner. (NELAC)

Laboratory Control Sample (however named, such as laboratory fortified blank, spiked blank, or QC check sample):

A sample matrix, free from the analytes of interest, spiked with verified known amounts of analytes or a material containing known and verified amounts of analytes, taken through all preparation and analysis steps. Where there is no preparation taken for an analysis (such as in

aqueous volatiles), or when all samples and standards undergo the same preparation and analysis process (such as Phosphorus), there is no LCS. It is generally used to establish intralaboratory or analyst specific precision and bias or to assess the performance of all or a portion of the measurement system.

An LCS shall be prepared at a minimum of 1 per batch of 20 or less samples per matrix type per sample extraction or preparation method except for analytes for which spiking solutions are not available such as total suspended solids, total dissolved solids, total volatile solids, total solids, pH, color, odor, temperature, dissolved oxygen or turbidity. The results of these samples shall be used to determine batch acceptance.

Note: NELAC standards allow a matrix spike to be used in place of this control as long as the acceptance criteria are as stringent as for the LCS. (NELAC)

### Laboratory Duplicate:

Aliquots of a sample taken from the same container under laboratory conditions and processed and analyzed independently. (NELAC)

# Least Squares Regression (1st Order Curve):

The least squares regression is a mathematical calculation of a straight line over two axes. The y axis represents the instrument response (or Response ratio) of a standard or sample and the x axis represents the concentration. The regression calculation will generate a correlation coefficient (r) that is a measure of the "goodness of fit" of the regression line to the data. A value of 1.00 indicates a perfect fit. In order to be used for quantitative purposes, r must be greater than or equal to 0.99 for organics and 0.995 for inorganics.

# Limit of Detection (LOD):

An estimate of the minimum amount of a substance that an analytical process can reliably detect. An LOD is analyte- and matrix-specific and may be laboratory dependent. (Analytical Chemistry, 55, p.2217, December 1983, modified) See also Method Detection Limit.

### Manager (however named):

The individual designed as being responsible for the overall operation, all personnel, and the physical plant of the environmental laboratory. A supervisor may report to the manager. In some cases, the supervisor and the manager may be the same individual. (NELAC)

#### Matrix:

The component or substrate that contains the analyte of interest. For purposes of batch and QC requirement determinations, the following matrix distinctions shall be used:

Aqueous: Any aqueous sample excluded from the definition of Drinking Water matrix or Saline/Estuarine source. Includes surface water, groundwater, effluents, and TCLP or other extracts.

Drinking Water: any aqueous sample that has been designated as a potable or potential potable water source.

Saline/Estuarine: any aqueous sample from an ocean or estuary, or other salt water source such as the Great Salt Lake.

Non-aqueous Liquid: any organic liquid with <15% settleable solids.

Biological Tissue: any sample of a biological origin such as fish tissue, shellfish, or plant material. Such samples shall be grouped according to origin.

Solids: includes soils, sediments, sludges, and other matrices with >15% settleable solids.

Chemical Waste: a product or by-product of an industrial process that results in a matrix not previously defined.

Air: whole gas or vapor samples including those contained in flexible or rigid wall containers and the extracted concentrated analytes of interest from a gas or vapor that are collected with a sorbant tube, impinger solution, filter, or other device. (NELAC)

Matrix Spike (spiked sample or fortified sample):

Prepared by adding a known mass of target analyte to a specified amount of matrix sample for which an independent estimate of target analyte concentration is available. Matrix spikes are used, for example, to determine the effect of the matrix on a method's recovery efficiency.

Matrix spikes shall be performed at a frequency of one in 20 samples per matrix type per sample extraction or preparation method except for analytes for which spiking solutions are not available such as, total suspended solids, total dissolved solids, total volatile solids, total solids, pH, color, odor, temperature, dissolved oxygen or turbidity. The selected sample(s) shall be rotated among client samples so that various matrix problems may be noted and/or addressed. Poor performance in a matrix spike may indicate a problem with the sample composition and shall be reported to the client whose sample was used for the spike. (QAMS)

Matrix Spike Duplicate (spiked sample or fortified sample duplicate):

A second replicate matrix spike is prepared in the laboratory and analyzed to obtain a measure of the precision of the recovery for each analyte.

Matrix spike duplicates or laboratory duplicates shall be analyzed at a minimum of 1 in 20 samples per matrix type per sample extraction or preparation method. The laboratory shall document their procedure to select the use of an appropriate type of duplicate. The selected sample(s) shall be rotated among client samples so that various matrix problems may be noted and/or addressed. Poor performance in the duplicates may indicate a problem with the sample composition and shall be reported to the client whose sample was used for the duplicate. (QAMS)

# Method Blank:

A sample of a matrix similar to the batch of associated samples (when available) that is free from the analytes of interest and is processed simultaneously with and under the same conditions as samples through all steps of the analytical procedures, and in which no target analytes or interferences are present at concentrations that impact the analytical results for sample analyses. (NELAC)

# Method Detection Limit:

The minimum concentration of a substance (an analyte) that can be measured and reported with 99% confidence that the analyte concentration is greater than zero and is determined from analysis of a sample in a given matrix containing the analyte. (40 CFR Part 136, Appendix B)

# National Environmental Laboratory Accreditation Conference (NELAC):

A voluntary organization of State and Federal environmental officials and interest groups purposed primarily to establish mutually acceptable standards for accrediting environmental laboratories. A subset of NELAP. (NELAC)

National Environmental Laboratory Accreditation Program (NELAP):

The overall National Environmental Laboratory Accreditation Program of which NELAC is a part. (NELAC)

### Negative Control:

Measures taken to ensure that a test, its components, or the environment do not cause undesired effects, or produce incorrect test results. (NELAC)

### NELAC Standards:

The plan of procedures for consistently evaluating and documenting the ability of laboratories performing environmental measurements to meet nationally defined standards established by the National Environmental Laboratory Accreditation Conference. (NELAC)

#### Performance Audit:

The routine comparison of independently obtained qualitative and quantitative measurement system data with routinely obtained data in order to evaluate the proficiency of an analyst or laboratory. (NELAC)

# Performance Based Measurement System (PBMS):

A set of processes wherein the data quality needs, mandates or limitations of a program or project are specified and serve as criteria for selecting appropriate test methods to meet those needs in a cost-effective manner. (NELAC)

#### Positive Control:

Measures taken to ensure that a test and/or its components are working properly and producing correct or expected results from positive test subjects. (NELAC)

### Precision:

The degree to which a set of observations or measurements of the same property, obtained under similar conditions, conform to themselves; a data quality indicator. Precision is usually expressed as standard deviation, variance or range, in either absolute or relative terms. (NELAC)

#### Preservation:

Refrigeration and/or reagents added at the time of sample collection (or later) to maintain the chemical and/or biological integrity of the sample. (NELAC)

#### Proficiency Testing:

A means of evaluating a laboratory's performance under controlled conditions relative to a given set of criteria through analysis of unknown samples provided by an external source. (NELAC) [2.1]

# Proficiency Testing Program:

The aggregate of providing rigorously controlled and standardized environmental samples to a laboratory for analysis, reporting of results, statistical evaluation of the results and the collective demographics and results summary of all participating laboratories. (NELAC)

### Proficiency Test Sample (PT):

A sample, the composition of which is unknown to the analyst and is provided to test whether the analyst/laboratory can produce analytical results within specified acceptance criteria. (QAMS)

### Quality Assurance:

An integrated system of activities involving planning, quality control, quality assessment, reporting and quality improvement to ensure that a product or service meets defined standards of quality with a stated level of confidence. (QAMS)

### Quality Assurance [Project] Plan (QAPP):

A formal document describing the detailed quality control procedures by which the quality requirements defined for the data and decisions pertaining to a specific project are to be achieved. (EAP-QAD)

### Quality Control:

The overall system of technical activities which purpose is to measure and control the quality of a product or service so that it meets the needs of users. (QAMS)

#### Quality Control Sample:

An uncontaminated sample matrix spiked with known amounts of analytes from a source independent from the calibration standards. It is generally used to establish intra-laboratory or analyst specific precision and bias or to assess the performance of all or a portion of the measurement system. (EPA-QAD)

#### Quality Manual:

A document stating the management policies, objectives, principles, organizational structure and authority, responsibilities, accountability, and implementation of an agency, organization, or laboratory, to ensure the quality of its product and the utility of its product to its users. (NELAC)

# Quality System:

A structured and documented management system describing the policies, objectives, principles, organizational authority, responsibilities, accountability, and implementation plan of an organization for ensuring quality in its work processes, products (items), and services. The quality system provides the framework for planning, implementing, and assessing work performed by the organization and for carrying out required QA and QC (ANSI/ASQC-E-41994)

#### Quantitation Limits:

The maximum or minimum levels, concentrations, or quantities of a target variable (e.g., target analyte) that can be quantified with the confidence level required by the data user. (NELAC)

#### Range:

The difference between the minimum and the maximum of a set of values. (EPA-QAD)

Reagent Blank (method reagent blank):

A sample consisting of reagent(s), without the target analyte or sample matrix, introduced into the analytical procedure at the appropriate point and carried through all subsequent steps to determine the contribution of the reagents and of the involved analytical steps. (QAMS)

### Reference Material:

A material or substance one or more properties of which are sufficiently well established to be used for the calibration of an apparatus, the assessment of a measurement method, or for assigning values to materials. (ISO Guide 30-2.1)

#### Reference Method:

A method of known and documented accuracy and precision issued by an organization recognized as competent to do so. (NELAC)

### Reference Standard:

A standard, generally of the highest metrological quality available at a given location, from which measurements made at that location are derived. (VIM-6.0-8)

### **Replicate Analyses:**

The measurements of the variable of interest performed identically on two or more sub-samples of the same sample within a short time interval. (NELAC)

### Requirement:

Denotes a mandatory specification; often designated by the term "shall". (NELAC).

# Resource Conservation and Recovery Act (RCRA):

The enabling legislation under 42 USC 321 et seq. (1976), that gives EPA the authority to control hazardous waste from the "cradle-to-grave", including its generation, transportation, treatment, storage, and disposal. (NELAC)

# Safe Drinking Water Act (SDWA):

The enabling legislation, 42 USC 300f et seq. (1974), (Public Law 93-523), that requires the EPA to protect the quality of drinking water in the U.S. by setting maximum allowable contaminant levels, monitoring, and enforcing violations. (NELAC)

#### Sample Duplicate:

Two samples taken from and representative of the same population and carried through all steps of the sampling and analytical procedures in an identical manner. Duplicate samples are used to assess variance of the total method including sampling and analysis. (EPA-QAD)

Second Order Polynomial Curve (Quadratic): The  $2^{nd}$  order curves are a mathematical calculation of a slightly curved line over two axis. The y axis represents the instrument response (or Response ratio) of a standard or sample and the x axis represents the concentration. The  $2^{nd}$  order regression will generate a coefficient of determination (COD or  $r^2$ ) that is a measure of the "goodness of fit" of the quadratic curvature the data. A value of 1.00 indicates a perfect fit. In order to be used for quantitative purposes,  $r^2$  must be greater than or equal to 0.99.

# Selectivity:

(Analytical chemistry) the capability of a test method or instrument to respond to a target substance of constituent in the presence of non-target substances. (EPA-QAD)

### Sensitivity:

The capability of a method or instrument to discriminate between measurement responses representing different levels (e.g., concentrations) of a variable of interest. (NELAC)

### Spike:

A known mass of target analyte added to a blank, sample or sub-sample; used to determine recovery efficiency or for other quality control purposes.

If the mandated or requested test method does not specify the spiking components, the laboratory shall spike all reportable components to be reported in the Laboratory Control Sample and Matrix Spike. However, in cases where the components interfere with accurate assessment (such as simultaneously spiking chlordane, toxaphene and PCBs in Method 608), the test method has an extremely long list of components or components are incompatible, a representative number (at a minimum 10%) of the listed components may be used to control the test method. The selected components of each spiking mix shall represent all chemistries, elution patterns and masses permit specified analytes and other client requested components. However, the laboratory shall ensure that all reported components are used in the spike mixture within a two-year time period.. (NELAC)

### Standard:

The document describing the elements of laboratory accreditation that has been developed and established within the consensus principles of NELAC and meets the approval requirements of NELAC procedures and policies. (ASQC)

# Standard Operating Procedures (SOPs):

A written document which details the method of an operation, analysis, or action whose techniques and procedures are thoroughly prescribed and which is accepted as the method for performing certain routine or repetitive tasks. (QAMS)

# Standardized Reference Material (SRM):

A certified reference material produced by the U.S. National Institute of Standards and Technology or other equivalent organization and characterized for absolute content, independent of analytical method. (EPA-QAD)

# Supervisor (however named):

The individual(s) designated as being responsible for a particular area or category of scientific analysis. This responsibility includes direct day-to-day supervision of technical employees, supply and instrument adequacy and upkeep, quality assurance/quality control duties, and ascertaining that technical employees have the required balance of education, training and experience to perform the required analyses. (NELAC)

#### Surrogate:

A substance with properties that mimic the analyte of interest. It is unlikely to be found in environment samples and is added to them for quality control purposes.

Surrogate compounds must be added to all samples, standards, and blanks, for all organic chromatography methods except when the matrix precludes its use or when a surrogate is not available. Poor surrogate recovery may indicate a problem with sample composition and shall be reported to the client whose sample produced poor recovery. (QAMS)

Systems Audit (also Technical Systems Audit):

A thorough, systematic, qualitative on-site assessment of the facilities, equipment, personnel, training, procedures, record keeping, data validation, data management, and reporting aspects of a total measurement system. (EPA-QAD)

# Technical Director:

Individuals(s) who has overall responsibility for the technical operation of the environmental testing laboratory. (NELAC)

# Test:

A technical operation that consists of the determination of one or more characteristics or performance of a given product, material, equipment, organism, physical phenomenon, process, or service according to a specified procedure. The result of a test is normally recorded in a document sometimes called a test report or a test certificate. (ISO/IEC Guide 2-12.1, amended)

# Test Method:

An adoption of a scientific technique for a specific measurement problem, as documented in a laboratory SOP. (NELAC)

# Toxic Substances Control Act (TSCA):

The enabling legislation in 15 USC 2601 et seq., (1976) that provides for testing, regulating, and screening all chemicals produced or imported into the United States for possible toxic effects prior to commercial manufacture. (NELAC)

# Traceability:

The property of a result of a measurement whereby it can be related to appropriate standards, generally international or national standards, through an unbroken chain of comparisons. (VIM-6.12)

# Uncertainty:

A parameter associated with the result of a measurement that characterizes the dispersion of the value that could reasonably be attributed to the measured value.

United States Environmental Protection Agency (EPA):

The Federal governmental agency with responsibility for protecting public health and safeguarding and improving the natural environment (i.e., the air, water, and land) upon which human life depends. (US-EPA)

# Validation:

The process of substantiating specified performance criteria. (EPA-QAD)

# Verification:

Confirmation by examination and provision of evidence that specified requirements have been met. (NELAC)

NOTE: In connection with the management of measuring equipment, verification provides a means for checking that the deviations between values indicated by a measuring instrument and corresponding known values of a measured quantity are consistently smaller than the maximum allowable error defined in a standard, regulation or specification peculiar to the management of the measuring equipment.

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 5 Page 13 of 14

The result of verification leads to a decision either to restore in service, to perform adjustment, to repair, to downgrade, or to declare obsolete. In all cases, it is required that a written trace of the verification performed shall be kept on the measuring instrument's individual record.

#### Work Cell:

A well-defined group of analysts that together perform the method analysis. The members of the group and their specific functions within the work cell must be fully documented. (NELAC)
### Acronyms:

BS – Blank Spike

BSD – Blank Spike Duplicate

CAR – Corrective Action Report

CCV – Continuing Calibration Verification

CF – Calibration Factor

CFR – Code of Federal Regulations

COC – Chain of Custody

CRS - Change Request Form

DOC - Demonstration of Capability

DQO – Data Quality Objectives

DU – Duplicate

DUP - Duplicate

EHS - Environment, Health and Safety

EPA – Environmental Protection Agency

GC - Gas Chromatography

GC/MS - Gas Chromatography/Mass Spectrometry

HPLC - High Performance Liquid Chromatography

ICP - Inductively Coupled Plasma Atomic Emission Spectroscopy

ICV – Initial Calibration Verification

IDL – Instrument Detection Limit

IH – Industrial Hygiene

IS – Internal Standard

LCS – Laboratory Control Sample

LCSD – Laboratory Control Sample Duplicate

LIMS – Laboratory Information Management System

MDL – Method Detection Limit

MS – Matrix Spike

MSD – Matrix Spike Duplicate

MSDS - Material Safety Data Sheet

NELAC - National Environmental Laboratory Accreditation Conference

NELAP - National Environmental Laboratory Accreditation Program

PT – Performance Testing

QAM - Quality Assurance Manual

QA/QC – Quality Assurance / Quality Control

QAPP – Quality Assurance Project Plan

RF – Response Factor

**RPD** – Relative Percent Difference

**RSD** – Relative Standard Deviation

SD – Standard Deviation

SOP: Standard Operating Procedure

TAT – Turn-Around-Time

VOA – Volatiles

VOC – Volatile Organic Compound

Appendix 6.

### Laboratory Certifications, Accreditations, Validations

**TestAmerica Buffalo** maintains certifications, accreditations, certifications, and validations with numerous state and national entities. Programs vary but may include on-site audits, reciprocal agreements with another entity, performance testing evaluations, review of the QA Manual, Standard Operating Procedures, Method Detection Limits, training records, etc. At the time of this QA Manual revision, the laboratory has accreditation/certification/licensing with the following organizations:

State	Program	Certification Number
Arkansas	SDWA, CWA, RCRA, SOIL	88-0686
California*	NELAP CWA, RCRA	01169CA
Connecticut	SDWA, CWA, RCRA, SOIL	PH-0568
Florida*	NELAP CWA, RCRA	E87672
Georgia*	SDWA, NELAP CWA, RCRA	956
Illinois*	NELAP SDWA, CWA, RCRA	200003
lowa	SW/CS	374
Kansas*	NELAP SDWA, CWA, RCRA	E-10187
Kentucky	SDWA	90029
Kentucky UST	UST	30
Louisiana*	NELAP CWA, RCRA	2031
Maine	SDWA, CWA	NY0044
Maryland	SDWA	294
Massachusetts	SDWA, CWA	M-NY044
Michigan	SDWA	9937
Minnesota	SDWA,CWA, RCRA	036-999-337
New Hampshire*	NELAP SDWA, CWA	233701
New Jersey*	NELAP, SDWA, CWA, RCRA,	NY455
New York*	NELAP, AIR, SDWA, CWA, RCRA, CLP	10026
Oklahoma	CWA, RCRA	9421
Pennsylvania*	Registration, NELAP CWA, RCRA	68-00281
Tennessee	SDWA	02970
USDA	FOREIGN SOIL PERMIT	S-41579
USDOE	Department of Energy	DOECAP-STB
Virginia	SDWA	278
Washington	CWA,RCRA	C1677
West Virginia	CWA,RCRA	252
Wisconsin	CWA, RCRA	998310390

The certificates and parameter lists (which may differ) for each organization may be found on the corporate web site, the laboratory's public server, and in the QA office.

## **Claims of Accreditation Status**

*TestAmerica Buffalo* has agreed to make only valid claims as to its accreditation/certification status by any authority by ensuring that the expiration dates are not exceeded and the method-specific scope or parameter lists are

Document No. BF-QAM Section Revision No.: 0 Section Effective Date: 02/01/2008 Appendix 6 Page 2 of 2

supportable, as required by each. Any false claims would be reported to that authority. The agreement covers the use of the authority's name, such as "Authority-Accredited," logo, or certificate number. The only valid proof of accreditation/certification is the current certificate and scope of the authority. It is the responsibility of the laboratory to make these documents available to all staff, and it is the staff's duty to reference only the current documents.

- A report with scope and non-scope analytes may only be presented on the same report if the non-accredited results are clearly and unambiguously identified. No report with non-scope analytes may be associated with the logo, "Authority accredited" phrase, or the certificate number. Only the analytes specified by a unique method are valid within the scope. There shall be no intentional misleading of the users of the laboratory's services in this regard.
- No opinions and/or interpretations based on results outside the laboratory's scope may be presented on a document referenced by "Authority-accredited, the logo, or the certificate number. If these are made, they must be written in a separate letter which is not endorsed by the authority.
- The "Authority-accredited" logo may only be affixed to equipment calibrated by a laboratory that is accredited by the authority. If calibration labels contain the logo, they must also show the calibration laboratory's name or its certificate number, the instrument's unique identification, the date of the last calibration, and a cross-reference to the last calibration certificate.
- Should the company decide to use the "Authority-accredited" logo in marketing activities, no misrepresentation may occur. Only reference to the accredited scope at a specific laboratory site is allowed. If any "Authority-accredited" language is used in proposals or quotations, any non-scope analytes must be clearly denoted as not accredited by that authority. The same is true for any use of laboratory letterhead with the "Authority-accredited" wording or logo. The logo may not be affixed to any material, item, product, part, or packaging, thereby implying accreditation status to that piece. In literature, any use of the logo must be positioned adjacent to the accredited laboratory's name and clearly state that the presence of the logo does not imply certification/approval of the products tested. At no time may the logo appear to suggest that a person is accredited. Misrepresentation of accreditation status is never allowed and must be reported if it occurs. If in doubt, the idea of the logo's use may be presented to the authority for approval.
- If accreditation is terminated or suspended, the laboratory will immediately cease to use the "Authority-accredited" wording, the logo, or the certificate number reference in any way and inform clients impacted by the change.

# Appendix 7. Data Qualifiers

Qualifiers	Footnote	
Organic		
Α	Tentatively identified compound is a suspected aldol condensation.	
U or ND	Compound analyzed but not detected.	
E	Compound was over the calibration range.	
D	Compound analyzed at a secondary dilution factor.	
J	Compound detected but below the reporting limit (the value given is an estimate).	
Ν	Identification of tentatively identified compound is based on a mass spectral library search.	
В	Compound was detected in the method blank.	
С	Mass spectral confirmation of compound.	
Р	The % difference between the results from both columns is >25% (CLP).	
NC	The recovery and or RPD was not calculated.	
*	Value outside QC limits.	
1	Indicates co-elution	
Qualifiers	Footnote	
Inorganic		
B or J	The reported values is less than the Reporting Limit but greater than the Instrument Detection Limit (IDL) or Method Detection Limit (MDL).	
E	The reported value is estimated because of the presence of interference	
NC	The recovery was not calculated due to the concentration of analyte in the sample being >4 times the concentration of spike added.	
N	Spiked sample recovery is not within control limits.	
S	The reported value was determined by the Method of Standard Additions (MSA).	
U or ND	The analyte was analyzed but not detected.	
Н	Indicates analytical holding time exceedance. Value should be considered estimated	
*	Spike or Duplicate analysis and or RPD is not within control limits.	
+	Correlation coefficient for the MSA is less than 0.995.	

# **APPENDIX F Existing Utility Drawings**



# EXPLANATION





# EXPLANATION

